# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

AMENDMENT NO. 3
TO
FORM S-1
REGISTRATION STATEMENT
UNDER
THE SECURITIES ACT OF 1933

# ADDUS HOMECARE CORPORATION

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization)	8082 (Primary Standard Industrial Classification Code Number)	20-5340172 (I.R.S. Employer
(Address, including zip code.	2401 South Plum Grove Road Palatine, Illinois 60067 (847) 303-5300 and telephone number, including area code, of registrant's princip	Identification No.) nal executive offices)
(	Mark S. Heaney	,
	President and Chief Executive Officer Addus HomeCare Corporation 2401 South Plum Grove Road Palatine, Illinois 60067 (847) 303-5300	
· · · · · · · · · · · · · · · · · · ·	ling zip code, and telephone number, including area code, of agen curities and Exchange Commission send copies of all notices, orc	· · · · · · · · · · · · · · · · · · ·
Dominick P. DeChiara, Esq. Lloyd H. Spencer, Esq. Nixon Peabody LLP 437 Madison Avenue New York, New York 10022 (212) 940-3000	Col V 1155 A	in J. Diamond, Esq. Vhite & Case LLP venue of the Americas York, New York 10036 (212) 819-8200
Approximate date of commencement of propertatement becomes effective.	posed sale of the securities to the public: As soon a	s practicable after this registration
If any of the securities being registered on this fact of 1933 check the following box. $\hfill\Box$	orm are to be offered on a delayed or continuous basis	pursuant to Rule 415 under the Securities
	s for an offering pursuant to Rule 462(b) under the Sec he earlier effective registration statement for the same	
If this form is a post-effective amendment filed pegistration statement number of the earlier effective	oursuant to Rule 462(c) under the Securities Act, check registration statement for the same offering. $\ \Box$	the following box and list the Securities Act
If this form is a post-effective amendment filed pegistration statement number of the earlier effective	oursuant to Rule 462(d) under the Securities Act, check registration statement for the same offering. $\ \Box$	the following box and list the Securities Act
Indicate by check mark whether the registrant is company. See definitions of "large accelerated filer," one):	a large accelerated filer, an accelerated filer, a non-ac "accelerated filer," and "smaller reporting company" in	ccelerated filer, or a smaller reporting Rule 12b-2 of the Exchange Act. (Check
□ Large Accelerated Filer	☐ Accelerated Filer	
<ul><li>Non-accelerated Filer</li><li>(Do not check if a smaller reporting cor</li></ul>	□ Smaller Reporting Conpany)	mpany

The registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933, or until the Registration Statement shall become effective on such date as the Commission, acting pursuant to said Section 8(a), may determine.

EXPLANATORY NOTE		
This Amendment is filed solely to file the exhibits indicated in Item 16 of Part II. No change is made to the preliminary prospectus constituting Part I of the Registration Statement or Items 13, 14, 15 or 17 of Part II of the Registration Statement.		

# PART II INFORMATION NOT REQUIRED IN PROSPECTUS

# Item 13. Other Expenses of Issuance and Distribution

The following table sets forth the various expenses, all of which will be borne by the registrant, in connection with the sale and distribution of the securities being registered, other than the underwriting discounts and commissions. All amounts shown are estimates except for the SEC registration fee, the FINRA filing fee and The Nasdaq Global Market initial listing fee.

SEC registration fee	\$3,850
FINRA filing fee	7,400
Nasdaq Global Market initial listing fee	5,000
Transfer agent and registrar fees	*
Accounting fees and expenses	*
Legal fees and expenses	*
Printing and engraving expenses	*
Directors' and officers' insurance	*
Blue sky qualification fees and expenses	*
Miscellaneous	*
Total	\$ *

To be provided by amendment.

## Item 14. Indemnification of Officers and Directors

Section 145 of the Delaware General Corporation Law authorizes a court to award, or a corporation's board of directors to grant, indemnity to officers, directors and other corporate agents in terms sufficiently broad to permit such indemnification under certain circumstances and subject to certain limitations.

As permitted by Section 145 of the Delaware General Corporation Law, the registrant's amended and restated certificate of incorporation includes a provision that eliminates the personal liability of its directors for monetary damages for breach of their fiduciary duty as directors.

In addition, as permitted by Section 145 of the Delaware General Corporation Law, the amended and restated bylaws of the registrant, which will be in effect prior to the completion of any offering pursuant to this registration statement, provide that:

- the registrant shall indemnify its directors and officers for serving the registrant in those capacities or for serving other business enterprises at the registrant's request, to the fullest extent permitted by Delaware law, if such person acted in good faith and in a manner such person reasonably believed to be in or not opposed to the best interests of the registrant, and, with respect to any criminal proceeding, had no reasonable cause to believe such person's conduct was unlawful;
- the registrant may, in its discretion, indemnify employees and agents in those circumstances where indemnification is not required by law;
- the registrant is required to advance expenses, as incurred, to its directors and officers in connection with defending a proceeding, except that such director or officer shall undertake to repay such advances if it is ultimately determined that such person is not entitled to indemnification;
- the registrant will not be obligated pursuant to the bylaws to indemnify a person with respect to proceedings initiated by that person, except with respect to proceedings authorized by the registrant's board of directors or brought to enforce a right to indemnification:
- the rights conferred in the bylaws are not exclusive, and the registrant is authorized to enter into indemnification agreements with its directors, officers, employees and agents and to obtain insurance to indemnify such persons; and
- the registrant may not retroactively amend the bylaw provisions to reduce its indemnification obligations to directors, officers, employees and agents.

The registrant is party to indemnification agreements with each of Brian D. Young, Mark L. First, Simon A. Bachleda, Mark S. Heaney and W. Andrew Wright, III in their capacities as officers and directors (each, an indemnitee). Pursuant to these agreements, the registrant has agreed to hold each indemnitee harmless and indemnify him to the fullest extent permitted by law against all expenses, judgments, penalties, fines and amounts paid in settlement including, without limitation, all liability arising out of the negligence or active or passive wrongdoing of the indemnitee. The registrant is not obligated to make any payment to any indemnitee that is finally determined to be unlawful. In respect of any threatened, pending or completed proceeding in which the registrant is jointly liable with an indemnitee, the registrant will pay the entire amount of any judgment or settlement without requiring the indemnitee to contribute. The registrant will advance, to the extent permitted by law, all expenses incurred by or on behalf of an indemnitee in connection with a proceeding. No amendment, alteration or repeal of the registrant's certificate of incorporation, bylaws or the indemnification agreement with any indemnitee will limit any right of that indemnitee in respect of any action taken or omitted by that indemnitee prior to such amendment. With respect to Messrs. Young, First and Bachleda, the registrant has agreed that, where the indemnitee has certain rights to indemnification, advancement of expenses and/or insurance provided by any of the Eos Funds or their affiliates, the registrant will be the indemnitor of first resort, the registrant will be required to advance the full amount of expenses incurred by the indemnitee and the registrant has waived and released the Eos Funds and their affiliates from any and all claims for contribution, subrogation or any other recovery of any kind. The registrant anticipates that it will enter into similar indemnification agreements with any new member elected to our board of d

#### Item 15. Recent Sales of Unregistered Securities

During the three years preceding the filing of this registration statement, the registrant sold the following securities, which were not registered under the Securities Act. The information below does not reflect the conversion of our series A preferred stock.

The registrant granted options under its stock option plan to purchase 74,265 shares of common stock (net of expirations and cancellations) to employees, directors and consultants, having exercise prices ranging from \$100.00 to \$110.00 per share. None of these options to purchase shares of common stock have been exercised. The offers, sales and issuances of the securities described above were deemed to be exempt from registration under the Securities Act in reliance on Rule 701 in that the transactions were under compensatory benefit plans and contracts relating to compensation as provided under Rule 701. The recipients of such securities were the registrant's employees, directors or bona fide consultants and received the securities under our stock option plan.

On September 19, 2006, the registrant issued: (a) 28,940 shares of its series A preferred stock to Eos Capital Partners III, L.P. in exchange for aggregate consideration of \$28,940,000; (b) 8,310 shares of its series A preferred stock to Eos Partners SBIC III, L.P. in exchange for aggregate consideration of \$8,310,000; and (c) 500 shares of its series A preferred stock to Freeport Loan Fund, LLC in exchange for aggregate consideration of \$500,000, the proceeds of which were used to fund a portion of the purchase price paid in connection with the registrant's acquisition of Addus HealthCare, Inc. The registrant issued these securities to these accredited investors in reliance upon Section 4(2) of the Securities Act of 1933 as a transaction by an issuer not involving a public offering. Each entity had adequate access to information about the registrant through its relationship with the registrant or through information provided to them.

On September 19, 2006, in connection with the registrant's acquisition of Addus HealthCare, Inc., the registrant issued the following shares of common stock to the following persons, former stockholders of Addus HealthCare, Inc., in exchange for their contribution of shares of Addus HealthCare, Inc. common stock, as follows:

Stockholder	Addus HealthCare, Inc. Common Stock Contributed	Addus HomeCare Corporation Common Stock Issued
W. Andrew Wright	72	71,536
Mark S. Heaney	5	5,285
James A. Wright	1	944
Courtney E. Panzer	1	944
Addus Term Trust	4	3,114
W. Andrew Wright Grantor Retained Annuity Trust	13	12,552

The foregoing shares of common stock described in the table above were issued in reliance upon Section 4(2) of the Securities Act of 1933 as a transaction by an issuer not involving a public offering. Each holder had adequate access to information about the registrant through his relationship with the registrant or through information provided to him.

The registrant did not, nor does it plan to, pay or give, directly or indirectly, any commission or other remuneration, including underwriting discounts or commissions, in connection with any of the issuances of securities listed above. In addition, each of the certificates issued representing the securities in the transactions listed above bears a restrictive legend permitting the transfer thereof only in compliance with applicable securities laws. The recipients of securities in each of the transactions listed above represented to the registrant their intention to acquire the securities for investment only and not with a view to or for sale in connection with any distribution thereof. All recipients had or have adequate access, through their employment or other relationship with the registrant or through other access to information provided by our company, to information about our company.

#### Item 16. Exhibits and Financial Statement Schedules.

# (a) Exhibits.

Exhibit <u>Numbe</u> r 1.1	Description of Document Form of Underwriting Agreement*
3.1	Restated Certificate of Incorporation of the Registrant**
3.2	Certificate of Amendment to Restated Certificate of Incorporation of the Registrant, dated as of July 10, 2009**
3.2(a)	Certificate of Amendment of Restated Certificate of Incorporation of the Registrant, dated as of September 22, 2009
3.3	Form of Amended and Restated Certificate of Incorporation of the Registrant, to be effective prior to the closing of the offering**
3.4	Bylaws of the Registrant**
3.5	Form of Amended and Restated Bylaws of the Registrant, to be effective prior to the closing of the offering**
4.1	Form of Common Stock Certificate*
4.2	Stockholders' Agreement, dated September 19, 2006, by and among the Registrant, Eos Capital Partners III, L.P., Eos Partners SBIC III, L.P., Freeport Loan Fund LLC, W. Andrew Wright, III, Addus Term Trust, W. Andrew Wright Grantor Retained Annuity Trust, Mark S. Heaney, James A. Wright and Courtney E. Panzer**
4.3	Registration Rights Agreement, dated September 19, 2006, by and among the Registrant, Eos Capital Partners III, L.P., Eos Partners SBIC III, L.P., Freeport Loan Fund LLC, W. Andrew Wright, III, Addus Term Trust, W. Andrew Wright Grantor Retained Annuity Trust, Mark S. Heaney, James A. Wright and Courtney E. Panzer**

4.4	Form of 10% Junior Subordinated Promissory Note**
5.1	Opinion of Nixon Peabody LLP*
10.1	Employment and Non-Competition Agreement, dated September 19, 2006, between Addus HealthCare, Inc. and W. Andrew Wright, III**
10.1(a)	Amendment to Employment and Non-Competition Agreement, dated May 6, 2008, between Addus HealthCare, Inc. and W. Andrew Wright, III**
10.1(b)	Separation and General Release Agreement, dated as of September 20, 2009, between Addus HealthCare, Inc. and W. Andrew Wright, III**
10.2	Amended and Restated Employment and Non-Competition Agreement, dated May 6, 2008, between Addus HealthCare, Inc. and Mark S. Heaney**
10.3	Employment and Non-Competition Agreement, dated July 31, 2008, between Addus HealthCare, Inc. and Francis J. Leonard**
10.4	Amended and Restated Employment and Non-Competition Agreement, dated August 27, 2007, between Addus HealthCare, Inc. and Darby Anderson**
10.5	Employment and Non-Competition Agreement, dated April 10, 2008, between Addus HealthCare, Inc. and Sharon Rudden**
10.6	Amended and Restated Employment and Non-Competition Agreement, dated October 8, 2008, between Addus HealthCare, Inc. and David W. Stasiewicz**
10.6(a)	Amendment No. 1 to Amended and Restated Employment and Non-Competition Agreement between Addus HealthCare, Inc. and David W. Stasiewicz*
10.7	Employment and Non-Competition Agreement, dated March 23, 2007, between Addus HealthCare, Inc. and Paul Diamond**
10.8	Credit Agreement, dated as of September 19, 2006, between Addus Acquisition Corporation, as borrower, Addus HealthCare, Inc., the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as lender, Residential Funding Corporation and Fifth Third Bank (Chicago), as lenders***
10.8(a)	Consent and First Amendment to Credit Agreement, dated as of July 29, 2007, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, Residential Funding Corporation and Fifth Third Bank (Chicago), as lenders
10.8(b)	Consent and Second Amendment to Credit Agreement, dated as of October 15, 2007, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders
10.8(c)	Consent and Third Amendment to Credit Agreement, dated as of November 13, 2007, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders

Exhibit Number Description of Document

Exhibit <u>Numbe</u> r 10.8(d)	Description of Document  Consent and Fourth Amendment to Credit Agreement, dated as of April 1, 2008, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders
10.8(e)	Consent and Fifth Amendment to Credit Agreement, dated as of June 9, 2008, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders
10.8(f)	Consent and Sixth Amendment to Credit Agreement, dated as of September 25, 2008, between Addus HealthCare, Inc., as borrower, the Registrant, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Offshore Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders
10.8(g)	Consent and Seventh Amendment to Credit Agreement, dated as of October 21, 2008, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders
10.9	Management Consulting Agreement, dated September 19, 2006, between Addus HealthCare, Inc. and Eos Management, Inc.**
10.9(a)	Amendment No. 1 to Management Consulting Agreement, dated July 2008, between Addus HealthCare, Inc. and Eos Management, Inc.**
10.9(b)	Termination Agreement between Addus HealthCare, Inc. and Eos Management, Inc.**
10.10	Addus HealthCare, Inc. Home Health and Home Care Division Vice President and Regional Director Bonus Plan**
10.11	Addus HealthCare, Inc. Support Center Vice President and Department Director Bonus Plan**
10.12	Addus Holding Corporation 2006 Stock Incentive Plan**
10.13	Director Form of Option Award Agreement under the 2006 Stock Incentive Plan**
10.14	Executive Form of Option Award Agreement under the 2006 Stock Incentive Plan**
10.15	Contingent Payment Agreement, dated September 19, 2006, among the Registrant, Addus Acquisition Corporation, Addus Management Corporation, Addus HealthCare, Inc., W. Andrew Wright, III, Addus Term Trust, W. Andrew Wright Grantor Retained Annuity Trust, Mark S. Heaney, James A. Wright and Courtney E. Panzer**
10.16	Form of Indemnification Agreement**
10.17	License Agreement, dated March 24, 2006, between McKesson Information Solutions, LLC and Addus HealthCare, Inc.**
10.17(a)	Contract Supplement to the License Agreement, dated March 24, 2006**

Contract Supplement to the License Agreement, dated March 28, 2006\*\*

10.17(b)

Exhibit Number 10.17(c)	Description of Document  Amendment to License Agreement, dated March 28, 2006, between McKesson Information Solutions, LLC and Addus HealthCare, Inc.**
10.18	Lease, dated April 1, 1999, between W. Andrew Wright, III and Addus HealthCare, Inc.**
10.18(a)	First Amendment to Lease, dated as of April 1, 2002, between W. Andrew Wright, III and Addus HealthCare, Inc.**
10.18(b)	Second Amendment to Lease, dated as of September 19, 2006, between W. Andrew Wright, III and Addus HealthCare, Inc.**
10.18(c)	Third Amendment to Lease, dated as of September 1, 2008, between W. Andrew Wright, III and Addus HealthCare, Inc.**
10.19	Consent Fee Agreement among the Registrant, Eos Capital Partners III, L.P. and Eos Partners SBIC III, L.P.**
10.20	Form of Addus HomeCare Corporation 2009 Stock Incentive Plan**
10.20(a)	Form of Incentive Stock Option Award Agreement under the 2009 Stock Incentive Plan**
10.20(b)	Form of Restricted Stock Award Agreement under the 2009 Stock Incentive Plan**
21.1	Subsidiaries of the Registrant**
23.1	Consent of BDO Seidman, LLP, Independent Registered Public Accounting Firm**
23.2	Consent of Dixon Hughes PLLC, Independent Public Accounting Firm**
23.3	Consent of Nixon Peabody LLP (included in Exhibit 5.1)*
24.1	Power of Attorney**
24.1(a)	Power of Attorney (Steven I. Geringer)**

<sup>\*</sup> To be filed by amendment.

(b) Financial Statement Schedules.

All other schedules have been omitted because they are either inapplicable or the required information has been given in the consolidated financial statements or the notes thereto.

#### Item 17. Undertakings

The undersigned registrant hereby undertakes to provide to the underwriters at the closing specified in the underwriting agreements certificates in such denominations and registered in such names as required by the underwriters to permit prompt delivery to each purchaser.

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

<sup>\*\*</sup> Previously filed.

<sup>\*\*\*</sup> Portions of this exhibit have been omitted pursuant to a request for confidential treatment, and the redacted provisions have been filed separately with the Securities and Exchange Commission in connection with such request.

The undersigned registrant hereby undertakes that:

- (1) for purposes of determining any liability under the Securities Act, the information omitted from the form of prospectus filed as part of this registration statement in reliance upon Rule 430A and contained in a form of prospectus filed by the registrant pursuant to Rule 424(b)(1) or (4) or 497(h) under the Securities Act shall be deemed to be part of this registration statement as of the time it was declared effective.
- (2) for the purpose of determining any liability under the Securities Act, each post-effective amendment that contains a form of prospectus shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

# **SIGNATURES**

Pursuant to the requirements of the Securities Act of 1933, as amended, the registrant has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Palatine, State of Illinois, on the 22nd day of September, 2009.

ADDUS HOMECARE CORPORATION

By: /s/ MARK S. HEANEY

Mark S. Heaney

President and Chief Executive Officer

Pursuant to the requirements of the Securities Act of 1933, as amended, this Registration Statement has been signed below by the following persons in the capacities indicated on September 22, 2009.

<u>Signatur</u> e	<u>Title</u>
/S/ MARK S. HEANEY Mark S. Heaney	President, Chief Executive Officer and Chairman of the board (principal executive officer)
/S/ FRANCIS J. LEONARD Francis J. Leonard	Chief Financial Officer, Vice President and Secretary (principal financial officer and principal accounting officer)
* Brian D. Young	— Director
* Mark L. First	— Director
* Simon A. Bachleda	— Director
W. Andrew Wright, III	— Director
* Steven I. Geringer	— Director
*By: /S/ MARK S. HEANEY Mark S. Heaney Attorney-in-Fact	

# **EXHIBIT INDEX**

Exhibit Number	Description of Document
1.1	Form of Underwriting Agreement*
3.1	Restated Certificate of Incorporation of the Registrant**
3.2	Certificate of Amendment to Restated Certificate of Incorporation of the Registrant, dated as of July 10, 2009**
3.2(a)	Certificate of Amendment of Restated Certificate of Incorporation of the Registrant, dated as of September 22, 2009
3.3	Form of Amended and Restated Certificate of Incorporation of the Registrant, to be effective prior to the closing of the offering**
3.4	Bylaws of the Registrant**
3.5	Form of Amended and Restated Bylaws of the Registrant, to be effective prior to the closing of the offering**
4.1	Form of Common Stock Certificate*
4.2	Stockholders' Agreement, dated September 19, 2006, by and among the Registrant, Eos Capital Partners III, L.P., Eos Partners SBIC III, L.P., Freeport Loan Fund LLC, W. Andrew Wright, III, Addus Term Trust, W. Andrew Wright Grantor Retained Annuity Trust, Mark S. Heaney, James A. Wright and Courtney E. Panzer**
4.3	Registration Rights Agreement, dated September 19, 2006, by and among the Registrant, Eos Capital Partners III, L.P., Eos Partners SBIC III, L.P., Freeport Loan Fund LLC, W. Andrew Wright, III, Addus Term Trust, W. Andrew Wright Grantor Retained Annuity Trust, Mark S. Heaney, James A. Wright and Courtney E. Panzer**
4.4	Form of 10% Junior Subordinated Promissory Note**
5.1	Opinion of Nixon Peabody LLP*
10.1	Employment and Non-Competition Agreement, dated September 19, 2006, between Addus HealthCare, Inc. and W. Andrew Wright**
10.1(a)	Amendment to Employment and Non-Competition Agreement, dated May 6, 2008, between Addus HealthCare, Inc. and W. Andrew Wright**
10.1(b)	Separation and General Release Agreement, dated as of September 20, 2009, between Addus HealthCare, Inc. and W. Andrew Wright, III**
10.2	Amended and Restated Employment and Non-Competition Agreement, dated May 6, 2008, between Addus HealthCare, Inc. and Mark S. Heaney**
10.3	Employment and Non-Competition Agreement, dated July 31, 2008, between Addus HealthCare, Inc. and Frank Leonard**
10.4	Amended and Restated Employment and Non-Competition Agreement, dated August 27, 2007, between Addus HealthCare, Inc. and Darby Anderson**
10.5	Employment and Non-Competition Agreement, dated April 10, 2008, between Addus HealthCare, Inc. and Sharon Rudden**
10.6	Amended and Restated Employment and Non-Competition Agreement, dated October 8, 2008, between Addus HealthCare, Inc. and David W. Stasiewicz**
10.6(a)	Amendment No. 1 to Amended and Restated Employment and Non-Competition Agreement between Addus HealthCare, Inc. and David W. Stasiewicz*
10.7	Employment and Non-Competition Agreement, dated March 23, 2007, between Addus HealthCare, Inc. and Paul Diamond**

Exhibit	Description of
Number	Document
10.8	Credit Agreement, dated as of September 19, 2006, between Addus Acquisition Corporation, as borrower, Addus HealthCare, Inc.,
	the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort
	Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing
	Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as lender, Residential Funding Corporation and Fifth
	Third Bank (Chicago), as lenders***

- 10.8(a) Consent and First Amendment to Credit Agreement, dated as of July 29, 2007, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Constitution Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as a lender, Freeport Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, Residential Funding Corporation and Fifth Third Bank (Chicago), as lenders
- 10.8(b) Consent and Second Amendment to Credit Agreement, dated as of October 15, 2007, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Offshore Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders
- 10.8(c) Consent and Third Amendment to Credit Agreement, dated as of November 13, 2007, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Offshore Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders
- Consent and Fourth Amendment to Credit Agreement, dated as of April 1, 2008, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Offshore Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders
- 10.8(e) Consent and Fifth Amendment to Credit Agreement, dated as of June 9, 2008, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Construction of Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders
- 10.8(f)

  Consent and Sixth Amendment to Credit Agreement, dated as of September 25, 2008, between Addus HealthCare, Inc., as borrower, the Registrant, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Offshore Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders
- 10.8(g)

  Consent and Seventh Amendment to Credit Agreement, dated as of October 21, 2008, between Addus HealthCare, Inc., as borrower, the Registrant, Addus Management Corporation, Lowell Home Health Agency, Inc., Little Rock Home Health Agency, Inc., Fort Smith Home Health Agency, Inc., Benefits Assurance Co., Inc., PHC Acquisition Corporation, Professional Reliable Nursing Service, Inc., Freeport Financial LLC, as agent, Freeport Loan Fund LLC, as a lender, Freeport Offshore Loan Fund LLC, as a lender, Freeport Loan Trust 2006-1, as a lender, CF Blackburn LLC and Fifth Third Bank (Chicago), as lenders

Exhibit Number	Description of Document
10.9	Management Consulting Agreement, dated September 19, 2006, between Addus HealthCare, Inc. and Eos Management, Inc.**
10.9(a)	Amendment No. 1 to Management Consulting Agreement, dated July 2008, between Addus HealthCare, Inc. and Eos Management, Inc.**
10.9(b)	Termination Agreement between Addus HealthCare, Inc. and Eos Management, Inc.**
10.10	Addus HealthCare, Inc. Home Health and Home Care Division Vice President and Regional Director Bonus Plan**
10.11	Addus HealthCare, Inc. Support Center Vice President and Department Director Bonus Plan**
10.12	Addus Holding Corporation 2006 Stock Incentive Plan**
10.13	Director Form of Option Award Agreement under the 2006 Stock Incentive Plan**
10.14	Executive Form of Option Award Agreement under the 2006 Stock Incentive Plan**
10.15	Contingent Payment Agreement, dated September 19, 2006, among the Registrant, Addus Acquisition Corporation, Addus Management Corporation, Addus HealthCare, Inc., W. Andrew Wright, III, Addus Term Trust, W. Andrew Wright Grantor Retained Annuity Trust, Mark S. Heaney, James A. Wright and Courtney E. Panzer**
10.16	Form of Indemnification Agreement**
10.17	License Agreement, dated March 24, 2006, between McKesson Information Solutions, LLC and Addus HealthCare, Inc.**
10.17(a)	Contract Supplement to the License Agreement, dated March 24, 2006**
10.17(b)	Contract Supplement to the License Agreement, dated March 28, 2006**
10.17(c)	Amendment to License Agreement, dated March 28, 2006, between McKesson Information Solutions, LLC and Addus HealthCare Inc.**
10.18	Lease, dated April 1, 1999, between W. Andrew Wright, III and Addus HealthCare, Inc.**
10.18(a)	First Amendment to Lease, dated as of April 1, 2002, between W. Andrew Wright, III and Addus HealthCare, Inc.**
10.18(b)	Second Amendment to Lease, dated as of September 19, 2006, between W. Andrew Wright, III and Addus HealthCare, Inc.**
10.18(c)	Third Amendment to Lease, dated as of September 1, 2008, between W. Andrew Wright, III and Addus HealthCare, Inc.**
10.19	Consent Fee Agreement among the Registrant, Eos Capital Partners III, L.P. and Eos Partners SBIC III, L.P.**
10.20	Form of Addus HomeCare Corporation 2009 Stock Incentive Plan**
10.20(a)	Form of Incentive Stock Option Award Agreement under the 2009 Stock Incentive Plan**
10.20(b)	Form of Restricted Stock Award Agreement under the 2009 Stock Incentive Plan**
21.1	Subsidiaries of the Registrant**
23.1	Consent of BDO Seidman, LLP, Independent Registered Public Accounting Firm**
23.2	Consent of Dixon Hughes PLLC, Independent Public Accounting Firm**
23.3	Consent of Nixon Peabody LLP (included in Exhibit 5.1)*
24.1	Power of Attorney**
24.1(a)	Power of Attorney (Steven I. Geringer)**

To be filed by amendment.

\*\* Previously filed.

\*\*\* Portions of this exhibit have been omitted pursuant to a request for confidential treatment, and the redacted provisions have been filed separately with the Securities and Exchange Commission in connection with such request.

#### CERTIFICATE OF AMENDMENT

ЭF

## RESTATED CERTIFICATE OF INCORPORATION

OF

## ADDUS HOMECARE CORPORATION

Under Section 242 of the Delaware General Corporation Law

Addus HomeCare Corporation, a corporation organized and existing under the General Corporation Law of the State of Delaware (hereinafter called the "Corporation"), hereby certifies that:

- 1. The name of the Corporation is Addus HomeCare Corporation.
- 2. The Certificate of Incorporation of the Corporation was originally filed under the name Addus Holding Corporation with the Secretary of State of the State of Delaware on the 27th day of July, 2006. The Restated Certificate of Incorporation of the Corporation was filed with the Secretary of State of the State of Delaware on the 19th day of September, 2006. A Certificate of Amendment of the Restated Certificate of Incorporation of the Corporation was filed with the Secretary of State of the State of Delaware on the 10th day of July, 2009.
- 3. The Restated Certificate of Incorporation of the Corporation is hereby further amended by deleting Section 3(a)(ii) of the Fourth Article thereof and replacing it with the following:
  - "Dividends on the shares of Series A Convertible Preferred Stock (when, as and if declared by the Board, out of funds legally available therefor), shall be payable in cash, pursuant to a promissory note issued by the Corporation, or by a combination thereof, in the sole discretion of the holder of such shares of Series A Convertible Preferred Stock, in each case, on each Series A Dividend Date."
- 4. The Restated Certificate of Incorporation of the Corporation is hereby further amended by deleting Section 7(b) of the Fourth Article thereof and replacing it with the following:
  - "All accrued and unpaid dividends on the shares being converted pursuant to this Section 7 shall be paid in cash, pursuant to a promissory note issued by the Corporation, or by a combination thereof, in the sole discretion of the holder of such shares of Series A Convertible Preferred Stock in each case, on the Conversion Date (as defined below)."
- 5. The Restated Certificate of Incorporation of the Corporation is hereby further amended by deleting Section 7(c) of the Fourth Article thereof and replacing it with the following:

"Any holder of shares of Series A Convertible Preferred Stock may exercise the conversion right pursuant to subsection (a) above by delivering to the Corporation the certificate or certificates for the shares to be converted, duly endorsed or assigned in blank or to the Corporation (if required by it), accompanied by written notice stating that the holder elects to convert such shares and stating the name or names (with address) in which the certificate or certificates for the shares of Common Stock are to be issued. Conversion shall be deemed to have been effected on the date when such delivery is made or as otherwise specified in such written notice (the "Conversion Date"). As promptly as practicable after the effective date, the Corporation shall issue and deliver to the holder at its address on the books of record of the Corporation or, upon the written order of such holder, to the place designated by such holder, a certificate or certificates for the number of full shares of Common Stock to which such holder is entitled, and an amount in cash, a promissory note (which promissory note may be combined with any promissory note issued in payment of accrued dividends), or a combination thereof, in the sole discretion of the holder, in respect of any fractional interest in a share of Common Stock as provided in subsection (d) below. The Person in whose name the certificate or certificates for Common Stock are to be issued shall be deemed to have become a holder of record of such Common Stock on the applicable Conversion Date unless the transfer books of the Corporation are closed on that date, in which event such Person shall be deemed to have become a holder of record of such Common Stock on the next succeeding date on which the transfer books are open, but the Series A Conversion Price shall be that in effect on the Conversion Date. Upon conversion of only a portion of the number of shares covered by a certificate representing shares of the Series A Convertible Preferred Stock surrendered for conversion, the Corporation shall issue and deliver to or upon the written order of the holder of the certificate so surrendered for conversion, at the expense of the Corporation, a new certificate covering the number of shares of such Series A Convertible Preferred Stock representing the unconverted portion of the certificate so surrendered."

6. The Restated Certificate of Incorporation of the Corporation is hereby further amended by deleting Section 7(d) of the Fourth Article thereof and replacing it with the following:

"Upon conversion, the Corporation will not issue fractional shares of its Common Stock. In lieu of any fractional shares of Common Stock which would otherwise be issuable upon the conversion of the shares of Series A Convertible Preferred Stock, the Corporation shall pay to the holder of the Series A Convertible Preferred Stock being so converted an adjustment in cash, a promissory note (which promissory note may be combined with any promissory note issued in payment of accrued dividends), or a combination thereof, in the sole discretion of the holder, in respect of such fractional interest in an amount equal to the then fair market value, as determined in good faith by the Board, of a share of Common Stock multiplied by such fractional interest."

7. The Restated Certificate of Incorporation of the Corporation is hereby further amended by deleting Section 7(e)(v) of the Fourth Article thereof and replacing it with the following:

"In any case in which the provisions of this subsection (e) shall require that an adjustment shall become effective immediately after a record date of an event, the Corporation may defer until the occurrence of such event (i) issuing to the holder of any share of Series A Convertible Preferred Stock converted after such record date and before the occurrence of such event the shares of capital stock issuable upon such conversion by reason of the adjustment required by such event in addition to the shares of capital stock issuable upon such conversion before giving effect to such adjustments, and (ii) paying to such holder any amount in lieu of a fractional share of capital stock pursuant to subsection (d) above; provided, however, that the Corporation shall deliver to such holder an appropriate instrument evidencing such holder's right to receive such additional shares and such amount in lieu of a fraction share of capital stock."

- 8. The Restated Certificate of Incorporation of the Corporation is hereby further amended by deleting Section 8(b) of the Fourth Article thereof and replacing it with the following:
  - "As promptly as practicable after the satisfaction of the condition set forth in Section 8(a) and the delivery to the Corporation of the certificate or certificates for the shares of Series A Convertible Preferred Stock which have been converted, duly endorsed or assigned in blank to the Corporation (if required by it), the Corporation shall issue and deliver to or upon the written order of each holder of Series A Convertible Preferred Stock, to the place designated by such holder, a certificate or certificates for the number of full shares of Common Stock to which such holder is entitled. The Person in whose name the certificate or certificates for Common Stock are to be issued shall be deemed to have become a stockholder of record on the date of such occurrence and on such date the shares of Series A Convertible Preferred Stock shall cease to be outstanding, whether or not the certificates representing such shares have been received by the Corporation."
- 9. The Restated Certificate of Incorporation of the Corporation is hereby further amended by adding a new Section 8(c) to the Fourth Article thereof to read in its entirety as follows:
  - "All accrued and unpaid dividends on the shares converted pursuant to this Section 8 shall be paid in cash, pursuant to a promissory note issued by the Corporation, or by a combination thereof, in the sole discretion of the holder of such shares, in each case, promptly following the consummation of a QIPO. The provisions of Section 7(d) shall apply to any conversion pursuant to this Section 8."
- 10. The Restated Certificate of Incorporation of the Corporation is hereby further amended by adding a new Section 10 to the Fourth Article thereof to read in its entirety as follows:

# "10. Promissory Notes.

The terms of any promissory note issued pursuant to Section 3(a)(ii), 7(b), 7(c) or 8(c) of this Article IV shall be negotiated in good faith by the Corporation and the applicable Series A Holder; *provided*, that no Series A Holder shall be required to accept a promissory note in lieu of cash without its consent."

11. The amendment to the Restated Certificate of Incorporation of the Corporation contained herein has been duly adopted in accordance with the provisions of Sections 228 and 242 of the General Corporation Law of the State of Delaware.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the undersigned authorized officer has caused this Certificate of Amendment to be executed this 22nd day of September, 2009.

# ADDUS HOMECARE CORPORATION

By: /s/ Mark S. Heaney

Name: Mark S. Heaney Title: President

 $[Signature\ Page\ for\ Addus\ Home Care\ Corporation\ Certificate\ of\ Amendment\ to\ Restated\ Certificate\ of\ Incorporation]$ 

PORTIONS OF THIS EXHIBIT HAVE BEEN REDACTED AND THIS EXHIBIT IS SUBJECT TO A REQUEST FOR CONFIDENTIAL TREATMENT. REDACTED MATERIAL IS MARKED WITH "[\*\*\*]" AND HAS BEEN FILED SEPARATELY WITH THE SECURITIES AND EXCHANGE COMMISSION.

**Execution Copy** 

# CREDIT AGREEMENT

## DATED AS OF SEPTEMBER 19, 2006

by and among

ADDUS ACQUISITION CORPORATION (to be merged as of the Closing Date into Addus HealthCare, Inc.) as Borrower

and

THE OTHER PERSONS PARTY HERETO THAT ARE DESIGNATED AS LOAN PARTIES

and

FREEPORT FINANCIAL LLC as Agent and Lead Arranger

FREEPORT LOAN FUND LLC, as a Lender

and

THE OTHER FINANCIAL INSTITUTIONS PARTY HERETO as Lenders

# TABLE OF CONTENTS

		Pag
SECTION	1. DEFINITIONS	1
1.1	Definitions	1
1.2	Rules of Construction	29
SECTION 2	2. AMOUNTS AND TERMS OF LOANS	29
2.1	Loans	29
2.2	Interest and Applicable Margins	34
2.3	Fees	37
2.4	Payments	38
2.5	Prepayments	38
2.6	Maturity	39
2.7	Loan Accounts	40
2.8	Yield Protection	40
2.9	Taxes	42
SECTION :	3. CONDITIONS TO LOANS	44
3.1	Conditions to Initial Loans	44
3.2	Conditions to All Loans	46
SECTION 4	4. REPRESENTATIONS AND WARRANTIES	43
4.1	Organization, Powers, Capitalization and Good Standing	47
4.2	Disclosure	47
4.3	No Material Adverse Effect	48
4.4	No Conflict	48
4.5	Financial Statements and Projections	48
4.6	Solvency	48
4.7	Use of Proceeds; Margin Regulations	48
4.8	Brokers	49
4.9	Compliance with Laws	49
4.10	1 5	49
4.11	Investigations, Audits, Etc.	49
4.12	1 )	49
4.13	Litigation; Adverse Facts	50
4.14	Ownership of Property; Liens	50
4.15		50
4.16		51
4.17	1	51
4.18	Agreements and Other Documents	51

i

51

	4.19	Insurance	52
	4.20	Taxes and Tax Returns	52
	4.21	No Earn-outs	52
	4.22	Compliance With Health Care Laws	52
SECTION 5. AFFIRMATIVE COVENANTS			54
	5.1	Compliance With Laws and Contractual Obligations	54
	5.2	Insurance	54
	5.3	Inspection; Coding and Billing Review; Lender Meeting	55
	5.4	Organizational Existence	55
	5.5	Environmental Matters	55
	5.6	Landlords' Agreements, Mortgagee Agreements, Bailee Letters and Real Estate Purchases	56
	5.7	Further Assurances	56
	5.8	Payment of Taxes	57
	5.9	Cash Management Systems	57
	5.10	Interest Rate Agreements	57
	5.11	Employee Benefit Plans	57
	5.12	Health Care Law Matters	57
	5.13	HIPAA	58
SEC	TION 6.	NEGATIVE COVENANTS	58
	6.1	Indebtedness	58
	6.2	Liens and Related Matters	59
	6.3	Investments	60
	6.4	Contingent Obligations	61
	6.5	Restricted Payments	61
	6.6	Restriction on Fundamental Changes	62
	6.7	Disposal of Assets or Subsidiary Stock	62
	6.8	Transactions with Affiliates	62
	6.9	Conduct of Business	63
	6.10	Fiscal Year	63
	6.11	Press Release; Public Offering Materials	63
	6.12	Subsidiaries	63
	6.13	Deposit Accounts	63
	6.14	ERISA	63
	6.15	Sale-Leasebacks	63
SECTION 7. FINANCIAL COVENANTS/REPORTING			64
	7.1	Financial Covenants	64
	7.2	Financial Statements and Other Reports	66

7.3	Accounting Terms; Utilization of GAAP for Purposes of Calculations Under Agreement	69		
SECTION 8. DEFAULT, RIGHTS AND REMEDIES				
8.1 8.2 8.3 8.4 8.5	Event of Default Suspension or Termination of Commitments Acceleration and other Remedies Performance by Agent Application of Proceeds	69 72 72 72 73		
SECTION 9.	ASSIGNMENT AND PARTICIPATION	73		
9.1 9.2 9.3 9.4 9.5	Assignment and Participations Agent Set Off and Sharing of Payments Disbursement of Funds Disbursements of Advances; Payment	73 75 81 81		
SECTION 10	O. MISCELLANEOUS	83		
10.1 10.2 10.3	Indemnities Amendments and Waivers Notices	83 83 84		
10.4 10.5	Failure or Indulgence Not Waiver; Remedies Cumulative Marshaling; Payments Set Aside	85 85		
10.6 10.7 10.8	Severability Lenders' Obligations Several; Independent Nature of Lenders' Rights Headings	85 85 86		
10.9 10.10	Applicable Law Successors and Assigns	86 86		
	No Fiduciary Relationship; Limited Liability Construction Confidentiality	86 86 86		
10.14 10.15	CONSENT TO JURISDICTION WAIVER OF JURY TRIAL	87 87		
10.17	Survival of Warranties and Certain Agreements Entire Agreement Counterparts; Effectiveness	87 87 88		
10.19	Replacement of Lenders Delivery of Termination Statements and Mortgage Releases	88 89		

89

10.21 Subordination of Intercompany Debt

#### **INDEX OF APPENDICES**

# Annexes

Annex A - Pro Rata Shares and Commitment Amounts

Annex B - Closing Checklist

Annex C - Pro Forma

Annex D - Lenders' Bank Accounts

Annex E - Compliance, Pricing and Excess Cash Flow Certificate
Annex F - Form of Borrower Assignment and Assumption Agreement

## **Exhibits**

Exhibit 2.1(a) - Term Notes
Exhibit 2.1(b)(i) - Revolving Note

Exhibit 2.1(b)(ii) - Notice of Revolving Credit Advance
Exhibit 2.1(c) - Request for Letter of Credit Issuance
Exhibit 2.2(e) - Notice of Continuation/Conversion
Exhibit 5.12 - HIPAA Business Associate Agreement
Exhibit 7.2(e) - Borrowing Availability Certificate

Exhibit 9.1 - Assignment Agreement

#### Schedules

Schedule 4.1(a) - Jurisdictions of Organization and Qualifications

Schedule 4.1(b) - Capitalization

Schedule 4.5 - Financial Statements and Projections

Schedule 4.7 - Use of Proceeds
Schedule 4.10 - Intellectual Property
Schedule 4.11 - Investigations and Audits
Schedule 4.12 - Employee Matters

Schedule 4.13 - Litigation Schedule 4.14 - Real Estate

Schedule 4.15 - Environmental Matters

Schedule 4.16 - ERISA

Schedule 4.17 - Deposit and Disbursement Accounts Schedule 4.18 - Agreements and Other Documents

Schedule 4.19 - Insurance

Schedule 4.22(a) - Compliance with Health Care Laws
Schedule 4.22(f) - Participation in Health Care Programs

Schedule 6.1 - Indebtedness

Schedule 6.2 - Liens

Schedule 6.4 - Contingent Obligations Schedule 6.8 - Affiliate Transactions Schedule 6.9 - Business Description

#### **CREDIT AGREEMENT**

This CREDIT AGREEMENT is dated as of September 19, 2006 and entered into by and among Addus Acquisition Corporation, a Delaware corporation ("Merger Sub", which will immediately following the consummation of the Addus Acquisition (as defined below) and the advance of the Loans (as defined below), merge (the "Merger") with and into Addus HealthCare, Inc., an Illinois corporation (the "Company"; prior to consummation of the Merger, Merger Sub, and from and after consummation of the Merger, the Company, being referred to in this Agreement as the "Borrower"), the other persons designated as "Loan Parties", the financial institutions who are or hereafter become parties to this Agreement as Lenders, FREEPORT FINANCIAL, LLC, a Delaware limited liability company (in its individual capacity "Freeport"), as Agent and FREEPORT LOAN FUND LLC, a Delaware limited liability company, as a Lender (in its individual capacity, "Freeport Loan").

#### RECITALS:

WHEREAS, Borrower desires that Lenders extend a term credit facility and a revolving credit facility to Borrower to fund the repayment of certain indebtedness of Borrower, to finance fees and expenses incurred by Holdings in connection with the Addus Acquisition, to finance future acquisitions by Borrower and its Subsidiaries, to provide working capital financing for Borrower and its Subsidiaries and to provide funds for other general corporate purposes of Borrower and its Subsidiaries; and

WHEREAS, Borrower desires to secure all of its Obligations (as hereinafter defined) under the Loan Documents (as hereinafter defined) by granting to Agent, for the benefit of Agent and Lenders, a security interest in and lien upon substantially all of its personal and owned real property; and

WHEREAS, Addus Management Corporation, a Delaware corporation ("<u>Intermediate Holdings</u>") that upon consummation of the Addus Acquisition will own 80% of the Stock of Borrower, which is a wholly-owned subsidiary of Addus Holding Corporation, a Delaware corporation ("<u>Holdings</u>") that upon consummation of the Addus Acquisition will own 20% of the Stock of Borrower, and Holdings and Intermediate Holdings are willing to guaranty all of the Obligations and to pledge to Agent, for the benefit of Agent and Lenders, all of the Stock of Borrower and their Subsidiaries and substantially all of its other personal and owned real property to secure the Obligations; and

WHEREAS, all capitalized terms herein shall have the meanings ascribed thereto in Section 1 hereof.

NOW, THEREFORE, in consideration of the premises and the agreements, provisions and covenants herein contained, Borrower, Loan Parties, Lenders and Agent agree as follows:

#### SECTION 1. DEFINITIONS

1.1 Definitions, Capitalized terms used in the Loan Documents shall have the following respective meanings:

Acceptable Standby Letter of Credit means a standby letter of credit, issued by a bank or financial institution acceptable to Agent in its sole discretion, in form and substance satisfactory to Agent in its sole discretion, in an amount equal to 105% of the aggregate outstanding Letter of Credit Obligations to be available to Agent to reimburse payments of drafts drawn under outstanding Letters of Credit and to pay any Fees and expenses related thereto.

Account <u>Debtor</u> means any Person who may become obligated to any Loan Party under, with respect to, or on account of, an Account, Chattel Paper or General Intangibles (including a payment intangible).

Accounting Changes means: (a) changes in accounting principles required by GAAP and implemented by Holdings or any of its Subsidiaries; (b) changes in accounting principles recommended by the Company's certified public accountants and implemented by the Company; and (c) changes in carrying value of the Company's or any of its Subsidiaries' assets, liabilities or equity accounts resulting from (i) the application of purchase accounting principles (A.P.B. 16 and/or 17, FASB 141 and EITF 88-16 and FASB 109) to the Related Transactions or (ii) as the result of any other adjustments that, in each case, were applicable to, but not included in, the Pro Forma.

Accounts means all "accounts," as such term is defined in the Code, now owned or hereafter acquired by any Loan Party.

Acquisition means any transaction or series of related transactions for the purpose of or resulting, directly or indirectly, in (a) the acquisition of all or substantially all of the assets of a Person or of all or substantially all of any business or division of a Person, (b) the acquisition of in excess of 50% of the capital stock, partnership interests, membership interest or equity of any Person, or otherwise causing any Person to become a Subsidiary, or (c) a merger or consolidation or any other combination with another Person (other than a Person that is already a Subsidiary).

Acquisition Note means that certain Limited Recourse Acquisition Note in the principal amount of \$35,666,880 issued on the Closing Date by Merger Sub in favor of the Sellers in form acceptable to the Agent.

Addus Acquisition means the acquisition by Merger Sub and Holdings of the Stock of the Company pursuant to the terms of the Purchase Agreement.

Advance has the meaning ascribed to it in Section 3.2.

Advance Multiple shall mean at any time, a multiple equal to the applicable maximum Leverage Ratio permitted by Section 7.1(d) for the last day of the then current Fiscal Quarter; provided that for the period from the Closing Date through September 30, 2006, the Advance Multiple shall be deemed to be 4.25:1.00.

Affected Lender has the meaning ascribed to it in Section 10.19(a).

Affiliate means, with respect to any Person, (a) each Person that, directly or indirectly, owns or controls, whether beneficially, or as a trustee, guardian or other fiduciary, 10% or more of the Stock having ordinary voting power in the election of directors of such Person, (b) each Person that controls, is controlled by or is under common control with such Person and (c) each of such Person's officers, directors, joint venturers and partners. For the purposes of this definition, "control" of a Person shall mean the possession, directly or indirectly, of the power to direct or cause the direction of its management or policies, whether through the ownership of voting securities, by contract or otherwise; provided, however, that the term "Affiliate" shall specifically exclude Agent and each Lender and each unrelated portfolio company of Sponsor and its Affiliates that is not Holdings or a Subsidiary of Holdings.

Agent means Freeport in its capacity as Agent for itself and Lenders or a successor agent.

Agent Indemnitee means all Affiliates, officers, directors, employees, agents, and attorneys of Agent.

Agreement means this Credit Agreement (including all schedules, subschedules, annexes and exhibits hereto), as the same may be amended, supplemented, restated or otherwise modified from time to time.

Anti-Terrorism Laws has the meaning ascribed to it in Section 4.9.

Applicable Margins means collectively the Applicable Revolver Index Margin, the Applicable Term Loan Index Margin, the Applicable Revolver LIBOR Margin and the Applicable Term Loan LIBOR Margin.

<u>Applicable Revolver Index Margin</u> means the per annum interest rate margin from time to time in effect and payable in addition to the Index Rate applicable to the Revolving Loan, as determined by reference to <u>Section 2.2(a)</u>.

Applicable Revolver LIBOR Margin means the per annum interest rate from time to time in effect and payable in addition to the LIBOR Rate applicable to the Revolving Loan, as determined by reference to Section 2.2(a).

<u>Applicable Term Loan Index Margin</u> means the per annum interest rate from time to time in effect and payable in addition to the Index Rate applicable to the Term Loan, as determined by reference to Section 2.2(a).

Applicable Term Loan LIBOR Margin means the per annum interest rate from time to time in effect and payable in addition to the LIBOR Rate applicable to the Term Loan, as determined by reference to Section 2.2(a).

Asset Disposition means the disposition, whether by sale, lease, transfer, loss, damage, destruction, casualty, condemnation or otherwise, of any of the following: (a) any of the Stock or other equity or ownership interest of any of Borrower's Subsidiaries or (b) any or all of the assets of Borrower or any of its Subsidiaries other than (1) sales and dispositions described in Section 6.7(a), (2) sales of Cash Equivalents, (3) licenses of Intellectual Property in the ordinary course of business, (4) any lease, sublease, license or sublicense of property not constituting Indebtedness and not entered into as part of a sale and leaseback transaction, (5) any disposition, by Borrower or any Guarantor to a Guarantor or to Borrower, (6) any sale or issuance by Borrower or any Subsidiary thereof of its own Stock to its direct parent, (7) the abandonment or termination of intellectual property of leaseholds in the ordinary course of business, (8) sales, discounts and forgiveness of accounts receivable in connection with the compromise thereof in the ordinary course of business, and (9) dispositions permitted under Section 6.6.

Assignment Agreement has the meaning ascribed to it in Section 9.1(a).

Average Daily Balance means, for any period, the quotient of (a) the sum of the daily closing balances of the Revolving Credit Advances for each day during such period divided by (b) the number of days in such period. For the purpose of calculating the Average Daily Balance for any period that includes days prior to the Closing Date, the daily closing balance of the Revolving Credit Advances for each day prior to the Closing Date shall be deemed to have been \$0.

Bankruptcy Code means the provisions of Title 11 of the United States Code, 11 U.S.C. §§ 101 et seq. or other applicable bankruptcy, insolvency or similar laws.

Borrower has the meaning ascribed to it in the preamble to the Agreement.

Borrower Assignment and Assumption Agreement means an agreement substantially in the form of Annex F.

Borrower Pledge Agreement means the Pledge Agreement of even date herewith executed by Borrower in favor of Agent, on behalf of itself and Lenders, pledging all Stock of certain of its Domestic Subsidiaries and 65% of the voting Stock of its first tier Foreign Subsidiaries.

Borrower Security Agreement means the Borrower Security Agreement of even date herewith entered into between Agent, on behalf of itself and Lenders, and Borrower.

Borrowing Availability means as of any date of determination the lesser of (i) the Maximum Amount less the outstanding Revolving Loans and (ii) the excess, if any, of (a) the product of (I) the Advance Multiple then in effect multiplied by (II) EBITDA for the most recent period of twelve months for which financial statements have been delivered pursuant to Section 7.2(a), over (b) the aggregate amount of Funded Debt (including, without duplication, the outstanding balance of Loans and Letter of Credit Obligations then outstanding, but excluding any Loan requested and not yet made) of Holdings and its Subsidiaries on a consolidated basis at such time.

Borrowing Availability Certificate has the meaning ascribed to it in Section 7.2(e).

Business Day means any day that is not a Saturday, a Sunday or a day on which banks are required or permitted to be closed in the States of Illinois or Wisconsin and in reference to LIBOR Loans shall mean any such day that is also a LIBOR Business Day.

Capex Limit has the meaning ascribed to it in Section 7.1.

<u>Capital Expenditures</u> means (i) all expenditures (by the expenditure of cash or (without duplication) the incurrence of Indebtedness) during any measuring period for any fixed asset or improvements or replacements, substitutions, or additions thereto that have a useful life of more than one year and are required to be capitalized under GAAP, <u>plus</u> (ii) (without duplication of (i)) deposits made during such measuring period for Capital Expenditures <u>less</u> (iii) such deposits during a prior measuring period and reflected in the amount calculated in clause (i) above, <u>less</u> (iv) Net Proceeds of Asset Dispositions which Borrower has reinvested under Section 2.5(c) that are reflected in the amount calculated in clause (i) above and less (v) all such expenditures in respect of which such Person is entitled to reimbursement in full in cash from a third party (including landlords) and has been so reimbursed in full in cash.

<u>Capital Lease</u> means, with respect to any Person, any lease of any property (whether real, personal or mixed) by such Person as lessee that, in accordance with GAAP, would be required to be classified and accounted for as a capital lease on a balance sheet of such Person.

<u>Capital Lease Obligation</u> means, with respect to any Capital Lease of any Person, the amount of the obligation of the lessee thereunder that, in accordance with GAAP, would appear on a balance sheet of such lessee in respect of such Capital Lease.

Carry Over Amount has the meaning ascribed to it Section 7.1(a).

Cash Equivalents means: (i) marketable securities (A) issued or directly and unconditionally guaranteed as to interest and principal by the United States government or (B) issued by any agency of the United States government the obligations of which are backed by the full faith and credit of the United States, in each case maturing within one (1) year after acquisition thereof; (ii) marketable direct obligations issued by any state of the United States of America or any political subdivision of any such state or any public instrumentality thereof, in each case maturing within one year after acquisition thereof and having, at the time of acquisition, a rating of at least A-1 from S&P or at least P-1 from Moody's; (iii) commercial paper maturing no more than one year from the date of acquisition and, at the time of acquisition, having a rating of at least A-1 from S&P or at least P-1 from Moody's; (iv) certificates of deposit or bankers' acceptances issued or accepted by any Lender or by any commercial bank organized under the laws of the United States of America or any state thereof or the District of Columbia that is at least (A) "adequately capitalized" (as defined in the regulations of its primary Federal banking regulator) and (B) has Tier 1 capital (as defined in such regulations) of not less than \$250,000,000, in each case maturing within one year after issuance or acceptance thereof; and (v) shares of any money market mutual or similar funds that (A) has substantially all of its assets invested continuously in the types of investments referred to in clauses (i) through (iv) above, (B) has net assets of not less than \$500,000,000 and (C) has the highest rating obtainable from either S&P or Moody's.

Certificate of Exemption has the meaning ascribed to it in Section 2.9(c).

CHAMPVA means, collectively, the Civilian Health and Medical Program of the Department of Veteran Affairs, a program of medical benefits covering retirees and dependents of former members of the armed services administered by the United States Department of Veteran Affairs, and all laws, rules, regulations, manuals, orders, guidelines or requirements pertaining to such program, including, without limitation, (i) all federal statutes (whether set forth in 38 U.S.C. §§ 1701 et seq. or elsewhere) affecting such program or, to the extent applicable to TRICARE, and (ii) all rules, regulations (including 38 C.F.R. § 1701 et seq.), manuals, orders and administrative, reimbursement and other guidelines of all Governmental Authorities promulgated in connection with such program (whether or not having the force of law), in each case as the same may be amended, supplemented or otherwise modified from time to time.

Change of Control means any event, transaction or occurrence as a result of which (a) the Sponsor ceases to own and control all of the economic and voting rights associated with ownership of at least fifty-five percent (55%) of all of the outstanding Voting Stock of Holdings on a fully diluted basis, (b) Holdings ceases to own and control all of the economic and voting rights associated with all of the outstanding Stock of Intermediate Holdings, (c) Intermediate Holdings and Holdings, together, cease to own and control all of the economic and voting rights associated with all of the outstanding Stock of Borrower, or (d) except as a result of transactions permitted under this Agreement, Borrower ceases to own and control all of the economic and voting rights associated with all of the outstanding Stock of any of its Subsidiaries.

Charges means all federal, state, county, city, municipal, local, foreign or other governmental premiums and other amounts (including premiums and other amounts owed to the PBGC at the time due and payable), levies, assessments, charges, liens, claims or encumbrances upon or relating to (a) the Collateral, (b) the Obligations, (c) the employees, payroll, income or gross receipts of any Loan Party, (d) any Loan Party's ownership or use of any properties or other assets, or (e) any other aspect of any Loan Party's business.

Chattel Paper means any "chattel paper," as such term is defined in the Code, including electronic chattel paper, now owned or hereafter acquired by any Loan Party, wherever located.

<u>Closing Checklist</u> means the schedule, including all appendices, exhibits or schedules thereto, listing certain documents and information to be delivered in connection with the Agreement, the other Loan Documents and the transactions contemplated thereunder, substantially in the form attached hereto as <u>Annex B</u>.

Closing Date means September 19, 2006.

CMS means the Department of Health and Human Services Centers for Medicare & Medicaid Services or any successor thereto.

Code means the Uniform Commercial Code as the same may, from time to time, be enacted and in effect in the State of New York; provided, that to the extent that the Code is used to define any term herein or in any Loan Document and such term is defined differently in different Articles or Divisions of the Code, the definition of such term contained in Article or Division 9 shall govern; provided further, that in the event that, by reason of mandatory provisions of law, any or all of the attachment, perfection or priority of, or remedies with respect to, Agent's or any Lender's Lien on any Collateral is governed by the Uniform Commercial Code as enacted and in effect in a jurisdiction other than the State of New York, the term "Code" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes of the provisions thereof relating to such attachment, perfection, priority or remedies and for purposes of definitions related to such provisions.

Coding and Billing Review has the meaning ascribed to it in Section 5.3(b).

<u>Collateral</u> means the property covered by the Security Agreements and the other Collateral Documents and any other property, real or personal, tangible or intangible, now existing or hereafter acquired, that may at any time be or become subject to a security interest or Lien in favor of Agent, on behalf of itself and Lenders, to secure the Obligations or any portion thereof.

<u>Collateral Documents</u> means the Security Agreements, the Pledge Agreements, the Guaranties, the Patent Security Agreements, the Trademark Security Agreements, the Copyright Security Agreements and all similar agreements entered into guaranteeing payment of, or granting a Lien upon property as security for payment of, the Obligations or any portion thereof.

Commitment Termination Date means the earliest of (a) as to the Revolving Loan and Term Loan, September 19, 2011, (b) the date of termination of Lenders' obligations to make Advances and to incur Letter of Credit Obligations or permit existing Loans to remain outstanding pursuant to Section 8.3, and (c) the date of (i) indefeasible prepayment in full by Borrower of the Loans, (ii) the cancellation and return (or stand-by guarantee) of all Letters of Credit or the cash collateralization or, with the consent of Agent in each instance, the backing with standby letters of credit acceptable to Agent, of all Letter of Credit Obligations pursuant to and in the amount required by Section 2.5(f), and (iii) the permanent reduction of the Commitments to zero dollars (\$0).

Commitments means (a) as to any Lender, the aggregate of such Lender's Revolving Loan Commitment and Term Loan Commitment as set forth on Annex A to the Agreement or in the most recent Assignment Agreement executed by such Lender and (b) as to all Lenders, the aggregate of all Lenders' Revolving Loan Commitments and Term Loan Commitments, which aggregate commitment shall be Fifty Seven Million Five Hundred Thousand Dollars (\$57,500,000) on the Closing Date, as such Commitments may be reduced, amortized or adjusted from time to time in accordance with this Agreement.

<u>Communication</u> means any notice, information or other communication required or permitted to be given or made under this Agreement, but excluding any Loan Document requested by Agent to be delivered solely in a signed writing, including without limitation, any Note, power of attorney, or Patent, Trademark or Copyright Security Agreement.

Compliance, Pricing, and Excess Cash Flow Certificate has the meaning ascribed to it in Section 7.2(n).

Consolidated Net Income means net income during the measuring period on a consolidated basis excluding: (i) the income (or deficit) of any Person accrued prior to the date it became a Subsidiary of, or was merged or consolidated into, Holdings, Borrower or any of Holdings' or Borrower's Subsidiaries, (ii) the income (or deficit) of any Person (other than a Subsidiary) in which Holdings has an ownership interest, except to the extent any such income has actually been received by Borrower or any of its Subsidiaries in the form of cash dividends or distributions, (iii) the undistributed earnings of any Subsidiary of Holdings to the extent that the declaration or payment of dividends or similar distributions by such Subsidiary is not at the time permitted by the terms of any contractual obligation or requirement of law applicable to such Subsidiary, (iv) any restoration to income of any contingency reserve, except to the extent that provision for such reserve was made out of income accrued during such period, (v) any net gain attributable to the write-up of any asset, (vi) any loss attributable to the write-down of any asset (other than Accounts and Inventory), (vii) any net gain from the collection of the proceeds of life insurance policies, (viii) any net gain arising from the acquisition of any securities, or the extinguishment of any Indebtedness, of Holdings or any of its Subsidiaries, (ix) any deferred credit representing the excess of equity in any Subsidiary of Holdings or Borrower of the investment in such Subsidiary.

Contingent Obligation means, as applied to any Person, any direct or indirect liability of that Person: (i) with respect to Guaranteed Indebtedness and with respect to any Indebtedness, lease, dividend or other obligation of another Person if the purpose or intent of the Person incurring such liability, or the effect thereof, is to provide assurance to the obligee of such liability that such liability will be paid or discharged, or that any agreements relating thereto will be complied with, or that the holders of such liability will be protected (in whole or in part) against loss with respect thereto; (ii) with respect to any letter of credit issued for the account of that Person or as to which that Person is otherwise liable for reimbursement of drawings; (iii) under any foreign exchange contract, currency swap agreement, interest rate swap agreement or other similar agreement or arrangement designed to alter the risks of that Person arising from fluctuations in currency values or interest rates, (iv) any agreement, contract or transaction involving commodity options or future contracts, (v) to make take-or-pay or similar payments if required regardless of nonperformance by any other party or parties to an agreement, or (vi) pursuant to any agreement to purchase, repurchase or otherwise acquire any obligation or any property constituting security therefor, to provide funds for the payment or discharge of such obligation or to maintain the solvency, financial condition or any balance sheet item or level of income of another. The amount of any Contingent Obligation shall be equal to the amount of the obligation so guaranteed or otherwise supported or, if not a fixed and determined amount, the maximum amount so guaranteed.

Contingent Payment Agreement means that certain Contingent Payment Agreement dated as of September 19, 2006 by and among Holdings, Intermediate Holdings, Merger Sub, the Company, W. Andrew Wright, III, as Sellers' Representative and each of the individuals and entities identified as "Contingent Payment Recipients" on Exhibit A thereto.

<u>Contingent Payments</u> means the amounts payable under section 1.2 of the Contingent Payment Agreement as such amounts are calculated under the Contingent Payment Agreement as originally in effect or as amended with the prior written consent of the Agent and the Required Lenders.

<u>Contractual Obligations</u> means, as applied to any Person, any indenture, mortgage, deed of trust, contract, undertaking, agreement or other instrument to which that Person is a party or by which it or any of its properties is bound or to which it or any of its properties is subject including the Related Transactions Documents.

Control Agreement means a tri-party deposit account, securities account or commodities account control agreements by and among the applicable Loan Party, Agent and the depository, securities intermediary or commodities intermediary, and each in form and substance reasonably satisfactory in all respects to Agent and in any event providing to Agent "control" of such deposit account, securities or commodities account within the meaning of Articles 8 and 9 of the Code.

<u>Copyright License</u> means any and all rights now owned or hereafter acquired by any Loan Party under any written agreement granting any right to such Loan Party to use any Copyright or Copyright registration owned by a third party.

Copyright Security Agreements means the Copyright Security Agreements made in favor of Agent, on behalf of itself and Lenders, by each applicable Loan Party.

<u>Copyrights</u> means all of the following now owned or hereafter adopted or acquired by any Loan Party: (a) all copyrights (whether registered or unregistered), all registrations and recordings thereof, and all applications in connection therewith, including all registrations, recordings and applications in the United States Copyright Office or in any similar office or agency of the United States, any state or territory thereof, or any other country or any political subdivision thereof; and (b) all reissues, extensions or renewals thereof.

<u>Current Assets</u> means, with respect to any Person, all current assets of such Person as of any date of determination calculated in accordance with GAAP, but excluding cash, cash equivalents and debts due from Affiliates.

<u>Current Liabilities</u> means, with respect to any Person, all liabilities that should, in accordance with GAAP, be classified as current liabilities, and in any event shall include all Indebtedness payable on demand or within one year from any date of determination without any option on the part of the obligor to extend or renew beyond such year, all accruals for federal or other taxes based on or measured by income and payable within such year, but excluding the current portion of long-term debt required to be paid within one year and the aggregate outstanding principal balances of the Revolving Loan.

Default means any event that, with the passage of time or notice or both, would, unless cured or waived, become an Event of Default.

Default Rate has the meaning ascribed to it in Section 2.2(d).

<u>Disbursement Account</u> has the meaning ascribed to it in <u>Section 2.1(d)</u>.

<u>Disclosure Schedules</u> means the Schedules prepared by Borrower and denominated as <u>Schedules 4.1(a) through 6.9</u> in the index to the Agreement.

<u>Documents</u> means any "document," as such term is defined in the Code, including electronic documents, now owned or hereafter acquired by any Loan Party, wherever located.

Dollars or \$\sqrt{\$} means lawful currency of the United States of America.

Domestic Loan Parties means any Loan Party organized under the laws of a jurisdiction in the United States of America.

<u>Domestic Subsidiaries</u> means any Subsidiary organized under the laws of a jurisdiction in the United States of America.

EBITDA means Consolidated Net Income less: (in each case to the extent included in the calculation of Consolidated Net Income, but without duplication): (a) income tax credits, (b) interest income, (c) gain from extraordinary items, (d) the aggregate net gain arising from the sale, exchange or other disposition of assets out of the ordinary course of business, other than Accounts and Inventory, (e) any other non-cash gains, (f) expenditures related to the Related Transactions and not reflected on the Pro Forma or the footnotes thereto, and (g) non-recurring gains; plus: (in each case to the extent deducted in the calculation of Consolidated Net Income, but without duplication): (i) any provision for income taxes or franchise taxes, (ii) Interest Expense, (iii) depreciation, depletion and amortization expense (including goodwill impairment charges), (iv) amortized debt discount (but in the case of amortization and expenses of Related Transactions, only to the extent included in the Pro Forma), (v) any deduction as the result of any grant to any members of the management of Holdings or Borrower or any of its Subsidiaries of any Stock, (vi) loss from extraordinary items (vii) any loss arising from the sale, exchange or other disposition of assets out of the ordinary course of business, other than Accounts and Inventory, (viii) any other non-cash losses (other than non-cash losses relating to write-offs, write-downs or reserves with respect to Accounts and Inventory), (ix) expenses of the Related Transactions (including, without limitation or duplication, bonuses paid to employees on or prior to the Closing Date), provided that such expenses were included in the Pro Forma, or disclosed in any notes thereto, (x) non-cash purchase accounting adjustments in connection with the Addus Acquisition and write-off of amortization related to write-up of assets due to purchase accounting in connection with the Addus Acquisition, (xi) management fees paid pursuant to the Management Consulting Agreement not to exceed \$100,000 in the aggregate in any Fiscal Quarter, (xii) any charges which have been reimbursed in cash through proceeds from the Escrow Account, (xiii) the McKesson Add-Back and (xiv) other expenses and items consistent with those expenses and items otherwise considered in computing add-backs hereunder incurred during the period commencing July 1, 2006 through and including the Closing Date in an amount not to exceed \$1,050,000; provided, that, notwithstanding anything to the contrary contained herein, for each of the Fiscal Quarters listed below, EBITDA shall be deemed to be the amount set forth below opposite such corresponding period:

Period	EBITDA
March 31, 2006	\$2,914,037
June 30, 2006	\$3,781,124

Environmental Laws means all applicable federal, state, local and foreign laws, statutes, ordinances, codes, rules, standards and regulations, now or hereafter in effect, and any legally binding applicable judicial or administrative interpretation thereof, including any applicable judicial or administrative order, consent decree, order or judgment, imposing liability or standards of conduct for or relating to the regulation and protection of human health, safety, and the environment (including ambient air, surface water, groundwater, wetlands, land surface or subsurface strata). Environmental Laws include the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (42 U.S.C. §§

9601 et seq.) ("CERCLA"); the Hazardous Materials Transportation Authorization Act of 1994 (49 U.S.C. §§ 5101 et seq.); the Federal Insecticide, Fungicide, and Rodenticide Act (7 U.S.C. §§ 136 et seq.); the Solid Waste Disposal Act (42 U.S.C. §§ 6901 et seq.); the Toxic Substances Control Act (15 U.S.C. §§ 2601 et seq.); the Clean Air Act (42 U.S.C. §§ 7401 et seq.); the Federal Water Pollution Control Act (33 U.S.C. §§ 1251 et seq.); the Occupational Safety and Health Act, to the extent it regulates Hazardous Materials (29 U.S.C. §§ 651 et seq.); the Safe Drinking Water Act (42 U.S.C. §§ 300(f) et seq.), laws applicable to medical waste, and any and all regulations promulgated thereunder, and all analogous state, local and foreign counterparts or equivalents and any transfer of ownership notification or approval statutes that relate to environmental contamination or Hazardous Materials.

Environmental Liabilities means, with respect to any Person, all liabilities, obligations, responsibilities, response, remedial and removal costs, investigation and feasibility study costs, capital costs, operation and maintenance costs, losses, damages, punitive damages, property damages, natural resource damages, consequential damages, treble damages, costs and expenses (including all reasonable fees, disbursements and expenses of counsel, experts and consultants), fines, penalties, sanctions and interest incurred as a result of or related to any claim, suit, action, investigation, proceeding or demand by any Person, whether based in contract, tort, implied or express warranty, strict liability, criminal or civil statute or common law, including any arising under or related to any Environmental Laws, Environmental Permits, or in connection with any Release or threatened Release or presence of a Hazardous Material whether on, at, in, under, from or about or in the vicinity of any real or personal property or to any real property to which Hazardous Materials originating from or generated by the Loan Parties or their Subsidiaries come to be located.

Environmental Permits means all permits, licenses, authorizations, certificates, approvals or registrations required by any Governmental Authority under any Environmental Laws.

Equipment means all "equipment," as such term is defined in the Code, now owned or hereafter acquired by any Loan Party, wherever located.

ERISA means the Employee Retirement Income Security Act of 1974, as amended from time to time, and any regulations promulgated thereunder.

ERISA Affiliate means, with respect to any Loan Party, any trade or business (whether or not incorporated) that, together with such Loan Party, are treated as a single employer within the meaning of Sections 414(b) or (c) (and Sections 414(m) and (o) of the IRC for purposes of provisions relating to Section 412 of the IRC).

ERISA Event means, with respect to any Loan Party or any ERISA Affiliate, (a) any event described in Section 4043(c) of ERISA with respect to a Title IV Plan (other than an event for which the 30-day notice period is waived); (b) the withdrawal of any Loan Party or ERISA Affiliate from a Title IV Plan subject to Section 4063 of ERISA during a plan year in which it was a substantial employer, as defined in Section 4001(a)(2) of ERISA; (c) the complete or partial withdrawal of any Loan Party or any ERISA Affiliate from any Multiemployer Plan; (d) the filing of a notice of intent to terminate a Title IV Plan or the treatment of a plan amendment as a termination under Section 4041 of ERISA; (e) the institution of proceedings to terminate a Title IV Plan or Multiemployer Plan by the PBGC; (f) the failure by any Loan Party or ERISA Affiliate to make when due required contributions to a Multiemployer Plan or Title IV Plan unless such failure is cured within 30 days; (g) any other event or condition that would reasonably be expected to constitute grounds for the termination of, or the appointment of a trustee to administer, any Title IV Plan or Multiemployer Plan under Section 4042 of ERISA or for the imposition of liability under Section 4069 or 4212(c) of ERISA; (h) the termination of a Multiemployer Plan under

Section 4041A of ERISA or the reorganization or insolvency of a Multiemployer Plan under Section 4241 or 4245 of ERISA; or (i) the loss of a Qualified Plan's qualification or tax exempt status; or (j) the termination of a Plan described in Section 4064 of ERISA.

Escrow Account means the account established on the Closing Date pursuant to the terms of the Purchase Agreement.

Escrow Agreement means that certain Escrow Agreement dated as of September 19, 2006, by and among Fifth Third Bank, as Escrow Agent, Holdings, Intermediate Holdings, Merger Sub, and W. Andrew Wright, III, as Sellers' Representative.

Event of Default has the meaning ascribed to it in Section 8.1.

Excess Cash Flow has the meaning ascribed to it in Schedule 2 to Annex E.

Excluded Tax has the meaning ascribed to it in Section 2.9(a).

Fair Labor Standards Act means the Fair Labor Standards Act, 29 U.S.C. §201 et seq.

<u>Federal Funds Rate</u> means, for any day, a floating rate equal to the weighted average of the rates on overnight federal funds transactions among members of the Federal Reserve System, as determined by Agent in its sole discretion, which determination shall be final, binding and conclusive (absent manifest error).

Federal Reserve Board means the Board of Governors of the Federal Reserve System.

Fees means any and all fees payable to Agent or any Lender pursuant to the Agreement or any of the other Loan Documents.

Field Review has the meaning ascribed to it in Section 5.3(a).

Fifth Third Bank means Fifth Third Bank (Chicago).

<u>Financial Statements</u> means the consolidated and consolidating income statements, statements of cash flows and balance sheets of Holdings and its Subsidiaries delivered in accordance with <u>Section 7.2</u>.

Fiscal Month means any of the monthly accounting periods of Holdings and its Subsidiaries of each Fiscal Year.

Fiscal Quarter means any of the quarterly accounting periods of Holdings and its Subsidiaries, ending on March 31, June 30, September 30 and December 31 of each year.

Fiscal Year means any of the annual accounting periods of Holdings and its Subsidiaries ending on December 31 of each year.

<u>Fixed Charges</u> means for any measuring period (i) Interest Expense <u>plus</u> (ii) scheduled payments of principal of Funded Debt <u>plus</u> (iii) management fees paid pursuant to the Management Consulting Agreement.

<u>Fixed Charge Coverage Ratio</u> means for any measuring period the ratio of (x) (i) EBITDA <u>less</u> (ii) Capital Expenditures, other than the portion thereof funded by third party financing and less the sum

of all federal, state and local income taxes and franchise taxes (excluding provisions for taxes in respect of gains on the sale of assets, and extraordinary and non-recurring gains) paid in cash (net of any credit for such taxes), to (y) Fixed Charges.

Fixtures means all "fixtures" as such term is defined in the Code, now owned or hereafter acquired by any Loan Party.

Freeport has the meaning ascribed to it in the Preamble.

Freeport Fee Letter has the meaning ascribed to it in Section 2.3(a).

Foreign Lender has the meaning ascribed to it in Section 2.9(c).

Foreign Subsidiary means any direct or indirect Subsidiary of Holdings organized under the laws of a jurisdiction outside of the United States.

Funded Debt means, with respect to any Person, without duplication, all Indebtedness for borrowed money evidenced by notes, bonds, debentures, or similar evidences of Indebtedness and that by its terms matures more than one year from, or is directly or indirectly renewable or extendible at such Person's option under a revolving credit or similar agreement obligating the lender or lenders to extend credit over a period of more than one year from the date of creation thereof, and specifically including Capital Lease Obligations and current portions thereof, current maturities of long-term debt, revolving credit and short-term debt extendible beyond one year at the option of the debtor, and also including, in the case of Borrower, the Loans and Letter of Credit Obligations and, without duplication, Guaranteed Indebtedness consisting of guaranties of Funded Debt of other Persons; provided, however that Indebtedness arising under the Commercial Insurance Premium Finance and Security Agreement, dated June 8, 2006, by and between the Borrower and Cananwill, Inc. ("Cananwill") wherein the Borrower assigned to Cananwill as security for the total amount payable any gross unearned premiums, shall not be deemed Funded Debt. Notwithstanding the foregoing, the obligation of Intermediate Holdings or any other Loan Party to pay the Contingent Payments shall not constitute Funded Debt to the extent no payment obligations have matured or otherwise become due and payable with respect thereto.

<u>Funding Date</u> has the meaning ascribed to it in <u>Section 3.2</u>.

GAAP means generally accepted accounting principles in the United States of America, consistently applied.

General Intangibles means "general intangibles," as such term is defined in the Code, now owned or hereafter acquired by any Loan Party.

Goods means any "goods," as such term is defined in the Code, now owned or hereafter acquired by any Loan Party, wherever located, including embedded software to the extent included in "goods" as defined in the Code.

Governmental Authority means any nation or government, any state or other political subdivision thereof, and any agency, department or other entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government.

<u>Guaranteed Indebtedness</u> means, as to any Person, any obligation of such Person guaranteeing, providing comfort for or otherwise supporting any Indebtedness, lease, dividend, or other obligation

("primary obligation") of any other Person (the "primary obligor") in any manner, including any obligation or arrangement of such Person to (a) purchase or repurchase any such primary obligation, (b) advance or supply funds (i) for the purchase or payment of any such primary obligation or (ii) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency or any balance sheet condition of the primary obligor, (c) purchase property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation, (d) protect the beneficiary of such arrangement from loss (other than product warranties given in the ordinary course of business) or (e) indemnify the owner of such primary obligation against loss in respect thereof. The amount of any Guaranteed Indebtedness at any time shall be deemed to be an amount equal to the lesser at such time of (x) the stated or determinable amount of the primary obligation in respect of which such Guaranteed Indebtedness is incurred and (y) the maximum amount for which such Person may be liable pursuant to the terms of the instrument embodying such Guaranteed Indebtedness, or, if not stated or determinable, the maximum reasonably anticipated liability (assuming full performance) in respect thereof.

<u>Guaranties</u> means, collectively, the Holdings Guaranty, each Subsidiary Guaranty and any other guaranty executed by any Guarantor in favor of Agent and Lenders in respect of the Obligations.

<u>Guarantors</u> means Holdings, Intermediate Holdings, each Domestic Subsidiary of Borrower and each other Person, if any, that executes a guaranty or other similar agreement in favor of Agent, for itself and the ratable benefit of Lenders, in connection with the transactions contemplated by the Agreement and the other Loan Documents.

<u>Hazardous Material</u> means any substance, material or waste that is regulated by, or forms the basis of liability now or hereafter under, any Environmental Laws, including any material or substance that is (a) defined as a "hazardous waste," "hazardous material," "hazardous substance," "dangerous goods," "extremely hazardous waste," "restricted hazardous waste," "pollutant," "contaminant," "hazardous constituent," "special waste," "toxic substance," "medical waste," "biohazardous" or other similar term or phrase under any Environmental Laws, or (b) petroleum or any fraction or by-product thereof, asbestos, polychlorinated biphenyls (PCB's), or any radioactive substance.

Health Care Laws means (i) all federal and state fraud and abuse laws, including, but not limited to, the Federal Anti-Kickback Statute (42 U.S.C. §1320a-7(b)), the Stark Law (42 U.S.C. §1395nn), the False Claims Act (31 U.S.C. §3729 et seq.), Sections 1320a-7 and 1320a-7a of Title 42 of the United States Code, and the regulations promulgated pursuant to such statutes; (ii) HIPAA; (iii) Medicare and the Medicare Regulations; (iv) Medicaid and the Medicaid Regulations; (v) TRICARE and the TRICARE regulations; (vi) CHAMPVA and the CHAMPVA regulations; (vii) the Clinical Laboratory Improvement Amendments of 1999 (42 U.S.C. §263a et seq.); (viii) quality, safety and accreditation standards and requirements of all applicable federal and state laws or regulatory bodies; (ix) licensure, registration and approval laws and regulations; (x) all laws and regulations governing billing for health care items and services; (xi) any and all other applicable health care laws, regulations, manual provisions, policies and administrative guidance; and (xii) each of (i) through (xi) as may be amended from time to time.

HHS means the United States Department of Health and Human Services or any successor thereto.

<u>HIPAA</u> means the privacy transactions and security provisions of the Health Insurance Portability and Accountability Act of 1996 (Pub. L. No. 104-191) and the regulations promulgated thereunder, as amended from time to time.

Holdings has the meaning ascribed thereto in the recitals to the Agreement.

Holdings Guaranty means the guaranty of even date herewith executed by each of Holdings and Intermediate Holdings in favor of Agent, on behalf of itself and Lenders.

Holdings Pledge Agreement means the Pledge Agreement of even date herewith executed by Holdings and Intermediate Holdings in favor of Agent, on behalf of itself and Lenders, pledging (i) all Stock of their Domestic Subsidiaries and 65% of the voting Stock of their first tier Foreign Subsidiaries, (ii) all stock of Intermediate Holdings and (iii) all stock of Borrower.

<u>Holdings Security Agreement</u> means the Holdings Security Agreement of even date herewith entered into by and among Agent, on behalf of itself and Lenders, Holdings and Intermediate Holdings.

Holdings Stockholder Agreement means that certain Stockholders' Agreement dated as of September 19, 2006 by and among Holdings, the investors signatory thereto and the management stockholders signatory thereto.

Indebtedness means, with respect to any Person, without duplication (a) all indebtedness of such Person for borrowed money or for the deferred purchase price of property payment for which is deferred six (6) months or more, but excluding obligations to trade creditors incurred in the ordinary course of business that are unsecured and not overdue by more than six (6) months unless being contested in good faith, (b) all reimbursement and other obligations with respect to letters of credit, bankers' acceptances and surety bonds, whether or not matured, (c) all obligations evidenced by notes, bonds, debentures or similar instruments, (d) all indebtedness created or arising under any conditional sale or other title retention agreement with respect to property acquired by such Person (even though the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such property), (e) all Capital Lease Obligations and the present value (discounted at the Index Rate as in effect on the date of determination) of future rental payments under all synthetic leases, (f) all obligations in respect of payments that would be required to be made upon termination of any hedging arrangements, (g) all net payment obligations of such Person under any foreign exchange contract, currency swap agreement, interest rate swap (including Interest Rate Agreements), cap or collar agreement or other similar agreement or arrangement designed to alter the risks of that Person arising from fluctuations in currency values or interest rates, in each case whether contingent or matured and any and all Rate Management Obligations, (h) all Indebtedness referred to above secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien upon or in property or other assets (including accounts and general intangibles) owned by such Person, even though such Person has not assumed or become liable for the payment of such Indebtedness, (i) "earnouts" (to the extent treated as Indebtedness under GAAP) and similar payment obligations excluding bonus, phantom stock or other similar compensation payments owed to employees, or officers and incurred in the ordinary course of business, and (j) the Obligations. Notwithstanding any of the foregoing, non-recourse Indebtedness of such Person shall only constitute "Indebtedness" in an amount which shall be the lesser of (x) the amount of such Person's liability for such Indebtedness and (y) the fair market value of such property securing such Indebtedness. In no event shall any obligations in respect of preferred stock issued on the Closing Date constitute Indebtedness except to the extent of any mandatory obligation to redeem such preferred stock on a date that is earlier than the date six months following the fifth anniversary of the Closing Date. In no event shall the Acquisition Note constitute Indebtedness so long as it is paid in full on or prior to its maturity date; provided that if such Acquisition Note is not paid in full as of its maturity date, such Acquisition Note shall constitute Indebtedness for all purposes hereunder (except for purposes of calculating Excess Cash Flow).

<u>Indemnitees</u> has the meaning ascribed to it in <u>Section 10.1</u>.

Index Rate means, for any day, a floating rate equal to the higher of (i) the rate publicly quoted from time to time by The Wall Street Journal and (ii) the Federal Funds Rate plus 50 basis points per annum.

Index Rate Loan means a Loan or portion thereof bearing interest by reference to the Index Rate.

<u>Instruments</u> means all "instruments," as such term is defined in the Code, now owned or hereafter acquired by any Loan Party, wherever located, and, in any event, including all promissory notes and other evidences of indebtedness, other than instruments that constitute, or are a part of a group of writings that constitute, Chattel Paper.

Intellectual Property means any and all Licenses, Patents, Copyrights, Trademarks, and the goodwill associated with such Trademarks.

Intercompany Debt has the meaning ascribed to it in Section 10.21.

<u>Intercompany Notes</u> means a promissory note contemplated by <u>Section 6.1(c)</u>.

Interest Expense means for any measuring period interest expense (whether cash or non-cash) determined in accordance with GAAP and deducted in the calculation of Consolidated Net Income, including capitalized interest expense, less the sum of (i) amortization of capitalized fees and expenses with respect to Related Transactions for such period, (ii) amortization of any original issue discount attributable to Funded Debt or warrants for such period, and (iii) interest paid inkind during such period.

Interest Payment Date means (a) as to any Index Rate Loan, the first Business Day of each calendar month to occur while such Loan is outstanding, and (b) as to any LIBOR Loan, the last day of the applicable LIBOR Period; provided, that in the case of any LIBOR Period greater than three months in duration, the last day of each three month interval within such LIBOR Period and the last day of such LIBOR Period shall be an Interest Payment Date; and provided further that, in addition to the foregoing, each of (x) the date upon which all of the Commitments have been terminated and the Loans have been paid in full and (y) the Commitment Termination Date shall be deemed to be an "Interest Payment Date" with respect to any interest that has then accrued under the Agreement.

Interest Rate Agreement means any interest rate swap agreement, interest rate cap agreement, interest rate collar agreement or similar agreement or arrangement designed to protect Borrower against fluctuations in interest rates entered into between Borrower and any Lender to the extent required by Section 5.10.

<u>Intermediate Holdings</u> has the meaning ascribed thereto in the recitals to the Agreement.

Inventory means any "inventory," as such term is defined in the Code, now owned or hereafter acquired by any Loan Party, wherever located.

Investment means (i) any direct or indirect purchase or other acquisition by Borrower or any of its Subsidiaries of any Stock, or other ownership interest in, any other Person, and (ii) any direct or indirect loan, advance or capital contribution by Borrower or any of its Subsidiaries to any other Person excluding Accounts and deposits arising in the ordinary course of business. For purposes of covenant compliance, the amount of any Investment shall be the aggregate cash Investment less all cash returns and cash distributions received by such Person.

<u>Investment Property</u> means all "investment property," as such term is defined in the Code, now owned or hereafter acquired by any Loan Party, wherever located.

IRC means the Internal Revenue Code of 1986, as amended, and all regulations promulgated thereunder.

IRS means the United States Internal Revenue Service.

<u>L/C Issuer</u> means Fifth Third Bank or a Subsidiary thereof or a bank or other legally authorized Person selected by or acceptable to Agent and Borrower, in such Person's capacity as an issuer of Letters of Credit hereunder.

L/C Sublimit has the meaning ascribed to it in Section 2.1(c).

Lenders means Freeport, the other lenders named on the signature pages of the Agreement (and, if any such Person shall decide to assign all or any portion of the Obligations, such term shall include any assignee of such Person); provided, that for the purposes of (i) the definitions of "Interest Rate Agreement" and "Obligations", (ii) Sections 9.2(a), (b), (c), (d) and (f) of this Agreement and (iii) the granting and perfection of Liens pursuant to one or more Loan Documents, each Qualified Counterparty shall be deemed to be a Lender (it being agreed that no such Qualified Counterparty shall have the right to vote on or consent to any matter requiring a vote or consent of one or more Lenders).

<u>Letters of Credit</u> means documentary or standby letters of credit issued for the account of Borrower by L/C Issuer, and bankers' acceptances issued by Borrower, for which Agent and Lenders have incurred Letter of Credit Obligations.

Letter of Credit Fee has the meaning ascribed to it in Section 2.3(c).

<u>Letter of Credit Obligations</u> means all outstanding obligations incurred by Agent and Lenders at the request of Borrower, whether direct or indirect, contingent or otherwise, due or not due, in connection with the issuance of Letters of Credit by L/C Issuers or the purchase of a participation or issuance of a guaranty as set forth in <u>Section 2.1(d)</u> with respect to any Letter of Credit. The amount of such Letter of Credit Obligations shall equal the maximum amount that may be payable by Agent and Lenders thereupon or pursuant thereto.

<u>Leverage Ratio</u> means for any measuring period the ratio of (x)(i) Funded Debt (including Letter of Credit Obligations, but otherwise excluding the Revolving Loan balance) as of the last day of such measuring period <u>plus</u> (ii) the Average Daily Balance for the 60 days prior to such date to (y) EBITDA.

<u>Liberty Letter of Credit</u> means the letter of credit, dated February 24, 2006, in the maximum amount of \$2,300,000, number ILS201407, issued by Fifth Third Bank for the benefit of Liberty Mutual Insurance Company in respect of workers compensation claims from and after January 1, 2006.

LIBOR Breakage Costs means an amount equal to the amount of any losses, expenses, liabilities (including, without limitation, any net loss or expense incurred by reason of the liquidation or redeployment of deposits or other funds acquired by a Lender to fund or maintain any LIBOR Loan) sustained by a Lender as a result of (i) any default by Borrower in making any borrowing of, conversion

into or continuation of any LIBOR Loan following Borrower's delivery to Agent of any LIBOR Loan request in respect thereof or (ii) any payment of a LIBOR Loan on any day that is not the last day of the LIBOR Period applicable thereto (regardless of the source of such prepayment and whether voluntary, by acceleration or otherwise). For purposes of calculating amounts payable to a Lender under Section 2.3(d), each Lender shall be deemed to have actually funded its relevant LIBOR Loan through the purchase of a deposit bearing interest at the LIBOR Rate in an amount equal to the amount of that LIBOR Loan and having a maturity and repricing characteristics comparable to the relevant LIBOR Period; provided, however, that each Lender may fund each of its LIBOR Loans in any manner it sees fit, and the foregoing assumption shall be utilized only for the calculation of amounts payable under Section 2.3(d).

<u>LIBOR Business Day</u> means a Business Day on which banks in the City of London are generally open for interbank or foreign exchange transactions.

<u>LIBOR Loans</u> means a Loan or any portion thereof bearing interest by reference to the LIBOR Rate.

<u>LIBOR Period</u> means, with respect to any LIBOR Loan, each period commencing on a LIBOR Business Day selected by Borrower pursuant to this Agreement and ending one, two, three or six months thereafter, as selected by Borrower's irrevocable notice to Agent as set forth in <u>Section 2.2(e)</u>; <u>provided</u>, that the foregoing provision relating to LIBOR Periods is subject to the following:

- (a) if any LIBOR Period would otherwise end on a day that is not a LIBOR Business Day, such LIBOR Period shall be extended to the next succeeding LIBOR Business Day unless the result of such extension would be to carry such LIBOR Period into another calendar month in which event such LIBOR Period shall end on the immediately preceding LIBOR Business Day;
- (b) any LIBOR Period that would otherwise extend beyond the date set forth in clause (a) of the definition of "Commitment Termination Date" shall end on such date;
- (c) any LIBOR Period that begins on the last LIBOR Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such LIBOR Period) shall end on the last LIBOR Business Day of a calendar month; and
  - (d) Borrower shall select LIBOR Periods so that there shall be no more than five (5) separate LIBOR Periods in existence at any one time.

#### LIBOR Rate means for each LIBOR Period, a rate of interest determined by Agent equal to:

- (a) the offered rate for deposits in United States Dollars for the applicable LIBOR Period that appears on Telerate Page 3750 as of 11:00 a.m. (London time), on the second full LIBOR Business Day next preceding the first day of such LIBOR Period (unless such date is not a Business Day, in which event the next succeeding Business Day will be used); divided by
- (b) a number equal to 1.0 <u>minus</u> the aggregate (but without duplication) of the rates (expressed as a decimal fraction) of reserve requirements in effect on the day that is two (2) LIBOR Business Days prior to the beginning of such LIBOR Period (including basic, supplemental, marginal and emergency reserves under any regulations of the Federal Reserve Board or other Governmental Authority having jurisdiction with respect thereto, as now and from time to time in effect) for Eurocurrency funding (currently referred to as "Eurocurrency Liabilities" in Regulation D of the Federal Reserve Board that are required to be maintained by a member bank of the Federal Reserve System.

If such interest rates shall cease to be available from Telerate News Service, the LIBOR Rate shall be determined from such internationally recognized financial reporting service or other information as shall be available to Agent.

<u>License</u> means any Copyright License, Patent License, Trademark License or other license of rights or interests now held or hereafter acquired by any Loan Party.

<u>Lien</u> means any mortgage or deed of trust, pledge, hypothecation, assignment, deposit arrangement, lien, charge, claim, security interest, easement or encumbrance, or preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever (including any lease or title retention agreement, any financing lease having substantially the same economic effect as any of the foregoing, and the filing of, or agreement to give, any financing statement perfecting a security interest under the Code or comparable law of any jurisdiction).

<u>Litigation</u> has the meaning ascribed to it in <u>Section 7.2(k)</u>.

Loan Account has the meaning ascribed to it in Section 2.7.

Loan Documents means the Agreement, the Notes, the Collateral Documents, the Freeport Fee Letter, the Subordination Agreement, the Interest Rate Agreements, and all other agreements, instruments, documents and certificates identified in the Closing Checklist executed and delivered to, or in favor of, Agent or any Lenders and including all other pledges, powers of attorney, consents, assignments, contracts, notices, and all other written matter whether heretofore, now or hereafter executed by or on behalf of any Loan Party, or any employee of any Loan Party, and delivered to Agent or any Lender in connection with the Agreement or the transactions contemplated thereby. Any reference in the Agreement or any other Loan Document to a Loan Document shall include all appendices, exhibits or schedules thereto, and all amendments, restatements, supplements or other modifications thereto, and shall refer to the Agreement or such Loan Document as the same may be in effect at any and all times such reference becomes operative.

Loan Parties means Holdings, Intermediate Holdings, Merger Sub, the Company, Borrower and each other Person (i) that executes this Agreement as a "Loan Party," (ii) that executes a Guaranty, (iii) that grants a Lien on all or substantially all of its assets to secure payment of the Obligations and (iv) all of the Stock of which is pledged to Agent for the benefit of itself and Lenders.

Loans means the Revolving Loan and the Term Loan.

Management Consulting Agreement means that certain Management Consulting Agreement between Borrower and Sponsor as in effect on the Closing Date.

Material Adverse Effect means a material adverse effect on (a) the business, assets, operations, industry, properties, prospects or financial or other condition of the Loan Parties considered as a whole, (b) Borrower's ability to pay any of the Loans or any of the other Obligations in accordance with the terms of the Agreement, (c) the Collateral or Agent's Liens, on behalf of itself and Lenders, on the Collateral or the priority of such Liens, or (d) Agent's or any Lender's rights and remedies under the Agreement and the other Loan Documents.

Maximum Amount means, as of any date of determination, an amount equal to the Revolving Loan Commitment of all Lenders as of that date.

Maximum Lawful Rate has the meaning ascribed to it in Section 2.2(f).

McKesson Add-Back means an amount equal to the expenses booked by Borrower and its Subsidiaries with respect to McKesson system implementation services in an amount not to exceed \$600,000 in the aggregate.

Medicaid means, collectively, the certain program of medical assistance, funded jointly by the federal government and states, for impoverished individuals who are aged, blind and/or disabled, and for members of families with dependent children, which program is more fully described in Title XIX of the Social Security Act (42 U.S.C. §§ 1396 et seq.), and the Medicaid Regulations.

Medicaid Regulations means, collectively, (i) all federal statutes (whether set forth in Title XIX of the Social Security Act or elsewhere) affecting Medicaid, (ii) all applicable provisions of all federal rules, regulations, manuals and orders of all Governmental Authorities promulgated pursuant to or in connection with the statutes described in clause (i) above and all federal administrative, reimbursement and other guidelines of all Governmental Authorities having the force of law promulgated pursuant to or in connection with the statutes described in clause (i) above, (iii) all state statutes and plans for medical assistance enacted in connection with the statutes and provisions described in clauses (i) and (ii) above and (iv) all applicable provisions of all rules, regulations, manuals and orders of all Governmental Authorities promulgated pursuant to or in connection with the statutes described in clause (iii) above and all state administrative, reimbursement and other guidelines of all Governmental Authorities having the force of law promulgated pursuant to or in connection with the statutes described in clause (iii) above, in each case as may be amended, supplemented or otherwise modified from time to time.

Medicare means, collectively, the certain federal program providing health insurance for eligible elderly and other individuals, under which physicians, hospitals, skilled nursing homes, home health care and other providers are reimbursed for certain covered services they provide to the beneficiaries of such program, which program is more fully described in Title XVIII of the Social Security Act (42 U.S.C. §§ 1395 *et seq.*), and the Medicare Regulations.

Medicare Regulations means, collectively, all federal statutes (whether set forth in Title XVIII of the Social Security Act or elsewhere) affecting Medicare and any successor statute(s), together with all applicable provisions of all rules, regulations, manuals and orders and administrative, reimbursement and other guidelines of all Governmental Authorities (including, without limitation, HHS, CMS, the Office of the Inspector General for HHS, or any other Person succeeding to the functions of any of the foregoing) promulgated pursuant to or in connection with any of the foregoing having the force of law, in each case as may be amended, supplemented or otherwise modified from time to time.

Merger has the meaning ascribed to such term in the Preamble.

Merger Agreement means that certain Agreement and Plan of Merger dated as of the Closing Date among Merger Sub and the Company as in effect on the date hereof and as modified in accordance with the terms of this Agreement.

Merger Documents means the Merger Agreement and all documents, instruments and agreements delivered in connection with the Merger Agreement.

Merger Sub has the meaning ascribed to such term in the Preamble.

Minimum EBITDA means, for any period of determination, EBITDA for the 12 Fiscal-Month period then ended.

Moody's means Moody's Investors Services, Inc.

Multiemployer Plan means a "multiemployer plan" as defined in Section 4001(a)(3) of ERISA, and to which any Loan Party or ERISA Affiliate is making or is obligated to make contributions on behalf of participants who are or were employed by any of them or withdrawal liability payments.

Net Proceeds means (i) cash proceeds received by Holdings or any of its Subsidiaries from any Asset Disposition (including insurance proceeds, awards of condemnation, and payments under notes or other debt securities received in connection with any Asset Disposition), net of (a) the costs of such Asset Disposition (including taxes attributable to such sale, lease or transfer) and any commissions and other customary transaction fees, costs and expenses), other than any costs payable to any Affiliate of a Loan Party (b) amounts applied to repayment of Indebtedness (other than the Obligations) secured by a Lien permitted under the Agreement on the asset or property disposed, (c) any amounts required to be held in escrow until such time as such amounts are released from escrow whereupon such amounts shall be considered Net Proceeds and (d) amounts reasonably and in good faith provided as a reserve in accordance with GAAP, against (x) any liabilities under any indemnification obligations associated with such Asset Disposition or (y) any other liabilities retained by such Loan Party associated with the properties disposed of in such Asset Disposition (provided that, to the extent and at the time any such amounts are released from such reserve, such amounts shall constitute Net Proceeds), and (ii) cash proceeds attributable to any working capital, earnings, balance sheet or similar adjustment under the Acquisition Agreement.

Non-Consenting Lender has the meaning ascribed to it in Section 10.19(c).

Non-Excluded Taxes has the meaning ascribed to it in Section 2.9(a).

Non-Funding Lender has the meaning ascribed to it in Section 9.5(a).

Notes means, collectively, the Revolving Notes and the Term Notes.

Notice of Conversion/Continuation has the meaning ascribed to it in Section 2.2(e).

Notice of Revolving Credit Advance has the meaning ascribed to it in Section 2.1(b).

Obligations means all loans, advances, debts, liabilities and obligations, for the performance of covenants, tasks or duties or for payment of monetary amounts (whether or not such performance is then required or contingent, or such amounts are liquidated or determinable), including obligations pursuant to Interest Rate Agreements, Rate Management Agreements and Letter of Credit Obligations, owing by any Loan Party to Agent or any Lender, and all covenants and duties regarding such amounts, of any kind or nature, present or future, whether or not evidenced by any note, agreement or other instrument, arising under the Agreement or any of the other Loan Documents. This term includes all principal, interest (including all interest that accrues after the commencement of any case or proceeding by or against any Loan Party in bankruptcy, whether or not allowed in such case or proceeding), Fees, Charges, expenses, attorneys' fees and any other sum chargeable to any Loan Party under the Agreement or any of the other Loan Documents.

Operating Cash Flow has the meaning ascribed to it in Section 7.1(e) of Schedule 1 to Annex E.

Other Lender has the meaning ascribed to it in Section 9.5(d).

Other Taxes has the meaning ascribed to it in Section 2.9(a).

Overadvance has the meaning ascribed to it in Section 2.1(b).

<u>Participation Agreements</u> means all participation and provider agreements with health maintenance organizations, insurance programs, Third Party Payors and preferred provider organizations with respect to the business of the Loan Parties.

<u>Patent License</u> means rights under any written agreement now owned or hereafter acquired by any Loan Party granting any right to such Loan Party with respect to a Patent owned by a third party.

Patent Security Agreements means the Patent Security Agreements made in favor of Agent, on behalf of itself and Lenders, by each applicable Loan Party.

<u>Patents</u> means all of the following in which any Loan Party now holds or hereafter acquires any interest: (a) all letters patent of the United States or any other country, all issuances and recordings thereof, and all applications for letters patent of the United States or of any other country, including issued patents, recordings and applications in the United States Patent and Trademark Office or in any similar office or agency of the United States or any other country, and (b) all reissues, continuations, continuations-in-part or extensions thereof.

<u>PBGC</u> means the Pension Benefit Guaranty Corporation.

Pension Plan means a Plan described in Section 3(2) of ERISA.

Permitted Encumbrances means the following encumbrances: (a) Liens for taxes, assessments or governmental charges or levies not yet due and payable or Liens for taxes, assessments or governmental charges or levies being contested in good faith and by appropriate proceedings for which adequate reserves have been established in accordance with GAAP, excluding federal income tax Liens and Liens in favor of the PBGC under ERISA; (b) Liens in respect of property or assets of the Borrower or any of its Subsidiaries imposed by law which were incurred in the ordinary course of business and which have not arisen to secure Indebtedness for borrowed money, such as carriers', materialmen's, warehousemen's and mechanics' Liens, statutory and common law landlord's Liens, and other similar Liens arising in the ordinary course of business, and which either (1) do not in the aggregate materially detract from the value of such property or assets or materially impair the use thereof in the operation of the business of the Borrower or any of its Subsidiaries or (2) are being contested in good faith by appropriate proceedings, which proceedings have the effect of preventing the forfeiture or sale of the property or asset subject to such Lien; (c) Liens created by or pursuant to this Agreement, the Collateral Documents or the other Loan Documents; (d) Liens in existence on the Closing Date which are listed, and the property subject thereto described, on Schedule 6.2, without giving effect to any extensions or renewals thereof; (e) Liens arising from judgments, decrees, awards or attachments in circumstances not constituting an Event of Default, provided that the amount of cash and property (determined on a fair market value basis) deposited or delivered to secure the respective judgment or decree or subject to attachment shall not exceed the limit for a separate judgment in Section 8.1(h); (f) Liens (other than any Lien imposed by ERISA) (1) incurred or deposits made in the ordinary course of business of the

Borrower and its Subsidiaries in connection with workers' compensation, unemployment insurance and other types of social security, (3) to secure the performance by the Borrower and its Subsidiaries of tenders, statutory obligations (other than excise taxes), surety, stay, customs and appeal bonds, statutory bonds, bids, leases, government contracts, trade contracts, performance and return of money bonds and other similar obligations (exclusive of obligations for the payment of borrowed money) to the extent incurred in the ordinary course of business, (4) to secure the performance by the Borrower and its Subsidiaries of leases of real property, to the extent incurred or made in the ordinary course of business consistent with past practices, and (5) other deposits made solely in the ordinary course of the Loan Parties' business; (g) licenses, sublicenses, leases or subleases granted to third Persons in the ordinary course of business not interfering in any material respect with the business of the Borrower or any of its Subsidiaries; (h) easements, rights-of-way, restrictions, minor defects or irregularities in title, encroachments and other similar charges or encumbrances, in each case not securing Indebtedness and not interfering in any material respect with the ordinary conduct of the business of the Borrower or any of its Subsidiaries; (i) Liens arising from precautionary UCC financing statements regarding operating leases; (j) Liens created pursuant to or in connection with leases or Capital Leases permitted pursuant to this Agreement, provided that (1) such Liens only serve to secure the payment of rent or Indebtedness arising under such leases or Capital Leases and (2) the Liens encumbering the assets leased or purported to be leased under such leases or Capital Leases do not encumber any other assets of the Borrower or any of its Subsidiaries (other than letters of credit, payment undertaking agreements, guaranteed investment contracts, deposits of cash or Cash Equivalents and other credit support arrangements, in each case having an aggregate value not exceeding the fair market value of the assets leased or purported to be leased under such leases or Capital Leases (each of such values determined at the time when the lease agreement relating to the relevant lease or Capital Lease is signed and delivered)); (k) (1) those liens, encumbrances, hypothecs and other matters affecting title to any real property and found reasonably acceptable by the Agent or insured against by title insurance, (2) as to any particular real property at any time, such easements, encroachments, covenants, rights of way, minor defects, irregularities or encumbrances on title which would not reasonably be expected to materially impair such real property for the purpose for which it is held by the mortgagor or grantor thereof, or the lien or hypothec held by the Agent, (3) zoning and other municipal ordinances which are not violated in any material respect by the existing improvements and the present use made by the mortgagor or grantor thereof of the premises, (4) general real estate taxes and assessments not yet delinquent, (5) any Lien that would be disclosed on a true, correct and complete survey of the real property that does not materially affect the use or enjoyment of the real property as it is currently being used, and (6) such other similar items as the Agent may consent to (such consent not to be unreasonably withheld); (1) Liens arising pursuant to purchase money mortgages or security interests securing Indebtedness representing the purchase price (or financing of the purchase price within 90 days after the respective purchase) of fixed assets acquired after the Closing Date, provided that (1) any such Liens attach only to the assets so purchased, upgrades thereon and, if the asset so purchased is an upgrade, the original asset itself (and such other assets financed by the same financing source), (2) the Indebtedness (other than Indebtedness incurred from the same financing source to purchase other assets and excluding Indebtedness representing obligations to pay installation and delivery charges for the property so purchased) secured by any such Lien does not exceed 100% of the lesser of the fair market value or the purchase price of the property being purchased at the time of the incurrence of such Indebtedness and (3) the Indebtedness secured thereby is permitted to be incurred pursuant to this Agreement; (m) Liens arising out of consignment or similar arrangements for the sale of goods entered into by the Borrower or any of its Subsidiaries in the ordinary course of business; (n) rights of setoff upon deposits of cash in favor of banks or other depository institutions as permitted by any Control Agreement or, with respect to deposits of cash not subject to a Control Agreement, customary rights of setoff in favor of such banks or depository institutions; (o) Liens on insurance premiums in favor of insurers and pledged to finance the payment thereof; and (p) Liens securing Indebtedness or leases that refinance, refund, extend, renew and/or replace Indebtedness or leases secured by Liens described in clauses (a) through (o) above, as long as such Indebtedness is permitted hereunder.

<u>Person</u> means any individual, sole proprietorship, partnership, joint venture, trust, unincorporated organization, association, corporation, limited liability company, institution, public benefit corporation, other entity or government (whether federal, state, county, city, municipal, local, foreign, or otherwise, including any instrumentality, division, agency, body or department thereof).

<u>Plan</u> means, at any time, an "employee benefit plan," as defined in Section 3(3) of ERISA, that any Loan Party or ERISA Affiliate maintains, contributes to or has an obligation to contribute to on behalf of participants who are or were employed by any Loan Party.

<u>Pledge Agreements</u> means the Borrower Pledge Agreement, the Holdings Pledge Agreement and any other pledge agreement entered into after the Closing Date by any Loan Party.

Prior Lender means Fifth Third Bank (Chicago).

<u>Prior Lender Obligations</u> means that certain Loan and Security Agreement dated as of November 22, 2005 by and among Borrower, the other borrowers party thereto and the Prior Lender.

<u>Pro Forma</u> means the unaudited consolidated and consolidating balance sheets of Holdings and its Subsidiaries prepared in accordance with GAAP as of the Closing Date after giving effect to the Related Transactions. The Pro Forma is annexed hereto as <u>Annex C</u>.

Pro Rata Share means with respect to all matters relating to any Lender (a) with respect to the Revolving Loan, the percentage obtained by dividing (i) the Revolving Loan Commitment of that Lender by (ii) the aggregate Revolving Loan Commitments of all Lenders, (b) with respect to the Term Loan, the percentage obtained by dividing (i) the Term Loan Commitment of that Lender by (ii) the aggregate Term Loan Commitments of all Lenders, (c) with respect to all Loans, the percentage obtained by dividing (i) the aggregate Commitments of that Lender by (ii) the aggregate Commitments of all Lenders, and (d) with respect to all Loans on and after the Commitment Termination Date, the percentage obtained by dividing (i) the aggregate outstanding principal balance of the Loans held by that Lender, by (ii) the outstanding principal balance of the Loans held by all Lenders, as any such percentages may be adjusted by assignments pursuant to Section 9.1.

<u>Proceeding</u> means a proceeding under the United States Bankruptcy Code, Insolvency Laws or any similar law in any jurisdiction, in which any Loan Party or any Subsidiary thereof is a debtor.

<u>Projections</u> means Borrower's and its Subsidiaries' forecasted consolidated and consolidating: (a) balance sheets; (b) profit and loss statements; (c) cash flow statements; and (d) capitalization statements, all prepared on a Subsidiary by Subsidiary or division-by-division basis, if applicable, and otherwise consistent with the historical Financial Statements of Borrower and its Subsidiaries, together with appropriate supporting details and a statement of underlying assumptions.

Proposed Change has the meaning ascribed to it in Section 10.19(c).

<u>Purchase Agreement</u> means that certain Stock Purchase Agreement dated as of September 19, 2006, by and among Holdings, Intermediate Holdings, Merger Sub, the Company, W. Andrew Wright, III, as sellers' representative and the other parties identified as sellers therein (collectively, the "Sellers").

Qualified Assignee means (a) any Lender, any Affiliate of any Lender and, with respect to any Lender, any investment fund that invests in commercial loans and that is managed or advised by such Lender, an Affiliate of such Lender or the same investment advisor as such Lender or by an Affiliate of such investment advisor (an "Approved Fund"), and (b) any commercial bank, savings and loan association or savings bank or any other entity which is an "accredited investor" (as defined in Regulation D under the Securities Act of 1933) which extends credit or buys loans as one of its businesses, including insurance companies, mutual funds, lease financing companies and commercial finance companies, which has a rating of BBB or higher from S&P and a rating of Baa2 or higher from Moody's at the date that it becomes a Lender and, in either case, which, through its applicable lending office, is capable of lending to Borrower without the imposition of any withholding or similar taxes.

Qualified Counterparty means a Person that is a party to an Interest Rate Agreement for the benefit of a Loan Party which (i) is an Affiliate of a Lender and (ii) has entered into an agreement, in form and substance to the Agent, pursuant to which such Person has, among other things, appointed the Agent as its agent and agreed to be bound by certain provisions of the Loan Documents.

Qualified Plan means a Pension Plan that is intended to be tax-qualified under Section 401(a) of the IRC.

Rate Management Agreement means any agreement, device or arrangement providing for payments which are related to fluctuations of interest rates, exchange rates, forward rates, or equity prices, including, but not limited to, dollar-denominated or cross-currency interest rate exchange agreements, forward currency exchange agreements, interest rate cap or collar protection agreements, forward rate currency or interest rate options, puts and warrants, and any agreement pertaining to equity derivative transactions (e.g., equity or equity index swaps, options, caps, floors, collars and forwards), including without limitation any ISDA Master Agreement between Borrower and Fifth Third Bank or any affiliate of Fifth Third Bank, and any schedules, confirmations and documents and other confirming evidence between the parties confirming transactions thereunder, all whether now existing or hereafter arising, and in each case as amended, modified or supplemented from time to time.

Rate Management Obligations means any and all obligations of Borrower to Fifth Third Bank or any affiliate of Fifth Third Bank, whether absolute, contingent or otherwise and howsoever and whensoever (whether now or hereafter) created, arising, evidenced or acquired (including all renewals, extensions and modifications thereof and substitutions therefore), under or in connection with (i) any and all Rate Management Agreements, and (ii) any and all cancellations, buybacks, reversals, terminations or assignments of any Rate Management Agreement.

Real Estate has the meaning ascribed to it in Section 4.14.

Refinancing means the payment in full by Borrower of the Prior Lender Obligations on the Closing Date.

Related Transactions means the Addus Acquisition, the Merger, the initial borrowing under the Revolving Loan and the Term Loan on the Closing Date, the Refinancing, the payment of all Fees, costs and expenses associated with all of the foregoing and the execution and delivery of all of the Related Transactions Documents, including, without limitation (i) the creation of five escrow accounts under the Escrow Agreement and the deposit of cash in each such escrow account in the amount contemplated by the Escrow Agreement as in effect on the date hereof and (ii) the release by Fifth Third Bank of all of its liens and security interests in all property of the Loan Parties, other than the security interest that Fifth Third Bank may have or acquire in the Restricted Cash Escrow Funds (as such term is defined in the

Escrow Agreement as in effect on the date hereof) and the release by Fifth Third Bank of the Company (except to the extent of its interest in such Restricted Cash Escrow Funds) and its Subsidiaries for their respective liability in respect of all letters of credit (other than the Liberty Letter of Credit) issued by Fifth Third Bank and constituting Prior Lender Obligations.

Related Transactions Documents means the Loan Documents, the Merger Documents, the Escrow Agreement, the Contingent Payment Agreement, the Purchase Agreement, the Acquisition Note, the Holdings Stockholder Agreement, the Securities Purchase Agreement, the Restated Certificate and all other agreements or instruments executed in connection with the Related Transactions.

Release means any release, threatened release, spill, emission, leaking, pumping, pouring, emitting, emptying, escape, injection, deposit, disposal, discharge, dispersal, dumping, leaching or migration of Hazardous Material in the indoor or outdoor environment, including the movement of Hazardous Material through or in the air, soil, surface water, ground water or property.

Replacement Lender has the meaning ascribed to it in Section 10.19(a).

<u>Requisite Lenders</u> means Lenders having (a) more than 50% of the Commitments of all Lenders, or (b) if the Commitments have been terminated, more than 50% of the aggregate outstanding amount of the Loans.

Requisite Revolving Lenders means Lenders having (a) more than 50% of the Revolving Loan Commitments of all Lenders, or (b) if the Revolving Loan Commitments have been terminated, more than 50% of the aggregate outstanding amount of the Revolving Loan.

Responsible Officer means, with respect to any Loan Party, any officer appointed by the Board of Directors.

Restated Certificate means the amended and restated certificate of incorporation of Holdings.

Restricted Payment means, with respect to any Loan Party (a) the declaration or payment of any dividend or the incurrence of any liability to make any other payment or distribution of cash or other property or assets in respect of Stock; (b) any payment on account of the purchase, redemption, defeasance, sinking fund or other retirement of such Loan Party's Stock or any other payment or distribution made in respect thereof, either directly or indirectly; (c) any payment made to redeem, purchase, repurchase or retire, or to obtain the surrender of, any outstanding warrants, options or other rights to acquire Stock of such Loan Party now or hereafter outstanding; (d) any payment of a claim for the rescission of the purchase or sale of, or for material damages arising from the purchase or sale of, any shares of such Loan Party's Stock or of a claim for reimbursement, indemnification or contribution arising out of or related to any such claim for damages or rescission; and (e) any payment of a Contingent Payment.

Revolving Credit Advance has the meaning ascribed to it in Section 2.1(b).

Revolving Lenders means those Lenders having a Revolving Loan Commitment, or, following the termination of the Revolving Loan Commitment, those Lenders holding any portion of the Revolving Loan.

Revolving Loan(s) means, at any time, the sum of (i) the aggregate amount of Revolving Credit Advances outstanding to Borrower <u>plus</u> (ii) the aggregate Letter of Credit Obligations incurred on behalf of Borrower. Unless the context otherwise requires, references to the outstanding principal balance of the Revolving Loan shall include the outstanding balance of Letter of Credit Obligations.

Revolving Loan Commitment means (a) as to any Lender, the commitment of such Lender to make its Pro Rata Share of Revolving Credit Advances or incur its Pro Rata Share of Letter of Credit Obligations as set forth on Annex A or in the most recent Assignment Agreement, if any, executed by such Lender and (b) as to all Lenders, the aggregate commitment of all Lenders to make the Revolving Credit Advances or incur Letter of Credit Obligations, which aggregate commitment shall be Twelve Million Five Hundred Dollars (\$12,500,000) on the Closing Date, as such amount may be adjusted, if at all, from time to time in accordance with the Agreement.

Revolving Notes has the meaning ascribed to it in Section 2.1(b).

S&P means Standard & Poor's Ratings Services, a division of the McGraw-Hill Companies, Inc.

Scheduled Installments has the meaning ascribed to it in Section 2.1(a).

Securities Purchase Agreement means that certain Securities Purchase Agreement to be entered into on September 19, 2006 by and among Holdings, Eos Capital Partners III, L.P., Eos Partners SBIC III, L.P. and Freeport Financial Loan Fund LLC.

Security Agreement means each of the Borrower Security Agreement, the Holdings Security Agreement and the Subsidiary Security Agreement.

Settlement Date has the meaning ascribed to it in Section 9.5(a)(ii).

<u>Software</u> means all "software" as such term is defined in the Code, now owned or hereafter acquired by any Loan Party, other than software embedded in any category of Goods, including all computer programs and all supporting information provided in connection with a transaction related to any program.

Solvent means, with respect to any Person on a particular date, that on such date (a) the fair value of the property of such Person is greater than the total amount of liabilities, including subordinated and contingent liabilities, of such Person; (b) the present fair saleable value of the assets of such Person is not less than the amount that will be required to pay the probable liability of such Person on its debts and liabilities, including subordinated and contingent liabilities as they become absolute and matured; (c) such Person does not intend to, and does not believe that it will, incur debts or liabilities beyond such Person's ability to pay as such debts and liabilities mature; and (d) such Person is not engaged in a business or transaction, and is not about to engage in a business or transaction, for which such Person's property would constitute an unreasonably small capital. The amount of contingent liabilities (such as Litigation, guaranties and pension plan liabilities) at any time shall be computed as the amount that, in light of all the facts and circumstances existing at the time, represents the amount that would reasonably be expected to become an actual or matured liability.

Sponsor means, collectively, Eos Partners SBIC III, L.P., a Delaware limited partnership, EOS Capital Partners III, L.P., a Delaware limited partnership and EOS Capital Partners, L.P., a Delaware limited partnership.

Statement has the meaning ascribed to it in Section 7.2(c).

Stock means all shares, options, warrants, general or limited partnership interests, membership interests or other equivalents (regardless of how designated) of or in a corporation, partnership, limited liability company or equivalent entity whether voting or nonvoting, including common stock, preferred stock or any other "equity security" (as such term is defined in Rule 3a11-1 of the General Rules and Regulations promulgated by the Securities and Exchange Commission under the Securities Exchange Act of 1934).

Stockholder means, with respect to any Person, each holder of Stock of such Person.

<u>Subordination Agreement</u> means the Subordination and Intercreditor Agreement, dated as of September 19, 2006 by and among the Company, Holdings, Merger Sub, Intermediate Holdings, the Agent, W. Andrew Wright, III, Addus Term Trust, W. Andrew Wright Grantor Retained Annuity Trust, Mark S. Heaney, James A. Wright and Courtney E. Panzer.

Subsidiary means, with respect to any Person, (a) any corporation of which an aggregate of more than 50% of the outstanding Stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether, at the time, Stock of any other class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at the time, directly or indirectly, owned legally or beneficially by such Person or one or more Subsidiaries of such Person, or with respect to which any such Person has the right to vote or designate the vote of more than 50% of such Stock whether by proxy, agreement, operation of law or otherwise, and (b) any partnership or limited liability company in which such Person and/or one or more Subsidiaries of such Person shall have an interest (whether in the form of voting or participation in profits or capital contribution) of more than 50% or of which any such Person is a general partner or may exercise the powers of a general partner. Unless the context otherwise requires, each reference to a Subsidiary shall be a reference to a Subsidiary of the Borrower.

<u>Subsidiary Guaranty</u> means the Subsidiary Guaranty of even date herewith executed by one or more Subsidiaries of Borrower in favor of Agent, on behalf of itself and Lenders.

<u>Subsidiary Security Agreement</u> means the Subsidiary Security Agreement of even date herewith entered into by and among Agent, on behalf of itself and Lenders, and one or more Subsidiaries of Borrower.

<u>Supermajority Revolving Lenders</u> means Lenders having (a) 80% or more of the Revolving Loan Commitments of all Lenders, or (b) if the Revolving Loan Commitments have been terminated, 80% or more of the aggregate outstanding amount of the Revolving Loan.

Tax Returns means all reports, returns, information returns, claims for refund, elections, estimated Tax filings or payments, requests for extension, documents, statements, declarations and certifications and other information required to be filed with respect to Taxes, including attachments thereto and amendments thereof.

Taxes has the meaning ascribed to it in Section 2.9(a).

Termination Date means the date on which (a) the Loans have been indefeasibly repaid in full, (b) all other Obligations under the Agreement and the other Loan Documents have been completely discharged (other than contingent indemnification obligations as to which no unsatisfied claim has been asserted), (c) all Letter of Credit Obligations have been cash collateralized in the amount set forth in Section 2.5(f), cancelled or, with the consent of Agent in each instance, backed by standby letters of credit acceptable to Agent, (d) all Commitments have been terminated and (e) Agent and Lenders have been released by Loan Parties of all claims against Agent and Lenders.

Term Lenders means those Lenders having Term Loan Commitments.

Term Loans has the meaning ascribed to it in Section 2.1 (a).

Term Notes has the meaning ascribed to it in Section 2.1 (a).

Term Loan Commitment means (a) as to any Lender, the commitment of such Lender to make its Pro Rata Share of the Term Loans (as set forth on Annex A) in the maximum aggregate amount set forth in Section 2.1(a) or in the most recent Assignment Agreement, if any, executed by such Lender and (b) as to all Lenders, the aggregate commitment of all Lenders to make the Term Loans. The Term Loan Commitment shall reduce automatically by the amount prepaid or repaid in respect of the Term Loan (but solely by the amount of such prepayment or repayment allocable to a Lender, for purposes of clause (a) of this definition). Immediately prior to the making of the Term Loan, the Term Loan Commitments of the Lenders aggregate \$45,000,000.

Third Party Payor means Medicare, Medicaid, TRICARE, CHAMPVA, Blue Cross and/or Blue Shield, state government insurers, and private insurers and any other Person which presently or in the future maintains Third Party Payor Programs.

Third Party Payor Programs means all third party payor programs in which any Loan Party participates (including, without limitation, Medicare, Medicaid, TRICARE, CHAMPVA, or any other federal or state health care programs, as well as Blue Cross and/or Blue Shield, managed care plans, and any other private insurance programs).

Title IV Plan means a "pension plan" as defined in Section 3(2) of ERISA (other than a Multiemployer Plan), that is covered by Title IV of ERISA or Section 412 of the IRC, and that any Loan Party or ERISA Affiliate maintains, contributes to or has an obligation to contribute to on behalf of participants who are or were employed by any of them.

<u>Trademark Security Agreements</u> means the Trademark Security Agreements made in favor of Agent, on behalf of itself and Lenders, by each applicable Loan Party.

<u>Trademark License</u> means rights under any written agreement now owned or hereafter acquired by any Loan Party granting any right to such Loan Party to use any Trademark owned by a third party.

Trademarks means all of the following now owned or hereafter adopted or acquired by any Loan Party: (a) all trademarks, trade names, corporate names, business names, trade styles, service marks, logos, internet domain names, other source or business identifiers, prints and labels on which any of the foregoing have appeared or appear and designs (whether registered or unregistered), all registrations and recordings thereof, and all applications in connection therewith, including registrations, recordings and applications in the United States Patent and Trademark Office or in any similar office or agency of the United States, any state or territory thereof, or any other country or any political subdivision thereof; (b) all reissues, extensions or renewals thereof; and (c) all goodwill associated with or symbolized by any of the foregoing.

TRICARE means the program of medical benefits covering former and active members of the uniformed services and certain of their dependents, formally known as CHAMPUS, financed and

administered by the United States Departments of Defense, Health and Human Services and Transportation, and all laws, rules, regulations, manuals, orders, guidelines or requirements pertaining to such program, including, without limitation, (i) all federal statutes (whether set forth in 10 U.S.C. §§ 1071 - 1106 or elsewhere) affecting such program and (ii) all rules, regulations (including 32 C.F.R. § 199), manuals, orders and administrative, reimbursement and other guidelines of all Governmental Authorities promulgated in connection with such program (whether or not having the force of law), in each case as the same may be amended, supplemented or otherwise modified from time to time.

Voting Stock means Stock having ordinary voting power to elect the board of directors (or similar body) of such Person.

1.2 <u>Rules of Construction</u>. Rules of construction with respect to accounting terms used in the Agreement or the other Loan Documents shall be as set forth or referred to in this <u>Section 1.2</u>. All other undefined terms contained in any of the Loan Documents shall, unless the context indicates otherwise, have the meanings provided for by the Code to the extent the same are used or defined therein; in the event that any term is defined differently in different Articles or Divisions of the Code, the definition contained in Article or Division 9 shall control. Unless otherwise specified, references in the Agreement or any of the Appendices to a Section, subsection or clause refer to such Section, subsection or clause as contained in the Agreement. The words "herein," "hereof" and "hereunder" and other words of similar import refer to the Agreement as a whole, including all Annexes, Exhibits and Schedules, as the same may from time to time be amended, restated, modified or supplemented, and not to any particular section, subsection or clause contained in the Agreement or any such Annex, Exhibit or Schedule.

Wherever from the context it appears appropriate, each term stated in either the singular or plural shall include the singular and the plural, and pronouns stated in the masculine, feminine or neuter gender shall include the masculine, feminine and neuter genders. The words "including", "includes" and "include" shall be deemed to be followed by the words "without limitation"; the word "or" is not exclusive; references to Persons include their respective successors and assigns (to the extent and only to the extent permitted by the Loan Documents) or, in the case of governmental Persons, Persons succeeding to the relevant functions of such Persons; and all references to statutes and related regulations shall include any amendments of the same and any successor statutes and regulations. Whenever any provision in any Loan Document refers to the knowledge (or an analogous phrase) of any Loan Party, such words are intended to signify that such Loan Party has actual knowledge or awareness of a particular fact or circumstance or, as applicable, actual knowledge of the existence of any Default, Event of Default or other event that has had a Material Adverse Effect. Definitions of agreements and instruments in Section 1 shall mean and refer to such agreements and instruments as amended, modified, supplemented, restated, substituted or replaced from time to time in accordance with their respective terms and the terms of this Agreement and the other Loan Documents.

### SECTION 2. AMOUNTS AND TERMS OF LOANS

- 2.1 Loans. Subject to the terms and conditions of this Agreement and in reliance upon the representations and warranties of Borrower and the other Loan Parties contained herein:
- (a) <u>Term Loans</u>. Each Term Lender agrees, severally and not jointly, to lend to Borrower in one draw, on the Closing Date, its Pro Rata Share of \$45,000,000 (the "<u>Term Loans</u>").

Borrower shall repay the Term Loans through periodic payments on the dates and in the amounts indicated below ("Scheduled Installments"), and in any event the entire remaining principal balance shall be repaid on the Commitment Termination Date.

### Term Loan

<u>Date</u>	Scheduled Installment
December 31, 2006	\$ 700,000
March 31, 2007	\$ 700,000
June 30, 2007	\$ 700,000
September 30, 2007	\$ 700,000
December 31, 2007	\$ 1,050,000
March 31, 2008	\$ 1,050,000
June 30, 2008	\$ 1,050,000
September 30, 2008	\$ 1,050,000
December 31, 2008	\$ 1,400,000
March 31, 2009	\$ 1,400,000
June 30, 2009	\$ 1,400,000
September 30, 2009	\$ 1,400,000
December 31, 2009	\$ 1,750,000
March 31, 2010	\$ 1,750,000
June 30, 2010	\$ 1,750,000
September 30, 2010	\$ 1,750,000
December 31, 2010	\$ 2,100,000
March 31, 2011	\$ 2,100,000
June 30, 2011	\$ 2,100,000
September 19, 2011	\$ 19,100,000

The final installment shall in all events equal the entire remaining principal balance of the Term Loan. Amounts borrowed under this <u>Section 2.1(a)</u> and repaid may not be reborrowed.

At the request of the applicable Lender, the Term Loans shall be evidenced by promissory notes substantially in the form of <a href="Exhibit 2.1(a)"><u>Exhibit 2.1(a)</u></a> (as amended, modified, extended, substituted or replaced from time to time, each a "<u>Term Note</u>" and, collectively, the "<u>Term Notes</u>"), and Borrower shall execute and deliver a Term Note to each such Term Lender. Each Term Note shall represent the obligation of Borrower to pay the amount of the applicable Term Lender's Term Loan Commitment, together with interest thereon.

#### (b) Revolving Loans.

(i) Each Revolving Lender agrees, severally and not jointly, to make available to Borrower from time to time until the Commitment Termination Date its Pro Rata Share of advances (each a "Revolving Credit Advance") requested by Borrower hereunder. The Pro Rata Share of the Revolving Loan of any Revolving Lender shall not at any time exceed its separate Revolving Loan Commitment. Revolving Credit Advances may be repaid and reborrowed; provided, that the amount of any Revolving Credit Advance to be made at any time shall not exceed Borrowing Availability. The Revolving Loans shall be repaid in full on the Commitment Termination Date. If requested by a Revolving Lender, Borrower shall execute and deliver to such Revolving Lender a note to evidence the Revolving Loan Commitment of that Revolving Lender. Each such note shall be in the maximum principal amount of the Revolving Loan Commitment of the applicable Revolving Lender, dated the Closing Date and substantially in the form of Exhibit 2.1(b)(i) (as amended, modified, extended, substituted or replaced from time to time, each a "Revolving Note" and, collectively, the "Revolving Notes"). Other than pursuant to Section 2.1(b)(ii), if at any time the aggregate outstanding Revolving Loan exceeds Borrowing Availability (without giving effect to any deduction therefrom for the outstanding Revolving Loans and Funded Debt, to the extent consisting of Revolving Loans) (any such excess Revolving Loan is herein referred to as an "Overadvance"), Lenders shall not be obligated to make Revolving Credit Advances, no additional Letters of Credit shall be issued and, except as provided in Section 2.1(b)(ii) below, the Revolving Loan must be repaid immediately and Letters of Credit cash collateralized in an amount sufficient to eliminate any Overadvance. All Overadvances shall constitute Index Rate Loans and at the written request of Agent or the Requisite Lenders shall bear interest at the Index Rate plus the Applicable Revolver Index Margin and shall bear interest at the Default Rate only if not repaid within three (3) Business Days. For funding requests for Revolving Credit Advances to be funded as Index Rate Loans of \$5,000,000 or less, written notice must be provided by noon (Chicago time) on the Business Day on which the Revolving Credit Advance is to be made; for funding requests for Revolving Credit Advances to be funded as Index Rate Loans of more than \$5,000,000, written notice must be provided by noon (Chicago time) on the Business Day before which the Revolving Credit Advance is to be made. All Revolving Credit Advances to be funded as LIBOR Loans require three (3) Business Days' prior written notice. Written notices for all funding requests shall be in the form attached as Exhibit 2.1(b)(ii) ("Notice of Revolving Credit Advance").

(ii) If Borrower requests that Revolving Lenders make, or permit to remain outstanding an Overadvance, Agent may, in its sole discretion, elect to make, or permit to remain outstanding such Overadvances; provided, however, that Agent may not cause Revolving Lenders to make, or permit to remain outstanding, (a) a Revolving Loan balance in excess of the Maximum Amount or (b) an Overadvance in an aggregate amount in excess of \$500,000. If an Overadvance is made, or permitted to remain outstanding, pursuant to the preceding sentence, then all Revolving Lenders shall be bound to make, or permit to remain outstanding such Overadvance based upon their Pro Rata Shares of the Revolving Loan Commitment in accordance with the terms of this Agreement. If an Overadvance

remains outstanding for more than ninety (90) days during any one hundred eighty (180) day period, the Revolving Loans must be repaid immediately in an amount sufficient to eliminate all of such Overadvances. Furthermore, holders of a majority of the Revolving Loan Commitment may prospectively revoke Agent's ability to make or permit Overadvances by written notice to Agent.

- (c) <u>Letters of Credit</u>. The Revolving Loan Commitment may, in addition to advances under the Revolving Loan, be utilized (subject to the limitations imposed by Section 2.1(b)), upon the request of Borrower, for the issuance of Letters of Credit. Immediately upon the issuance by an L/C Issuer of a Letter of Credit, and without further action on the part of Agent or any of the Lenders, each Revolving Lender shall be deemed to have purchased from such L/C Issuer a participation in such Letter of Credit (or in its obligation under a risk participation agreement with respect thereto) equal to such Revolving Lender's Pro Rata Share of the aggregate amount available to be drawn under such Letter of Credit. For purposes of clarification, the Liberty Letter of Credit shall be deemed to be a Letter of Credit for all purposes hereunder.
- (i) <u>Maximum Amount</u>. The aggregate amount of Letter of Credit Obligations with respect to all Letters of Credit outstanding or unreimbursed at any time shall not exceed \$8,000,000 ("<u>L/C Sublimit</u>").
- (ii) Reimbursement. Borrower shall be irrevocably and unconditionally obligated forthwith without presentment, demand, protest or other formalities of any kind, to reimburse any L/C Issuer on demand in immediately available funds for any amounts paid by such L/C Issuer with respect to a Letter of Credit, including all reimbursement payments, Fees, Charges, costs and expenses paid by such L/C Issuer, without duplication of fees otherwise paid by Borrower. Borrower hereby authorizes and directs Agent, at Agent's option, to debit Borrower's account (by increasing the outstanding principal balance of the Revolving Credit Advances made to Borrower) in the amount of any payment made by an L/C Issuer with respect to any Letter of Credit, and a notice of Revolving Credit Advance requesting an Index Rate Loan in such amount shall be deemed to have been timely given on such date. All amounts paid by an L/C Issuer with respect to any Letter of Credit that are not immediately repaid by Borrower with the proceeds of a Revolving Credit Advance or otherwise shall bear interest payable on demand at the interest rate applicable to Revolving Credit Advances which are Index Rate Loans plus, at the election of Agent or Requisite Revolving Lenders, an additional two percent (2.00%) per annum. Each Revolving Lender agrees to fund its Pro Rata Share of any Revolving Loan made pursuant to this Section 2.1(c)(ii). In the event Agent elects not to debit Borrower's account and Borrower fails to reimburse the L/C Issuer in full on the date of any payment in respect of a Letter of Credit, Agent shall promptly notify each Revolving Lender of the amount of such unreimbursed payment and the accrued interest thereon and each Revolving Lender, on the next Business Day prior to 2:00 p.m. (Chicago time), shall deliver to Agent an amount equal to its Pro Rata Share thereof in same day funds. Each Revolving Lender hereby absolutely and unconditionally agrees to pay to the L/C Issuer upon demand by the L/C Issuer such Revolving Lender's Pro Rata Share of each payment made by the L/C Issuer in respect of a Letter of Credit and reimbursed within one (1) Business Day by Borrower or satisfied through a debit of Borrower's account. Each Revolving Lender acknowledges and agrees that its obligations pursuant to this subsection in respect of Letters of Credit are absolute and unconditional and shall not be affected by any circumstance whatsoever, including setoff, counterclaim, the occurrence and continuance of a Default or an Event of Default or any failure by Borrower to satisfy any of the conditions set forth in Section 3.2. If any Revolving Lender fails to make available to the L/C Issuer the amount of such Revolving Lender's Pro Rata Share of any payments made by the L/C Issuer in respect of a Letter of Credit as provided in this Section 2.1(e)(ii), the L/C Issuer shall be entitled to recover such amount on demand from such Revolving Lender together with interest at the Index Rate.

(iii) Request for Letters of Credit. Borrower shall give Agent at least three (3) Business Days' prior written notice specifying the date a Letter of Credit is requested to be issued, the amount and the name and address of the beneficiary and a description of the transactions proposed to be supported thereby, and the expiry date (or extended expiry date or the increased amount) of the Letter of Credit. Each request by Borrower for the issuance, extension or increase (each, an "issuance") of a Letter of Credit shall be in the form of Exhibit 2.1(c). If Agent informs Borrower that the L/C Issuer cannot issue the requested Letter of Credit directly, Borrower may request that L/C Issuer arrange for the issuance of the requested Letter of Credit under a guaranty or risk participation agreement with another financial institution reasonably acceptable to Agent, L/C Issuer and Borrower. The issuance of any Letter of Credit under this Agreement shall be subject to satisfaction of the conditions set forth in Section 3.2 and the conditions that the Letter of Credit (i) supports a transaction benefiting Borrower or its wholly-owned Subsidiaries and (ii) is in a form, is for an amount and contains such terms and conditions as are reasonably satisfactory to the L/C Issuer and, in the case of standby letters of credit, Agent. The initial notice requesting the issuance of a Letter of Credit shall be accompanied by the form of the Letter of Credit and the Master Standby Agreement or Master Documentary Agreement, as applicable, and an application or reimbursement agreement is inconsistent with the terms of this Agreement, then the provisions of this Agreement, to the extent of such inconsistency, shall control.

(iv) Expiration Dates of Letters of Credit. The expiration date of each Letter of Credit shall be on a date which is not later than the earlier of (a) one year from its date of issuance or (b) the date set forth in clause (a) of the definition of the term Commitment Termination Date. Notwithstanding the foregoing, a Letter of Credit may provide for automatic extensions of its expiration date for one (1) or more successive one (1) year periods provided that upon not less than 60 days' written notice to Borrower, the L/C Issuer has the right to terminate such Letter of Credit on each such annual expiration date and no renewal term may extend the term of the Letter of Credit to a date that is later than the date set forth in clause (a) of the definition of the term Commitment Termination Date. The L/C Issuer may elect not to renew any such Letter of Credit and, upon direction by Agent or Requisite Revolving Lenders, shall not renew any such Letter of Credit at any time during the continuance of an Event of Default, provided that, in the case of a direction by Agent or Requisite Revolving Lenders, the L/C Issuer receives such directions prior to the date notice of non-renewal is required to be given by the L/C Issuer and the L/C Issuer has had a reasonable period of time to act on such notice.

(v) Obligations Absolute. The obligation of Borrower to reimburse the L/C Issuer, Agent and Lenders for payments made in respect of Letters of Credit issued by the L/C Issuer shall be unconditional and irrevocable and shall be paid under all circumstances strictly in accordance with the terms of this Agreement, including the following circumstances: (a) any lack of validity or enforceability of any Letter of Credit; (b) any amendment or waiver of or any consent or departure from all or any of the provisions of any Letter of Credit or any Loan Document; (c) the existence of any claim, set-off, defense or other right which Borrower, any of its Subsidiaries or Affiliates or any other Person may at any time have against any beneficiary of any Letter of Credit, Agent, any L/C Issuer, any Lender or any other Person, whether in connection with this Agreement, any other Loan Document or any other related or unrelated agreements or transactions; (d) any draft or other document presented under any Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; (e) payment under any Letter of Credit against presentation of a draft or other document that does not substantially comply with the terms of such Letter of Credit; or (f) any other act or omission to act or delay of any kind of any L/C Issuer, Agent, any Lender or any other Person or any other event or circumstance whatsoever that might, but for the provisions of this Section  $2.1(\varepsilon)(v)$ , constitute a legal or equitable discharge of Borrower's obligations hereunder, in each case, other than

gross negligence or willful misconduct on the part of the L/C Issuer and other than the payment by the L/C Issuer of a draft drawn under a Letter of Credit that on its face does not substantially comply with the terms of the Letter of Credit. Without limiting the generality of the foregoing, it is expressly understood and agreed by Borrower that the absolute and unconditional obligation of Borrower to Agent and Lenders hereunder to reimburse payments made under a Letter of Credit will not be excused by the gross negligence or willful misconduct of the L/C Issuer. However, the foregoing shall not be construed to excuse an L/C Issuer from claims which Borrower may assert against the L/C Issuer subject to the terms of the Master Standby Agreement or the Master Documentary Agreement.

(vi) Obligations of L/C Issuers. Each L/C Issuer (other than Freeport) hereby agrees that it will not issue a Letter of Credit hereunder until it has provided Agent with written notice specifying the amount and intended issuance date of such Letter of Credit and Agent has returned a written acknowledgment of such notice to L/C Issuer (such notices and acknowledgments to be given promptly). Each L/C Issuer (other than Freeport) further agrees to provide to Agent: (a) a copy of each Letter of Credit issued by such L/C Issuer promptly after its issuance; (b) a monthly report summarizing available amounts under Letters of Credit issued by such L/C Issuer, the dates and amounts of any draws under such Letters of Credit, the effective date of any increase or decrease in the face amount of any Letters of Credit during such week and the amount of any unreimbursed draws under such Letters of Credit; and (c) such additional information reasonably requested by Agent from time to time with respect to the Letters of Credit issued by such L/C Issuer.

(d) <u>Funding Authorization</u>. The proceeds of all Loans made pursuant to this Agreement subsequent to the Closing Date are to be funded by Agent by wire transfer to the account designated by Borrower below<sup>1</sup> (the "Disbursement Account"):

Bank: Fifth Third Bank
ABA No.: 042-000-314
Account No.: 072-312-606-34
Reference: Addus Transaction

Borrower shall provide Agent with written notice of any change in the foregoing instructions at least three (3) Business Days before the desired effective date of such change.

#### 2.2 Interest and Applicable Margins.

(a) Borrower shall pay interest to Agent, for the ratable benefit of Lenders with respect to the various Loans made by each Lender, in arrears on each applicable Interest Payment Date, at the following rates: (i) with respect to the Revolving Loans which are designated as Index Rate Loans (and for all other Obligations not otherwise set forth below), the Index Rate plus the Applicable Revolver Index Margin per annum or, with respect to Revolving Loans which are designated as LIBOR Loans, at the election of Borrower, the applicable LIBOR Rate plus the Applicable Revolver LIBOR Margin per annum; and (ii) with respect to such portion of the Term Loans designated as an Index Rate Loan, the Index Rate plus the Applicable Term Loan Index Margin per annum or, with respect to such portion of the Term Loans designated as a LIBOR Loan, the applicable LIBOR Rate plus the Applicable Term Loan LIBOR Margin per annum.

The Applicable Margins shall be as follows:

Applicable Revolver Index Margin	2.75%
Applicable Revolver LIBOR Margin	3.75%
Applicable Term Loan Index Margin	2.75%
Applicable Term Loan LIBOR Margin	3.75%

Borrower to supply account information.

provided; however, the Applicable Margins, with respect to the Term Loan, shall be adjusted (up or down) prospectively on a quarterly basis as determined by Holdings' and its Subsidiaries' consolidated financial performance. Adjustments in Applicable Margins will be determined by reference to the following grids:

Level of Applicable Margin Level I	Leverage Ratio  3 4.00 to 1.00	Applicable Term Loan Index Margin 3.25%	Applicable Term Loan LIBOR Margin 4.25%
Level II	<sup>3</sup> 2.50 to 1.00, and < 4.00 to 1.00	2.75%	3.75%
Level III	< 2.50 to 1.00	2.25%	3.25%

All adjustments in the Applicable Margins shall be implemented quarterly on a prospective basis, five (5) Business Days after the date of delivery to Lenders of the quarterly unaudited Financial Statements evidencing the need for an adjustment. Concurrently with the delivery of those Financial Statements, Borrower shall deliver to Agent and Lenders a certificate, signed by its chief financial officer, setting forth in reasonable detail the basis for the continuance of, or any change in, the Applicable Margins. If any Default or an Event of Default has occurred and is continuing at the time any reduction in the Applicable Margins is to be implemented, that reduction shall be deferred until the first day of the first calendar month following the date on which all Defaults or Events of Default are waived or cured.

- (b) If any payment on any Loan becomes due and payable on a day other than a Business Day, the maturity thereof will be extended to the next succeeding Business Day (except as set forth in the definition of LIBOR Period) and, with respect to payments of principal, interest thereon shall be payable at the then applicable rate during such extension.
- (c) All computations of Fees calculated on a per annum basis and interest shall be made by Agent on the basis of a 360-day year, in each case for the actual number of days occurring in the period for which such Fees and interest are payable. The Index Rate is a floating rate determined for each day. Each determination by Agent of an interest rate and Fees hereunder shall be presumptive evidence of the correctness of such rates and Fees.
- (d) So long as an Event of Default has occurred and is continuing under Section 8.1(a), (f) or (g) and without notice of any kind, or so long as any other Event of Default has occurred and is continuing and at the election of Agent (or upon the written request of Requisite Lenders) confirmed by written notice from Agent to Borrower, the interest rates applicable to the Loans and the Letter of Credit Fee shall be increased by two percentage points (2%) per annum above the rates of interest or the rate of such Fee otherwise applicable hereunder ("Default Rate"), and the outstanding principal balance of the

Loans shall bear interest at the Default Rate applicable to such Obligations. Interest and Letter of Credit Fees at the Default Rate shall accrue from the initial date of such Event of Default until that Event of Default is cured or waived and shall be payable upon demand, but in any event, shall be payable on the next regularly scheduled payment date set forth herein for such Obligation.

- (e) After the earlier of sixty days following the Closing Date and the completion of the primary syndication of the credit facility, Borrower shall have the option to (i) request that any Revolving Credit Advance be made as a LIBOR Loan, (ii) convert at any time all or any part of outstanding Loans from Index Rate Loans to LIBOR Loans, (iii) convert any LIBOR Loan to an Index Rate Loan, subject to payment of the LIBOR Breakage Costs in accordance with Section 2.3(d) if such conversion is made prior to the expiration of the LIBOR Period applicable thereto, or (iv) continue all or any portion of any Loan as a LIBOR Loan upon the expiration of the applicable LIBOR Period and the succeeding LIBOR Period of that continued Loan shall commence on the first day after the last day of the LIBOR Period of the Loan to be continued. Any Loan or group of Loans having the same proposed LIBOR Period to be made or continued as, or converted into, a LIBOR Loan must be in a minimum amount of \$1,000,000 and integral multiples of \$500,000 in excess of such amount. Any such election must be made by noon (Chicago time) on the 3rd Business Day prior to (1) the date of any proposed Revolving Credit Advance which is to bear interest at the LIBOR Rate, (2) the end of each LIBOR Period with respect to any LIBOR Loans to be continued as such, or (3) the date on which Borrower wishes to convert any Index Rate Loan to a LIBOR Loan for a LIBOR Period designated by Borrower in such election. If no election is received with respect to a LIBOR Loan by noon (Chicago time) on the 3rd Business Day prior to the end of the LIBOR Period with respect thereto, that LIBOR Loan shall be converted to an Index Rate Loan at the end of its LIBOR Period. Borrower must make such election by notice to Agent in writing, by fax or overnight courier (or by telephone, to be promptly confirmed in writing). In the case of any conversion or continuation, such election must be made pursuant to a written notice (a "Notice of Conversion/Continuation") in the form of Exhib
- (f) Notwithstanding anything to the contrary set forth in this Section 2.2, if a court of competent jurisdiction determines in a final order that the rate of interest payable hereunder exceeds the highest rate of interest permissible under law (the "Maximum Lawful Rate"), then so long as the Maximum Lawful Rate would be so exceeded, the rate of interest payable hereunder shall be equal to the Maximum Lawful Rate; provided, however, that if at any time thereafter the rate of interest payable hereunder is less than the Maximum Lawful Rate, Borrower shall continue to pay interest hereunder at the Maximum Lawful Rate until such time as the total interest received by Agent, on behalf of Lenders, is equal to the total interest that would have been received had the interest rate payable hereunder been (but for the operation of this paragraph) the interest rate payable since the Closing Date as otherwise provided in this Agreement. Thereafter, interest hereunder shall be paid at the rate(s) of interest and in the manner provided in Sections 2.2(a) through (e), unless and until the rate of interest again exceeds the Maximum Lawful Rate, and at that time this paragraph shall again apply. In no event shall the total interest received by any Lender pursuant to the terms hereof exceed the amount that such Lender could lawfully have received had the interest due hereunder been calculated for the full term hereof at the Maximum Lawful Rate. If the Maximum Lawful Rate is calculated pursuant to this paragraph, such interest shall be calculated at a daily rate equal to the Maximum Lawful Rate divided by the number of days in the year in which such calculation is made. If, notwithstanding the provisions of this Section 2.2(f), a court of competent jurisdiction shall determine by a final, non-appealable order that a Lender has received interest hereunder in excess of the Maximum Lawful Rate, Agent shall, to the extent permitted by applicable law, promptly apply such excess as specified in Section 2.5(e) and thereafter shall refund

#### 2.3 Fees.

- (a) Fee Letter. Borrower shall pay to Freeport, individually, the Fees specified in that certain fee letter dated as of July 27, 2006 between Sponsor and Freeport (the "Freeport Fee Letter"), at the times specified for payment therein. Borrower hereby expressly assumes the obligations of Sponsor under the Freeport Fee Letter and Sponsor is hereby released from all obligations under the Freeport Fee Letter.
- (b) <u>Unused Line Fee</u>. As additional compensation for the Revolving Lenders, Borrower shall pay to Agent, for the ratable benefit of such Lenders, in arrears, on the first Business Day of each month prior to the Commitment Termination Date and on the Commitment Termination Date, a fee for each such month or other period for Borrower's non-use of available funds in an amount equal to one-half percent (0.50%) per annum multiplied by the difference between (x) the Maximum Amount (as it may be reduced from time to time) and (y) the sum of (A) the Average Daily Balance for such month or other period plus (B) the average daily amount of undrawn and unreimbursed Letter of Credit Obligations during such month or other period.
- (c) Letter of Credit Fee. Borrower agrees to pay to Agent for the benefit of Revolving Lenders, as compensation to such Revolving Lenders for Letter of Credit Obligations incurred hereunder, (i) without duplication of costs and expenses otherwise payable to Agent or Lenders hereunder, all reasonable costs and expenses, without duplication of fees otherwise paid by Borrower, incurred by Agent or any Lender on account of such Letter of Credit Obligations, and (ii) for each month during which any Letter of Credit Obligation shall remain outstanding, a fee (the "Letter of Credit Fee") in an amount equal to the product of the average daily undrawn face amount of all Letters of Credit issued, guaranteed or supported by risk participation agreements multiplied by a per annum rate equal to the Applicable Revolver LIBOR Margin. Such fee shall be paid to Agent for the benefit of the Revolving Lenders in arrears, on the first Business Day of each month and on the Commitment Termination Date. In addition, Borrower shall pay to any L/C Issuer, on demand, such reasonable fees, without duplication of fees otherwise payable hereunder (including all per annum fees), charges and expenses of such L/C Issuer in respect of the issuance, negotiation, acceptance, amendment, transfer and payment of Such Letter of Credit or otherwise payable pursuant to the application and related documentation under which such Letter of Credit is issued.
- (d) <u>LIBOR Breakage Costs</u>. Upon (i) any default by Borrower in making any borrowing of, conversion into or continuation of any LIBOR Loan following Borrower's delivery to Agent of any LIBOR Loan request in respect thereof or (ii) any payment of a LIBOR Loan on any day that is not the last day of the LIBOR Period applicable thereto (regardless of the source of such prepayment and whether voluntary, by acceleration or otherwise), Borrower shall pay Agent, for the benefit of all Lenders that funded or were prepared to fund any such LIBOR Loan, LIBOR Breakage Costs, if applicable.
- (e) Expenses and Attorneys' Fees. Borrower agrees to pay all reasonable and documented, out-of-pocket fees, charges, costs and expenses (including reasonable attorneys' fees and expenses) incurred by Agent in connection with any matters contemplated by or arising out of the Loan Documents, in connection with the examination, review, due diligence investigation, documentation, negotiation, closing and syndication of the transactions contemplated herein and in connection with the continued administration of the Loan Documents including any amendments, modifications, consents and

waivers. Borrower agrees to promptly pay all reasonable and documented, out-of-pocket fees, charges, costs and expenses (including reasonable fees, charges, costs and expenses of attorneys, auditors, appraisers, consultants and advisors) incurred by Agent in connection with any Field Review (provided that except for any Field Review conducted while an Event of Default is continuing, Borrower shall be obligated to pay for only one Field Review per Fiscal Year), amendment, waiver or consent with respect to the Loan Documents, any Event of Default, work-out or action to enforce any Loan Document or to collect any payments due from Borrower or any other Loan Party. In addition, in connection with any work-out or action to enforce any Loan Document or to collect any payments due and owing from Borrower or any other Loan Party, Borrower agrees to promptly pay all reasonable and documented, out-of-pocket fees, charges, costs and expenses, including, without limitation, reasonable attorneys' fees, incurred by Lenders, provided that all Lenders, other than Agent, shall be entitled to reimbursement hereunder for only one outside counsel. All fees, charges, costs and expenses for which Borrower is responsible under this Section 2.3(e) shall be deemed part of the Obligations when incurred, payable in accordance with the final sentence of Section 2.4 and secured by the Collateral.

2.4 <u>Payments</u>. All payments by Borrower of the Obligations shall be without deduction, defense, setoff or counterclaim (except as otherwise provided in Section 2.9) and shall be made in same day funds and delivered to Agent, for the benefit of Agent and Lenders, as applicable, by wire transfer to the account identified below or such other place as Agent may from time to time designate in writing.

#### Freeport Financial LLC

Bank: US Bank
ABA: 075-000-022
Account #: 182380366340

Account Name: Freeport Financial LLC Reference: Addus HealthCare, Inc.

Borrower shall receive credit on the day of receipt for funds received by Agent by 1:00 p.m. (Chicago time). In the absence of timely receipt, such funds shall be deemed to have been paid on the next Business Day. Whenever any payment to be made hereunder shall be stated to be due on a day that is not a Business Day, the payment may be made on the next succeeding Business Day and such extension of time shall be included in the computation of the amount of interest and Fees due hereunder.

Borrower hereby authorizes Lenders to make Revolving Credit Advances for the payment of Scheduled Installments, interest, Fees and reasonable and documented expenses, Letter of Credit reimbursement obligations and any amounts required to be deposited with respect to outstanding Letter of Credit Obligations pursuant to Sections 2.5(f) or 8.3; provided, that so long as no Event of Default has occurred and is continuing, expense reimbursements pursuant to Section 2.3(e) shall be payable 30 days after notice thereof to Borrower (and otherwise such expense reimbursements shall be payable upon demand).

#### 2.5 Prepayments.

- (a) <u>Voluntary Prepayments of Loans</u>. At any time, Borrower may prepay the Loans, in whole or in part, without premium or penalty subject to the payment of LIBOR Breakage Costs, if applicable. Prepayments of Term Loans shall be applied in accordance with Section 2.5(e).
- (b) <u>Prepayments from Excess Cash Flow</u>. Within one hundred (100) days after the end of each Fiscal Year commencing with the Fiscal Year ended December 31, 2007, Borrower shall prepay the Loans in an amount equal to (i) seventy-five percent (75%) of the Excess Cash Flow for such

Fiscal Year if the Leverage Ratio for such Fiscal Year was greater than or equal to 2.50 to 1.00, or (ii) fifty percent (50%) of the Excess Cash Flow for such Fiscal Year if the Leverage Ratio for such Fiscal Year was less than 2.50 to 1.00. The calculation shall be based on the audited Financial Statements for Holdings and its Subsidiaries. Any prepayments from Excess Cash Flow paid pursuant to this Section 2.5(b) shall be applied in accordance with Section 2.5(e).

- (c) <u>Prepayments from Asset Dispositions</u>. Immediately upon receipt of any Net Proceeds in excess of \$100,000 in the aggregate during any Fiscal Year, Borrower shall prepay the Loans in an amount equal to such Net Proceeds, except that Borrower or its Subsidiaries may reinvest all Net Proceeds of any such Asset Disposition, within one hundred eighty (180) days or irrevocably commit to a third party in writing to reinvest such Net Proceeds within 180 days, in assets usable in the business of the Borrower and its Subsidiaries. If Borrower does not intend to so reinvest such Net Proceeds or if the period set forth in the immediately preceding sentence expires without Borrower having reinvested or irrevocably committed to a third party in writing to reinvest the Net Proceeds of any such Asset Disposition or if such Net Proceeds are attributable to a working capital, earnings, balance sheet or similar adjustment under the Acquisition Agreement, Borrower shall prepay the Loans in an amount equal to such remaining Net Proceeds in accordance with Section 2.5(e).
- (d) <u>Prepayments from Issuance of Securities</u>. Immediately upon the receipt by Holdings, Borrower or any of their Subsidiaries of the cash proceeds of the issuance of Stock or payments under notes or other securities received in connection with the issuance of Stock, Borrower shall prepay the Loans in an amount equal to such cash proceeds or note payments, net of underwriting discounts and commissions and other reasonable out-of-pocket costs associated therewith. The payments shall be applied in accordance with Section 2.5(e). Notwithstanding the foregoing, the following proceeds of Stock issuance shall be excluded from any mandatory prepayment: (i) proceeds of issuances of Stock by Holdings to management of Holdings or its Subsidiaries and (ii) proceeds of issuances of Stock by Holdings, Intermediate Holdings, Merger Sub, the Company or any Subsidiary of the Company on or prior to the Closing Date.
- (e) <u>Application of Proceeds</u>. With respect to any prepayments made by Borrower pursuant to Sections 2.5(b), 2.5(c) and Section 2.5(d) and any payments of the Term Loan pursuant to Section 2.5(a), such prepayments shall be applied as follows: first, in payment of the Term Loan to the next Scheduled Installment and thereafter pro rata against all remaining Scheduled Installments until such Term Loan shall have been prepaid in full; and second, to the Revolving Credit Advances outstanding to Borrower until the same have been repaid in full but not as a permanent reduction of the Revolving Loan Commitment and thereafter to all other Obligations then due and owing. Considering each type of Loan being prepaid separately, any such prepayment shall be applied first to Index Rate Loans of the type required to be prepaid before application to LIBOR Loans of the type required to be prepaid, in each case in a manner which minimizes any resulting LIBOR Breakage Costs.
- (f) <u>Letter of Credit Obligations</u>. In the event any Letters of Credit are outstanding at the time that the Revolving Loan Commitment is terminated, Borrower shall deposit with Agent for the benefit of all Revolving Lenders cash in an amount equal to 105% of the aggregate outstanding Letter of Credit Obligations or an Acceptable Standby Letter of Credit to be available to Agent to reimburse payments of drafts drawn under such Letters of Credit and pay any Fees and expenses related thereto.
- 2.6 <u>Maturity.</u> All of the Obligations shall become due and payable as otherwise set forth herein, but in any event all of the remaining Obligations (other than contingent indemnification obligations as to which no claim has been asserted) shall become due and payable upon the Commitment Termination Date or pursuant to <u>Section 8.3</u>. Until the Termination Date, Agent shall be entitled to retain the Liens on the

Collateral granted under the Collateral Documents and the ability to exercise all rights and remedies available to it under the Loan Documents and applicable laws. Notwithstanding anything contained in this Agreement to the contrary, upon any termination of the Revolving Loan Commitment, all of the Obligations (other than contingent indemnification obligations as to which no claim has been asserted) shall be due and payable.

2.7 Loan Accounts. Agent shall maintain a loan account (the "Loan Account") on its books to record: the name and federal employer identification number of each Lender, all Advances and the Term Loans, all payments made by Borrower, and all other debits and credits as provided in this Agreement with respect to the Loans or any other Obligations. All entries in the Loan Account shall be made in accordance with Agent's customary accounting practices as in effect from time to time. The balance in the Loan Account, as recorded on Agent's most recent printout or other written statement, shall, absent manifest error, be presumptive evidence of the amounts due and owing to Agent and Lenders by Borrower; provided that any failure to so record or any error in so recording shall not limit or otherwise affect (other than to the extent of such error) Borrower's duty to pay the Obligations. Agent shall render to Borrower a monthly accounting of transactions with respect to the Loans setting forth the balance of the Loan Account for the immediately preceding month. Unless Borrower notifies Agent in writing of any objection to any such accounting (specifically describing the basis for such objection), within forty-five (45) days after the date thereof, each and every such accounting shall, absent manifest error, be deemed presumptive evidence of all matters reflected therein. Only those items expressly objected to in such notice shall be deemed to be disputed by Borrower. Notwithstanding any provision herein contained to the contrary, any Lender may elect (which election may be revoked) to dispense with the issuance of Notes to that Lender and may rely on the Loan Account as evidence of the amount of Obligations from time to time owing to it.

#### 2.8 Yield Protection.

(a) <u>Capital Adequacy and Other Adjustments</u>. In the event that any Lender shall have determined that the adoption after the date hereof of any law, treaty, governmental (or quasi-governmental) rule, regulation, guideline or order regarding capital adequacy, reserve requirements or similar requirements or any corporation controlling such Lender with any request or directive regarding capital adequacy, reserve requirements or similar requirements (whether or not having the force of law and whether or not failure to comply therewith would be unlawful) from any central bank or governmental agency or body having jurisdiction does or shall have the effect of increasing the amount of capital, reserves or other funds required to be maintained by such Lender or any corporation controlling such Lender and thereby reducing the rate of return on such Lender's or such corporation's capital as a consequence of its

Commitments or Letter of Credit Obligations hereunder, then Borrower shall from time to time within fifteen (15) days after notice and demand from such Lender (together with the certificate referred to in the next sentence and with a copy to Agent) pay to Agent, for the account of such Lender, additional amounts sufficient to compensate such Lender for such reduction; provided that the respective Lender shall not be entitled to receive additional payments pursuant to this Section 2.8(a) for periods occurring prior to the 180<sup>th</sup> day before the receipt of such notice and demand (provided that this limitation shall not apply to reductions arising out of the retroactive application of any law, treaty, rule, regulation, guideline or order which arises during such 180 day period); and provided further, that such Lender shall not be entitled to any such additional amounts, unless such Lender is imposing similar types of assessments on other similarly situated borrowers. A certificate as to the amount of such cost and showing the basis of the computation of such cost submitted by such Lender to Borrower and Agent shall

such Lender's internal policies of general application, use reasonable commercial efforts to minimize the redirect rate of return, costs and expenses incurred by it and payable to it by Borrower pursuant to this Section 2.8(a).

(b) Increased LIBOR Funding Costs; Illegality. Notwithstanding anything to the contrary contained herein, if the introduction of or any change in any law, rule, regulation, treaty or directive (or any change in the interpretation thereof) after the date hereof shall make it unlawful, or any central bank or other Governmental Authority shall assert that it is unlawful, for any Lender to agree to make or to make or to continue to fund or maintain any LIBOR Loan, then, unless that Lender is able to make or to continue to fund or to maintain such LIBOR Loan at another branch or office of that Lender without, in that Lender's opinion, adversely affecting it or its Loans or the income obtained therefrom, on notice thereof and demand therefor by such Lender to Borrower through Agent, (i) the obligation of such Lender to agree to make or to make or to continue to fund or maintain LIBOR Loans shall terminate and (ii) Borrower shall, at the end of each respective LIBOR Period, repay each outstanding LIBOR Loan of such Lender or convert such LIBOR Loans into Index Rate Loans; provided that if the continued existence of any such LIBOR Loan through the end of its respective LIBOR Period is illegal, then Borrower shall forthwith prepay in full each such outstanding LIBOR Loan (without payment of any LIBOR Breakage Costs) owing by Borrower to such Lender, together with interest accrued thereon, unless Borrower, on behalf of Lender, within five (5) Business Days after the delivery of such notice and demand, converts all LIBOR Loans into Index Rate Loans. If the introduction of, change in or interpretation (in each case made after the date hereof) of any law, rule, regulation, treaty or directive would impose or increase reserve requirements (other than as taken into account in the definition of LIBOR) or otherwise increase the cost to any such Lender of making or maintaining a LIBOR Loan, then Borrower shall from time to time within fifteen (15) days after notice and demand from Agent to Borrower (together with the certificate referred to in the next sentence) pay to Agent, for the account of all such affected Lenders, additional amounts sufficient to compensate such Lenders for such increased cost; provided that Borrower shall not be liable to pay for any such amounts incurred or accrued more than one hundred eighty (180) days prior to the date on which notice of the event giving rise to the obligation to make such payment is given to Borrower (provided that this limitation shall not apply to increased costs arising out of the retroactive application of any law, treaty, rule, regulation or directive (or any change in interpretation thereof) which arises during such 180 day period), provided further, that such Lender shall not be entitled to any such additional amounts unless such Lender is imposing similar types of assessments on other similarly situated borrowers. and provided, further, however, that any such additional amounts shall be without duplication of amounts to which such Lender may be entitled under Section 2.9 and shall exclude Excluded Taxes. A certificate as to the amount of such cost and showing the basis of the computation of such cost submitted by Agent on behalf of all such affected Lenders to Borrower shall be presumptive evidence of the matters set forth therein. Each Lender agrees that, as promptly as is practicable after it becomes aware of any circumstances referred to above which would result in any such increased cost, the affected Lender shall, to the extent not inconsistent with such Lender's internal policies of general application, use reasonable commercial efforts to minimize costs and expenses incurred by it and payable to it by Borrower pursuant to this Section 2.8(b).

(c) <u>Increased Costs</u>. In the event that, after the date hereof (or in the case of a Qualified Assignee or participant, the date of the relevant assignment or sale of a participation), (1) any changes in any existing law, regulation, treaty or directive or in the administration, interpretation or application thereof, (2) any new law, regulation, treaty or directive enacted or any administration, interpretation or application thereof, or (3) compliance with any request, guideline or directive (whether or not having the force of law) from any Governmental Authority does or shall impose on Agent or any Lender any other condition, cost or expense in connection with any LIBOR Loan or Letter of Credit hereunder; and the result of any of the foregoing is to increase the cost to Agent or any such Lender of

issuing or maintaining any Letter of Credit or making or continuing any LIBOR Loan hereunder, as the case may be, or to reduce any amount receivable hereunder or under any other Loan Document, then, in any such case, Borrower shall promptly pay to Agent or such Lender, upon its demand with reasonable documentation thereof, any additional amounts necessary to compensate Agent or such Lender for such additional cost or reduced amount receivable, as reasonably determined by Agent or such Lender; provided that Agent or such Lender shall not be entitled to any such amounts to the extent that the event giving rise to such additional cost or reduced amount receivable occurred more than six (6) months prior to the date such notice and demand is given to the Borrower; provided, however, that if the event giving rise to such additional cost or reduced amount receivable has a retroactive effect, then such 6-month period shall be extended to include the period of such retroactive effect; and provided further, however, that any such additional amounts payable under this Section 2.8(c) shall be without duplication of amounts to which such Lender may be entitled under Section 2.9 and shall exclude Excluded Taxes. If Agent or such Lender becomes aware that it is entitled to claim any additional amounts pursuant to this Section 2.9(b), it shall promptly notify Borrower of the event by reason of which Agent or such Lender has become so entitled. A certificate as to any additional amounts payable pursuant to the foregoing sentence submitted by Agent or such Lender to Borrower (with a copy to Agent if applicable) shall be presumptive evidence of the amount due. Borrower shall pay Agent or such Lender, as the case may be, the amount shown as due on any such certificate within ten (10) days after the receipt thereof.

#### 2.9 Taxes.

(a) No Deductions; Other Taxes. Except as required by law or as otherwise provided in this Section 2.9, any and all payments or reimbursements made hereunder or under any other Loan Documents shall be made free and clear of and without deduction for any and all charges, present or future, taxes, levies, imposts, deductions or withholdings, and all liabilities with respect thereto (including any interest, additions to tax or penalties applicable thereto) of any nature whatsoever imposed by any Governmental Authority ("Taxes"), excluding (a) such Taxes to the extent imposed on or measured by Agent's or a Lender's net income (and franchise taxes, branch profits taxes, taxes on doing business or other taxes imposed in lieu thereof) as a result of a connection between such Agent or Lender and the jurisdiction of the Governmental Authority imposing such Tax or any political subdivision or taxing authority thereof or therein (other than any such connection resulting from such Agent or Lender having executed, delivered, performed its obligations or received a payment under, or enforced, this Agreement or any other Loan Document) and (b) any United States federal withholding tax that is (i) imposed on amounts payable to a Foreign Lender at the time such Foreign Lender becomes a party to this Agreement, except to the extent of any additional amounts to which such Foreign Lender's assignor (if any) was entitled, at the time of assignment, to receive from the Borrower with respect to such withholding tax pursuant to this Section 2.9(a) or (ii) is attributable to such Foreign Lender's failure (other than as a result of a change in law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority) to comply with Section 2.9(c) (collectively, "Excluded Taxes," and all such non-Excluded Taxes being referred to herein as "Non-Excluded Taxes"). If Borrower shall be required by law to deduct any Taxes or Other Taxes from or in respect of any sum payable hereunder or under any other Loan Document to any Lender or Agent, then, to the extent of any Non-Excluded Taxes or Other Taxes, the sum payable hereunder shall be increased as may be necessary so that, after making all required deductions (including deductions applicable to additional sums payable pursuant to this Section 2.9), such Lender or Agent receives an amount equal to the sum it would have received had no such deductions been made. All required deductions shall be withheld and timely paid over to the relevant Governmental Authority in accordance with applicable law. In addition, Borrower agrees to timely pay to the relevant Governmental Authority in accordance with applicable law any present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies which arise from any payment made hereunder or under any other Loan Document or from the execution, delivery, enforcement or registration of, or otherwise with respect to, this Agreement or any other Loan Document ("Other Taxes").

#### (b) Intentionally Omitted

#### (c) Tax Forms.

- (i) Prior to becoming a Lender under this Agreement, after the occurrence of any event requiring a change in the most recent form of certificate previously delivered or within fifteen (15) days after a reasonable written request of Borrower or Agent from time to time, each such Person or Lender that is not in each case a "United States person" (as such term is defined in IRC Section 7701(a)(30)) for U.S. federal income tax purposes (a "Foreign Lender") shall provide to Borrower and Agent, if it is legally entitled to, a properly completed and executed IRS Form W-8BEN or Form W-8ECI or other applicable form, certificate or document prescribed by the IRS (including all required attachments), certifying as to such Foreign Lender's entitlement to an exemption from, or reduction in, United States withholding tax with respect to interest payments to be made to such Foreign Lender under this Agreement and under the Notes (a "Certificate of Exemption"). Any Foreign Lender that is claiming an exemption from U.S. withholding tax under Section 871(h) or 881(c) of the IRC shall provide in addition to the IRS Form W-8BEN a properly executed certificate representing that such Foreign Lender is not a "bank" for purposes of 881(c) of the IRC, is not a ten percent (10%) stockholder of the Borrower within the meaning of Section 871(h)(3)(B) of the IRC, and is not a controlled foreign corporation related to the Borrower within the meaning of Section 864(d)(4) of the IRC. No Foreign Lender shall be entitled to additional payments under Section 2.9(a) or indemnification under Section 2.9(d) for any Non-Excluded Taxes to the extent such Non-Excluded Taxes would have not been imposed had the Foreign Lender complied with this Section 2.9(c).
- (ii) Prior to becoming a Lender, after the occurrence of any event requiring a change in the most recent form or certificate previously delivered, or within 15 days after a reasonable written request of the Borrower or Agent from time to time, each such Lender or Person that is a "United States person" (as such term is defined in IRC Section 7701(a)(30) for U.S. tax purposes, other than any such Lender that is an "exempt recipient" under IRC Section 1.6049-4(c)(1) shall provide to Borrower and Agent, if it is legally entitled to, a properly completed and executed IRS form W-9 (or any successor form), certifying as to such Lender's entitlement to an exemption from U.S. backup withholding tax.
- (d) <u>Indemnification</u>. Borrower will indemnify each Lender and Agent for the full amount of Non-Excluded Taxes and Other Taxes (including any Non-Excluded Taxes and Other Taxes imposed by any jurisdiction on amounts payable under this Section 2.9) paid by such Lender or Agent, as the case may be, and any liability (including penalties, interest and expenses including reasonable attorney's fees and expenses) arising therefrom or with respect thereto, whether or not such Non-Excluded Taxes or Other Taxes were correctly or legally asserted by the relevant Governmental Authority. A certificate as to the amount of such payments or liabilities submitted by Lender or Agent to Borrower (with a copy to Agent if applicable) shall be presumptive evidence of the amount due. Borrower shall pay Agent or such Lender, as the case may be, the amount shown as due on any such certificate within ten (10) day s after the receipt thereof.
- (e) Evidence of Payments. As soon as practicable after any payment of Non-Excluded Taxes or Other Taxes by the Borrower to a Governmental Authority, the Borrower shall deliver to the Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Agent.

- (f) <u>Change in Lending Office</u>. Any Lender claiming any additional amounts payable pursuant to this Section 2.9 shall use its reasonable efforts (consistent with its internal policies and applicable law) to change the jurisdiction of its lending office if such a change would reduce any such additional amounts (or any similar amount that may thereafter accrue) and would not, in the good faith determination of such Lender, be otherwise materially disadvantageous to such Lender; provided, however that if the Borrower requests such Lender to change its lending office, the Borrower shall pay the reasonable costs and expenses of the Lender in making such change in lending office.
- (g) Refunds. If the Agent or any Lender becomes aware that it is entitled to claim a refund from a Governmental Authority or other taxation authority in respect of Non-Excluded Taxes or Other Taxes as to which it has been indemnified by the Borrower or with respect to which the Borrower has paid additional amounts pursuant to this Section 2.9 it shall promptly notify the Borrower of the availability of such refund claim and shall, within 30 days after receipt of a request by the Borrower, make a claim to such Governmental Authority or taxation authority for such refund at the Borrower's expense. If the Agent or any Lender receives a refund (including pursuant to a claim made pursuant to the preceding sentence) in respect of any Non-Excluded Taxes or Other Taxes as to which it has been indemnified by the Borrower or with respect to which the Borrower has paid additional amounts pursuant to this Section 2.9, it shall pay over such refund to the Borrower (but only to the extent of indemnity payments made, or additional amounts paid, by the Borrower under this Section 2.9 with respect to the Non-Excluded Taxes or Other Taxes giving rise to such refund), net of all out-of-pocket expenses of the Agent or such Lender and without interest (other than any interest paid by the relevant Governmental Authority or taxation authority with respect to such refund); provided, that the Borrower, upon the request of the Agent or such Lender, agrees to repay the amount paid over to the Borrower (plus any penalties, interest or other charges imposed by the relevant Governmental Authority or taxation authority) to Agent or such Lender in the event the Agent or such Lender is required to repay such refund to such Governmental Authority. This paragraph shall not be construed to require the Agent or any Lender to make available its tax returns (or any other information relating to its taxes which it deems confidential) to the Borrower or any other Person.

# SECTION 3. CONDITIONS TO LOANS

The obligations of Lenders and L/C Issuers to make Loans and to issue or cause to be issued Letters of Credit are subject to satisfaction of all of the applicable conditions set forth below.

- 3.1 Conditions to Initial Loans. The obligations of Lenders and L/C Issuers to make the initial Loans and to issue or cause to be issued Letters of Credit on the Closing Date are, in addition to the conditions precedent specified in Section 3.2, subject to
- (a) Holdings and Merger Sub having received equity contributions from Sponsor, management or other investors in an amount not less than \$47,750,000, of which \$37,750,000 will be in cash (with Sponsor being the controlling shareholder on a fully-diluted basis), and the proceeds of all such equity contributions shall have been contributed to the capital of the Borrower; provided that up to \$10,000,000 of the investment by management shall consist of obligations to make Contingent Payments;
- (b) Funded Debt shall not exceed \$48,000,000 on the Closing Date after giving effect to the Related Transactions, and the Leverage Ratio calculated using Funded Debt at the Closing Date and EBITDA for the twelve month period ending on June 30, 2006 shall not exceed 3.5 to 1.0;

- (c) no more than \$3,000,000 of Revolving Loans shall be advanced on the Closing Date (including issued Letters of Credit), and after giving effect to the Loans and the Related Transactions and the payment of all fees and expenses in connection therewith, Borrowing Availability shall be at least \$9,500,000;
- (d) the Related Transactions shall have been consummated in accordance with their respective terms, except as may have been consented to in writing by Agent;
- (e) Borrower shall have entered into employment, confidentiality, stock purchase and non-compete documents with stockholders of Holdings reasonably satisfactory to Agent;
- (f) Agent shall have completed background and reference checks with results satisfactory to Agent on: (i) Borrower and any of its Affiliates, and (ii) the CEO, CFO, COO, shareholders, officers, and directors of each of Holdings, Borrower, their Subsidiaries and each of their respective Affiliates, in each case as determined by Agent in its sole discretion;
- (g) no material adverse change in the business, financial condition, collateral, operations, industry, properties or prospects of the Borrower or any of its Subsidiaries shall have occurred and be continuing and no litigation shall have commenced which could reasonably be expected to have a Material Adverse Effect on any of the foregoing;
- (h) no material disruption of, or material adverse change in, the financial, banking or capital markets shall have resulted that could, in Agent's judgment, reasonably be expected to materially impair Agent's ability to complete the primary syndication (as described in the Freeport Fee Letter) of the credit facility;
- (i) the purchase price of the Addus Acquisition (inclusive of aggregate fees and closing costs, including those payable to Agent and Lenders, and inclusive of \$10,000,000 of obligations to make Contingent Payments) will not exceed \$92,750,000, and Agent and Lenders shall have had the opportunity to review and shall be reasonably satisfied with the terms of the documentation for the Addus Acquisition, including, without limitation, all disclosure schedules and exhibits, and Agent shall have received a collateral assignment of Borrower's rights under such documents;
- (j) Company and Merger Sub shall have delivered the Borrower Assignment and Assumption Agreement on the Closing Date upon the consummation of the Merger;
  - (k) Borrower shall have delivered financial projections, which shall be reasonably satisfactory in form and substance to Agent and Lenders;
- (l) Agent and Lenders shall be satisfied, based on financial statements (actual and pro forma), projections and other evidence provided by Borrower, or requested by Agent, that Borrower after incurring the Indebtedness contemplated hereunder, will be Solvent;
- (m) Borrower shall have delivered any material third-party and regulatory approvals and consents necessary to consummate the Addus Acquisition which shall be final and non-appealable;
- (n) Borrower shall have delivered, no later than ten (10) Business Days prior to Closing, all environmental audit reports required by Agent which shall have been prepared by a nationally recognized environmental engineering firm acceptable to Agent, and Agent and its environmental

consultants shall have approved the scope and content of all such environmental audit reports with respect to real property owned or leased by Borrower or any of its Subsidiaries and shall be satisfied that there are no existing or potential Environmental Liabilities which could have an adverse impact on the financial condition of Borrower;

- (o) Borrower shall have delivered all policies or binders for property and casualty, liability, business interruption and other insurance satisfying the requirements of Section 5.2;
- (p) Borrower shall have delivered executed copies of the legal opinion of King & Spalding, LLP, special counsel to the Loan Parties, in addition to such local counsel opinions reasonably requested by Agent, all dated as of the date hereof and in form and substance reasonably satisfactory to Agent;
- (q) Borrower shall have delivered (i) copies of the organizational documents of each of the Loan Parties, certified by the Secretary of State of its jurisdiction of organization or, if such document is of a type that may not be so certified, certified by the secretary or similar officer of each such Loan Party, together with a good standing certificate from the Secretary of State of its jurisdiction of organization dated a recent date prior to the date hereof, (ii) resolutions of the governing body of each Loan Party approving and authorizing the execution, delivery and performance of the Loan Documents to which it is a party, certified as of the date hereof by the secretary or similar officer of Borrower as being in full force and effect without modification or amendment and (iii) signature and incumbency certificates of the officers of Holdings, Borrower or such Subsidiary executing the Loan Documents; and
- (r) Borrower shall deliver all documents listed on, take all actions set forth on and satisfy all other conditions precedent listed in the Closing Checklist attached hereto as Annex B, all in form and substance, or in a manner, satisfactory to Agent and Lenders.
- 3.2 <u>Conditions to All Loans</u>. Except as otherwise expressly provided herein, no Lender or L/C Issuer shall be obligated to fund any Revolving Credit Advance or Term Loan (each, an "Advance") or incur any Letter of Credit Obligation, if, as of the date thereof (the "Funding Date"):
- (a) any representation or warranty by any Loan Party contained herein or in any other Loan Document is untrue or incorrect in any material respect (without duplication of any materiality qualifier contained therein) as of such date, except to the extent that such representation or warranty expressly relates to an earlier date and Agent or Requisite Revolving Lenders have determined not to make such Advance or incur such Letter of Credit Obligation as a result thereof;
- (b) any Default or Event of Default has occurred and is continuing or would result after giving effect to any Advance (or the incurrence of any Letter of Credit Obligation) and Agent or Requisite Revolving Lenders have determined not to make such Advance or incur such Letter of Credit Obligation as a result thereof; or
- (c) after giving effect to any Advance (or the incurrence of any Letter of Credit Obligations), the outstanding amount of the Revolving Loan would exceed Borrowing Availability (without deducting therefrom the outstanding Revolving Loans) (except as provided in Section 2.1(b)(ii)).

The request and acceptance by Borrower of the proceeds of any Advance or the incurrence of any Letter of Credit Obligations shall be deemed to constitute, as of the date thereof, (i) a representation and warranty by Borrower that the conditions in this <u>Section 3.2</u> have been satisfied and (ii) a reaffirmation by Borrower of the granting and continuance of Agent's Liens, on behalf of itself and Lenders, pursuant to the Collateral Documents.

# SECTION 4. REPRESENTATIONS AND WARRANTIES

To induce Agent and Lenders to enter into the Loan Documents, to make Loans and to issue or cause to be issued Letters of Credit, Borrower and the other Loan Parties executing this Agreement, jointly and severally, represent, warrant and covenant to Agent and each Lender that the following statements, after giving effect to the consummation of the Related Transactions, are true, correct and complete.

#### 4.1 Organization, Powers, Capitalization and Good Standing.

- (a) <u>Organization and Powers</u>. Each of the Loan Parties is duly organized, validly existing and in good standing under the laws of its jurisdiction of organization and qualified to do business in all states where such qualification is required except where failure to be so qualified or be in good standing would not reasonably be expected to have a Material Adverse Effect. The jurisdiction of organization and all jurisdictions in which each Loan Party is qualified to do business as of the Closing Date are set forth on Schedule 4.1(a). Each of the Loan Parties has all requisite organizational power and authority to own and operate its properties, to carry on its business as now conducted and proposed to be conducted, to enter into each Loan Document and Related Transactions Document to which it is a party and to incur the Obligations, grant liens and security interests in the Collateral and carry out the Related Transactions.
- (b) <u>Capitalization</u>. As of the Closing Date: (i) the authorized Stock of each of the Loan Parties and each of their Subsidiaries is as set forth on Schedule 4.1(b); (ii) all issued and outstanding Stock of each of the Loan Parties and each of their Subsidiaries is duly authorized and validly issued, fully paid, nonassessable (as applicable), free and clear of all Liens other than those in favor of Agent for the benefit of Agent and Lenders, and such Stock was issued in compliance with all applicable state, federal and foreign laws concerning the issuance of securities; (iii) the identity of the holders of the Stock of each of the Loan Parties and the percentage of their fully-diluted ownership of the Stock of each of the Loan Parties is set forth on Schedule 4.1(b); and (iv) no Stock of any Loan Party or any of their Subsidiaries, other than those described above, are issued and outstanding. Except as provided in Schedule 4.1(b), as of the Closing Date, there are no preemptive or other outstanding rights, options, warrants, conversion rights or similar agreements or understandings for the purchase or acquisition from any Loan Party or any of their Subsidiaries of any Stock of any such entity.
- (c) <u>Binding Obligation</u>. This Agreement is, and the other Loan Documents and Related Transactions Documents when executed and delivered will be, the legally valid and binding obligations of the Loan Parties, each enforceable against each Loan Party, as applicable, in accordance with their respective terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or other laws affecting, creditors' rights generally and the effects of general principles of equity.
- 4.2 <u>Disclosure</u>. No representation or warranty of any Loan Party contained in this Agreement, the Financial Statements referred to in <u>Section 4.5</u>, the other Related Transactions (other than Projections, as to which the only representation and warranty made is as set forth in <u>Section 4.5</u> hereof), the other Loan Documents or any other document, certificate or written statement furnished to Agent or any Lender by or on behalf of any such Person for use in connection with the Loan Documents or the Related Transactions

Documents contains any untrue statement of a material fact or omitted, omits or will omit to state a material fact necessary in order to make the statements contained herein or therein not misleading in any material respect in light of the circumstances in which the same were made.

- 4.3 No Material Adverse Effect. Since December 31, 2005 there have been no events or changes in facts or circumstances affecting any Loan Party or any of its Subsidiaries which individually or in the aggregate have had or would reasonably be expected to have a Material Adverse Effect.
- 4.4 No Conflict. The consummation of the Related Transactions does not and will not violate or conflict with any laws, rules, regulations or orders of any Governmental Authority or violate, conflict with, result in a breach of, or constitute a default (with due notice or lapse of time or both) under any Contractual Obligation or organizational documents of any Loan Party or any of its Subsidiaries, except if such violations, conflicts, breaches or defaults have not had and would not reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect.
- 4.5 <u>Financial Statements and Projections</u>. Except as disclosed on Schedule 4.5, all Financial Statements concerning Holdings and its Subsidiaries which have been or will hereafter be furnished to Agent pursuant to this Agreement, including those listed below, have been or will be prepared in accordance with GAAP consistently applied (except as disclosed therein) and do or will present fairly in all material respects the financial condition of the entities covered thereby as at the dates thereof and the results of their operations for the periods then ended, subject to, in the case of unaudited Financial Statements, the absence of footnotes and normal year-end adjustments.
- (a) The consolidated balance sheets at December 31, 2005 and the related statement of income of the Company and its Subsidiaries, for the Fiscal Year then ended, audited by BDO Seidman, LLP.
- (b) The consolidated balance sheet at June 30, 2006 and the related statement of income of the Company and its Subsidiaries for the six (6) months then ended.

The Projections delivered on or prior to the Closing Date were prepared on the basis of the assumptions stated therein and such assumptions were believed by the Loan Parties to be reasonable at the time prepared. It is understood by all parties hereto that uncertainty is inherent in any forecasts or projections and that no assurance can be given that the results set forth in the Projections will actually be obtained.

4.6 Solvency. Each of the Loan Parties is Solvent.

#### 4.7 Use of Proceeds; Margin Regulations.

(a) No part of the proceeds of any Loan will be used for "buying" or "carrying" "margin stock" within the respective meanings of such terms under Regulation U of the Board of Governors of the Federal Reserve System as now and from time to time hereafter in effect or for any other purpose that violates the provisions of the regulations of the Board of Governors of the Federal Reserve System. If requested by Agent, each Loan Party will furnish to Agent and each Lender a statement to the foregoing effect in conformity with the requirements of FR Form G-3 or FR Form U-1, as applicable, referred to in Regulation U.

- (b) Borrower shall utilize the proceeds of the Loans solely for the Refinancing (and to pay any related transaction expenses), to finance fees and expenses incurred by Holdings in connection with the Addus Acquisition, to finance future acquisitions by the Borrower or any of its Subsidiaries and for the financing of Borrower's ordinary working capital and general corporate needs. Schedule 4.7 contains a description of Borrower's sources and uses of funds as of the Closing Date, including Loans and Letter of Credit Obligations to be made or incurred on that date, and a funds flow memorandum detailing how funds from each source are to be transferred for particular uses.
- (c) None of Holdings, Borrower or any of its Subsidiaries is subject to regulation as an "investment company" as defined in, or subject to regulation under, the Investment Company Act of 1940.
- 4.8 <u>Brokers</u>. No broker or finder acting on behalf of any Loan Party or Affiliate thereof brought about the obtaining, making or closing of the Loans or the Related Transactions, and no Loan Party or Affiliate thereof has any obligation to any Person in respect of any finder's or brokerage fees in connection therewith.
- 4.9 <u>Compliance with Laws</u>. Each Loan Party represents and warrants that it (i) is in compliance and each of its Subsidiaries is in compliance with the requirements of all applicable laws, rules, regulations and orders of any Governmental Authority (including, without limitation, the regulations administered by the Office of Foreign Assets Control, 31 C.F.R. Chapter V, and all underlying executive and administrative orders, and the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56) and the obligations, covenants and conditions contained in all Contractual Obligations other than those laws, rules, regulations, orders and provisions of such Contractual Obligations the noncompliance with which would not be reasonably expected to have, either individually or in the aggregate, a Material Adverse Effect, and (ii) maintains and each of its Subsidiaries maintains all licenses, qualifications and permits referred to above other than any failure to maintain which would not reasonably be expected to have a Material Adverse Effect.
- 4.10 Intellectual Property. As of the Closing Date, each of the Loan Parties and its Subsidiaries owns, is licensed to use or otherwise has the right to use, all material Intellectual Property used in or necessary for the conduct of its business as currently conducted that is material to the financial condition, business or operations of such Loan Party and its Subsidiaries and all such Intellectual Property that is federally registered as of the Closing Date is identified on Schedule 4.10 and duly and properly registered, filed or issued in the applicable office and jurisdictions for such registrations, filings or issuances. As of the Closing Date, except as disclosed in Schedule 4.10, to the knowledge of each of the Loan Parties and its Subsidiaries, the use of such Intellectual Property by the Loan Parties and their Subsidiaries and the conduct of their businesses does not and has not been alleged in writing by any Person to infringe on the rights of any Person.
- 4.11 <u>Investigations, Audits, Etc.</u> As of the Closing Date, except as set forth on <u>Schedule 4.11</u>, no Loan Party or any of their Subsidiaries is the subject of an audit by the IRS or, to each Loan Party's knowledge, any review by the IRS or any governmental investigation concerning the violation or possible violation of any law.
- 4.12 Employee Matters. Except as set forth on Schedule 4.12, (a) as of the Closing Date, no Loan Party or Subsidiary of a Loan Party nor any of their respective employees is subject to any collective bargaining agreement, (b) as of the Closing Date, no petition for certification or union election is pending

with respect to the employees of any Loan Party or any of their Subsidiaries and no union or collective bargaining unit has sought such certification or recognition with respect to the employees of any Loan Party or any of their Subsidiaries, (c) there are no strikes, slowdowns, work stoppages or controversies pending between any Loan Party or any of their Subsidiaries and its respective employees, other than employee grievances arising in the ordinary course of business which would not reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect and (d) as of the Closing Date, hours worked by and payment made to employees of each Loan Party and each of their Subsidiaries comply in all material respects with the Fair Labor Standards Act and each other federal, state, provincial, local or foreign law applicable to such matters. As of the Closing Date, except as set forth on Schedule 4.12, neither Borrower nor any of its Subsidiaries is party to an employment contract.

- 4.13 <u>Litigation</u>; <u>Adverse Facts</u>. Except as set forth on <u>Schedule 4.13</u>, there are no judgments outstanding against any Loan Party or any of its Subsidiaries or affecting any property of any Loan Party or any of its Subsidiaries as of the Closing Date, nor is there any Litigation pending, or to the best knowledge of any Loan Party threatened, against any Loan Party or any of its Subsidiaries, in each case which would reasonably be expected to result in a Material Adverse Effect.
- 4.14 Ownership of Property; Liens. As of the Closing Date, the real estate ("Real Estate") listed in Schedule 4.14 constitutes all of the real property owned, leased, subleased, or used by any Loan Party or any of its Subsidiaries. As of the Closing Date, each of the Loan Parties and each of its Subsidiaries owns good and marketable fee simple title to all of its owned Real Estate, and valid and marketable leasehold interests in all of its leased Real Estate, all as described on Schedule 4.14 and in each case subject to Permitted Encumbrances, and copies of all such leases owned or partially owned by an Affiliate or a summary of terms thereof reasonably satisfactory to Agent have been delivered to Agent. Schedule 4.14 further describes any Real Estate with respect to which any Loan Party or any of its Subsidiaries is a lessor, sublessor or assignor as of the Closing Date. As of the Closing Date, each of the Loan Parties and each of its Subsidiaries also has good and marketable title to, or valid leasehold interests in, all of its personal property and assets subject to applicable Permitted Encumbrances. As of the Closing Date, none of the properties and assets of any Loan Party or any of its Subsidiaries are subject to any Liens other than Permitted Encumbrances, and there are no facts, circumstances or conditions known to Borrower that are reasonably likely to result in any Liens (including Liens arising under Environmental Laws) other than Permitted Encumbrances against the properties or assets of any Loan Party or any of its Subsidiaries. As of the Closing Date, Schedule 4.14 also describes any purchase options, rights of first refusal or other similar contractual rights pertaining to any Real Estate. As of the Closing Date, all material permits required to have been issued or appropriate to enable the Real Estate to be lawfully occupied and used for all of the purposes for which it is currently occupied and used have been lawfully issued and are in full force and effect including, without limitation, the us
- 4.15 Environmental Matters. Except as set forth in Schedule 4.15, as of the Closing Date: (i) the Loan Parties and their Subsidiaries are and have been in compliance with all Environmental Laws, except for such noncompliance that would not reasonably be expected to result in Environmental Liabilities of the Loan Parties or their Subsidiaries in excess of \$1,000,000 in the aggregate; (ii) the Loan Parties and their Subsidiaries have obtained, and are in compliance with, all Environmental Permits required by Environmental Laws for the operations of their respective businesses as presently conducted or as proposed to be conducted, except where the failure to so obtain or comply with such Environmental Permits could not reasonably be expected to result in Environmental Liabilities of the Loan Parties or their Subsidiaries in excess of \$1,000,000 in the aggregate, and all such Environmental Permits are valid, uncontested and in good standing; (iii) no Loan Party and no Subsidiary of a Loan Party is involved in operations under Environmental Laws or knows of any facts, circumstances or conditions under

Environmental Laws, including any Releases of Hazardous Materials, that are likely to result in any Environmental Liabilities of such Loan Party or Subsidiary which could reasonably be expected to be in excess of \$1,000,000 in the aggregate, and no Loan Party or Subsidiary of a Loan Party has permitted any current or former tenant or occupant of the Real Estate to engage in any such operations; (iv) there is no Litigation arising under or related to any Environmental Laws, Environmental Permits or Hazardous Material that seeks damages, penalties, fines, costs or expenses in excess of \$1,000,000 in the aggregate or injunctive relief against, or that alleges criminal misconduct by any Loan Party or any Subsidiary of a Loan Party; and (v) no notice has been received by any Loan Party or any Subsidiary of a Loan Party identifying any of them as a "potentially responsible party" or requesting information under CERCLA or analogous state statutes, and to the knowledge of the Loan Parties, there are no facts, circumstances or conditions that would reasonably be expected to result in any of the Loan Parties or their Subsidiaries being identified as a "potentially responsible party" under CERCLA or analogous state statutes.

### 4.16 ERISA

- (a) Except with respect to Multiemployer Plans or as set forth on Schedule 4.16, each on-going Qualified Plan received a favorable determination or opinion letter from the IRS or is within the applicable remedial amendment period. Except as would not reasonably be expected to have a Material Adverse Effect, each Plan is in compliance with the applicable provisions of ERISA and the IRC. Except as would not reasonably be expected to have a Material Adverse Effect, neither any Loan Party nor ERISA Affiliate has failed to make any contribution or pay any amount due as required by either Section 412 of the IRC or Section 302 of ERISA or the terms of any such Title IV Plan. No Loan Party has engaged in a "prohibited transaction," as defined in Section 406 of ERISA and Section 4975 of the IRC, in connection with any Plan, that would subject any Loan Party to a material tax on prohibited transactions imposed by Section 502(i) of ERISA or Section 4975 of the IRC in an amount that would reasonably be expected to have a Material Adverse Effect.
- (b) As of the Closing Date, except as set forth in Schedule 4.16: (i) except as would not reasonably be expected to have a Material Adverse Effect, no ERISA Event or event described in Section 4062(e) of ERISA with respect to any Title IV Plan has occurred; (ii) except as would not reasonably be expected to have a Material Adverse Effect, no Loan Party or ERISA Affiliate has incurred any liability as a result of a complete or partial withdrawal from a Multiemployer Plan; and (iii) except as would not reasonably be expected to have a Material Adverse Effect, within the last five years no Title IV Plan of any Loan Party or ERISA Affiliate has been terminated, whether or not in a "standard termination" as that term is used in Section 4041(b)(1) of ERISA.
- 4.17 <u>Deposit and Disbursement Accounts</u>. <u>Schedule 4.17</u> lists all banks and other financial institutions at which any Loan Party maintains deposit or other accounts as of the Closing Date, including any Disbursement Accounts, and such <u>Schedule</u> correctly identifies the name, address and telephone number of each depository, the name in which the account is held, a description of the purpose of the account, and the complete account number therefor.

### 4.18 Agreements and Other Documents.

As of the Closing Date, each Loan Party has provided to Agent or its counsel, on behalf of Lenders, accurate and complete copies (or summaries) of all of the following agreements or documents to which it is subject and each of which is listed in <u>Schedule 4.18</u>: licenses and permits held by the Loan Parties, the absence of which would reasonably be expected to have a Material Adverse Effect; instruments and documents evidencing any Indebtedness or Guaranteed Indebtedness of such Loan Party and any Lien granted by such Loan Party with respect thereto; and instruments and agreements

evidencing the issuance of any equity securities, warrants, rights or options to purchase equity securities of such Loan Party; and agreements with referral sources or referral recipients (including physicians, hospite providers, and therapy providers).

### 4.19 Insurance.

Each Loan Party represents and warrants that it and each of its Subsidiaries currently maintains in good repair, working order and condition (normal wear and tear excepted) all material properties as set forth in Section 5.2 and maintains all insurance described in such Section. Schedule 4.19 lists all insurance policies of any nature maintained, as of the Closing Date, for current occurrences by each Loan Party, as well as a summary of the key business terms of each such policy such as deductibles, coverage limits and term of policy.

## 4.20 Taxes and Tax Returns.

As of the Closing Date, (i) all Tax Returns required to be filed by the Loan Parties have been timely and properly filed and (ii) all taxes that are due (other than taxes being contested in good faith by appropriate proceedings and for which adequate reserves have been provided for in accordance with GAAP) have been paid, in each case except where the failure to file Tax Returns or pay Taxes would not reasonably be expected to have a Material Adverse Effect. No Governmental Authority has asserted any claim for taxes, or to any Loan Party's knowledge, has threatened to assert any claim for taxes that would, if not paid by a Loan Party, have a Material Adverse Effect. All taxes required by law to be withheld or collected and remitted (including, without limitation, income tax, unemployment insurance and workmen's compensation premiums) with respect to the Loan Parties have been withheld or collected and paid to the appropriate Governmental Authorities (or are properly being held for such payment), except for amounts the nonpayment of which would not be reasonably likely to have a Material Adverse Effect.

(a) None of the Loan Parties has been notified that either the IRS, or any other Governmental Authority, has raised or intends to raise, any adjustments with respect to Taxes of the Loan Parties, which adjustments would be reasonably expected to have a Material Adverse Effect.

#### 4.21 No Earn-outs.

Other than Contingent Payments, no "earn-outs" or any similar payment obligations are payable by any Loan Party or any of their respective Subsidiaries in connection with the Addus Acquisition or any transaction occurring in connection therewith.

## 4.22 Compliance With Health Care Laws.

Without limiting the generality of any other representation or warranty set forth in this Agreement:

(a) Except as set forth on Schedule 4.22(a), each Loan Party and each Subsidiary of a Loan Party (i) is in material compliance with all Health Care Laws and all Third Party Payor Programs and (ii) is not in material violation of any order of CMS, the HHS, or any other Governmental Authority or other board or tribunal regulating, enforcing or overseeing compliance with Health Care Laws. Since September 30, 2001, except as set forth on Schedule 4.22(a), no Loan Party or Subsidiary of a Loan Party has received any written notice that it is not in material compliance in any respect with any of the requirements of any of the foregoing. Without limiting the generality of the foregoing, no Loan Party or Subsidiary of a Loan Party has received a subpoena or other notice that it is or will be subject to any

investigation by any Governmental Authority with respect to any Health Care Law, nor, to the knowledge of any Loan Party, is any investigation threatened or are there circumstances which, if known to a Governmental Authority, would lead the Governmental Authority to investigate such Loan Party or Subsidiary of a Loan Party. Each Loan Party and each Subsidiary of a Loan Party has received and maintains accreditation in good standing and without limitation or impairment by all applicable accrediting organizations, to the extent required by applicable law (including all Health Care Laws), all Third Party Payor Programs and all Participation Agreements, except where the failure to receive and/or maintain such accreditation would not reasonably be expected to have or result in, either individually or in the aggregate, a Material Adverse Effect.

- (b) Each Loan Party and each Subsidiary of a Loan Party has the requisite provider numbers or other authorizations to bill the Medicare program (to the extent such entity participates in the Medicare program), the respective Medicaid program in the state or states in which such entity operates, and all other Third Party Payor Programs that such Loan Party or such Subsidiary of a Loan Party currently bills or in the past has billed except where the failure to have such authorization could not reasonably be expected to be, have or result in, either individually or in the aggregate, a Material Adverse Effect. To the knowledge of the Borrower, there is no investigation, audit, claim review or other action pending or threatened which could result in a revocation, suspension, termination, probation, restriction, limitation, or non-renewal of any Third Party Payor, Participation Agreement, provider number or authorization or result in the exclusion of any Loan Party or a Subsidiary of a Loan Party from any Third Party Payor Program, which revocation, suspension, termination, probation, restriction, limitation, non-renewal or exclusion would reasonably be expected to be, have or result in, either individually or in the aggregate, a Material Adverse Effect. The Borrower has no knowledge that any condition exists or event has occurred which, in itself or with the giving of notice or lapse of time or both, reasonably would be expected to result in the suspension, revocation, forfeiture, non-renewal of any governmental consent applicable to any Loan Party or any Subsidiary of a Loan Party or any health care facility operated by any Loan Party or any Subsidiary of a Loan Party or such facility's participation in any Third Party Payor Program, or of any Participation Agreements, which suspension, revocation, forfeiture or non-renewal would reasonably be expected to be, have or result in, either individually or in the aggregate, a Material Adverse Effect.
- (c) Each Loan Party and each Subsidiary of a Loan Party has filed with all applicable federal, state and local Governmental Authorities for and received approval of all registrations, applications, licenses, certificates or determinations of need, requests for exemptions, permits and other regulatory authorizations necessary to conduct the businesses of such Loan Party or such Subsidiary of a Loan Party as currently conducted, the absence of which would not reasonably be expected to be, have or result in, either individually or in the aggregate, a Material Adverse Effect. Each Loan Party and each Subsidiary of a Loan Party is, and at all relevant times has been, in compliance in all respects with all such registrations, applications, licenses, requests for exemptions, permits and other regulatory authorizations, except to the extent that the failure to be in compliance would not reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect.
- (d) All billing practices of the Loan Parties and Subsidiary of the Loan Parties to all Third Party Payors have been true, fair and correct in all material respects, and in material compliance with all applicable requirements of Law (including all Health Care Laws), and no Loan Party or Subsidiary of a Loan Party has knowingly and willfully billed for or received any payment or reimbursement in excess of amounts allowed by law, except for such payments or reimbursements as such Loan Party or Subsidiary of a Loan Party may be disputing or contesting in good faith, except to the extent that the failure to be in compliance would not reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect.

- (e) Each Loan Party and each Subsidiary of a Loan Party has established or will establish a compliance plan, the purpose of which is to assure that each such Person is in compliance with Health Care Laws.
- (f) Except as set forth on Schedule 4.22(f), no Loan Party and no Subsidiary of a Loan Party and none of their respective employees and independent contractors is, or to the knowledge of such Loan Party or Subsidiary has been, excluded from participation from any federal or state health care program.

# SECTION 5. AFFIRMATIVE COVENANTS

Each Loan Party executing this Agreement jointly and severally agrees as to all Loan Parties that from and after the date hereof and until the Termination Date:

### 5.1 Compliance With Laws and Contractual Obligations.

Each Loan Party will (a) comply with and shall cause each of its Subsidiaries to comply with (i) the requirements of all applicable material laws, rules, regulations and orders of any Governmental Authority (including, without limitation, laws, rules, regulations and orders relating to taxes, employer and employee contributions, securities, employee retirement and welfare benefits, environmental protection matters and employee health and safety) as now in effect and which may be imposed in the future in all jurisdictions in which any Loan Party or any of its Subsidiaries is now doing business or may hereafter be doing business, other than the noncompliance with which would not be reasonably expected to have, either individually or in the aggregate, a Material Adverse Effect, and (ii) the obligations, covenants and conditions contained in all Contractual Obligations of such Loan Party or any of its Subsidiaries other than the noncompliance with which would not be reasonably expected to have, either individually or in the aggregate, a Material Adverse Effect, and (b) maintain or obtain and shall cause each of its Subsidiaries to maintain or obtain all licenses, certifications, qualifications and permits now held or hereafter required to be held by such Loan Party or any of its Subsidiaries, for which the loss, suspension, revocation or failure to obtain or renew, would reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect. This Section 5.1 shall not preclude any Loan Party or its Subsidiaries from contesting any taxes or other payments, if they are being diligently contested in good faith in a manner which stays enforcement thereof and if appropriate expense provisions have been recorded in conformity with GAAP and, subject to Section 6.2, no Lien (other than a Permitted Encumbrance) in respect thereof has been created.

### 5.2 Insurance.

(a) Each Loan Party will maintain or cause to be maintained, with insurers reasonably acceptable to Agent, public liability, malpractice and property damage insurance with respect to its business and properties and the business and properties of its Subsidiaries against loss or damage of the kinds customarily carried or maintained by corporations of established reputation engaged in substantially similar businesses and in amounts reasonably acceptable to Agent and will deliver evidence thereof to Agent. Agent confirms that the insurance in effect on the Closing Date is reasonably acceptable to it. The Loan Parties shall maintain business interruption insurance providing coverage for a period of at least six (6) months and in an amount not less than \$1,000,000. Each Loan Party shall, pursuant to endorsements and/or assignments in form and substance reasonably satisfactory to Agent, (i) cause Agent to be named as lender's loss payee in the case of casualty insurance and assignee in the case of all business interruption insurance, in each case for the benefit of Agent and Lenders and (ii) cause

Agent and each Lender to be named as additional insureds in the case of all liability insurance. So long as no Event of Default has occurred and is continuing, proceeds of business interruption insurance will be released to the Loan Parties. In the event any Loan Party fails to provide Agent with evidence of the insurance coverage required by this Agreement, Agent may purchase insurance at such Loan Party's expense to protect Agent's interests in the Collateral. This insurance may, but need not, protect such Loan Party's interests. The coverage purchased by Agent may not pay any claim made by such Loan Party or any claim that is made against such Loan Party in connection with the Collateral. If Agent purchases insurance for the Collateral, such Loan Party will be responsible for the costs of that insurance (which may exceed the cost of insurance that such Loan Party could obtain on its own), including interest and other Charges imposed by Agent in connection with the placement of the insurance, until the effective date of the cancellation or expiration of the insurance. The costs of the insurance may be added to the Obligations.

## 5.3 Inspection; Coding and Billing Review; Lender Meeting.

- (a) <u>Inspection</u>. Upon three (3) days' prior written notice to the Loan Parties and the Lenders, each Loan Party shall permit any authorized representatives of Agent to visit, audit and inspect any of the properties of such Loan Party and its Subsidiaries, including its and their financial and accounting records, and to make copies and take extracts therefrom, and to discuss its and their affairs, finances and business with its and their officers and certified public accountants, at such reasonable times during normal business hours and as often as may be reasonably requested (collectively a "<u>Field Review</u>"); provided, that, upon the occurrence and continuance of an Event of Default, Agent shall not be required to provide any notice to the Loan Parties prior to the performance of a Field Review; provided further, that, so long as no Event of Default has occurred and is continuing, Agent shall only be reimbursed by Borrower for one (1) Field Review during any Fiscal Year. Representatives of each Lender will be permitted to accompany representatives of Agent during each Field Review at such Lender's expense.
- (b) <u>Coding and Billing Review</u>. Each Loan Party shall permit an independent third party selected by Agent in its sole discretion to conduct a review of the clinical coding and billing procedures of such Loan Party and its Subsidiaries (a "<u>Coding and Billing Review</u>"); provided that so long as no Event of Default has occurred and is continuing, (a) Agent shall only be reimbursed by Borrower for four (4) Coding and Billing Reviews during any Fiscal Year and (b) the Borrower's reimbursement obligations for each such Coding and Billing Review shall not exceed \$15,000.
- (c) <u>Lender Meeting</u>. Each Loan Party will participate and will cause key management personnel of each Loan Party and its Subsidiaries to participate in a meeting with Agent and Lenders at least once during each Fiscal Year, which meeting shall be held at such time and such place as may be reasonably requested by Agent and reasonably convenient to Borrower.

## 5.4 Organizational Existence.

Except as otherwise permitted by <u>Section 6.6</u>, each Loan Party will at all times preserve and keep in full force and effect its organizational existence and all rights and franchises material to its business.

### 5.5 Environmental Matters.

Each Loan Party shall and shall cause each Person within its control to: (a) conduct its operations and keep and maintain its Real Estate in compliance with all Environmental Laws and Environmental Permits other than noncompliance that would not reasonably be expected to have a Material Adverse

Effect; (b) notify Agent promptly after such Loan Party or any Person within its control becomes aware of any violation of Environmental Laws or Environmental Permits or any Release on, at, in, under, above, to, from or about any Real Estate that is reasonably likely to result in Environmental Liabilities to a Loan Party or its Subsidiaries in excess of \$1,000,000; and (c) promptly forward to Agent a copy of any order, notice of actual or alleged violation or liability, request for information or any communication or report received by such Loan Party or any Person within its control that relates to any Environmental Laws or Environmental Permits and that could reasonably be expected to result in Environmental Liabilities in excess of \$1,000,000.

### 5.6 <u>Landlords' Agreements, Mortgagee Agreements, Bailee Letters and Real Estate Purchases</u>.

Each Loan Party shall use reasonable efforts to obtain a landlord's agreement, mortgagee agreement or bailee letter, as applicable, from the lessor of each leased property, mortgagee of owned property or bailee with respect to any warehouse, processor or converter facility or other location where Collateral with a book value greater than \$100,000 is stored or located, which agreement or letter shall contain a waiver or subordination of all Liens or claims that the landlord, mortgagee or bailee may assert against the Collateral at that location, and shall otherwise be reasonably satisfactory in form and substance to Agent.

#### 5.7 Further Assurances.

- (a) Each Loan Party shall, from time to time, execute such guaranties, financing statements, documents, security agreements and reports as Agent or Requisite Lenders at any time may reasonably request to evidence, perfect or otherwise implement the guaranties and security for repayment of the Obligations contemplated by the Loan Documents, provided, that subsequent guaranties and security agreements shall be substantially in the same form as those delivered as of the Closing Date.
- (b) In the event any Loan Party acquires a fee ownership interest in real property after the Closing Date having a fair market value or purchase price in excess of \$100,000, such Loan Party shall deliver to Agent a fully executed mortgage or deed of trust over such real property in form and substance satisfactory to Agent, together with such title insurance policies, surveys, appraisals, evidence of insurance, legal opinions, environmental assessments and other documents and certificates as shall be reasonably required by Agent.
- (c) Each Loan Party shall (i) cause each Person, upon its becoming a Domestic Subsidiary of such Loan Party (provided that this shall not be construed to constitute consent by any of the Lenders to any transaction not expressly permitted by the terms of this Agreement), promptly to guaranty the Obligations and to grant to Agent, for the benefit of Agent and Lenders, a security interest in the real (to the extent owned in fee), personal and mixed property of such Domestic Subsidiary to secure the Obligations and (ii) pledge, or cause to be pledged, to Agent, for the benefit of Agent and Lenders, all of the Stock of such Domestic Subsidiary to secure the Obligations. Each Loan Party shall pledge, or cause to be pledged, to Agent, for the benefit of Agent and Lenders, 65% of the outstanding Voting Stock and 100% of the outstanding nonvoting Stock of any person upon its becoming a first tier Foreign Subsidiary of any Loan Party. The documentation for such guaranty, security and pledge shall be substantially similar to the Loan Documents executed concurrently herewith with such modifications as are reasonably requested by Agent in order to accommodate changes to or differences in applicable law.

### 5.8 Payment of Taxes

Each Loan Party shall timely pay and discharge (or cause to be paid and discharged) all material taxes, assessments and governmental and other charges or levies imposed upon it or upon its income or profits, or upon property belonging to it; provided that such Loan Party shall not be required to pay any such tax, assessment, charge or levy that is being contested in good faith by appropriate proceedings and for which the affected Loan Party shall have set aside on its books adequate reserves with respect thereto in conformance with GAAP and for which a Lien has not been imposed on the assets of any Loan Party.

### 5.9 Cash Management Systems.

Borrower shall, and shall cause each other Loan Party to, enter into Control Agreements with respect to each deposit account maintained by Borrower or any Subsidiary of Borrower (other than any zero balance account) as of or after the Closing Date. Each such deposit account control agreement shall be in form and substance satisfactory to Agent. In the event that Borrower or another Loan Party establishes a new deposit account after the Closing Date (other than a zero balance account), such Loan Party shall enter into such a Control Agreement within thirty (30) days following the establishment thereof. No such Control Agreement shall, in the absence of a continuing Event of Default, entitle Agent or any Lender to direct the applicable depository bank to transfer any funds in the applicable deposit account to Agent or any other Person.

### 5.10 Interest Rate Agreements.

Within one hundred and twenty (120) days after the Closing Date, Borrower shall enter into, and shall thereafter maintain, Interest Rate Agreements providing for interest rate protection for an aggregate amount of \$22,500,000 of the principal amount of the outstanding Term Loans for a period of three years on terms and conditions reasonably satisfactory to Agent.

### 5.11 Employee Benefit Plans.

Each Loan Party shall (i) maintain each Pension Plan in substantial compliance with all applicable requirements of law and regulations and (ii) make, on a timely basis, all required contributions to any Multiemployer Plan. No Loan Party shall (i) seek a waiver of the minimum funding standards of ERISA, (ii) terminate or withdraw from any Pension Plan or Multiemployer Plan or (iii) take any other action with respect to any Pension Plan that would reasonably be expected to entitle the PBGC to terminate, impose liability in respect of, or cause a trustee to be appointed to administer, any Pension Plan, unless the actions or events described in clauses (i), (ii) and (iii) individually or in the aggregate would not have a Material Adverse Effect.

### 5.12 Health Care Law Matters.

(a) Each Loan Party shall and shall cause each Person within its control to: (a) conduct its operations in material compliance with all Health Care Laws; (b) to the extent permitted by applicable law and so long as no privilege is compromised, notify Agent promptly after such Loan Party or any Person within its control becomes aware of any violation of any material violation of the Anti-Kickback Statute, Stark Law or False Claims Act, or any violation of other Health Care Law that could reasonably likely result in a claim, fine or settlement in excess of \$250,000; and (c) to the extent permitted by applicable law, promptly forward to Agent any subpoena or other request or other investigation by a Governmental Authority with respect to a possible material violation of any Health Care Laws.

(b) As soon as practicable but in no event later than 180 days following the Closing Date, each Loan Party shall, and shall cause each Person within its control to establish, deliver and enforce policies and procedures in its compliance plan the purpose of which is to assure that each Loan Party and each Person within the control of each Loan Party are in compliance in all material respects with all requirements of applicable Healthcare Laws.

## 5.13 HIPAA.

Each Loan Party and each Subsidiary of a Loan Party that is a "covered entity" as defined under HIPAA shall enter into a Business Associate Agreement (as defined under HIPAA) with Agent, substantially in the form of Exhibit 5.12, and shall maintain such Business Associate Agreement in full force and effect during the term of this Agreement. In the event that Agent or any of its authorized representatives shall make any inspection pursuant to Section 5.3 or otherwise, or shall take possession of any Collateral that would involve such Person having access to "protected health information" as defined under HIPAA, each Loan Party or Subsidiary of a Loan Party that is a "covered entity" shall permit disclosure of the protected health information pursuant to the Business Associate Agreement to the extent permitted by HIPAA.

## SECTION 6. NEGATIVE COVENANTS

Each Loan Party executing this Agreement jointly and severally agrees as to all Loan Parties that from and after the date hereof until the Termination Date:

### 6.1 Indebtedness

The Loan Parties shall not and shall not cause or permit their Subsidiaries directly or indirectly to create, incur, assume, or otherwise become or remain directly or indirectly liable with respect to any Indebtedness (other than pursuant to a Contingent Obligation permitted under Section 6.4) except:

- (a) Indebtedness described on Schedule 6.1;
- (b) the Obligations;
- (c) intercompany Indebtedness arising from loans made by any Loan Party to any other Loan Party other than Holdings; provided, however, that upon the request of Agent at any time, such intercompany Indebtedness shall be evidenced by promissory notes having terms reasonably satisfactory to Agent, the sole originally executed counterparts of which shall be pledged and delivered to Agent, for the benefit of Agent and Lenders, as security for the Obligations;
- (d) Indebtedness, not exceeding \$250,000 in aggregate principal amount outstanding at any time, secured by Liens described in and permitted by clause (l) of the definition of "Permitted Encumbrances";
- (e) Indebtedness, not exceeding \$1,750,000 in aggregate notional principal amount outstanding at any time, secured by Liens described in and permitted by clause (j) of the definition of "Permitted Encumbrances";

- (f) refinancings of Indebtedness permitted under clauses (a) and (e) that do not accelerate the scheduled dates for payment thereof, increase the principal amounts thereof, materially increase any interest rate or fees applicable thereto, add additional obligors therefor, or enhance the collateral therefor or the priority thereof;
- (g) Indebtedness under (i) Interest Rate Agreements and Rate Management Agreements entered into to protect Borrower or Subsidiary thereof against fluctuations in interest rates in respect of Indebtedness otherwise permitted under this Agreement or (ii) other hedging agreements providing protection against fluctuations in currency values or commodity prices in connection with Borrower's or any of its Subsidiaries' operations, so long as the purpose of any such agreement is a bona fide hedging activity (and is not for speculative purposes);
  - (h) Indebtedness under the Acquisition Note;
- (i) Indebtedness consisting of the obligation of any Loan Party to reimburse an insurer for any payment made by such insurer in respect of a worker's compensation claim with respect to an employee of the Borrower or its Subsidiaries; and
  - (j) Indebtedness in respect of the Contingent Payments.

### 6.2 Liens and Related Matters.

- (a) No Liens. The Loan Parties shall not and shall not cause or permit their Subsidiaries to directly or indirectly create, incur, assume or permit to exist any Lien on or with respect to any property or asset of such Loan Party or any such Subsidiary, whether now owned or hereafter acquired, or any income or profits therefrom, except Permitted Encumbrances (including, without limitation, those Liens constituting Permitted Encumbrances existing on the date hereof and renewals and extensions thereof, as set forth on Schedule 6.2).
- (b) No Negative Pledges. The Loan Parties shall not and shall not cause or permit their Subsidiaries to directly or indirectly enter into or assume any agreement (other than the Loan Documents) prohibiting the creation or assumption of any Lien upon its properties or assets, whether now owned or hereafter acquired and other than (i) provisions restricting subletting or assignment under any lease governing a leasehold interest or lease of personal property; (ii) restrictions with respect to a Subsidiary imposed pursuant to any agreement which has been entered into for the sale of disposition of all or substantially all of the equity interests or assets of such Subsidiary, so long as such sale or disposition of all or substantially all of the equity interests or assets of such Subsidiary is permitted under this Agreement; (iii) restrictions on assignments or sublicensing of licensed Intellectual Property and (iv) restrictions in agreements or Capital Leases governing or creating any Indebtedness referred to in paragraphs (d) or (e) of Section 6.1, so long as such restrictions apply only to the assets financed or leased pursuant to such agreement or Capital Lease.
- (c) No Restrictions on Subsidiary Distributions to Borrower. Except as provided herein, the Loan Parties shall not and shall not cause or permit their Subsidiaries to directly or indirectly create or otherwise cause or suffer to exist or become effective any consensual encumbrance or restriction of any kind on the ability of any such Subsidiary to: (1) pay dividends or make any other distribution on any of such Subsidiary's Stock owned by Borrower or any other Subsidiary; (2) pay any Indebtedness owed to Borrower or any other Subsidiary; (3) make loans or advances to Borrower or any other Subsidiary; or (4) transfer any of its property or assets to Borrower or any other Subsidiary, other than

restrictions contained in this Agreement and restrictions contained in any agreement existing on the Closing Date, or any agreement entered into as permitted by paragraph (f) of Section 6.1 (in respect of the refinancing of an agreement that contained such a encumbrance or restriction as of the Closing Date) or in the Contingent Payment Agreement as in effect on the Closing Date, so long as such restriction is no less favorable to the Loan Parties than those contained in this Agreement as of the Closing Date.

## 6.3 Investments

The Loan Parties shall not and shall not cause or permit their Subsidiaries to directly or indirectly make or own any Investment in any Person except:

- (a) Borrower and its Subsidiaries may make and own Investments in Cash Equivalents subject to (except as otherwise provided in Section 5.9) Control Agreements in favor of Agent; provided that such Cash Equivalents (except as consented to by Agent, such consent not to be unreasonably withheld) are not subject to setoff rights;
  - (b) any Loan Party may make intercompany loans to any other Loan Party other than Holdings to the extent permitted under Section 6.1;
- (c) Borrower and its Subsidiaries which are Loan Parties may make loans and advances to employees of the Borrower and its Subsidiaries which are Loan Parties for moving, entertainment, travel and other similar expenses in the ordinary course of business not to exceed \$100,000 in the aggregate at any time outstanding;
- (d) Loan Parties and their Subsidiaries may make (i) capital contributions to their wholly-owned Subsidiaries that are Loan Parties and (ii) capital contributions to their wholly-owned Subsidiaries that are not Loan Parties in an amount (under this clause (ii)) not to exceed \$100,000 in the aggregate reduced by the amount of Investments made pursuant to Section 6.3(b) in the form of intercompany loans made by a Loan Party to a Subsidiary that is not a Loan Party;
  - (e) non-cash consideration received in accordance with Section 6.7;
- (f) Investments existing on the Closing Date, as set forth on Schedule 6.3 and any renewals, amendments and replacements thereof that do not increase the amount thereof;
- (g) each Loan Party may hold investments comprised of notes payable, or stock or other securities issued by financially troubled Account Debtors (excluding Affiliates) to such Loan Party pursuant to agreements with respect to settlement of such Account Debtor's Accounts with such Loan Party negotiated in the ordinary course of business;
- (h) Investments consisting of loans by Borrower to employees of Borrower which are used solely by such employees to simultaneously purchase the Stock of Holdings, provided that Holdings contemporaneously contributes the proceeds of such Stock to the capital of Borrower;
  - (i) Interest Rate Agreements and other hedging agreements subject to Section 6.1;
- (j) Borrower and its Subsidiaries may make advances in the form of a prepayment of expenses, so long as such expenses were incurred in the ordinary course of business and are being paid in accordance with customary trade terms of Borrower or such Subsidiary; and

- (k) other Investments not exceeding \$100,000 at any time outstanding.
- 6.4 <u>Contingent Obligations</u>. The Loan Parties shall not and shall not cause or permit their Subsidiaries to directly or indirectly create or become or be liable with respect to any Contingent Obligation except:
  - (a) Letter of Credit Obligations;
- (b) Contingent Obligations as set forth on Schedule 6.4 arising under the Loan Documents and the Purchase Agreement and other documents related to the Addus Acquisition;
  - (c) those arising under Interest Rate Agreements or other hedging agreements entered into in compliance with Section 6.1;
  - (d) those resulting from endorsement of negotiable instruments for collection in the ordinary course of business;
  - (e) those existing on the Closing Date and described in Schedule 6.4;
- (f) those arising with respect to customary indemnification obligations incurred in connection with Asset Dispositions permitted hereunder or in connection with obtaining title insurance; and
- (g) those incurred with respect to Indebtedness permitted by Section 6.1 provided that (i) any such Contingent Obligation is subordinated to the Obligations to the same extent as the Indebtedness to which it relates is subordinated to the Obligations and (ii) no Loan Party may incur Contingent Obligations in respect of Indebtedness incurred by any Person that is not a Loan Party under this clause (g).
- 6.5 <u>Restricted Payments</u>. The Loan Parties shall not and shall not cause or permit their Subsidiaries to directly or indirectly declare, order, pay, make or set apart any sum for any Restricted Payment, except that:
- (a) Each Loan Party other than Holdings may make payments and distributions to Holdings (whether directly or through sequential upstream Restricted Payments) that are used by Holdings to pay federal, state and local income taxes then due and owing, estimated taxes then due and owing, franchise taxes and other similar licensing expenses incurred in the ordinary course of business; provided that each Loan Party's aggregate contribution to taxes as a result of the filing of a consolidated, unitary or combined return with Holdings or another party shall not be greater than they would have been had such Loan Party not filed a consolidated, unitary or combined return with Holdings;
- (b) The Loan Parties may make Restricted Payments contemplated to be made on the Closing Date under the Purchase Agreement and Merger Agreement;
- (c) Direct or indirect wholly-owned Subsidiaries of Borrower may make Restricted Payments to the entity which is the direct owner of the equity of such wholly-owned Subsidiary; and
- (d) Borrower may pay dividends to Holdings, and Borrower may pay dividends to Intermediate Holdings and Intermediate Holdings may pay dividends to Holdings, in each case, to permit

Holdings to repurchase Stock owned by employees of Borrower whose employment with Borrower and its Subsidiaries has been terminated, provided that such Restricted Payments shall not exceed \$500,000 in any Fiscal Year or \$1,000,000 during the term of this Agreement in the aggregate and provided that no Event of Default exists at the time of such Restricted Payment or would occur as a result thereof.

- 6.6 <u>Restriction on Fundamental Changes</u>. The Loan Parties shall not and shall not cause or permit their Subsidiaries to directly or indirectly: (a) amend, modify or waive any term or provision of its organizational documents in a manner adverse to the Lenders, including its articles of incorporation, certificates of designations pertaining to preferred stock, by-laws, partnership agreement or operating agreement in any manner adverse to the Agent or Lenders unless required by law; (b) enter into any transaction of merger or consolidation except, upon not less than five (5) Business Days' prior written notice to Agent, any wholly-owned Subsidiary of Borrower may be merged with or into Borrower (<u>provided</u> that Borrower is the surviving entity) or any other wholly-owned Subsidiary of Borrower (provided that, in the case of any such merger of any Domestic Subsidiary with or into a Foreign Subsidiary, the Domestic Subsidiary is the surviving entity); (c) liquidate, wind-up or dissolve itself (or suffer any liquidation or dissolution); or (d) acquire by purchase or otherwise all or any part of the Stock, business or assets of any other Person.
- 6.7 <u>Disposal of Assets or Subsidiary Stock.</u> The Loan Parties shall not and shall not cause or permit their Subsidiaries to directly or indirectly convey, sell, lease, sublease, transfer or otherwise dispose of, or grant any Person an option to acquire, in one transaction or a series of related transactions, any of its property, business or assets, whether now owned or hereafter acquired, except for (a) sales of inventory to customers in the ordinary course of business and dispositions of obsolete equipment not used or useful in the business; (b) any condemnation or taking of such assets by eminent domain proceedings; (c) Asset Dispositions by Borrower and its Subsidiaries (excluding sales of Accounts and Stock of any of Holdings' Subsidiaries) if all of the following conditions are met: (i) the aggregate fair market value of assets sold or otherwise disposed of in any Fiscal Year does not exceed \$250,000; (ii) the consideration received is at least equal to the fair market value of such assets (as determined by the board of directors of the applicable Loan Party in good faith); (iii) at least 85% of the consideration received is cash; (iv) the Net Proceeds of such Asset Disposition are applied as required by <u>Section 2.5(c)</u>; and (v) no Event of Default has occurred and is continuing or would result from such Asset Disposition and (d) other sales, leases, subleases, transfers or dispositions that are not Asset Dispositions by virtue of clauses (2) through (9) of the definition of Asset Disposition.
- 6.8 <u>Transactions with Affiliates</u>. The Loan Parties shall not and shall not cause or permit their Subsidiaries to directly or indirectly enter into or permit to exist any transaction (including the purchase, sale, lease or exchange of any property or the rendering of any management, consulting, investment banking, advisory or other similar services) with any Affiliate or with any director, officer or employee of any Loan Party, except (a) as set forth on Schedule 6.8, (b) transactions in the ordinary course of and pursuant to the reasonable requirements of the business of any such Loan Party or any of its Subsidiaries and upon fair and reasonable terms which are (x) to the extent exceeding \$1,000,000, fully disclosed to Agent and (y) are no less favorable to any such Loan Party or any of its Subsidiaries than would be obtained in a comparable arm's length transaction with a Person that is not an Affiliate, (c) payment of reasonable compensation to officers and employees for services actually rendered to any such Loan Party or any of its Subsidiaries; (d) payment of director's fees not to exceed \$50,000 in the aggregate for any Fiscal Year of Borrower plus reasonable and documented expenses of the directors; (e) loans to employees permitted in Section 6.3, (f) reimbursement of employee travel, lodging and other costs incurred in the ordinary course of business, (g) the guaranty of the Obligations by Loan Parties, (h) employment agreements, equity incentive agreements and other employee and management arrangements in the ordinary course of business and (i) Borrower may pay (x) quarterly management fees to Eos

Management, Inc. or an Affiliate thereof pursuant to the Management Consulting Agreement not to exceed \$100,000 in the aggregate for any Fiscal Quarter and (y) indemnities pursuant to the terms of the Management Consulting Agreement; provided that no Event of Default has occurred and is continuing at the time that any such management fee or indemnities are paid or would result after giving effect thereto; provided further that it is expressly agreed that any such management fees and indemnities not permitted to be so paid shall be accrued and paid when such Event of Default has been cured or waived, and Borrower may reimburse the Sponsor for reasonable, out-of-pocket expenses pursuant to the terms of the Management Consulting Agreement.

- 6.9 <u>Conduct of Business</u>. Holdings and Intermediate Holdings shall not engage in any business activity other than its ownership of the Stock of its Subsidiaries and its performance of the Related Transaction Documents; provided, however that Intermediate Holdings may be a party to and perform its obligations under any employment agreement entered into in connection with the Related Transactions. The Loan Parties shall not and shall not cause or permit their Subsidiaries to directly or indirectly engage in any business other than businesses of the type described on <u>Schedule 6.9</u> and businesses substantially similar thereto.
- 6.10 Fiscal Year. The Loan Parties shall not and shall not cause or permit their Subsidiaries to change their Fiscal Year.
- 6.11 <u>Press Release; Public Offering Materials</u>. No Loan Party shall cause or permit the issuance of any press releases or other similar public disclosure, using the name of Freeport or its affiliates or referring to this Agreement, without giving Freeport a reasonable opportunity to review and comment thereon prior to issuance.
- 6.12 <u>Subsidiaries</u>. The Loan Parties shall not and shall not cause or permit their Subsidiaries to directly or indirectly establish, create or acquire any new Subsidiary unless such Subsidiary complies with the requirement of Section 5.7(c) with respect to such Subsidiary.
- 6.13 <u>Deposit Accounts</u>. The Loan Parties shall not and shall not cause or permit their Subsidiaries to establish any new deposit accounts (other than any zero balance account) without prior written notice to Agent and unless Agent and the bank at which the account is to be opened enter into a Control Agreement as required by Section 5.9.
- 6.14 ERISA. The Loan Parties shall not and shall not cause or permit any ERISA Affiliate to, cause or permit to occur an ERISA Event to the extent such ERISA Event would reasonably be expected to have a Material Adverse Effect.
- 6.15 <u>Sale-Leasebacks</u>. The Loan Parties shall not and shall not cause or permit their Subsidiaries to engage in any sale-leaseback, synthetic lease or similar transaction.

### 6.16 Changes to Material Agreements.

None of the Loan Parties shall change or amend the terms of the Acquisition Note if such changes or amendment would be adverse in any material respect to the rights or interests of the Loan Parties, the Agent or the Lenders.

## SECTION 7. FINANCIAL COVENANTS/REPORTING

Borrower covenants and agrees that from and after the date hereof until the Termination Date, Borrower shall perform and comply with, and shall cause each of the other Loan Parties to perform and comply with, all covenants in this <u>Section 7</u> applicable to such Person.

## 7.1 Financial Covenants.

(a) <u>Capital Expenditure Limits</u>. Holdings and its Subsidiaries on a consolidated basis shall not make Capital Expenditures during the following periods that exceed in the aggregate the amounts set forth opposite each of such periods (the "Capex Limit"):

	Maximum Capital
<u>Period</u>	Expenditures per Period
Closing Date through Fiscal Year ending December 31, 2006	\$ 1,000,000
Each Fiscal Year ending thereafter	\$ 1.250.000

<u>provided</u>, <u>however</u>, that commencing with Fiscal Year 2007, the Capex Limit referenced above will be increased in any period by the positive amount equal to the lesser of (i) 50% of the Capex Limit for the immediately prior period, and (ii) the amount (if any), equal to the difference obtained by taking the Capex Limit <u>minus</u> the actual amount of any Capital Expenditures expended during such prior period (the "<u>Carry Over Amount</u>"), and for purposes of measuring compliance herewith, the Carry Over Amount shall be deemed to be the last amount spent on Capital Expenditures in that succeeding period.

(b) Minimum EBITDA. Holdings and its Subsidiaries on a consolidated basis shall have, at the end of each Fiscal Quarter set forth below, Minimum EBITDA for the 12-Fiscal Month period then ended calculated of not less than the following:

Period	Mir	nimum EBITDA
December 31, 2006	\$	12,750,000
March 31, 2007	\$	12,500,000
June 30, 2007	\$	12,500,000
September 30, 2007	\$	12,500,000
December 31, 2007	\$	12,500,000
March 31, 2008	\$	12,750,000
June 30, 2008	\$	12,750,000
September 30, 2008	\$	13,000,000
December 31, 2008	\$	13,250,000
March 31, 2009	\$	13,500,000
June 30, 2009	\$	13,500,000
September 30, 2009	\$	13,750,000
December 31, 2009	\$	14,000,000
March 31, 2010	\$	14,000,000
June 30, 2010	\$	14,000,000
September 30, 2010	\$	14,250,000
December 31, 2010	\$	14,250,000
March 31, 2011 and each Fiscal Quarter ending thereafter	\$	14,500,000

- (c) <u>Minimum Fixed Charge Coverage Ratio</u>. Holdings and its Subsidiaries shall have on a consolidated basis at the end of each Fiscal Quarter set forth below, a Fixed Charge Coverage Ratio for the 12-Fiscal Month period then ended, of not less than the following:
  - 1.20:1.00 for the Fiscal Quarter ending December 31, 2006;
  - $1.20{:}1.00 \ for \ the \ Fiscal \ Quarter \ ending \ March \ 31, \ 2007;$
  - 1.20:1.00 for the Fiscal Quarter ending June 30, 2007;
  - 1.20:1.00 for the Fiscal Quarter ending September 30, 2007;
  - 1.20:1.00 for the Fiscal Quarter ending December 31, 2007;
  - 1.15:1.00 for the Fiscal Quarter ending March 31, 2008;
  - 1.15:1.00 for the Fiscal Quarter ending June 30, 2008;
  - 1.15:1.00 for the Fiscal Quarter ending September 30, 2008;
  - 1.15:1.00 for the Fiscal Quarter ending December 31, 2008;
  - 1.15:1.00 for the Fiscal Quarter ending March 31, 2009;
  - 1.15:1.00 for the Fiscal Quarter ending June 30, 2009; and
  - 1.10 for each Fiscal Quarter ending thereafter.
- (d) <u>Maximum Leverage Ratio</u>. Holdings and its Subsidiaries on a consolidated basis shall have, at the end of each Fiscal Quarter set forth below, a Leverage Ratio as of the last day of such Fiscal Quarter and for the 12-Fiscal Month period then ended, of not more than the following:
  - 4.25:1.00 for the Fiscal Quarter ending December 31, 2006;
  - 4.25:1.00 for the Fiscal Quarter ending March 31, 2007;
  - 4.25:1.00 for the Fiscal Quarter ending June 30, 2007;
  - 4.25:1.00 for the Fiscal Quarter ending September 30, 2007;
  - 4.25:1.00 for the Fiscal Quarter ending December 31, 2007;
  - 4.00:1.00 for the Fiscal Quarter ending March 31, 2008;

```
4.00:1.00 for the Fiscal Quarter ending June 30, 2008;
3.75:1.00 for the Fiscal Quarter ending September 30, 2008;
3.75:1.00 for the Fiscal Quarter ending December 31, 2008;
3.50:1.00 for the Fiscal Quarter ending March 31, 2009;
3.50:1.00 for the Fiscal Quarter ending June 30, 2009;
3.25:1.00 for the Fiscal Quarter ending September 30, 2009;
3.25:1.00 for the Fiscal Quarter ending December 31, 2009;
3.00:1.00 for the Fiscal Quarter ending March 31, 2010;
3.00:1.00 for the Fiscal Quarter ending June 30, 2010;
2.75:1.00 for the Fiscal Quarter ending September 30, 2010;
2.75:1.00 for the Fiscal Quarter ending December 31, 2010; and
2.50 for each Fiscal Quarter ending thereafter.
```

- 7.2 <u>Financial Statements and Other Reports</u>. Holdings will maintain, and cause each of its Subsidiaries to maintain, a system of accounting established and administered in accordance with sound business practices to permit preparation of Financial Statements in conformity with GAAP (it being understood that monthly Financial Statements are not required to have footnote disclosures). Borrower will deliver each of the Financial Statements and other reports described below to Agent and, upon receipt, Agent shall deliver such Financial Statements and other reports to each Lender.
- (a) Monthly Financials. As soon as available and in any event within thirty (30) days (provided that for the Fiscal Months ending September 30, 2006 and October 31, 2006, Borrower shall have forty-five (45) days to deliver such financial statements) after the end of each Fiscal Month (including the last Fiscal Month of Borrower's Fiscal Year), Borrower will deliver (1) the consolidated and consolidating balance sheets of Holdings and its Subsidiaries, as at the end of such month, and the related consolidated and consolidating statements of income, stockholders' equity and cash flow for such Fiscal Month and for the period from the beginning of the then current Fiscal Year of Holdings to the end of such Fiscal Month, (2) a report setting forth in comparative form the corresponding figures for the current Fiscal Year and the corresponding figures from the most recent Projections for the current Fiscal Year delivered pursuant to Section 7.2(h), and (3) a schedule of the outstanding Indebtedness for borrowed money of Holdings and its Subsidiaries describing in reasonable detail each such debt issue or loan outstanding and the principal amount and amount of accrued and unpaid interest with respect to each such debt issue or loan.
- (b) Quarterly Financials. As soon as available and in any event within forty-five (45) days after the end of each Fiscal Quarter (including the last Fiscal Quarter of Borrower's Fiscal Year), Borrower will deliver (1) the consolidated and consolidating balance sheets of Holdings and its Subsidiaries, as at the end of such quarter, and the related consolidated and consolidating statements of income (including separate statements of income by location consistent with past practices), stockholders' equity and cash flow for such Fiscal Quarter, (2) a report setting forth in comparative form the corresponding figures for the corresponding periods of the previous Fiscal Year and the corresponding figures from the most recent Projections for the current Fiscal Year delivered pursuant to Section 7.2(h), and (3) a schedule of the outstanding Indebtedness for borrowed money of Holdings and its Subsidiaries describing in reasonable detail each such debt issue or loan outstanding and the principal amount and amount of accrued and unpaid interest with respect to each such debt issue or loan.
- (c) <u>Year-End Financials</u>. As soon as available and in any event within one hundred and twenty (120) days after the end of each Fiscal Year of Holdings, Borrower will deliver (1) the consolidated and consolidating balance sheets of Holdings and its Subsidiaries, as at the end of such year,

and the related consolidated and consolidating statements of income (including separate statements of income by location consistent with past practices), stockholders' equity and cash flow for such Fiscal Year, (2) a report with respect to the consolidated Financial Statements from a firm of Certified Public Accountants selected by Borrower and reasonably acceptable to Agent, which report shall be prepared in accordance with Statement of Auditing Standards No. 58 (the "Statement") "Reports on Audited Financial Statements" and such report shall be "Unqualified" (as such term is defined in such Statement).

- (d) <u>Accountants' Reports</u>. Promptly upon receipt thereof, Borrower will deliver copies of all significant reports submitted by Borrower's firm of certified public accountants in connection with each annual, interim or special audit or review of any type of the Financial Statements or related internal control systems of Holdings or its Subsidiaries made by such accountants, including any comment letter submitted by such accountants to management in connection with their services.
- (e) <u>Borrowing Availability Certificate</u>. Together with each delivery of Financial Statements of Holdings pursuant to <u>Sections 7.2(a), (b)</u> and <u>(c)</u>, Borrower will deliver a Borrowing Availability Certificate (in substantially the same form as Exhibit 7.2(e), the "Borrowing Availability Certificate") as at the last day of such period.
- (f) Management Report. Together with each delivery of Financial Statements of Holdings pursuant to Sections 7.2(b) and (c), Borrower will deliver a management report (1) describing the operations and financial condition of Holdings and its Subsidiaries for the Fiscal Quarter then ended and the portion of the current Fiscal Year then elapsed (or for the Fiscal Year then ended in the case of year-end financials) and (2) discussing the reasons for any significant variations. The information above shall be presented in reasonable detail and shall be certified by the chief financial officer of Holdings or Borrower to the effect that such information fairly presents the results of operations and financial condition of Holdings and its Subsidiaries as at the dates and for the periods indicated.
- (g) <u>Appraisals</u>. From time to time, if Agent or any Lender determines that obtaining appraisals is necessary in order for Agent or such Lender to comply with applicable laws or regulations, Agent will, at Borrower's expense, obtain appraisal reports in form and substance and from appraisers reasonably satisfactory to Agent stating the then current fair market values of all or any portion of the Real Estate owned by Loan Parties.
- (h) <u>Projections</u>. As soon as available and in any event no later than the day thirty (30) days after the last day of each of Holdings' Fiscal Years, Borrower will deliver Projections of Holdings and its Subsidiaries for the forthcoming Fiscal Year month by month.
- (i) <u>SEC Filings and Press Releases</u>. Promptly upon their becoming available, Borrower will deliver copies of (1) all Financial Statements, reports, notices and proxy statements, material reports and material notices sent or made available by Holdings or any of its Subsidiaries to its Stockholders, (2) all regular and periodic reports and all registration statements and prospectuses, if any, filed by Holdings or any of its Subsidiaries with any securities exchange or with the Securities and Exchange Commission, any Governmental Authority or any private regulatory authority, and (3) all press releases and other statements made available by Holdings or any of its Subsidiaries to the public concerning developments in the business of any such Person.
- (j) Events of Default, Etc. Promptly upon any Responsible Officer of any Loan Party obtaining knowledge of any of the following events or conditions, Borrower shall deliver copies of all notices given or received by Borrower or Holdings or any of their Subsidiaries with respect to any

such event or condition and a certificate of Borrower's chief executive officer specifying the nature and period of existence of such event or condition and what action Holdings, Borrower or any of their Subsidiaries has taken, is taking and proposes to take with respect thereto: (1) any condition or event that constitutes, or which would reasonably be expected to result in the occurrence of, an Event of Default or Default; (2) any notice that any Person has given to Borrower or any of its Subsidiaries or any other action taken with respect to a claimed default or event or condition of the type referred to in Section 8.1(b); or (3) any event or condition that would reasonably be expected to result in any Material Adverse Effect.

- (k) <u>Litigation</u>. Promptly upon any officer of any Loan Party obtaining knowledge of (1) the institution of any action, charge, claim, demand, suit, proceeding, petition, governmental investigation, tax audit or arbitration now pending or, to the best knowledge of such Loan Party after due inquiry, threatened against or affecting any Loan Party or any of its Subsidiaries or any property of any Loan Party or any of its Subsidiaries ("Litigation") not previously disclosed by Borrower to Agent or (2) any material development in any action, suit, proceeding, governmental investigation or arbitration at any time pending against or affecting any Loan Party or any property of any Loan Party which, in each case, would reasonably be expected to have a Material Adverse Effect, Borrower will promptly give notice thereof to Agent and provide such other information as may be reasonably available to them to enable Agent and its counsel to evaluate such matter.
- (I) Notice of Corporate and other Changes. Borrower shall provide prompt written notice of (1) any change after the Closing Date in the authorized and issued Stock of any Loan Party or any amendment to their articles or certificate of incorporation, by-laws, partnership agreement or other organizational documents, (2) any Subsidiary created or acquired by any Loan Party or any of its Subsidiaries after the Closing Date, such notice, in each case, to identify the applicable jurisdictions, capital structures or Subsidiaries, as applicable, and (3) any other event that occurs after the Closing Date which would cause any of the representations and warranties in Section 4 of this Agreement or in any other Loan Document to be untrue or misleading in any material respect. The foregoing notice requirement shall not be construed to constitute consent by any of the Lenders to any transaction referred to above which is not expressly permitted by the terms of this Agreement.
- (m) Other Information. With reasonable promptness, Borrower will deliver such other information and data with respect to any Loan Party or any Subsidiary of any Loan Party as from time to time may be reasonably requested by Agent.
- (n) <u>Compliance, Pricing and Excess Cash Flow Certificate</u>. Together with each delivery of Financial Statements pursuant to Sections 7.2(b) and (c), Borrower will deliver a fully and properly completed Compliance, Pricing and Excess Cash Flow Certificate (in substantially the same form as Annex F (the "Compliance, Pricing and Excess Cash Flow Certificate") signed by Borrower's chief executive officer or chief financial officer; provided that the Excess Cash Flow portion of such certificate is only required to be delivered annually; provided further that Schedule 2 of the Compliance, Pricing and Excess Cash Flow Certificate shall be delivered only in connection with the Financial Statements of Borrower and its Subsidiaries delivered pursuant to Section 7.2(c).
- (o) <u>Taxes</u>. Borrower shall provide prompt written notice of (i) the execution or filing with the IRS or any other Governmental Authority of any agreement or other document extending, or having the effect of extending, the period for assessment or collection of any Charges by any Loan Party or any of its Subsidiaries and (ii) any agreement by any Loan Party or any of its Subsidiaries or request directed to any Loan Party or any of its Subsidiaries to make any adjustment under IRC Section 481(a), by reason of a change in accounting method or otherwise, which would reasonably be expected to have a Material Adverse Effect.

- (p) New Contracts. Borrower shall provide prompt written notice, together with copies thereof, of any material new national labor contracts to which it or any other Loan Party becomes a party or any material amendment to any material national labor contract to which it or any other Loan Party is a party.
- 7.3 Accounting Terms; Utilization of GAAP for Purposes of Calculations Under Agreement. For purposes of this Agreement, all accounting terms not otherwise defined herein shall have the meanings assigned to such terms in conformity with GAAP. Financial statements and other information furnished to Agent pursuant to Section 7.2 or any other section (unless specifically indicated otherwise) shall be prepared in accordance with GAAP as in effect at the time of such preparation; provided that no Accounting Change shall affect financial covenants, standards or terms in this Agreement; provided further that Borrower shall prepare footnotes to the Financial Statements required to be delivered hereunder that show the differences between the Financial Statements delivered (which reflect such Accounting Changes) and the basis for calculating financial covenant compliance (without reflecting such Accounting Changes). All such adjustments described in clause (c) of the definition of the term Accounting Changes resulting from expenditures made subsequent to the Closing Date (including capitalization of costs and expenses or payment of pre-Closing Date liabilities) shall be treated as expenses in the period the expenditures are made. Notwithstanding the foregoing, in the event that any Accounting Change shall occur and such change results in a change in the method of calculation of the financial covenants, standards or terms in this Agreement, then Borrower and Agent agree to negotiate in good faith in order to amend such provisions of this Agreement so as to equitably reflect such Accounting Changes with the desired result that the criteria for evaluating the financial condition of the Loan Parties shall be the same after such Accounting Changes as if such Accounting Changes had not been made. Until such time as such an amendment shall have been executed and delivered by Borrower, Agent and the Requisite Lenders, all financial covenants, standards and terms in this Agreement shall continue to be calculated or construed as if

# SECTION 8. <u>DEFAULT, RIGHTS AND REMEDIES</u>

- 8.1 Event of Default. "Event of Default" shall mean the occurrence or existence of any one or more of the following:
- (a) <u>Payment</u>. (1) Failure to pay any installment or other payment of principal of any Loan when due, or to timely repay Revolving Loans to reduce their balance to the maximum amount of Revolving Loans then permitted to be outstanding in accordance with Section 2.1(b)(i) or to reimburse any L/C Issuer for any payment made by such L/C Issuer under or in respect of any Letter of Credit when due or (2) failure to pay, within three (3) days after the due date, any interest or Fees on any Loan or within ten (10) days after request for payment thereof, any other amount due under this Agreement or any of the other Loan Documents; or
- (b) <u>Default in Other Agreements</u>. (1) Any Loan Party or any of its Subsidiaries fails to pay when due or within any applicable grace period any principal or interest on Indebtedness (other than the Loans) or any Contingent Obligations having an individual principal amount in excess of \$50,000 or having an aggregate principal amount in excess of \$50,000 or (2) breach or default of any Loan Party or any of its Subsidiaries, or the occurrence of any condition or event, with respect to any Indebtedness or any Contingent Obligations, if the effect of such breach, default or occurrence is to cause

or to permit the holder or holders then to cause, Indebtedness and/or Contingent Obligations having an individual principal amount in excess of \$50,000 or having an aggregate principal amount in excess of \$50,000 to become or be declared due prior to their stated maturity; or

- (c) <u>Breach of Certain Provisions</u>. Failure of any Loan Party to perform or comply with any term or condition contained in (1) the Freeport Fee Letter, (2) Section 7.2 which failure continues for more than five (5) Business Days after the date specified for performance or compliance with such term or condition, (3) that portion of Section 5.2 relating to the Loan Parties' obligation to maintain insurance, or (4) Section 5.3, Section 5.4, Section 5.12, Section 6 or Section 7.1; or
- (d) <u>Breach of Warranty</u>. Any representation, warranty, certification or other statement made by any Loan Party in any Loan Document or in any statement or certificate at any time given by such Person in writing pursuant or in connection with any Loan Document is false in any material respect (without duplication of materiality qualifiers contained therein) on the date made; or
- (e) Other Defaults Under Loan Documents. Any Loan Party defaults in the performance of or compliance with any term contained in this Agreement or the other Loan Documents (other than occurrences described in other provisions of this Section 8.1 for which a different grace or cure period is specified, or for which no cure period is specified and which constitute immediate Events of Default) and such default is not remedied or waived within thirty (30) days after the earlier of (1) receipt by Borrower of notice from Agent or Requisite Lenders of such default or (2) actual knowledge of an officer of Borrower or any other Loan Party of such default; or
- (f) <u>Involuntary Bankruptcy</u>; <u>Appointment of Receiver</u>, <u>Etc.</u> (1) A court enters a decree or order for relief with respect to any Loan Party in an involuntary case under the Bankruptcy Code, which decree or order is not stayed or other similar relief is not granted under any applicable federal or state law; or (2) the continuance of any of the following events for sixty (60) days unless dismissed, bonded or discharged: (a) an involuntary case is commenced against any Loan Party, under any applicable bankruptcy, insolvency or other similar law now or hereafter in effect; or (b) a decree or order of a court for the appointment of a receiver, liquidator, sequestrator, trustee, custodian or other officer having similar powers over any Loan Party, or over all or a substantial part of its property, is entered; or (c) a receiver, trustee or other custodian is appointed without the consent of a Loan Party, for all or a substantial part of the property of the Loan Party; or
- (g) <u>Voluntary Bankruptcy</u>; <u>Appointment of Receiver, Etc.</u> (1) any Loan Party commences a voluntary case under the Bankruptcy Code, or consents to the entry of an order for relief in an involuntary case or to the conversion of an involuntary case to a voluntary case under any such law or consents to the appointment of or taking possession by a receiver, trustee or other custodian for all or a substantial part of its property; or (2) any Loan Party makes any assignment for the benefit of creditors; or (3) the Board of Directors of any Loan Party adopts any resolution or otherwise authorizes action to approve any of the actions referred to in this Section 8.1(g); or
- (h) <u>Judgment and Attachments</u>. Any money judgment, writ or warrant of attachment, settlement, or similar process (other than those described elsewhere in this Section 8.1) involving (1) an amount in any individual case in excess of \$250,000 or (2) an amount in the aggregate at any time in excess of \$250,000 (in either case to the extent not covered by insurance) is entered or filed against one or more of the Loan Parties or any of their respective assets and remains undischarged, unvacated, unbonded or unstayed for a period of thirty (30) days or in any event later than five (5) Business Days prior to the date of any proposed sale thereunder; or

- (i) <u>Dissolution</u>. Any order, judgment or decree is entered against any Loan Party decreeing the dissolution or split up of such Loan Party and such order remains undischarged or unstayed for a period in excess of sixty (60) days; or
- (j) Solvency. Any Loan Party ceases to be Solvent, fails to pay its debts as they become due or admits in writing its present or prospective inability to pay its debts as they become due; or
- (k) <u>Invalidity of Loan Documents</u>. Any of the Loan Documents for any reason, other than a partial or full release in accordance with the terms thereof, ceases to be in full force and effect or is declared to be null and void, or any Loan Party denies that it has any further liability under any Loan Documents to which it is party, or gives notice to such effect;
  - (l) Change of Control. A Change of Control occurs;
- (m) Healthcare Defaults. (1) Any Loan Party or any Person within its control is required to pay a fine, penalty, settlement amount or other payment (whether imposed by judicial order or settlement) which, individually or in the aggregate, is in excess of \$5,000,000 for any violation or alleged violation of any Healthcare Law, including violation of the Federal Anti-Kickback Statute, Stark Law or False Claims Act, (2) any Loan Party or any Person within its control is required to pay a fine, penalty, settlement amount or other payment (whether imposed by judicial order or settlement) for any violation or alleged violation of any Healthcare Law, including violation of the Anti-kickback Statute, Stark Law or False Claims Act which, individually or in the aggregate, would result in, upon payment of such amount, the Borrower having Borrowing Availability of less than \$3,000,000, (3) any Loan Party or other Subsidiary of a Loan Party receives notice of the revocation of its Medicare or Medicaid certification and the loss of such certification which, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect, or (4) one or more Loan Parties or Subsidiaries of a Loan Party receives notice that such Person(s) will be excluded from participation in Medicare, Medicaid or other federal or state healthcare program and such exclusion would reasonably be expected to have a Material Adverse Effect;
- (n) <u>Termination of Contracts</u>. Any contract or contracts directly or indirectly between a Loan Party and the Illinois Department on Aging or any related entity, agency or sub-agency thereof are terminated, amended or modified, the result of which termination, amendment or modification would reasonably be expected to have a Material Adverse Effect; or
- (o) <u>Rate Management Agreement</u>. Nonpayment by Borrower of any Rate Management Obligation (i) consisting of a termination or equivalent payment, when due, (ii) consisting of a payment of interest or equivalent payment within three (3) days of the date on which such payment is due, or the breach by Borrower of any term, provision or condition contained in any Rate Management Agreement and the continuation of such breach for thirty (30) days.

- 8.2 <u>Suspension or Termination of Commitments</u>. Upon the occurrence of any Default or Event of Default, Agent may, and at the request of Requisite Revolving Lenders Agent shall, without notice or demand, immediately suspend or terminate all or any portion of Lenders' obligations to make additional Advances or issue or cause to be issued Letters of Credit under the Revolving Loan Commitment; <u>provided</u> that, in the case of a Default, if the subject condition or event is waived by Requisite Revolving Lenders or cured within any applicable grace or cure period, the Revolving Loan Commitment shall be reinstated.
- 8.3 Acceleration and other Remedies. Upon the occurrence of any Event of Default described in Sections 8.1(f) or 8.1(g), the Commitments shall be immediately terminated and all of the Obligations, including the Revolving Loans, shall automatically become immediately due and payable, without presentment, demand, protest, notice of intent to accelerate, notice of acceleration or other requirements of any kind, all of which are hereby expressly waived by Borrower, and the Commitments shall thereupon terminate. Upon the occurrence and during the continuance of any other Event of Default, Agent may, and at the request of the Requisite Lenders, Agent shall, by written notice to Borrower (a) reduce the aggregate amount of the Commitments from time to time, (b) declare all or any portion of the Loans and all or any portion of the other Obligations to be, and the same shall forthwith become, immediately due and payable together with accrued interest thereon, (c) terminate all or any portion of the obligations of Agent, L/C Issuers and Lenders to make Revolving Credit Advances and issue Letters of Credit, (d) demand that Borrower immediately deliver to Agent either (i) cash for the benefit of L/C Issuers (and Borrower shall then immediately so deliver) in an amount equal to 105% of the aggregate outstanding Letter of Credit Obligations or (ii) an Acceptable Standby Letter of Credit for the benefit of the L/C Issuers (and Borrower shall then immediately so deliver), and (e) exercise any other remedies which may be available under the Loan Documents or applicable law. Borrower hereby grants to Agent, for the benefit of L/C Issuers and each Lender with a participation in any Letters of Credit then outstanding, a security interest in such cash collateral or Acceptable Standby Letter of Credit to secure all of the Letter of Credit Obligations. Any such cash collateral or Acceptable Standby Letter of Credit shall be made available by Agent to L/C Issuers to reimburse L/C Issuers for payments of drafts drawn under such Letters of Credit and any Fees, Charges and expenses of L/C Issuers with respect to such Letters of Credit and the unused portion thereof, after all such Letters of Credit shall have expired or been fully drawn upon, shall be applied to pay any other Obligations. After all such Letters of Credit shall have expired or been fully drawn upon and the Termination Date shall have occurred, the balance, if any, of such cash collateral or Acceptable Standby Letter of Credit shall be (subject to any rights of third parties and except as otherwise directed by a court of competent jurisdiction) returned to Borrower. Borrower shall from time to time execute and deliver to Agent such further documents and instruments as Agent may reasonably request with respect to such cash collateral.

8.4 <u>Performance by Agent</u>. If any Loan Party shall fail to perform any covenant, duty or agreement contained in any of the Loan Documents which failure constitutes an Event of Default, Agent may perform or attempt to perform such covenant, duty or agreement on behalf of such Loan Party after the expiration of any cure or grace periods set forth herein. In such event, such Loan Party shall, at the request of Agent, promptly pay any amount reasonably expended by Agent in such performance or attempted performance to Agent, together with interest thereon at the highest rate of interest in effect upon the occurrence of an Event of Default as specified in <u>Section 2.2(d)</u> from the date of such expenditure until paid. Notwithstanding the foregoing, it is expressly agreed that Agent shall not have any liability or responsibility for the performance of any obligation of any Loan Party under this Agreement or any other Loan Document.

8.5 Application of Proceeds. Notwithstanding anything to the contrary contained in this Agreement, upon the occurrence and during the continuance of an Event of Default, Borrower irrevocably waives the right to direct the application of any and all payments at any time or times thereafter received by Agent from or on behalf of Borrower, and Agent shall have the continuing and exclusive right to apply and to reapply any and all payments received at any time or times after the occurrence and during the continuance of an Event of Default. Notwithstanding anything to the contrary contained in this Agreement (including, without limitation, Section 2.1 and Section 2.5 hereof), all payments (including the proceeds of any Asset Disposition or other sale of, or other realization upon, all or any part of the Collateral) received after acceleration of the Obligations shall be applied as follows: first, to all costs and expenses incurred by or owing to Agent and any Lender with respect to this Agreement, the other Loan Documents or the Collateral; second, to accrued and unpaid interest and Fees with respect to the Obligations (including any interest which but for the provisions of the Bankruptcy Code, would have accrued on such amounts); third, to the principal amount of the Obligations outstanding (other than Obligations owed to any Lender under an Interest Rate Agreement) and to cash collateralize outstanding Letters of Credit (pro rata among all such Obligations (based upon the principal amount thereof or the outstanding face amount of such Letters of Credit, as applicable) (and with respect to amounts applied to Term Loans, pro rata among all remaining Scheduled Installments thereof); and fourth to any other obligations of Borrower owing to Agent or any Lender under the Loan Documents or any Interest Rate Agreement. Any balance remaining shall be delivered to Borrower or to whomever may be lawfully entitled to receive such balance or as a court of competent jurisdiction may direct.

## SECTION 9. ASSIGNMENT AND PARTICIPATION

### 9.1 Assignment and Participations

(a) Subject to the terms of this Section 9.1, any Lender may make an assignment to a Qualified Assignee of, or sale of participations in, at any time or times, the Loan Documents, Loans, Letter of Credit Obligations and any Commitment or any portion thereof or interest therein, including any Lender's rights, title, interests, remedies, powers or duties thereunder. Any assignment by a Lender shall: (i) require the consent of Agent (which consent shall not be unreasonably withheld or delayed with respect to a Qualified Assignee and which consent is not required by an assignment between Lenders or from a Lender to an Affiliate of a Lender or an Approved Fund) and the execution of an assignment agreement (an "Assignment Agreement" substantially in the form attached hereto as Exhibit 9.1 and otherwise in form and substance reasonably satisfactory to, and acknowledged by, Agent); (ii) be conditioned on such assignee Lender representing to the assigning Lender and Agent and the Borrower that it is purchasing the applicable Loans to be assigned to it for its own account, for investment purposes and not with a view to the distribution thereof; (iii) except with respect to any assignment by a Lender to an Affiliate or an Approved Fund of such Lender, after giving effect to any such partial assignment, the assignee Lender shall have Commitments in an amount at least equal to \$5,000,000 and the assigning Lender shall have retained Commitments in an amount at least equal to \$2,500,000; (iv) require a payment to Agent of an assignment between Lenders or from a Lender to an Approved Fund of a Lender), which shall not be unreasonably withheld or delayed. Notwithstanding the above, Agent may in its sole and absolute discretion permit any assignment by a Lender to a Person or Persons that are not Qualified Assignees, subject to Borrower's consent rights as set forth above. In the case of an assignment by a Lender that has become effective under this Section 9.1, (i) the assignee shall have, to the extent of such assignment, the

relieved of its obligations hereunder with respect to its Commitments or assigned portion thereof and the Loans, Letter of Credit Obligations and other interests assigned by it from and after the effective date of such assignment. Borrower hereby acknowledges and agrees that any assignment shall give rise to a direct obligation of Borrower to the assignee and that the assignee shall be considered to be a "Lender." In all instances, each Lender's liability to make Loans hereunder shall be several and not joint and shall be limited to such Lender's Pro Rata Share of the applicable Commitment. In the event Agent or any Lender assigns or otherwise transfers all or any part of the Obligations, Agent or any such Lender shall so notify Borrower and Borrower shall, upon the request of Agent or such Lender, execute new Notes in exchange for the Notes, if any, being assigned. Notwithstanding the foregoing provisions of this Section 9.1(a), (a) any Lender may at any time pledge the Obligations held by it and such Lender's rights under this Agreement and the other Loan Documents to a Federal Reserve Bank, (b) any Lender that is an investment fund may assign the Obligations held by it and such Lender's rights under this Agreement and the other Loan Documents to another investment fund managed by the same investment advisor or pledge such Obligations and rights to a trustee for the benefit of its investors and (c) any Lender may assign the Obligations to an Affiliate of such Lender or to a Person that is a Lender prior to the date of such assignment. Notwithstanding any other provision of this Agreement to the contrary, neither the Lenders nor any of their successors or assigns shall assign or transfer any interest herein without obtaining a representation from such successor or assign that any such assignment or transfer would not result in a non-exempt prohibited transaction under Section 406 of ERISA or Section 4975 of the IRC with respect to the Plans.

- (b) Any participation by a Lender of all or any part of its Commitments shall be made with the understanding that all amounts payable by Borrower hereunder shall be determined as if that Lender had not sold such participation, and that the holder of any such participation shall not be entitled to require such Lender to take or omit to take any action hereunder except actions directly affecting (i) any reduction in the principal amount of, or interest rate or Fees payable with respect to, any Loan in which such holder participates, (ii) any extension of the scheduled amortization of the principal amount of any Loan in which such holder participates or the final maturity date thereof, and (iii) any release of all or substantially all of the Collateral (other than in accordance with the terms of this Agreement, the Collateral Documents or the other Loan Documents). Solely for purposes of Sections 2.8, 2.9, 9.3 and 10.1, Borrower acknowledges and agrees that a participation shall give rise to a direct obligation of Borrower to the participant and the participant shall be considered to be a "Lender"; provided, (i) that any such participant shall not be entitled to receive any greater payment under Section 2.8 or Section 2.9 than the Lender granting such participation would have been entitled to receive with respect to the portion of its Commitment and Loans so participated; and (ii) with respect to Section 2.9, only to the extent such participant delivers the tax forms such Lender is required to collect under Section 2.9. Except as set forth in the preceding sentence no Borrower or any other Loan Party shall have any obligation or duty to any participant. Neither Agent nor any Lender (other than the Lender selling a participation) shall have any duty to any participant and may continue to deal solely with the Lender selling a participation as if no such sale had occurred.
- (c) Except as expressly provided in this Section 9.1, no Lender shall, as between Borrower and that Lender, or Agent and that Lender, be relieved of any of its obligations hereunder as a result of any sale, assignment, transfer or negotiation of, or granting of participation in, all or any part of the Loans, the Notes or other Obligations owed to such Lender.
- (d) Each Loan Party shall assist each Lender permitted to sell assignments or participations under this Section 9.1 as required to enable the assigning or selling Lender to effect any such assignment or participation, including the execution and delivery of any and all agreements, notes and other documents and instruments on substantially the same terms as and consistent with this

Agreement and the other Loan Documents as shall be requested and, prior to completion of the primary syndication (as described in the Freeport Fee Letter), the prompt preparation of informational materials for, and the participation of management in meetings with, potential assignees or participants, all on a timetable reasonably established by Agent in its sole discretion. Each Loan Party executing this Agreement shall certify the correctness, completeness and accuracy of all descriptions of the Loan Parties and their respective affairs contained in any selling materials provided by it and all other information provided by it and included in such materials, except that any Projections delivered by Borrower shall only be certified by Borrower as having been prepared by Borrower in compliance with the representations contained in Section 4.5. Agent shall maintain, on behalf of Borrower, in its offices located in Chicago, Illinois a "register" for recording the name, address, commitment and Loans owing to each Lender. The entries in such register shall be conclusive evidence of the amounts due and owing to each Lender in the absence of manifest error. Borrower, Agent and each Lender may treat each Person whose name is recorded in such register pursuant to the terms hereof as a Lender for all purposes of this Agreement. The register described herein shall be available for inspection by Borrower and any Lender, at any reasonable time upon reasonable prior notice. Notwithstanding any other provision of this Agreement to the contrary, neither the Lenders nor any of their successors or assigns shall assign or transfer any interest herein without obtaining a representation from such successor or assign that any such assignment or transfer would not result in a non-exempt prohibited transaction under Section 406 of ERISA or Section 4975 of the IRC with respect to the Plans.

(e) A Lender may furnish any information concerning Loan Parties in the possession of such Lender from time to time to assignees and participants (including prospective assignees and participants); provided that such Lender shall obtain from assignees or participants (and prospective assignees and prospective participants) confidentiality covenants substantially equivalent to those contained in Section 10.13.

### 9.2 Agent.

- (a) <u>Appointment</u>. Each Lender hereby designates and appoints Freeport as its Agent under this Agreement and the other Loan Documents, and each Lender hereby irrevocably authorizes Agent to execute and deliver the Collateral Documents and to take such action or to refrain from taking such action on its behalf under the provisions of this Agreement and the other Loan Documents and to exercise such powers as are set forth herein or therein, together with such other powers as are reasonably incidental thereto. Agent is authorized and empowered to amend, modify, or waive any provisions of this Agreement or the other Loan Documents on behalf of Lenders subject to the requirement that certain of Lenders' consent be obtained in certain instances as provided in this Section 9.2 and Section 10.2. The provisions of this Section 9.2 are solely for the benefit of Agent and Lenders and neither Borrower nor any other Loan Party shall have any rights as a third party beneficiary of any of the provisions hereof. In performing its functions and duties under this Agreement, Agent shall act solely as agent of Lenders and does not assume and shall not be deemed to have assumed any obligation toward or relationship of agency or trust with or for Borrower or any other Loan Party. Agent may perform any of its duties hereunder, or under the Loan Documents, by or through its agents or employees.
- (b) Nature of Duties. The duties of Agent shall be mechanical and administrative in nature. Agent shall not have by reason of this Agreement a fiduciary relationship in respect of any Lender. Nothing in this Agreement or any of the Loan Documents, express or implied, is intended to or shall be construed to impose upon Agent any obligations in respect of this Agreement or any of the Loan Documents except as expressly set forth herein or therein. Each Lender shall make its own independent investigation of the financial condition and affairs of each Loan Party in connection with the extension of credit hereunder and shall make its own appraisal of the creditworthiness of each Loan Party, and Agent

shall have no duty or responsibility, either initially or on a continuing basis, to provide any Lender with any credit or other information with respect thereto (other than as expressly required herein). If Agent seeks the consent or approval of any Lenders to the taking or refraining from taking any action hereunder, then Agent shall send notice thereof to each Lender. Agent shall promptly notify each Lender any time that the Requisite Lenders, Requisite Revolving Lenders or Supermajority Revolving Lenders have instructed Agent to act or refrain from acting pursuant hereto.

- (c) Rights, Exculpation, Etc. Neither Agent nor any of its officers, directors, employees or agents shall be liable to any Lender for any action taken or omitted by them hereunder or under any of the Loan Documents, or in connection herewith or therewith, except that Agent shall be liable to the extent of its own gross negligence or willful misconduct as determined by a final non-appealable order by a court of competent jurisdiction. Agent shall not be liable for any apportionment or distribution of payments made by it in good faith and if any such apportionment or distribution is subsequently determined to have been made in error the sole recourse of any Lender to whom payment was due but not made, shall be to recover from other Lenders any payment in excess of the amount to which they are determined to be entitled (and such other Lenders hereby agree to return to such Lender any such erroneous payments received by them). In no event shall Agent be liable for punitive, special, consequential, incidental, exemplary or other similar damages. In performing its functions and duties hereunder, Agent shall exercise the same care which it would in dealing with loans for its own account, but neither Agent nor any of its agents or representatives shall be responsible to any Lender for any recitals, statements, representations or warranties herein or for the execution, effectiveness, genuineness, validity, enforceability, collectibility, or sufficiency of this Agreement or any of the Loan Documents or the transactions contemplated thereby, or for the financial condition of any Loan Party. Agent shall not be required to make any inquiry concerning either the performance or observance of any of the terms, provisions or conditions of this Agreement or any of the Loan Documents or the financial condition of any Loan Party, or the existence or possible existence of any Default or Event of Default. Agent may at any time request instructions from Requisite Lenders, Requisite Revolving Lenders, Supermajority Revolving Lenders or all affected Lenders with respect to any actions or approvals which by the terms of this Agreement or of any of the Loan Documents Agent is permitted or required to take or to grant. If such instructions are promptly requested, Agent shall be absolutely entitled to refrain from taking any action or to withhold any approval and shall not be under any liability whatsoever to any Person for refraining from any action or withholding any approval under any of the Loan Documents until it shall have received such instructions from the Requisite Lenders, Requisite Revolving Lenders, Supermajority Revolving Lenders or such other portion of the Lenders as shall be prescribed by this Agreement. Without limiting the foregoing, no Lender shall have any right of action whatsoever against Agent as a result of Agent acting or refraining from acting under this Agreement or any of the other Loan Documents in accordance with the instructions of Requisite Lenders, Requisite Revolving Lenders, Supermajority Revolving Lenders or all affected Lenders, as applicable; and, notwithstanding the instructions of Requisite Lenders, Requisite Revolving Lenders, Supermajority Revolving Lenders or all affected Lenders, as applicable, Agent shall have no obligation to take any action if it believes, in good faith, that such action is deemed to be illegal by Agent or exposes Agent to any liability for which it has not received satisfactory indemnification in accordance with Section 9.2(e).
- (d) Reliance. Agent shall be entitled to rely, and shall be fully protected in relying, upon any written or oral notices, statements, certificates, orders or other documents or any telephone message or other communication (including any writing, telex, fax or telegram) believed by it in good faith to be genuine and correct and to have been signed, sent or made by the proper Person, and with respect to all matters pertaining to this Agreement or any of the Loan Documents and its duties hereunder or thereunder. Agent shall be entitled to rely upon the advice of legal counsel, independent accountants, and other experts selected by Agent in its sole discretion

- (e) <u>Indemnification</u>. Lenders will reimburse and indemnify Agent for and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses (including, without limitation, attorneys' fees and expenses), advances or disbursements of any kind or nature whatsoever which may be imposed on, incurred by, or asserted against Agent in its capacity as such in any way relating to or arising out of this Agreement or any of the Loan Documents or any action taken or omitted by Agent in its capacity as such under this Agreement or any of the Loan Documents, in proportion to each Lender's Pro Rata Share, but only to the extent that any of the foregoing is not reimbursed by Borrower; provided, however, that no Lender shall be liable for any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses, advances or disbursements to the extent resulting from Agent's gross negligence or willful misconduct as determined by a final non-appealable order by a court of competent jurisdiction. If any indemnity furnished to Agent for any purpose shall, in the opinion of Agent, be insufficient or become impaired, Agent may call for additional indemnity and cease, or not commence, to do the acts indemnified against even if so directed by the Requisite Lenders, Requisite Revolving Lenders, Supermajority Revolving Lenders or such other portion of the Lenders as shall be prescribed by this Agreement until such additional indemnity is furnished. The obligations of Lenders under this Section 9.2(e) shall survive the payment in full of the Obligations and the termination of this Agreement.
- (f) Freeport (or any successor Agent) Individually. With respect to its Commitments hereunder, Freeport (or any successor Agent) shall have and may exercise the same rights and powers hereunder and is subject to the same obligations and liabilities as and to the extent set forth herein for any other Lender. The terms "Lenders," "Requisite Lenders," "Requisite Revolving Lenders," "Supermajority Revolving Lenders" or any similar terms shall, unless the context clearly otherwise indicates, include Freeport (or any successor Agent) in its individual capacity as a Lender or one of the Requisite Lenders, Requisite Revolving Lenders or Supermajority Revolving Lenders. Freeport (or any successor Agent), either directly or through strategic affiliations, may lend money to, acquire equity or other ownership interests in, provide advisory services to and generally engage in any kind of banking, trust or other business with any Loan Party as if it were not acting as Agent pursuant hereto and without any duty to account therefor to Lenders. Freeport (or any successor Agent), either directly or through strategic affiliations, may accept fees and other consideration from any Loan Party for services in connection with this Agreement or otherwise without having to account for the same to Lenders.

### (g) Successor Agent.

- (i) <u>Resignation</u>. Agent may resign from the performance of all its agency functions and duties hereunder at any time by giving at least thirty (30) Business Days' prior written notice to Borrower and Lenders. Such resignation shall take effect upon the acceptance by a successor Agent of appointment pursuant to clause (ii) below or as otherwise provided in clause (ii) below.
- (ii) <u>Appointment of Successor</u>. Upon any such notice of resignation pursuant to clause (i) above, Requisite Lenders shall appoint a successor Agent which, unless an Event of Default has occurred and is continuing, shall be subject to Borrower's approval (which shall not be unreasonably withheld or delayed). If a successor Agent shall not have been so appointed within the thirty (30) Business Day period referred to in clause (i) above, the retiring Agent, upon notice to Borrower, shall then appoint a successor Agent which, unless an Event of Default has occurred and is continuing shall be reasonably acceptable to Borrower (such consent not to be unreasonably withheld) who shall serve as Agent until such time, if any, as Requisite Lenders appoint a successor Agent as provided above.

(iii) <u>Successor Agent</u>. Upon the acceptance of any appointment as Agent under the Loan Documents by a successor Agent, such successor Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Agent, and the retiring Agent shall be discharged from its duties and obligations under the Loan Documents. After any retiring Agent's resignation as Agent, the provisions of this <u>Section 9.2</u> shall continue to inure to its benefit as to any actions taken or omitted to be taken by it in its capacity as Agent.

## (h) Collateral Matters.

- (i) Release of Collateral. Lenders hereby irrevocably authorize Agent, at its option and in its discretion, to release any Lien granted to or held by Agent upon any Collateral (x) on the Termination Date, (y) constituting property being sold or disposed of if Borrower certifies to Agent that the sale or disposition is made in compliance with the provisions of this Agreement (and Agent may rely in good faith conclusively on any such certificate, without further inquiry) or (z) in accordance with the provisions of the next sentence. In addition, with the consent of Requisite Lenders, during any Fiscal Year Agent may release any Lien granted to or held by Agent upon any Collateral having a book value not greater than ten percent (10%) of the total book value of all Collateral as of the first day of such Fiscal Year.
- (ii) Confirmation of Authority; Execution of Releases. Without in any manner limiting Agent's authority to act without any specific or further authorization or consent by Lenders (as set forth in Section 9.2(h)(i)), each Lender agrees to confirm in writing, upon request by Agent or Borrower, the authority to release any Collateral conferred upon Agent under clauses (x) and (y) of Section 9.2(h)(i). Upon not less than five (5) Business Days' prior written request by Borrower, Agent shall (and is hereby irrevocably authorized by Lenders to) execute such documents as may be necessary to evidence the release of the Liens granted to Agent upon such Collateral; provided, however, that (x) Agent shall not be required to execute any such document on terms which, in Agent's opinion, would expose Agent to liability or create any obligation or entail any consequence other than the release of such Liens without recourse or warranty, and (y) such release shall not in any manner discharge, affect or impair the Obligations or any Liens upon (or obligations of any Loan Party, in respect of), all interests retained by any Loan Party, including the proceeds of any sale, all of which shall continue to constitute part of the Collateral.
- (iii) Absence of Duty. Agent shall have no obligation whatsoever to any Lender or any other Person to assure that the property covered by the Collateral Documents exists or is owned by Borrower or any other Loan Party or is cared for, protected or insured or has been encumbered or that the Liens granted to Agent have been properly or sufficiently or lawfully created, perfected, protected or enforced or are entitled to any particular priority, or to exercise at all or in any particular manner or under any duty of care, disclosure or fidelity, or to continue exercising, any of the rights, authorities and powers granted or available to Agent in this Section 9.2(h) or in any of the Loan Documents, it being understood and agreed that in respect of the property covered by the Collateral Documents or any act, omission or event related thereto, Agent may act in any manner it may deem appropriate, in its discretion, given Agent's own interest in property covered by the Collateral Documents as one of the Lenders and that Agent shall have no duty or liability whatsoever to any of the other Lenders, provided that Agent shall exercise the same care which it would in dealing with loans for its own account.
- (i) Agency for Perfection. Agent and each Lender hereby appoint each other Lender as agent for the purpose of perfecting Agent's security interest in assets which, in accordance with the

Code in any applicable jurisdiction, can be perfected by possession or control. Should any Lender (other than Agent) obtain possession or control of any such assets, such Lender shall notify Agent thereof, and, promptly upon Agent's request therefor, shall deliver such assets to Agent or in accordance with Agent's instructions or transfer control to Agent in accordance with Agent's instructions. Each Lender agrees that it will not have any right individually to enforce or seek to enforce any Collateral Document or to realize upon any collateral security for the Loans unless instructed to do so by Agent in writing, it being understood and agreed that such rights and remedies may be exercised only by Agent.

- (j) Notice of Default. Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default except with respect to defaults in the payment of principal, interest and Fees required to be paid to Agent for the account of Lenders, unless Agent shall have received written notice from a Lender or Borrower referring to this Agreement, describing such Default or Event of Default and stating that such notice is a "notice of default". Agent will use reasonable efforts to notify each Lender of its receipt of any such notice, unless such notice is with respect to defaults in the payment of principal, interest and fees, in which case Agent will notify each Lender of its receipt of such notice. Agent shall take such action with respect to such Default or Event of Default as may be requested by Requisite Lenders in accordance with Section 8. Unless and until Agent has received any such request, Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable or in the best interests of Lenders.
- (k) Lender Actions Against Collateral. Each Lender agrees that it will not take any enforcement action, nor institute any actions or proceedings, with respect to the Loans, against Borrower or any Loan Party hereunder or under the other Loan Documents or against any Collateral (including the exercise of any right of set-off) without the consent of the Agent or Requisite Lenders. All such enforcement actions and proceedings shall be (i) taken in concert and (ii) at the direction of or with the consent of Agent or Requisite Lenders. With respect to any action by Agent to enforce the rights and remedies of Agent and the Lenders under this Agreement and the other Loan Documents, each Lender hereby consents to the jurisdiction of the court in which such action is maintained, and agrees to deliver its Notes to Agent to the extent necessary to enforce the rights and remedies of Agent for the benefit of the Lenders under the Collateral Documents in accordance with the provisions hereof.
- (l) <u>Agent Reports</u>. Each Lender may from time to time receive one or more reports or other information (each, a "<u>Report</u>") prepared by or on behalf of Agent (or one or more of Agent's Affiliates). With respect to each Report, each Lender hereby agrees that:
- (i) Agent (and Agent's Affiliates) shall have no duties or obligations in connection with or as a result of a Lender receiving a copy of a Report, which will be provided solely as a courtesy, without consideration. Each Lender will perform its own diligence and will make its own independent investigation of the operations, financial conditions and affairs of the Loan Parties and will not rely on any Report or make any claim that it has done so. In addition, each Lender releases, and agrees that it will not assert, any claim against Agent (or one or more of Agent's Affiliates) that in any way relates to any Report or arises out of a Lender having access to any Report or any discussion of its contents, and each Lender agrees to indemnify and hold harmless Agent (and Agent's Affiliates) and their respective officers, directors, employees, agents and attorneys from all claims, liabilities and expenses relating to a breach by a Lender or any of its personnel of this Section or otherwise arising out of a Lender's access to any Report or any discussion of its contents;
- (ii) Each Report may not be complete and certain information and findings obtained by Agent (or one or more of Agent's Affiliates) regarding the operations and condition of the

Loan Parties may not be reflected in each Report. Agent (and Agent's Affiliates) makes no representations or warranties of any kind with respect to (i) any existing or proposed financing; (ii) the accuracy or completeness of the information contained in any Report or in any other related documentation; (iii) the scope or adequacy of Agent's (and Agent's Affiliates') due diligence, or the presence or absence of any errors or omissions contained in any Report or in any other related documentation; and (iv) any work performed by Agent (or one or more of Agent's Affiliates) in connection with or using any Report or any related documentation; and

(iii) Each Lender agrees to safeguard each Report and any related documentation with the same care which it uses with respect to information of its own which it does not desire to disseminate or publish, and agrees not to reproduce or distribute or provide copies of or disclose any Report or any other related documentation or any related discussions to anyone.

9.3 Set Off and Sharing of Payments. Subject to Section 9.2(k), in addition to any rights now or hereafter granted under applicable law and not by way of limitation of any such rights, during the continuance of any Event of Default, each Lender is hereby authorized by Borrower at any time or from time to time, with reasonably prompt subsequent notice to Borrower (any prior or contemporaneous notice being hereby expressly waived) to the extent permitted by law to set off and to appropriate and to apply any and all (A) balances held by such Lender at any of its offices for the account of Borrower or any of its Subsidiaries (regardless of whether such balances are then due to Borrower or its Subsidiaries), and (B) other property at any time held or owing by such Lender to or for the account of Borrower or any of its Subsidiaries, against and on account of any of the Obligations; except that no Lender shall exercise any such right without the prior written consent of Agent. Notwithstanding anything herein to the contrary, the failure to give notice of any set off and application made by such Lender to Borrower shall not affect the validity of such set off and application. Any Lender exercising a right to set off shall purchase for cash (and the other Lenders shall sell) interests in each of such other Lender's Pro Rata Share of the Obligations as would be necessary to cause all Lenders to share the amount so set off with each other Lender may exercise its right to set off with respect to amounts in excess of its Pro Rata Share of the Obligations and upon doing so shall deliver such amount so set off to the Agent for the benefit of all Lenders entitled to share in the amount so set off in accordance with their Pro Rata Shares.

9.4 <u>Disbursement of Funds</u>. Agent may, on behalf of Lenders, disburse funds to Borrower for Loans requested. Each Lender shall reimburse Agent on demand for all funds disbursed on its behalf by Agent, or if Agent so requests, each Lender will remit to Agent its Pro Rata Share of any Loan before Agent disburses same to Borrower. If Agent elects to require that each Lender make funds available to Agent prior to a disbursement by Agent to Borrower, Agent shall advise each Lender by telephone or fax of the amount of such Lender's Pro Rata Share of the Loan requested by Borrower no later than noon (Chicago time) on the Funding Date applicable thereto, and each such Lender shall pay Agent such Lender's Pro Rata Share of such requested Loan, in same day funds, by wire transfer to Agent's account on such Funding Date. If any Lender fails to pay the amount of its Pro Rata Share within one (1) Business Day after Agent's demand, Agent shall promptly notify Borrower, and to the extent that Agent has funded the Pro Rata Share of such Lender, Borrower shall immediately repay such amount to Agent. Any repayment required pursuant to this <u>Section 9.4</u> shall be without premium or penalty and without payment of any LIBOR Breakage Costs. Nothing in this <u>Section 9.4</u> or elsewhere in this Agreement or the other Loan Documents, including the provisions of <u>Section 9.5</u>, shall be deemed to require Agent to advance funds on behalf of any Lender or to relieve any Lender from its obligation to fulfill its commitments hereunder or to prejudice any rights that Agent or Borrower may have against any Lender as a result of any default by such Lender hereunder.

## 9.5 Disbursements of Advances; Payment.

## (a) Advances; Payments.

(i) Each Revolving Lender shall make the amount of such Lender's Pro Rata Share of such Revolving Credit Advance available to Agent in same day funds by wire transfer to Agent's account as set forth in Section 2.1(e) not later than 2:00 p.m. (Chicago time) on the requested Funding Date in the case of an Index Rate Loans and not later than 10:00 a.m. (Chicago time) on the requested Funding Date in the case of a LIBOR Loan. After receipt of such wire transfers (or, in the Agent's sole discretion, before receipt of such wire transfers), subject to the terms hereof, Agent shall make the requested Revolving Credit Advance to Borrower as designated by Borrower in the Notice of Revolving Credit Advance. All payments by each Revolving Lender shall be made without setoff, counterclaim or deduction of any kind.

(ii) At least once each calendar week or more frequently at Agent's election (each, a "Settlement Date"), Agent shall advise each Lender by telephone or fax of the amount of such Lender's Pro Rata Share of principal, interest and Fees paid for the benefit of Lenders with respect to each applicable Loan. Provided that each Lender has funded all payments and Advances required to be made by it and funded all purchases of participations required to be funded by it under this Agreement and the other Loan Documents as of such Settlement Date, Agent shall pay to each Lender such Lender's Pro Rata Share of principal, interest and Fees paid by Borrower since the previous Settlement Date for the benefit of such Lender on the Loans held by it. Such payments shall be made by wire transfer to such Lender's account (as specified by such Lender in Annex D or the applicable Assignment Agreement) not later than 1:00 p.m. (Chicago time) on the next Business Day following each Settlement Date. To the extent that any Lender (a "Non-Funding Lender") has failed to fund all such payments and Advances or failed to fund the purchase of all such participations required to be funded by such Lender pursuant to this Agreement, Agent shall be entitled to set off the funding shortfall against that Non-Funding Lender's Pro Rata Share of all payments received from Borrower.

(b) Availability of Lender's Pro Rata Share. Agent may assume that each Revolving Lender will make its Pro Rata Share of each Revolving Credit Advance available to Agent on each Funding Date. If such Pro Rata Share is not, in fact, paid to Agent by such Revolving Lender when due, Agent will be entitled to recover such amount on demand from such Revolving Lender without setoff, counterclaim or deduction of any kind. If any Revolving Lender fails to pay the amount of its Pro Rata Share forthwith upon Agent's demand, Agent shall promptly notify Borrower and to the extent that Agent advanced such funds on behalf of such Lender to the Borrower, the Borrower shall immediately repay such amount to Agent without premium or penalty and without payment of LIBOR Breakage Costs. Nothing in this Section 9.5(b) or elsewhere in this Agreement or the other Loan Documents shall be deemed to require Agent to advance funds on behalf of any Revolving Lender or to relieve any Revolving Lender from its obligation to fulfill its Commitments hereunder or to prejudice any rights that Borrower may have against any Revolving Lender as a result of any default by such Revolving Lender hereunder. To the extent that Agent advances funds to Borrower on behalf of any Revolving Lender and is not reimbursed therefor on the same Business Day as such Advance is made, Agent shall be entitled to retain for its account all interest accrued on such Advance until reimbursed by the applicable Revolving Lender.

## (c) Return of Payments.

- (i) If Agent pays an amount to a Lender under this Agreement in the belief or expectation that a related payment has been or will be received by Agent from Borrower and such related payment is not received by Agent, then Agent will be entitled to recover such amount from such Lender on demand without setoff, counterclaim or deduction of any kind.
- (ii) If Agent determines at any time that any amount received by Agent under this Agreement must be returned to Loan Party or paid to any other Person pursuant to any insolvency law or otherwise, then, notwithstanding any other term or condition of this Agreement or any other Loan Document, Agent will not be required to distribute any portion thereof to any Lender. In addition, each Lender will repay to Agent on demand any portion of such amount that Agent has distributed to such Lender, together with interest at such rate, if any, as Agent is required to pay to Borrower or such other Person, without setoff, counterclaim or deduction of any kind.

(d) Non-Funding Lenders. The failure of any Non-Funding Lender to make any Revolving Credit Advance or any payment required by it hereunder, shall not relieve any other Lender (each such other Revolving Lender, an "Other Lender") of its obligations to make such Advance or fund the purchase of any such participation on such date, but neither any Other Lender nor Agent shall be responsible for the failure of any Non-Funding Lender to make an Advance, fund the purchase of a participation or make any other payment required hereunder. Notwithstanding anything set forth herein to the contrary, a Non-Funding Lender shall not have any voting or consent rights under or with respect to any Loan Document or constitute a "Lender" or a "Revolving Lender" (or be included in the calculation of "Requisite Lenders", "Requisite Revolving Lenders" or "Supermajority Revolving Lenders" hereunder) for any voting or consent rights under or with respect to any Loan Document.

## SECTION 10. MISCELLANEOUS

10.1 <u>Indemnities</u>. Borrower agrees to indemnify, pay, and hold Agent, each Lender, each L/C Issuer and their respective Affiliates, officers, directors, employees, agents, and attorneys (the "Indemnitees") harmless from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, claims, costs and expenses (including all reasonable fees and expenses of counsel to such Indemnitees) of any kind or nature whatsoever that may be imposed on, incurred by, or asserted against the Indemnitee as a result of such Indemnitees being a party to this Agreement or the transactions consummated pursuant to this Agreement or otherwise relating to any of the Loan Documents or Related Transactions; provided, that Borrower shall have no obligation to an Indemnitee hereunder (w) with respect to any costs and expenses which are specifically excluded in Section 2.3(e), (x) with respect to liabilities to the extent resulting from the gross negligence or willful misconduct of that Indemnitee as determined by a court of competent jurisdiction or (y) other than any Agent Indemnitee in its capacity as such, from any investigation or proceeding solely among Lenders or participants or potential Lenders or participants relating to assignments of the Loans and/or Commitments or other intra-lender issues by any one or more Lenders, participants or potential Lenders or participants. If and to the extent that the foregoing undertaking may be unenforceable for any reason, Borrower agrees to make the maximum contribution to the payment and satisfaction thereof which is permissible under applicable law.

### 10.2 Amendments and Waivers.

- (a) Except for actions expressly permitted to be taken by Agent, no amendment, modification, termination or waiver of any provision of this Agreement or any other Loan Document, or any consent to any departure by any Loan Party therefrom, shall in any event be effective unless the same shall be in writing and signed by Borrower, and by Requisite Lenders, Requisite Revolving Lenders, Supermajority Revolving Lenders or all affected Lenders, as applicable. Except as set forth in clauses (b) and (c) below, all such amendments, modifications, terminations or waivers requiring the consent of any Lenders shall require the written consent of Requisite Lenders.
- (b) No amendment, modification, termination or waiver of or consent with respect to any provision of this Agreement that amends the definition of Advance Multiple in a manner which results in an increase in the amount of Revolving Loans which may be made available to Borrower shall be effective unless the same shall be in writing and signed by Agent, Supermajority Revolving Lenders and Borrower. No amendment, modification, termination or waiver of or consent with respect to any provision of this Agreement that waives compliance with the conditions precedent set forth in Section 3.2 to the making of any Loan or the incurrence of any Letter of Credit Obligations shall be effective unless the same shall be in writing and signed by Agent, Requisite Revolving Lenders and Borrower.

Notwithstanding anything contained in this Agreement to the contrary, no waiver or consent with respect to any Default or any Event of Default shall be effective for purposes of the conditions precedent to the making of Loans or the incurrence of Letter of Credit Obligations set forth in Section 3.2 unless the same shall be in writing and signed by Agent, Requisite Revolving Lenders and Borrower.

(c) No amendment, modification, termination or waiver shall, unless in writing and signed by Agent and each Lender directly affected thereby: (i) increase the principal amount, or postpone or extend the scheduled date of expiration, of any Lender's Commitment (which action shall be deemed only to affect those Lenders whose Commitments are increased or the scheduled date of expiration of whose Commitments are postponed or extended and may be approved by Requisite Lenders, including those Lenders whose Commitments are increased or the scheduled date of expiration of whose Commitments are postponed or extended); (ii) reduce the principal of, rate of interest on (other than any determination or waiver to charge or not charge interest at the Default Rate) or Fees payable with respect to any Loan or Letter of Credit Obligations of any affected Lender; (iii) extend any scheduled payment date or final maturity date of the principal amount of any Loan of any affected Lender or postpone or extend the scheduled date of expiration of any Letter of Credit beyond the date set forth in clause (b) of the initial sentence of Section 2.1(c)(iv); (iv) waive, forgive, defer, extend or postpone any payment of interest or Fees as to any affected Lender (which action shall be deemed only to affect those Lenders to whom such payments are made); (v) release any Guaranty or, except as otherwise permitted in Section 6.7 or Section 9.2(h), release Collateral (which action shall be deemed to directly affect all Lenders); (vi) change the percentage of the Commitments or of the aggregate unpaid principal amount of the Loans that shall be required for Lenders or any of them to take any action hereunder (which action shall be deemed to directly affect all Lenders); and (vii) amend or waive this Section 10.2 or the definitions of the terms "Requisite Lenders", "Requisite Revolving Lenders" or "Supermajority Revolving Lenders" insofar as such definitions affect the substance of this Section 10.2 or the term "Pro Rata Share" (which action shall be deemed to directly affect all Lenders). Furthermore, no amendment, modification, termination or waiver affecting the rights or duties of Agent or L/C Issuers under this Agreement or any other Loan Document shall be effective unless in writing and signed by Agent or L/C Issuers, as the case may be, in addition to Lenders required hereinabove to take such action. Each amendment, modification, termination or waiver shall be effective only in the specific instance and for the specific purpose for which it was given. No amendment, modification, termination or waiver shall be required for Agent to take additional Collateral pursuant to any Loan Document. No amendment, modification, termination or waiver of any provision of any Note (other than by virtue of the incorporation therein of any term from this Agreement that can be amended with less than all affected Lenders) shall be effective without the written concurrence of the holder of that Note. No notice to or demand on any Loan Party in any case shall entitle such Loan Party or any other Loan Party to any other or further notice or demand in similar or other circumstances. Any amendment, modification, termination, waiver or consent effected in accordance with this Section 10.2 shall be binding upon each holder of the Notes at the time outstanding and each future holder of the Notes.

10.3 Notices. Any notice or other communication required shall be in writing addressed to the respective party as set forth below and may be personally served, telecopied, sent by overnight courier service or U.S. mail and shall be deemed to have been given: (a) if delivered in person, when delivered; (b) if delivered by fax, on the date of transmission if transmitted on a Business Day before 4:00 p.m. Chicago time; (c) if delivered by overnight courier, one (1) Business Day after delivery to the courier properly addressed; or (d) if delivered by U.S. mail, four (4) Business Days after deposit with postage prepaid and properly addressed.

Notices shall be addressed as follows:

If to Borrower: Addus HealthCare, Inc.

2401 South Plum Grove Road Palatine, Illinois 60067 ATTN: Edward Busy, Esq.

Fax: (847) 303-5376

With a copy to: King & Spalding LLP

1185 Avenue of the Americas New York, New York 10036 ATTN: Robert S. Finley Fax: (212) 556-2222

If to Agent: FREEPORT FINANCIAL LLC

500 West Madison Street, Suite 1710

Chicago, Illinois 60661

ATTN: Addus HealthCare, Inc. Account Officer

Fax: (312) 281-4646

With a copy to: Winston & Strawn LLP

35 West Wacker Drive Chicago, Illinois 60601 ATTN: Patrick Hardiman Fax: (312) 558-5700

If to a Lender: To the address set forth on the signature page hereto or in the applicable Assignment

Agreement

10.4 <u>Failure or Indulgence Not Waiver</u>; <u>Remedies Cumulative</u>. No failure or delay on the part of Agent or any Lender to exercise, nor any partial exercise of, any power, right or privilege hereunder or under any other Loan Documents shall impair such power, right, or privilege or be construed to be a waiver of any Default or Event of Default. All rights and remedies existing hereunder or under any other Loan Document are cumulative to and not exclusive of any rights or remedies otherwise available.

10.5 Marshaling; Payments Set Aside. Neither Agent nor any Lender shall be under any obligation to marshal any assets in payment of any or all of the Obligations. To the extent that Borrower makes payment(s) or Agent enforces its Liens or Agent or any Lender exercises its right of set-off, and such payment(s) or the proceeds of such enforcement or set-off is subsequently invalidated, declared to be fraudulent or preferential, set aside, or required to be repaid by anyone (whether as a result of any demand, litigation, settlement or otherwise), then to the extent of such recovery, the Obligations or part thereof originally intended to be satisfied, and all Liens, rights and remedies therefor, shall be revived and continued in full force and effect as if such payment had not been made or such enforcement or set-off had not occurred

10.6 Severability. The invalidity, illegality, or unenforceability in any jurisdiction of any provision under the Loan Documents shall not affect or impair the remaining provisions in the Loan Documents.

10.7 <u>Lenders' Obligations Several</u>; <u>Independent Nature of Lenders' Rights</u>. The obligation of each Lender hereunder is several and not joint and no Lender shall be responsible for the obligation or commitment of any other Lender hereunder. In the event that any Lender at any time should fail to make

a Loan as herein provided, the Lenders, or any of them, at their sole option, may make the Loan that was to have been made by the Lender so failing to make such Loan. Nothing contained in any Loan Document and no action taken by Agent or any Lender pursuant hereto or thereto shall be deemed to constitute Lenders to be a partnership, an association, a joint venture or any other kind of entity. The amounts payable at any time hereunder to each Lender shall be a separate and independent debt.

- 10.8 <u>Headings</u>. Section and subsection headings are included herein for convenience of reference only and shall not constitute a part of this Agreement for any other purposes or be given substantive effect.
- 10.9 <u>Applicable Law</u>. THIS AGREEMENT AND EACH OF THE OTHER LOAN DOCUMENTS WHICH DOES NOT EXPRESSLY SET FORTH APPLICABLE LAW SHALL BE GOVERNED BY AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICTS OF LAWS PRINCIPLES, OTHER THAN SECTION 5-1401 OF THE NEW YORK GENERAL OBLIGATIONS LAW.
- 10.10 Successors and Assigns. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns except that Borrower may not assign its rights or obligations hereunder without the written consent of all Lenders and any such purported assignment without such written consent shall be void.
- 10.11 No Fiduciary Relationship; Limited Liability. No provision in the Loan Documents and no course of dealing between the parties shall be deemed to create any fiduciary duty owing to any Loan Party by Agent or any Lender. Borrower and each other Loan Party agree that neither Agent nor any Lender shall have liability to Borrower or any other Loan Party (whether sounding in tort, contract or otherwise) for losses suffered by Borrower or any other Loan Party in connection with, arising out of, or in any way related to the transactions contemplated and the relationship established by the Loan Documents, or any act, omission or event occurring in connection therewith, unless and to the extent that it is determined that such losses resulted from the gross negligence or willful misconduct of the party from which recovery is sought as determined by a final non-appealable order by a court of competent jurisdiction. Neither Agent nor any Lender shall have any liability with respect to, and Borrower and each other Loan Party hereby waive, release and agree not to sue for, any special, indirect or consequential damages suffered by Borrower and any other Loan Party in connection with, arising out of, or in any way related to the Loan Documents or the transactions contemplated thereby.
- 10.12 <u>Construction</u>. Agent, each Lender, Borrower and each other Loan Party acknowledge that each of them has had the benefit of legal counsel of its own choice and has been afforded an opportunity to review the Loan Documents with its legal counsel and that the Loan Documents shall be construed as if jointly drafted by Agent, each Lender, Borrower and each other Loan Party.
- 10.13 Confidentiality. Until the Termination Date, Agent and each Lender agree to exercise their best efforts to keep confidential any non-public information delivered pursuant to the Loan Documents and identified as such by Borrower and not to disclose such information to Persons other than (so long as any such Person shall have agreed to be bound by the terms of this Section 10.13 in a writing acceptable to the Borrower) to potential assignees or participants or to any Affiliate of, or Persons employed by or engaged, by Agent, a Lender or any of their respective Affiliates or a Lender's assignees or participants including attorneys, auditors, professional consultants, rating agencies, insurance industry associations and portfolio management services or to a Person that is a trustee, investment advisor, collateral manager, servicer, noteholder or secured party in a Securitization (as hereinafter defined) in connection with the administration, servicing and reporting on the assets serving as collateral for such Securitization. For the purposes of this section, "Securitization" shall mean a public or private offering by a Lender or any of its

Affiliates or their respective successors and assigns, of securities which represent an interest in, or which are collateralized, in whole or in part, by the Loans. The confidentiality provisions contained in this Section 10.13 shall not apply to disclosures (i) required to be made by Agent or any Lender to any regulatory or governmental agency or pursuant to law, rule, regulations or legal process or (ii) consisting of general portfolio information that does not specifically identify Borrower. Each Loan Party agrees that Agent or any Lender may, subject to the Loan Parties' consent (with such consent not to be unreasonably withheld), publish a tombstone or similar advertising material relating to the financing transactions contemplated by this Agreement. Agent or such Lender shall provide a draft of any such tombstone or similar advertising material to each Loan Party for review and comment prior to the publication thereof. Agent may provide to industry trade organizations information with respect to the credit facility that is necessary and customary for inclusion in league table measurements. The obligations of Agent and Lenders under this Section 10.13 shall supersede and replace the obligations of Agent and Lenders under any confidentiality agreement in respect of this financing executed and delivered by Agent or any Lender prior to the date hereof.

10.14 CONSENT TO JURISDICTION. AGENT, LENDERS, BORROWER AND LOAN PARTIES HEREBY CONSENT TO THE JURISDICTION OF ANY STATE OR FEDERAL COURT LOCATED WITHIN COOK COUNTY, STATE OF ILLINOIS AND IRREVOCABLY AGREE THAT, SUBJECT TO AGENT'S ELECTION, ALL ACTIONS OR PROCEEDINGS ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE OTHER LOAN DOCUMENTS SHALL BE LITIGATED IN SUCH COURTS. AGENT, LENDERS, BORROWER AND LOAN PARTIES EXPRESSLY SUBMIT AND CONSENT TO THE JURISDICTION OF THE AFORESAID COURTS AND WAIVE ANY DEFENSE OF FORUM NON CONVENIENS. BORROWER AND LOAN PARTIES HEREBY WAIVE PERSONAL SERVICE OF ANY AND ALL PROCESS AND AGREES THAT ALL SUCH SERVICE OF PROCESS MAY BE MADE UPON BORROWER AND LOAN PARTIES BY CERTIFIED OR REGISTERED MAIL, RETURN RECEIPT REQUESTED, ADDRESSED TO BORROWER, AT THE ADDRESS SET FORTH IN THIS AGREEMENT AND SERVICE SO MADE SHALL BE COMPLETE TEN (10) DAYS AFTER THE SAME HAS BEEN POSTED.

10.15 WAIVER OF JURY TRIAL. BORROWER, LOAN PARTIES, AGENT AND EACH LENDER HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS. BORROWER, LOAN PARTIES, AGENT AND EACH LENDER ACKNOWLEDGE THAT THIS WAIVER IS A MATERIAL INDUCEMENT TO ENTER INTO A BUSINESS RELATIONSHIP, THAT EACH HAS RELIED ON THE WAIVER IN ENTERING INTO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS AND THAT EACH WILL CONTINUE TO RELY ON THE WAIVER IN THEIR RELATED FUTURE DEALINGS. BORROWER, LOAN PARTIES, AGENT AND EACH LENDER WARRANT AND REPRESENT THAT EACH HAS HAD THE OPPORTUNITY OF REVIEWING THIS JURY WAIVER WITH LEGAL COUNSEL, AND THAT EACH KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS.

10.16 <u>Survival of Warranties and Certain Agreements</u>. All agreements, representations and warranties made herein shall survive the execution and delivery of this Agreement, the making of the Loans, issuances of Letters of Credit and the execution and delivery of the Notes. Notwithstanding anything in this Agreement or implied by law to the contrary, the agreements of Borrower set forth in <u>Sections 2.8</u>, <u>2.9</u> and <u>10.1</u> shall survive the repayment of the Obligations and the termination of this Agreement.

10.17 Entire Agreement. This Agreement, the Notes and the other Loan Documents embody the entire agreement among the parties hereto and supersede all prior commitments, agreements, representations, and understandings, whether oral or written, relating to the subject matter hereof (other than the Freeport

Fee Letter), and may not be contradicted or varied by evidence of prior, contemporaneous, or subsequent oral agreements or discussions of the parties hereto. All Exhibits, Schedules and Annexes referred to herein are incorporated in this Agreement by reference and constitute a part of this Agreement.

10.18 <u>Counterparts</u>; <u>Effectiveness</u>. This Agreement and any amendments, waivers, consents or supplements may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed and delivered shall be deemed an original, but all of which counterparts together shall constitute but one in the same instrument. This Agreement shall become effective upon the execution of a counterpart hereof by each of the parties hereto.

#### 10.19 Replacement of Lenders.

- (a) Within fifteen (15) days after receipt by Borrower of written notice and demand from any Lender for payment pursuant to Section 2.8 or 2.9 or, as provided in Section 10.19(c), in the case of certain refusals by any Lender to consent to certain proposed amendments, modifications, terminations or waivers with respect to this Agreement that have been approved by Requisite Lenders, Requisite Revolving Lenders, Supermajority Revolving Lenders or all affected Lenders, as applicable (any such Lender demanding such payment or refusing to so consent being referred to herein as an "Affected Lender"), Borrower may, at its option, notify Agent and such Affected Lender of its intention to do one of the following:
- (i) Borrower may obtain, at Borrower's expense, a replacement Lender ("Replacement Lender") for such Affected Lender, which Replacement Lender shall be reasonably satisfactory to Agent. In the event Borrower obtains a Replacement Lender that will purchase all outstanding Obligations owed to such Affected Lender and assume its Commitments hereunder within ninety (90) days following notice of Borrower's intention to do so, the Affected Lender shall sell and assign all of its rights and delegate all of its obligations under this Agreement to such Replacement Lender in accordance with the provisions of Section 9.1, provided that Borrower has reimbursed such Affected Lender for any administrative fee payable pursuant to Section 9.1 and, in any case where such replacement occurs as the result of a demand for payment pursuant to Section 2.8 or 2.9, paid all amounts required to be paid to such Affected Lender pursuant to Section 2.8 or 2.9 through the date of such sale and assignment; or
- (ii) Borrower may, with Agent's consent, give written notice of its intention to prepay in full all outstanding Obligations owed to such Affected Lender and terminate such Affected Lender's Pro Rata Share of the Revolving Loan Commitment and Pro Rata Share of the Term Loan Commitment, in which case, upon the effectiveness of such termination, the Revolving Loan Commitment and Term Loan Commitment will be reduced by the amount of such Pro Rata Share. Borrower shall, within ninety (90) days following notice of its intention to do so, prepay in full all outstanding Obligations owed to such Affected Lender (including, in any case where such prepayment occurs as the result of a demand for payment for increased costs, such Affected Lender's increased costs for which it is entitled to reimbursement under this Agreement through the date of such prepayment), and terminate such Affected Lender's obligations under the Revolving Loan Commitment and Term Loan Commitment.
- (b) In the case of a Non-Funding Lender pursuant to Section 9.5(a), at Borrower's request, Agent or a Person acceptable to Agent shall have the right with Agent's consent and in Agent's sole discretion (but shall have no obligation) to purchase from any Non-Funding Lender, and each Non-Funding Lender agrees that it shall, at Agent's request, sell and assign to Agent or such Person, all of the Loans and Commitments of that Non-Funding Lender for an amount equal to the principal balance of all

Loans held by such Non-Funding Lender and all accrued interest and Fees with respect thereto through the date of sale, such purchase and sale to be consummated pursuant to an executed Assignment Agreement.

- (c) If, in connection with any proposed amendment, modification, waiver or termination pursuant to Section 10.2 (a "Proposed Change"):
- (i) requiring the consent of all affected Lenders, the consent of Requisite Lenders is obtained, but the consent of other Lenders whose consent is required is not obtained (any such Lender whose consent is not obtained as described in this clause (ii) and in clause (ii) below being referred to as a "Non-Consenting Lender"), or
- (ii) requiring the consent of Supermajority Revolving Lenders, the consent of Requisite Revolving Lenders is obtained, but the consent of Supermajority Revolving Lenders is not obtained,

then, so long as Agent is not a Non-Consenting Lender, at Borrower's request Agent, or a Person reasonably acceptable to Agent, shall have the right with Agent's consent and in Agent's sole discretion (but shall have no obligation) to purchase from such Non-Consenting Lenders, and such Non-Consenting Lenders agree that they shall, upon Agent's request, sell and assign to Agent or such Person, all of the Loans and Commitments of such Non-Consenting Lenders for an amount equal to the principal balance of all Loans held by the Non-Consenting Lenders and all accrued interest and Fees and other Obligations owing with respect thereto through the date of sale, such purchase and sale to be consummated pursuant to an executed Assignment Agreement.

10.20 <u>Delivery of Termination Statements and Mortgage Releases</u>. On the Termination Date, and so long as no suits, actions, proceedings, or claims are pending or asserted against any Indemnitee asserting any damages, losses or liabilities that are indemnified liabilities hereunder, Agent shall deliver to Borrower termination statements, mortgage releases and other documents necessary or appropriate to evidence the termination of the Liens securing payment of the Obligations.

#### 10.21 Subordination of Intercompany Debt.

- (a) Each Loan Party hereby agrees that any intercompany Indebtedness or other intercompany payables or receivables, or intercompany advances directly or indirectly made by or owed to such Loan Party by any other Loan Party (collectively, "Intercompany Debt"), of whatever nature at any time outstanding shall be subordinate and subject in right of payment to the prior payment in full in cash of the Obligations. Each Loan Party hereby agrees that it will not, while any Event of Default is continuing, accept any payment, including by offset, on any Intercompany Debt until the Termination Date, in each case, except with the prior written consent of Agent.
- (b) In the event that any payment on any Intercompany Debt shall be received by a Loan Party other than as permitted by this Section 10.21 before the Termination Date, such Loan Party shall receive such payments and hold the same in trust for, segregate the same from its own assets and shall immediately pay over to, the Agent for the benefit of the Agent and Lenders all such sums to the extent necessary so that Agent and the Lenders shall have been paid in full, in cash, all Obligations owed or which may become owing.
- (c) Upon any payment or distribution of any assets of any Loan Party of any kind or character, whether in cash, property or securities by set-off, recoupment or otherwise, to creditors in any liquidation or other winding-up of such Loan Party or in the event of any Proceeding, Agent and Lenders shall first be entitled to receive payment in full in cash, in accordance with the terms of the Obligations

and of this Agreement, of all amounts payable under or in respect of such Obligations, before any payment or distribution is made on, or in respect of, any Intercompany Debt, in any such Proceeding, any distribution or payment, to which Agent or any Lender would be entitled except for the provisions hereof shall be paid by such Loan Party, or by any receiver, trustee in bankruptcy, liquidating trustee, agent or other person making such payment or distribution directly to Agent (for the benefit of Agent and the Lenders) to the extent necessary to pay all such Obligations in full in cash, after giving effect to any concurrent payment or distribution to Agent and Lenders (or to Agent for the benefit of Agent and Lenders).

[Signature Pages Follow]

Witness the due execution hereof by the respective duly authorized officers of the undersigned as of the date first written above.

#### ADDUS HEALTHCARE, INC.

By: /s/ Ed Budy
Name: Ed Budy
Title: Assistant Secretary

## ADDUS ACQUISITION CORPORATION

By: /s/ Simon Bachleda

Name: Simon Bachleda

Title: Secretary

### ADDUS HOLDING CORPORATION

By: /s/ Simon Bachleda
Name: Simon Bachleda
Title: Secretary

#### ADDUS MANAGEMENT CORPORATION

By: /s/ Simon Bachleda
Name: Simon Bachleda
Title: Secretary

## LOWELL HOME HEALTH AGENCY, INC.

 By:
 /s/ Ed Budy

 Name:
 Ed Budy

 Title:
 Secretary

Name:	Ed Budy
	Secretary
FORT	SMITH HOME HEALTH AGENCY, INC.
By:	/s/ Ed Budy
Name:	Ed Budy
Title:	Secretary
BENEI	FITS ASSURANCE CO., INC.
	,
By:	/s/ Ed Budy
By: Name:	/s/ Ed Budy Ed Budy
By: Name:	/s/ Ed Budy
By: Name: Title: PHC A	/s/ Ed Budy Ed Budy
By: Name: Title: PHC A	/s/ Ed Budy Ed Budy Secretary  CQUISITION CORPORATION

LITTLE ROCK HOME HEALTH AGENCY, INC.

By:

/s/ Ed Budy

/s/ Ed Budy

Secretary

Name: Ed Budy

By:

Title:

The undersigned hereby assumes and agrees to perform all obligations, liabilities indebtedness, covenants and agreements of Addus Acquisition Corporation as Borrower under the foregoing Credit Agreement

# ADDUS HEALTHCARE, INC.

By: /s/ Ed Budy

Name: Ed Budy

Title: Assistant Secretary

# FREEPORT FINANCIAL LLC, as Agent

By: /s/ Chad Blakeman

Title: Duly Authorized Signatory

# FREEPORT LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman

Title: Duly Authorized Signatory

#### FIFTH THIRD BANK, CHICAGO

By: /s/ Michael E. May
Name: Michael E. May

Title: Vice President

Address: 222 South Riverside Plaza

33rd Floor, MD GRVR3I Chicago, IL 60606 Attn: Gregory H. Bork Fax: (312) 704-4374

ABA No.: 042000314 Account No.: 89922553

Bank: Fifth Third Bank

Bank Address: 222 South Riverside Plaza

33rd Floor, MD GRVR3I Chicago, IL 60606

#### RESIDENTIAL FUNDING CORPORATION

By: /s/ Dennis M. Hansen
Name: Dennis M. Hansen

Title: SVP

Address: 2711 N. Haskell Ave., Ste. 900

Dallas, TX 75204 Attn: Dennis Hansen Fax: (214) 874-2075

ABA No.: Account No.: Bank:

Bank Address:

With a copy to:

Residential Funding Corporation c/o GMAC-RFC Health Capital 8400 Normandale Lake Blvd.

Suite 250

Minneapolis, MN 55437 Attn: Laura Mollet Fax: (952) 857-6949

# ANNEX A to CREDIT AGREEMENT

# PRO RATA SHARES AND COMMITMENT AMOUNTS

	REVOLVER		TERM LO	AN	
Lender	\$	%	\$	%	
Freeport Loan Fund LLC	5,978,260.86	47.8260870	21,521,739.14	47.8260870	
Residential Funding	3,260,869.57	26.0869565	11,739,130.43	26.0869565	
Corporation					
Fifth Third Bank	3,260,869.57	26.0869565	11,739,130.43	26.0869565	
(Chicago)					
Totals	12,500,000.00	100.00	45,000,000.00	100.00	

ANNEX B to CREDIT AGREEMENT CLOSING CHECKLIST

[INSERT CLOSING CHECKLIST]

B-1

#### PRELIMINARY LIST OF CLOSING DOCUMENTS AND DELIVERIES

relating to the

## CREDIT AGREEMENT

among

# ADDUS ACQUISITION CORPORATION,

 $(to\ be\ merged\ as\ of\ the\ Closing\ Date\ into\ Addus\ Health Care,\ Inc.)$ 

as Borrower

and

# THE OTHER PERSONS PARTY THERETO THAT ARE DESIGNATED AS LOAN PARTIES

and

## THE VARIOUS FINANCIAL INSTITUTIONS PARTY THERETO,

as the Lenders

## FREEPORT FINANCIAL LLC,

as Agent and Lead Arranger

and

#### FREEPORT LOAN FUND LLC,

as a Lender

DATED AS OF SEPTEMBER 19, 2006

#### LIST OF CLOSING DOCUMENTS AND DELIVERIES

Set forth below is an initial List of Closing Documents and Deliveries which lists the documents to be delivered in connection with the closing of the transactions contemplated by the Credit Agreement listed herein as Document No. 1 (the "<u>Credit Agreement</u>"). Each capitalized term used but not defined herein shall have the meaning ascribed to such term in the Credit Agreement.

#### PARTIES AND COUNSEL

Agent Freeport Financial LLC

Borrower Addus Acquisition Corporation, a Delaware corporation (to be merged as of the Closing Date into Addus

HealthCare, Inc., an Illinois corporation)

Holdings Addus Holding Corporation, a Delaware corporation
Intermediate Holdings Addus Management Corporation, a Delaware corporation

W&S Winston & Strawn LLP, counsel to the Agent

K&S King & Spalding LLP, counsel to Borrower and the other Loan Parties

#### **GENERAL LOAN DOCUMENTS**

Credit Agreement

#### ANNEXES

ANNEX A	Pro Rata Shares and Commitment Amounts
ANNEX B	Closing Checklist
ANNEX C	Pro Forma
ANNEX D	Lenders' Bank Accounts
ANNEX E	Compliance, Pricing and Excess Cash Flow Certificate
ANNEX F	Form of Borrower Assignment and Assumption Agreement
	SCHEDULES
Schedule 4.1(a)	Jurisdictions of Organization and Qualification
Schedule 4.1(b)	Capitalization
Schedule 4.5	Financial Statement Matters
Deficatio 4.5	Thiancial Statement Watters
Schedule 4.7	Use of Proceeds

Schedule 4.11 Schedule 4.12 **Employee Matters** Schedule 4.13 Litigation

Schedule 4.14 Real Estate

Schedule 4.15 **Environmental Matters** 

Schedule 4.16 **ERISA** 

Schedule 4.17 Deposit and Disbursement Accounts Schedule 4.18 Agreements and Other Documents Insurance

Schedule 4.19

Compliance with Health Care Laws Schedule 4.22(a) Participation in Health Care Programs Schedule 4.22(f)

Schedule 6.1 Indebtedness Schedule 6.2 Liens

Schedule 6.4 Contingent Obligations Schedule 6.8 Affiliate Transactions Schedule 6.9 **Business Description** 

#### **EXHIBITS**

Exhibit 2.1(a)	Term Notes
Exhibit 2.1(b)(i)	Revolving Note
Exhibit 2.l(b)(ii)	Notice of Revolving Credit Advance
Exhibit 2.1(c)	Request for Letter of Credit Issuance
Exhibit 2.2(e)	Notice of Continuation/Conversion
Exhibit 5.12	HIPPA Business Associate Agreement
Exhibit 7.2(e)	Borrowing Availability Certificate
Exhibit 9.1	Assignment Agreement

2. Notes for each of Fifth Third and GMAC

# **COLLATERAL DOCUMENTS**

UCC financing statements and other filings and documents required by law or the Loan Documents to provide the Agent first priority perfected Liens

- 4. UCC, judgment and tax lien search results for Holdings, Intermediate Holdings, Borrower and its Subsidiaries listing all effective financing statements and liens filed against Holdings, Intermediate Holdings, Borrower and its Subsidiaries, with copies of such financing statements
- 5. Perfection Certificate executed by each of Holdings, Intermediate Holdings, Borrower and each other Loan Party
- 6. Holdings Security Agreement executed by Holdings and Intermediate Holdings

Schedules to Holdings Security Agreement

- 7. Power of Attorney executed by Holdings and Intermediate Holdings
- 8. Borrower Security Agreement

Schedules to Borrower Security Agreement

- 9. Power of Attorney executed by Borrower
- 10. Subsidiary Security Agreement

Schedules to Subsidiary Security Agreement

- 11. Power of Attorney executed by Subsidiaries
- 12. Trademark Security Agreement

Schedules to Trademark Security Agreement

13. Copyright Security Agreement

Schedules to Copyright Security Agreement

- 14. Holdings Guaranty executed by Holdings and Intermediate Holdings
- 15. Subsidiary Guaranty
- 16. Holdings Pledge Agreement executed by Holdings and Intermediate Holdings

Schedules to Holdings Pledge Agreement

17. Borrower Pledge Agreement

Schedules to Borrower Pledge Agreement

- 18. Stock Certificates, together with stock powers duly executed in blank
- 19. Deposit Account Control Agreements with Fifth Third Bank for each of Addus HealthCare, Inc., Fort Smith, Lowell and Little Rock
- 20. Landlord Waiver with respect to the following location:
  - 2401 South Plum Grove Road
- 21. Payoff letters and releases issued by Fifth Third Bank, together with UCC-3 Termination Statements
- 22. Subordination Agreement

### IV. BORROWING DOCUMENTS

- 23. Letter of Direction with attached funds flow memorandum
- 24. Borrowing Availability Certificate

#### V. CORPORATE AUTHORIZATION AND RELATED DOCUMENTS

25. For Holdings, Intermediate Holdings, the Borrower and each Subsidiary, (i) articles of incorporation or formation, certified by the applicable Secretary of State, (ii) bylaws, (iii) resolutions of its board of directors approving and authorizing the execution, delivery and performance of the Loan Documents to which it is party and the transactions contemplated thereby, and (iv) signature and incumbency certificates of its officers executing any of the Loan Documents, all certified by its secretary or an assistant secretary (or similar officer) as being in full force and effect without modification, together with a good-standing certificate from the Secretary of State of its jurisdiction of formation or any other jurisdiction that it is qualified to conduct business

#### VI. LEGAL OPINIONS

- 26. Legal opinion of King & Spalding LLP, counsel for the Loan Parties
- 27. Illinois local counsel opinion
- 28. California local counsel opinion
- 29. Arkansas local counsel opinion
- 30. Legal opinion of King & Spalding LLP re Stock Purchase Agreement
- 31. Legal opinion of General Counsel of Addus HealthCare, Inc.

#### VII. MISCELLANEOUS DOCUMENTS

- 32. Certificates of insurance (property, liability and business interruption) with endorsements naming the Agent as lender loss payee and additional insured, as applicable
- 33. Collateral Assignment of Contract Rights
- 34. Borrower Assignment and Assumption Agreement
- 35. Detailed Financial Projections, including Pro Formas
- 36. Solvency Certificate
- 37. Certified copies of Related Transaction Documents
  - · Stock Purchase Agreement
  - · Merger Agreement
  - · Merger Documents
  - · Acquisition Note
  - Registration Rights Agreement
  - · Stockholders Agreement
  - · Escrow Agreement
  - Contingent Payment Agreement
  - · Restated Certificate
  - Securities Purchase Agreement
  - · Management Services Agreement
- 38. Patriot Act Certifications

- 39. Completion of background and reference checks with results satisfactory to Agent in accordance with Section 3.1(f) of the Credit Agreement
- 40. Receipt of all material third-party and regulatory approvals and consents
- 41. Fee Letter
- 42. Business Associate Agreement
- 43. Post-Closing Agreement

ANNEX C
to
CREDIT AGREEMENT
PRO FORMA

[INSERT PRO FORMA]

C-1

#### Consolidated Balance Sheet September 19, 2006 ASSETS **Current Assets** 4,808,411 Cash Accounts receivable, net 30,565,000 Prepaid expenses and deposits 2,163,000 **Total Current Assets** 37,536,411 Property and Equipment, Net 2,551,000 **Other Assets** 69,993,589 Goodwill and other intangibles, net 2,000,000 Debt issuance costs Other Total Other Assets 71,993,589 112,081,000 LIABILITIES & STOCKHOLDERS EQUITY **Current Liabilities** Current maturities of long-term debt: 2,100,000 Revolving credit and term loans Other 872,000 Accounts payable 5,548,000 Accrued income taxes 800,000 Accrued workers compensation 1,422,000 Other accrued expenses 10,601,000 21,343,000 **Total Current Liabilities Long-Term Liabilities** Long-term debt: Term loan 42,900,000 Contingent Payment Obligation & Other 10,087,000

52,987,000

37,750,000

37,751,000 112,081,000

1,000

ADDUS HOLDING CORPORATION

Total Long-Term Liabilities

Total Stockholders' Equity

Stockholders' Equity
Preferred stock

Common stock

Retained earnings

**ADDUS HOLDING CORPORATION** Consolidating Balance Sheet September 19, 2006

	Addus Holding Corp	Addus Mgmt Corp	Addus HealthCare	Subtotal	Eliminations	Consolidated Totals
ASSETS						
Current Assets						
Cash			4,808,411	4,808,411		4,808,411
Accounts receivable, net			30,565,000	30,565,000		30,565,000
Intercompany rec (pay)	37,750,000		(37,750,000)	0		0
Prepaid expenses and deposits			2,163,000	2,163,000		2,163,000
Total Current Assets	37,750,000	0	(213,589)	37,536,411	0	37,536,411
Property and Equipment, Net			2,551,000	2,551,000		2,551,000
Other Assets						
Goodwill and other intangibles, net			69,993,589	69,993,589		69,993,589
Investment in subsidiaries	1,000	1,000	0	2,000	(2,000)	0
Debt issuance costs			2,000,000	2,000,000		2,000,000
Other			0	0		0
Total Other Assets	1,000	1,000	71,993,589	71,995,589	(2,000)	71,993,589
	37,751,000	1,000	74,331,000	112,083,000	(2,000)	112,081,000
LIABILITIES & STOCKHOLDERS EQUITY						
Current Liabilities						
Current maturities of long-term debt:						
Revolving credit and term loans			2,100,000	2,100,000		2,100,000
Other			872,000	872,000		872,000
Accounts payable			5,548,000	5,548,000		5,548,000
Accrued income taxes			800,000	800,000		800,000
Accrued workers compensation			1,422,000	1,422,000		1,422,000
Other accrued expenses			10,601,000	10,601,000		10,601,000
Total Current Liabilities	0	0	21,343,000	21,343,000	0	21,343,000
Long-Term Liabilities						
Long-term debt:						
Term loan			42,900,000	42,900,000		42,900,000
Other			10,087,000	10,087,000		10,087,000
Total Long-Term Liabilities	0	0	52,987,000	52,987,000	0	52,987,000
Stockholders' Equity						
Preferred stock	37,750,000		0	37,750,000		37,750,000
Common stock	1,000	1,000	1,000	3,000	(2,000)	1,000
Retained earnings			0	0		0
Total Stockholders' Equity	37,751,000	1,000	1,000	37,753,000	(2,000)	37,751,000
	37,751,000	1,000	74,331,000	112,083,000	(2,000)	112,081,000

ANNEX D to CREDIT AGREEMENT

# WIRE TRANSFER INFORMATION

Name: Bank: ABA #:

Account #:
Account Name:

Reference:

#### **Lender Bank Info**

#### Freeport Financial

Bank U.S. Bank N.A., Milwaukee, WI

ABA 075-000-022

Account Name Freeport Financial, LLC

Account Number 182380366340 Tax ID 20-2163563

## Fifth Third

Bank Fifth Third Bank ABA 042-000-314

Account Name Fifth Third Bank, Addus Healthcare Participation

Account Number 89922553 Tax ID 38-0892650

## **GMAC**

Bank GMAC RFC Capital, Minneapolis, MN

ABA 021-000-021

Account Name GMAC RFC Capital Operating Account

Account Number 109-9829 Tax ID 93-0891336

#### ANNEX E

to

#### CREDIT AGREEMENT

#### COMPLIANCE, PRICING AND EXCESS CASH FLOW CERTIFICATE

ADDUS HEALTHCARE, INC.

This Certificate is given by Addus HealthCare, Inc., an Illinois corporation ("Borrower") pursuant to Section 7.2(n) of that certain Credit Agreement dated as of September , 2006 among Borrower, the other Loan Parties party thereto, the Lenders from time to time party thereto and Freeport Financial LLC, as agent for the Lenders (as such agreement may have been amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"). Capitalized terms used herein without definition shall have the meanings set forth in the Credit Agreement.

The undersigned is duly authorized to execute and deliver this Certificate on behalf of Borrower. By executing this Certificate such officer hereby certifies to Agent and Lenders that:

- (a) the financial statements delivered with this Certificate in accordance with Section 7.2(b) and 7.2(c) of the Credit Agreement fairly present in all material respects the results of operations and financial condition of Holdings and its Subsidiaries as of the dates of such financial statements;
- (b) I have reviewed the terms of the Credit Agreement and have made, or caused to be made under my supervision, a review in reasonable detail of the transactions and conditions of the Loan Parties during the accounting period covered by such financial statements;
- (c) except as set forth on <u>Schedule 1</u> hereto, Borrower is in compliance with the covenants contained in Section 6 and, as set forth on <u>Schedule 1</u> hereto, Section 7 of the Credit Agreement. [NOTE TO BORROWER: THE PORTION OF <u>SCHEDULE 1</u> PERTAINING TO FINANCIAL COVENANTS IS ONLY REQUIRED TO BE COMPLETED AS OF THE END OF EACH FISCAL QUARTER];
- (d) based upon the calculation of Leverage Ratio as set forth on <u>Schedule 1</u> hereto, Applicable Margins shall be calculated using Level . [NOTE TO BORROWER: THIS CLAUSE IS ONLY REQUIRED TO BE COMPLETED AS OF THE END OF EACH FISCAL QUARTER];
- (e) Excess Cash Flow, as demonstrated by the calculation on <u>Schedule 2</u> hereto, for the Fiscal Year ending equals \$ . [NOTE TO BORROWER: THIS CLAUSE IS ONLY REQUIRED TO BE COMPLETED IN CONNECTION WITH THE DELIVERY OF AUDITED FINANCIAL STATEMENTS PURSUANT TO SECTION 7.2(c) OF THE CREDIT AGREEMENT];
- (f) such review has not disclosed the existence during or at the end of such accounting period, and I have no knowledge of the existence as of the date hereof, of any condition or event that constitutes a Default or an Event of Default, except as set forth on <u>Schedule 3</u> hereto, which includes a description of the nature and period of existence of such Default or an Event of Default and what action Borrower has taken, is taking and proposes to take with respect thereto;

(g) except as set forth on Schedule 3 hereto, subsequent to the date of the most recent Certificate submitted by Borrower pursuant to Section 7.2(n) of the Credit Agreement, no Loan Party has (i) changed its name as it appears in official filings in the jurisdiction of its organization, (ii) changed its chief executive office, principal place of business, corporate offices, warehouses, or the location of its records concerning Collateral, (iii) changed the type of entity that it is, (iv) changed (or has had changed) its organization identification number, if any, issued by its jurisdiction of organization, (v) changed its jurisdiction of organization, (vi) changed the end of its Fiscal Year, or (vii) formed any new Subsidiary or entered into any partnership or joint venture with any other Person; and
(h) except as set forth on <u>Schedule 3</u> hereto, subsequent to the date of the most recent Certificate submitted by Borrower pursuant to <u>Section 7.2(n)</u> of the Credit Agreement, there has been no event which would alter any of the disclosures set forth on <u>Schedule 4.1(b)</u> of the Credit Agreement.

ne Credit Agreement, there has been no event which would alter any of the disclosures set forth of	n Schedule 4.1(b) of	he Credit Agre	eement.	
IN WITNESS WHEREOF, Borrower has caused this Certificate to be executed by it	s this	day of	, .	
	ADDUS HEALTHO	CARE, INC.		

By Its

# CAPITAL EXPENDITURE LIMIT (Section 7.1(a))

Capital Expenditures during such period

Permitted Capital Expenditures (including Carry Over Amount of \$	from prior fiscal year)	\$
In Compliance		Yes/No
	MINIMUM EBITDA (Section 7.1(b))	
EBITDA <sup>1</sup>		\$
Required EBITDA		\$
In Compliance		Yes/No

# MINIMUM FIXED CHARGE COVERAGE RATIO (Section 7.1(c))

Fixed Charge Coverage Ratio	
Required Fixed Charge Coverage Ratio	
In compliance	Yes/No

# MAXIMUM LEVERAGE RATIO (Section 7.1(d))

Leverage Ratio	
Required Leverage Ratio	
In Compliance	Yes/No

- 1. For purposes of calculating Minimum EBITDA, Minimum Fixed Charge Coverage Ratio and Maximum Leverage Ratio, for the Fiscal Quarters ending March 31, 2006 and June 30, 2006 EBITDA shall be deemed to be \$2,914,037 and \$3,781,124, respectively.
- 2. For purposes of calculating the Fixed Charge Coverage Ratio, for the Fiscal Quarters ending March 31, 2006 and June 30, 2006, Capital Expenditures shall be deemed to be \$238,000 and \$78,000, respectively.
- 3. For purposes of calculating the Fixed Charge Coverage Ratio, for the Fiscal Quarters ending March 31, 2006, June 30, 2006 and September 30, 2006, Interest Expense shall be deemed to be \$1,050,000 for each such Fiscal Quarter.
- 4. For purposes of calculating the Fixed Charge Coverage Ratio, for the Fiscal Quarters ending March 31, 2006, June 30, 2006 and September 30, 2006, scheduled payments of principal on the Funded Debt shall be deemed to be \$700,000 for each such Fiscal Quarter.

# EXCESS CASH FLOW (Section 2.5)

Excess 0	Cash Flow for any Fiscal Year is calculated as follows:	
EBITDA	A for such Fiscal Year	\$
Plus:	Net decreases in Working Capital <sup>(1)</sup> of Holdings and its Subsidiaries during such Fiscal Year	
	Extraordinary gains during such Fiscal Year which are cash items and were deducted from net income for purposes of calculating EBITDA	
	Any payments received during such Fiscal Year under the Purchase Agreement as a result of the actual working capital of the Company and its Subsidiaries exceeding the targets set forth therein	
Less:	cash payments made by Holdings and its Subsidiaries during such Fiscal Year with respect to Capital Expenditures (excluding any Capital Expenditures in such Fiscal Year to the extent in excess of the Capex Limit)	
	Interest Expense of Holdings and its Subsidiaries for such Fiscal Year	
	scheduled principal payments paid or payable in respect of Funded Debt of Holdings and its Subsidiaries during such Fiscal Year	
	cash income taxes paid by Holdings and its Subsidiaries during such Fiscal Year	
	net increases in Working Capital <sup>(2)</sup> of Holdings and its Subsidiaries during such Fiscal Year	
	losses during such Fiscal Year which are cash items which were added back to net income for purposes of determining EBITDA	
	aggregate amounts of all voluntary prepayments of the Term Loan applied in accordance with the provisions of the Credit Agreement during such Fiscal Year	
	an amount equal to the aggregate amount of working capital adjustments paid by Holdings and its Subsidiaries under the Purchase Agreement during such Fiscal Year	
Subtotal	I	\$
Require	d Prepayment Percentage	[75]2%
Excess (	Cash Flow	\$

Prepayment Percentage shall be 75% if the Leverage Ratio for such Fiscal Year was greater than or equal to 2.50 to 1.00 or 50% if the Leverage Ratio for such Fiscal Year was less than 2.50 to 1.00.

# ANNEX F FORM OF BORROWER ASSIGNMENT AND ASSUMPTION AGREEMENT

Please refer to the Credit Agreement, dated as of September 19, 2006 as amended or otherwise modified from time to time, the "<u>Credit Agreement</u>"), among the undersigned ("<u>Borrower</u>"), the financial institutions party thereto from time to time, as Lenders, and Madison Capital Funding LLC, as Agent. This notice is given pursuant to <u>Section 3.1(j)</u> of the Credit Agreement. Capitalized terms used herein but not otherwise defined shall have the meanings ascribed thereto in the Credit Agreement.

For good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound,

- (a) Merger Sub hereby assigns, transfers, conveys and delivers to Company, and Company hereby accepts delivery of, all of Merger Sub's benefits, privileges, rights, title and interest in and to the foregoing Credit Agreement, the other Loan Documents (including the Notes and Collateral Documents) and all other agreements to which Merger Sub is a party in connection therewith (as each of the foregoing may be amended, restated, supplemented or otherwise modified from time to time, collectively, the "Assumed Documents"); and
- (b) Company hereby fully and completely succeeds to, assumes and agrees to pay, perform and discharge, in accordance with its terms, the Obligations under the Assumed Documents (including the granting of security interests in the properties of Company in connection therewith), and Company hereby fully makes the representations and warranties contained in the Assumed Documents, all as if Company were an original party thereto.

IN WITNESS WHEREOF, the parties hereto have caused this Borrower Assignment and Assumption Agreement to be duly executed and delivered by their

This Borrower Assignment and Assumption Agreement is hereby attached to and made a part of said Credit Agreement. Capitalized terms used but not defined herein are used as defined in said Credit Agreement.

duly authorized officers as of \_\_\_\_\_\_\_\_, 2006.

Merger Sub

ADDUS ACQUISITION CORPORATION

By:
Title:

Company.

ADDUS HEALTHCARE, INC.

By: Title: EXHIBIT 2.1(a) to CREDIT AGREEMENT

#### FORM OF TERM NOTE

Chicago, Illinois

FOR VALUE RECEIVED, the undersigned, ADDUS ACQUISITION CORPORATION, a Delaware corporation (to be merged as of the Closing Date into
Addus HealthCare, Inc.) ("Borrower"), HEREBY PROMISES TO PAY to the order of ("Lender") at the offices of FREEPORT FINANCIAL LLC,
a Delaware limited liability company, as Agent for Lenders ("Agent"), at its address at 500 West Madison Street, Suite 1710, Chicago, Illinois 60661, or at such other
place as Agent may designate from time to time in writing, in lawful money of the United States of America and in immediately available funds, the amount of
DOLLARS AND CENTS (\$,). All capitalized terms used but not otherwise defined herein have the meanings given to them
in the "Credit Agreement" (as hereinafter defined).

This Term Note is one of the Term Notes issued pursuant to that certain Credit Agreement dated as of September 19, 2006 by and among Borrower, the other Persons named therein as Loan Parties, Agent, Lender and the other Persons signatory thereto from time to time as Lenders (including all annexes, exhibits and schedules thereto and as from time to time amended, restated, supplemented or otherwise modified, the "Credit Agreement"), and is entitled to the benefit and security of the Credit Agreement, the Security Agreement and all of the other Loan Documents referred to therein. Reference is hereby made to the Credit Agreement for a statement of all of the terms and conditions under which the Loans evidenced hereby are made and are to be repaid. The principal balance of the Term Loan, the rates of interest applicable thereto and the date and amount of each payment made on account of the principal thereof, shall be recorded by Agent on its books; provided that the failure of Agent to make any such recordation shall not affect the obligations of Borrower to make a payment when due of any amount owing under the Credit Agreement or this Term Note.

The principal amount of the indebtedness evidenced hereby shall be payable in the amounts and on the dates specified in the Credit Agreement. Interest thereon shall be paid until such principal amount is paid in full at such interest rates and at such times, and pursuant to such calculations, as are specified in the Credit Agreement. The terms of the Credit Agreement are hereby incorporated herein by reference.

If any payment on this Term Note becomes due and payable on a day other than a Business Day, the payment thereof shall be extended to the next succeeding Business Day and, with respect to payments of principal, interest thereon shall be payable at the then applicable rate during such extension.

Upon and after the occurrence of any Event of Default, this Term Note may, as provided in the Credit Agreement, and without presentment, demand, protest, notice of intent to accelerate, notice of acceleration or other legal requirement of any kind (all of which are hereby expressly waived by Borrower), be declared, and immediately shall become, due and payable.

Time is of the essence of this Term Note.

Except as provided in the Credit Agreement, this Term Note may not be assigned by Lender to any Person.

# THIS TERM NOTE SHALL BE GOVERNED BY AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH THE INTERNAL LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICTS OF LAW PRINCIPLES.

ADDUS ACQUISITION CORPORATION

	By: Name: Title:	
The undersigned hereby assumes and agrees to perform all obligations, liabilities indebtedness, covenants and agreements of Addus Acquisition Corporation as Borrower under the foregoing Term Note		
ADDUS HEALTHCARE, INC.		
By: Name: Title:	2	

#### EXHIBIT 2.1(b)(i) to CREDIT AGREEMENT

#### FORM OF REVOLVING NOTE

Chicago, Illinois

FOR VALUE RECEIVED, the undersigned ADDUS ACQUISITION COR	PORATION, a Delaware corporation (to be merged as of the Closing Date into
Addus HealthCare, Inc.) ("Borrower"), HEREBY PROMISES TO PAY to the orde	, , , , , , , , , , , , , , , , , , , ,
LLC, a Delaware limited liability company, as Agent for Lenders ("Agent"), at its	
other place as Agent may designate from time to time in writing, in lawful money of	of the United States of America and in immediately available funds, the amount of
DOLLARS AND CENTS (\$,,) or,	if less, the aggregate unpaid amount of all Revolving Credit Advances made to the
undersigned under the "Credit Agreement" (as hereinafter defined). All capitalized	terms used but not otherwise defined herein have the meanings given to them in
the Credit Agreement.	

This Revolving Note is one of the Revolving Notes issued pursuant to that certain Credit Agreement dated as of September 19, 2006 by and among Borrower, the other Persons named therein as Loan Parties, Agent, Lender and the other Persons signatory thereto from time to time as Lenders (including all annexes, exhibits and schedules thereto, and as from time to time amended, restated, supplemented or otherwise modified, the "Credit Agreement"), and is entitled to the benefit and security of the Credit Agreement, the Security Agreement and all of the other Loan Documents referred to therein. Reference is hereby made to the Credit Agreement for a statement of all of the terms and conditions under which the Loans evidenced hereby are made and are to be repaid. The date and amount of each Revolving Credit Advance made by Lenders to Borrower, the rates of interest applicable thereto and each payment made on account of the principal thereof, shall be recorded by Agent on its books; provided that the failure of Agent to make any such recordation shall not affect the obligations of Borrower to make a payment when due of any amount owing under the Credit Agreement or this Revolving Note in respect of the Revolving Credit Advances made by Lender to Borrower.

The principal amount of the indebtedness evidenced hereby shall be payable in the amounts and on the dates specified in the Credit Agreement, the terms of which are hereby incorporated herein by reference. Interest thereon shall be paid until such principal amount is paid in full at such interest rates and at such times, and pursuant to such calculations, as are specified in the Credit Agreement. The terms of the Credit Agreement are hereby incorporated herein by reference.

If any payment on this Revolving Note becomes due and payable on a day other than a Business Day, the payment thereof shall be extended to the next succeeding Business Day and, with respect to payments of principal, interest thereon shall be payable at the then applicable rate during such extension.

Upon and after the occurrence of any Event of Default, this Revolving Note may, as provided in the Credit Agreement, and without presentment, demand, protest, notice of intent to accelerate, notice of acceleration or other legal requirement of any kind (all of which are hereby expressly waived by Borrower), be declared, and immediately shall become, due and payable.

Time is of the essence of this Revolving Note.

Except as provided in the Credit Agreement, this Revolving Note may not be assigned by Lender to any Person.

# THIS REVOLVING NOTE SHALL BE GOVERNED BY AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH THE INTERNAL LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICTS OF LAW PRINCIPLES.

ADDUS ACQUISITION CORPORATION

	By: Name: Title:	
The undersigned hereby assumes and agrees to perform all obligations, liabilities indebtedness, covenants and agreements of Addus Acquisition Corporation as Borrower under the foregoing Revolving Note		
ADDUS HEALTHCARE, INC.		
By: Name: Title:		

#### EXHIBIT 2.1(b)(ii) to CREDIT AGREEMENT

#### FORM OF NOTICE OF REVOLVING CREDIT ADVANCE

Freeport Financial LLC, 500 West Madison Street, Suite 1710 Chicago, Illinois 60661

Attention: Addus HealthCare, Inc.

Account Manager

Ladies and Gentlemen:

The undersigned, Addus HealthCare, Inc., an Illinois corporation ("<u>Borrower</u>") refers to the Credit Agreement, dated as of September 19, 2006 (the "<u>Credit Agreement</u>," the terms defined therein being used herein as therein defined), by and among Borrower, the other Loan Parties signatory thereto, Freeport Financial LLC, for itself, as Lender, and as Agent for Lenders, and Lenders, and hereby gives you notice, irrevocably, pursuant to <u>Section 2.1(b)(i)</u> of the Credit Agreement, that the undersigned hereby requests a Revolving Credit Advance under the Credit Agreement, and in that connection sets forth below the information relating to such Revolving Credit Advance as required by <u>Section 2.1(b)(i)</u> of the Credit Agreement:

(i) The date of the requested Revolving Credit Advance is \_\_\_\_\_\_, \_\_\_\_\_\_

- (ii) The aggregate amount of the requested Revolving Credit Advance is \$\_\_\_\_\_
- (iii) The requested Revolving Credit Advance is [an Index Rate Loan] [a LIBOR Loan with a LIBOR Period of \_\_\_\_\_\_].
- (iv) The requested Revolving Credit Advance is to be sent to:

[Name of Bank]
[City of Bank]
Beneficiary:
Account No.: [number]
ABA No.: [number]
Attn: [name]

The undersigned hereby certifies that all of the conditions contained in Section 3.2 of the Credit Agreement have been satisfied.

ADDUS	HEALTHCARE, INC.
By:	
Name:	
Title:	

#### EXHIBIT 2.1(c) to CREDIT AGREEMENT

#### REQUEST FOR ISSUANCE OF LETTER OF CREDIT

Freeport Financial LLC 500 West Madison Street, Suite 1710 Chicago, Illinois 60661

Attention: Addus HealthCare, Inc.

Account Manager

Ladies and Gentlemen:

The undersigned, Addus HealthCare, Inc., an Illinois corporation ("Borrower") refers to the Credit Agreement, dated as of September 19, 2006 (the "Credit Agreement," the terms defined therein being used herein as therein defined), by and among the undersigned, Borrower, the other Loan Parties signatory thereto, Freeport Financial LLC, for itself, as Lender, and as Agent for Lenders, and Lenders, and hereby requests, pursuant to Section 2.1(c) of the Credit Agreement, that the undersigned hereby requests the issuance of a Letter of Credit under the Credit Agreement, and in that connection sets forth below the information relating to such Letter of Credit as required by Section 2.1(c) of the Credit Agreement:

(i)	The date of issuance [or effective date of increase or extension] of the requested Letter of Credit is
(ii)	The amount [or the amount of increase] of the Letter of Credit is \$
(iii)	The name of the beneficiary of the Letter of Credit is:
(iv)	The transaction for which such Letter of Credit is to be issued is described as follows:].
(vi)	The expiry [or extended expiry] date of such Letter of Credit is:

The undersigned hereby certifies that all of the statements contained in Section 3.2 of the Credit Agreement are true and correct on the date hereof, and will be true on the date of the requested Letter of Credit, before and after giving effect thereto and to the issuance thereof.

ADDUS	HEALTHCARE, INC.
By:	
Name: Title:	

# EXHIBIT 2.2(e) to CREDIT AGREEMENT

#### FORM OF NOTICE OF CONVERSION/CONTINUATION

Reference is made to that certain Credit Agreement dated as of September 19, 2006 by and among the undersigned ("Borrower"), the other Persons named therein as Loan Parties, Freeport Financial LLC ("Agent") and the Lenders from time to time signatory thereto (including all annexes, exhibits or schedules thereto, and as from time to time amended, restated, supplemented or otherwise modified, the "Credit Agreement"). Capitalized terms used herein without definition are so used as defined in the Credit Agreement.

Borrower hereby gives irrevocable notice, pursuant to Section 2.2(e) of the Cre	dit Agreement, of its request to:
(a) on [date] convert \$[] of the aggregate outstanding principal a [] Loan [and, in the case of a LIBOR Loan, having a LIBOR Period of [	amount of the [] Loan, bearing interest at the [] Rate, into a(n) [] month(s)];
[(b) on [date] continue \$[] of the aggregate outstanding principal Loan having a LIBOR Period of [] month(s)].	l amount of the [] Loan, bearing interest at the LIBOR Rate, as a LIBOR
Borrower certifies that the conversion and/or continuation of the Loans request \$].	ed above is for the separate account of the Borrower in the following amount:
Borrower hereby (i) certifies that all of the statements contained in <u>Section 3.2</u> nereof, and will be true in all material respects on the date of the requested conversion guaranty and continuance of Agent's Liens, on behalf of itself and Lenders, pursuant	,
	ADDUS HEALTHCARE, INC.
	Ву:
	Name:
	Title:

#### Exhibit 5.12 to CREDIT AGREEMENT

#### **BUSINESS ASSOCIATE AGREEMENT**

# ADDUS HEALTHCARE, INC., AND SUBSIDIARIES AND FREEPORT FINANCIAL, LLC, AS AGENT

This Business Associate Agreement ("B.A. Agreement"), effective as of September 19, 2006 ("Effective Date"), is entered into by and among FREEPORT FINANCIAL LLC, as agent and lead arranger on behalf of the lenders party to the Credit Agreement described below ("Agent"), ADDUS HEALTHCARE, INC., a Illinois corporation ("Borrower") and each of Borrower's subsidiaries signatory hereto from time to time (Borrower and such subsidiaries that meet the Privacy Rule definition of "covered entity" are collectively referred to as "Covered Entities" and each individually is a "Covered Entity") (Agent and each of the Covered Entities are each individually a "Party" and collectively the "Parties").

- 1. BACKGROUND AND PURPOSE. Agent, Borrower, Addus Acquisition Corporation, Freeport Loan Fund LLC and certain financial institutions party thereto as lenders ("Lenders") are parties to that certain Credit Agreement dated as of September 19, 2006 (as amended, restated, modified or supplemented from time to time, the "Credit Agreement"). In connection with the Credit Agreement and transactions contemplated thereby, the Parties have entered into one or more additional contracts or agreements (collectively with the Credit Agreement, all such contracts or agreements, as they may from time to time be amended, restated, modified or supplemented, are referred to as the "Underlying Contracts"). Performance and/or enforcement of the Underlying Contracts may involve the access, use and/or disclosure by Agent of Protected Health Information (as defined in 45 C.F.R. §160.103 and limited for purposes of this B.A. Agreement solely to the information (a) received by Agent from any Covered Entity or (b) created or received by Agent on behalf of any Covered Entity, in each case with respect to the Underlying Contracts, in accordance with the terms of this B.A. Agreement) (the "PHI") that is subject to the federal privacy regulations issued pursuant to the Health Insurance Portability and Accountability Act ("HIPAA") and codified at 45 C.F.R. parts 160 and 164 (the "Privacy Rule"). The PHI may include Electronic Protected Health Information (as defined in 45 C.F.R. §160.103) ("Electronic PHI") that is subject to the federal security regulations issued pursuant to HIPAA and codified at 45 C.F.R. parts 160 and 164 ("Security Rule"). The purpose of this B.A. Agreement is to allow for each Covered Entity's compliance with the Privacy Rule and Security Rule with respect to Agent's access to PHI related to the performance of and Agent's rights under the Underlying Contracts and the use and/or disclosure of PHI by Agent.
- 2. <u>DEFINITIONS</u>. Unless otherwise defined in this B.A. Agreement, all capitalized terms used in this B.A. Agreement shall have the meanings ascribed in HIPAA, the Privacy Rule, and/or the Security Rule, as applicable.

#### 3. OBLIGATIONS OF THE PARTIES WITH RESPECT TO PHI.

3.1 <u>Uses and Disclosures of PHI by Agent.</u> Except as otherwise specified in this B.A. Agreement, Agent may make any and all uses and disclosures of PHI necessary to

perform and enforce the Underlying Contracts. Notwithstanding anything to the contrary herein, nothing in this B.A. Agreement shall authorize Agent to use or further disclose PHI in a manner that would violate the Privacy Rule or the Security Rule if done by the applicable Covered Entity, except that, unless otherwise limited in this B.A. Agreement, Agent may (a) use the PHI in its possession for its proper management and administration and to carry out the legal responsibilities of Agent, (b) disclose the PHI in its possession to a third party for the purpose of Agent's proper management and administration or to carry out the legal responsibilities of Agent, provided, that such disclosure is required by law or Agent obtains reasonable assurances from the third party regarding the confidential handling of such PHI as required under the Privacy Rule and requires the third party to which it provides Electronic PHI to agree to implement reasonable and appropriate safeguards to protect that Electronic PHI, (c) provide Data Aggregation services relating to the health care operations of the Covered Entities, (d) deidentify any and all PHI obtained by Agent under this B.A. Agreement, and use such de-identified data, all in accordance with the de-identification requirements of the Privacy Rule, and (e) use and disclose the PHI in its possession to report violations of law to appropriate Federal and State authorities, consistent with HIPAA.

- 3.2 Obligations of Agent. With regard to its access, use and/or disclosure of the PHI, Agent agrees to:
  - a. not use or further disclose the PHI other than as permitted or required by this B.A. Agreement or as permitted by law;
  - b. use appropriate safeguards to prevent use or disclosure of the PHI other than as permitted in Section 3.2(a);
- c. implement administrative, physical, and technical safeguards that reasonably and appropriately protect the confidentiality, integrity, and availability of the Electronic PHI that it creates, receives, maintains, or transmits on behalf of a Covered Entity; and makes its policies, procedures, and documentation, to the extent required by the Security Rule to be maintained relating to such safeguards, available to the Secretary of the Department of Health and Human Services ("HHS") for purposes of determining the Covered Entity's compliance with the Security Rule;
- d. report to the applicable Covered Entity in writing any use or disclosure of the PHI not permitted in Section 3.2(a) of which Agent's management becomes aware;
- e. report to the applicable Covered Entity any Security Incident of which Agent management becomes aware, provided that for purposes of this B.A. Agreement, the Parties agree that any attempted or threatened incident that does not result in a security breach, including but not limited to "pings" and other request-response utilities, does not constitute a Security Incident;
- f. ensure that any agents and subcontractors to which Agent provides the PHI agree to the same restrictions and conditions that apply to Agent with respect to such PHI;

- g. make available the PHI necessary for a Covered Entity to respond to individuals' requests for access to the PHI about them in the event that the PHI in Agent's possession constitutes a Designated Record Set;
- h. make available the PHI for amendment and incorporate any amendments to the PHI in accordance with the Privacy Rule in the event that the PHI in Agent's possession constitutes a Designated Record Set;
- i. if applicable, make available the information as would be required to allow a Covered Entity to respond to a request by an individual for an accounting of disclosures in accordance with the Privacy Rule;
- j. make its internal practices, books and records relating to the use and disclosure of PHI available to the Secretary of HHS for purposes of determining a Covered Entity's compliance with the Privacy Rule; and
- k. return to the applicable Covered Entity or destroy, within ninety (90) days of the termination of this B.A. Agreement, the PHI in its possession and retain no copies, if it is feasible to do so. If Agent in its discretion determines that return or destruction is infeasible, Agent shall provide to Covered Entities notification of the conditions that make the return or destruction infeasible, and Agent agrees to extend all protections contained in this B.A. Agreement to Agent's use and/or disclosure of any retained PHI, and to limit any further uses and/or disclosures to the purposes that make the return or destruction of the PHI infeasible for so long as Agent maintains such PHI.
- 3.3 Obligations of Covered Entities. Each Covered Entity agrees that it shall not agree with an individual to any restrictions, or use a notice of privacy practices, that would limit Agent's access, use and disclosure of PHI in order to perform and/or enforce the Underlying Contracts.
- 3.4 Effect of Changes to the Privacy Rule. To the extent that any relevant provision of the Privacy Rule and or Security Rule is materially amended in a manner that changes the obligations of Business Associates or Covered Entities that are embodied in the terms of this B.A. Agreement, the Parties agree to negotiate in good faith appropriate amendment(s) to this B.A. Agreement to give effect to these revised obligations.

#### 4. TERM AND TERMINATION BY COVERED ENTITIES.

- 4.1 The Term of this Agreement shall be effective as of the Effective Date and, unless otherwise terminated in accordance with this Section 4, shall continue in effect until 90 days after the termination of the Underlying Contracts.
- 4.2 Upon any Covered Entity's knowledge of a material breach of the terms of this B.A. Agreement by Agent, such Covered Entity shall provide Agent written notice of that breach in sufficient detail to enable Agent to understand the specific nature of that breach and afford Agent an opportunity to cure the breach. If Agent fails to cure the breach within a reasonable time specified by such Covered Entity (but in any event not less than thirty (30)

days), such Covered Entity may terminate this B.A. Agreement but in the sole discretion of Agent such termination shall result in termination of the Underlying Contracts.

#### 5. MISCELLANEOUS.

- 5.1 <u>Underlying Contracts</u>. The terms of this B.A. Agreement shall prevail in the case of any conflict with the terms of any Underlying Contract to the extent and only to the extent necessary to allow Covered Entities to comply with the Privacy Rule.
- 5.2 <u>Survival</u>. In the event that Agent determines that return of PHI is infeasible, Sections 1, 2, 3.2, 3.3, 4 and 5 of this B.A. Agreement shall survive termination of this B.A. Agreement and continue in effect solely with respect to PHI Agent retains in accordance with Section 3.2.i until such PHI is destroyed or returned.
- 5.3 No Third Party Beneficiaries. Nothing in this B.A. Agreement shall confer upon any person other than the Parties and their respective successors or permitted assigns any rights, remedies, obligations, or liabilities whatsoever.
- 5.4 <u>Joinder to B.A. Agreement</u>. Additional subsidiaries of Borrower may from time to time agree to become Parties hereto and be bound by the terms hereof by executing and delivering to Agent a Joinder to B.A. Agreement (each a "<u>Joinder</u>") substantially in the form of <u>Exhibit A</u> attached hereto. Any such Joinder shall become effective and any such subsidiary executing the same shall be considered a Party hereto for all purposes hereof when Agent shall have received, and executed an acknowledgment of, a counterpart of such Joinder duly executed by such subsidiary. Neither Borrower nor any other Party need receive any notice of such Joinder nor shall the consent or approval of any Party hereto, other than the consent of Agent, be required for any such Joinder to be effective.
- 5.5 <u>Counterparts</u>. This B.A. Agreement may be executed in counterparts, each of which shall constitute an original, but all of which taken together, along with any Joinder executed in accordance with <u>Section 5.4</u>, shall constitute a single contract.

[SIGNATURES APPEAR ON NEXT PAGE]

IN WITNESS WHEREOF, each of the undersigned has caused this B	B.A. Agreement to be duly executed in its name and on its behalf.			
BORROWER:	AGENT:			
ADDUS HEALTHCARE, INC.	FREEPORT FINANCIAL LLC			
Ву:	Ву:			
Name: Ed Budy	Name:			
Title: Assistant Secretary	Title:			
SUBSIDIARIES OF BORROWER and				
OTHER LOAN PARTIES:				
ADDUS ACQUISITION CORPORATION				
By:				
Name: Simon Bachleda				
Title: Secretary				
LOWELL HOME HEALTH AGENCY, INC.	LITTLE ROCK HOME HEALTH AGENCY, INC.			
Ву:	Ву:			
Name: Ed Budy	Name: Ed Budy			
Title: Secretary	Title: Secretary			
FORT SMITH HOME HEALTH AGENCY, INC.	BENEFITS ASSURANCE CO., INC.			
By:	Ву:			
Name: Ed Budy	Name: Ed Budy			
Title: Secretary	Title: Secretary			
PROFESSIONAL RELIABLE NURSING SERVICE, INC.	PHC ACQUISITION CORPORATION			
Ву:	By:			
Name: Ed Budy	Name: Ed Budy			
Title: Secretary	Title: Secretary			

[Signature page to Business Associate Agreement]

# EXHIBIT 7.2(e) **CREDIT AGREEMENT**

#### BORROWING AVAILABILITY CERTIFICATE

ADDUS HEALTHCARE, INC.

Date:	,	
-------	---	--

This Certificate is given by Addus HealthCare, Inc., an Illinois corporation ("Borrower") pursuant to Section 7.2(e) of that certain Credit Agreement dated as of September 19, 2006 among Borrower, the other Loan Parties party thereto, the Lenders from time to time party thereto and Freeport Financial LLC, as agent for the Lenders (as such agreement may have been amended, restated, supplemented or otherwise modified from time to time the "Credit Agreement"). Capitalized terms used herein without definition shall have the meanings set forth in the Credit Agreement.

The undersigned is duly authorized to execute and deliver this Certificate on behalf of Borrower. By executing this Certificate such officer hereby certifies to Ager

	Lenders that:				
(a)	Attached hereto as Schedule 1 is a calculation of the Borrowing Availability for Borrower as of the above date;				
(b)	Based on such schedule, the Borrowing Availability as of the above date is:				
	\$				
IN W	VITNESS WHEREOF, Borrower has caused this Certificate to be executed by its this day of				
	ADDUS HEALTHCARE, INC.				
	Ву:				
	Its:				

## SCHEDULE 1

# BORROWING AVAILABILITY

1.	Advance Multiple in effect	
2.	EBITDA of Holdings and its Subsidiaries	\$
3.	1 multiplied by 2	\$
4.	Aggregate amount of Funded Debt (including, without limitation, the outstanding principal amount of Loans and the face amount of Letter of Credit Obligations then outstanding) of Holdings and its Subsidiaries on a consolidated basis	\$
5.	3 minus 4	\$
6.	Maximum Amount less the outstanding Revolving Loans	\$
7.	Lesser of 5 and 6	\$
8.	Borrowing Availability	\$

# Exhibit 9.1 to CREDIT AGREEMENT

#### ASSIGNMENT AGREEMENT

This Assignment Agreement (this "Agreement") is made as of	, by and between	("Assignor Lender") and
("Assignee Lender"). All capitalized terms used in this Agreement and not othe	erwise defined herein will have	e the respective meanings set forth in the Credit
Agreement as hereinafter defined.		

#### RECITALS:

WHEREAS, Addus HealthCare, Inc., an Illinois corporation, ("<u>Borrower</u>"), Agent, Assignor Lender and other Persons signatory thereto as Lenders have entered into that certain Credit Agreement dated as of September 19, 2006 (as amended, restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>") pursuant to which Assignor Lender has agreed to make certain Loans to, and incur certain Letter of Credit Obligations for, Borrower;

WHEREAS, Assignor Lender desires to assign to Assignee Lender [all/a portion] of its interest in the Loans (as described below), the Letter of Credit Obligations and the Collateral and to delegate to Assignee Lender [all/a portion] of its Commitments and other duties with respect to such Loans, Letter of Credit Obligations and Collateral;

WHEREAS, Assignee Lender desires to continue as a Lender under the Credit Agreement and to accept such assignment and delegation from Assignor Lender; and

NOW, THEREFORE, in consideration of the premises and the agreements, provisions, and covenants herein contained, Assignor Lender and Assignee Lender agree as follows:

#### 1. ASSIGNMENT, DELEGATION, AND ACCEPTANCE

1.1 <u>Assignment</u>. Assignor Lender hereby transfers and assigns to Assignee Lender, without recourse and without representations or warranties of any kind (except as set forth in <u>Section 3.2</u>), [all/such percentage] of Assignor Lender's right, title, and interest in the Revolving Loan], [the Term Loan], [the Loans], [Letter of Credit Obligations], Loan Documents and the Collateral as will result in Assignee Lender having as of the Effective Date (as hereinafter defined) a Pro Rata Share thereof, as follows:

Assignee Lender's Loans	Commitment Amount	Pro Rata Share
Revolving Loan	\$	%
Term Loan	\$	%

1.2 <u>Delegation</u>. Assignor Lender hereby irrevocably assigns and delegates to Assignee Lender [all/a portion] of its Commitments and its other duties and obligations as a Lender under the Loan Documents equivalent to the Pro Rata Share set forth above.

- 1.3 <u>Acceptance by Assignee Lender</u>. By its execution of this Agreement, Assignee Lender irrevocably purchases, assumes and accepts such assignment and delegation and agrees to be a Lender with respect to the delegated interest under the Loan Documents and to be bound by the terms and conditions thereof. By its execution of this Agreement, Assignor Lender agrees, to the extent provided herein, to relinquish its rights and be released from its obligations and duties under the Credit Agreement.
- 1.4 Effective Date. Such assignment and delegation by Assignor Lender and acceptance by Assignee Lender will be effective and Assignee Lender will become a Lender under the Loan Documents as of [the date of this Agreement] [\_\_\_\_\_\_\_, \_\_\_\_] ("Effective Date") and upon payment of the Assigned Amount and the Assignment Fee (as each term is defined below). Interest and Fees accrued prior to the Effective Date are for the account of Assignor Lender, and Interest and Fees accrued from and after the Effective Date are for the account of Assignee Lender.

#### 2. INITIAL PAYMENT AND DELIVERY OF NOTES

- 2.1 <u>Payment of the Assigned Amount</u>. Assignee Lender will pay to Assignor Lender, in immediately available funds, not later than 11:00 a.m. (Chicago time) on the Effective Date, an amount equal to its Pro Rata Share of the then outstanding principal amount of the Loans as set forth above in <u>Section 1.1</u> (the "<u>Assigned Amount</u>").
- 2.2 <u>Payment of Assignment Fee</u>. [Assignor Lender and/or Assignee Lender] will pay to Agent, for its own account in immediately available funds, not later than 11:00 a.m. (Chicago time) on the Effective Date, the assignment fee in the amount of \$3,500 (the "<u>Assignment Fee</u>") as required pursuant to Section 9.1(a) of the Credit Agreement.
- 2.3 Execution and Delivery of Notes. Following payment of the Assigned Amount and the Assignment Fee, Assignor Lender will deliver to Agent the Notes previously delivered to Assignor Lender for redelivery to Borrower and, if requested by Assignee Lender, Agent will obtain from Borrower for delivery to [Assignor Lender and] Assignee Lender, new executed Notes evidencing Assignee Lender's [and Assignor Lender's respective] Pro Rata Share[s] in the Loans after giving effect to the assignment described in Section 1. Each new Note will be issued in the aggregate maximum principal amount of the [applicable] Commitment [of the Lender to whom such Note is issued] OR [the Assignee Lender].
  - 3. REPRESENTATIONS, WARRANTIES AND COVENANTS
- 3.1 <u>Assignee Lender's Representations, Warranties and Covenants</u>. Assignee Lender hereby represents, warrants, and covenants the following to Assignor Lender and Agent:
  - (a) This Agreement is a legal, valid, and binding agreement of Assignee Lender, enforceable according to its terms;
  - (b) The execution and performance by Assignee Lender of its duties and obligations under this Agreement and the Loan Documents will not require any registration with, notice to, or consent or approval by any Governmental Authority;

- (c) Assignee Lender is familiar with transactions of the kind and scope reflected in the Loan Documents and in this Agreement;
- (d) Assignee Lender has made its own independent investigation and appraisal of the financial condition and affairs of each Loan Party, has conducted its own evaluation of the Loans and Letter of Credit Obligations, the Loan Documents and each Loan Party's creditworthiness, has made its decision to become a Lender to Borrower under the Credit Agreement independently and without reliance upon Assignor Lender or Agent, and will continue to do so;
- (e) Assignee Lender is entering into this Agreement in the ordinary course of its business, and is acquiring its interest in the Loans and Letter of Credit Obligations for its own account and not with a view to or for sale in connection with any subsequent distribution; provided, however, that at all times the distribution of Assignee Lender's property shall, subject to the terms of the Credit Agreement, be and remain within its control;
- (f) No future assignment or participation granted by Assignee Lender pursuant to Section 9.1 of the Credit Agreement will require Assignor Lender, Agent, or Borrower to file any registration statement with the Securities and Exchange Commission or to apply to qualify under the blue sky laws of any state;
  - (g) Assignee Lender has no loans to, written or oral agreements with, or equity or other ownership interest in any Loan Party;
- (h) Assignee Lender will not enter into any written or oral agreement with, or acquire any equity or other ownership interest in, any Loan Party without the prior written consent of Agent; and
- (i) As of the Effective Date, Assignee Lender (i) is entitled to receive payments of principal and interest in respect of the Obligations without deduction for or on account of any taxes imposed by the United States of America or any political subdivision thereof, (ii) is not subject to capital adequacy or similar requirements under Section 2.8(a) of the Credit Agreement, (iii) does not require the payment of any increased costs under Section 2.8(b) of the Credit Agreement, and (iv) is not unable to fund LIBOR Loans under Section 2.8(b) of the Credit Agreement, and Assignee Lender will indemnify Agent from and against all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, or expenses that result from Assignee Lender's failure to fulfill its obligations under the terms of Section 2.9(c) of the Credit Agreement or from any other inaccuracy in the foregoing.

- 3.2 <u>Assignor Lender's Representations, Warranties and Covenants</u>. Assignor Lender hereby represents, warrants and covenants the following to Assignee Lender:
  - (a) Assignor Lender is the legal and beneficial owner of the Assigned Amount;
  - (b) This Agreement is a legal, valid and binding agreement of Assignor Lender, enforceable according to its terms;
  - (c) The execution and performance by Assignor Lender of its duties and obligations under this Agreement and the Loan Documents will not require any registration with, notice to or consent or approval by any Governmental Authority;
  - (d) Assignor Lender has full power and authority, and has taken all action necessary to execute and deliver this Agreement and to fulfill the obligations hereunder and to consummate the transactions contemplated hereby;
  - (e) Assignor Lender is the legal and beneficial owner of the interests being assigned hereby, free and clear of any adverse claim, lien, encumbrance, security interest, restriction on transfer, purchase option, call or similar right of a third party; and
    - (f) This Assignment by Assignor Lender to Assignee Lender complies, in all material respects, with the terms of the Loan Documents.

#### 4. LIMITATIONS OF LIABILITY

Neither Assignor Lender (except as provided in Section 3.2) nor Agent makes any representations or warranties of any kind, nor assumes any responsibility or liability whatsoever, with regard to (a) the Loan Documents or any other document or instrument furnished pursuant thereto or the Loans, Letter of Credit Obligations or other Obligations, (b) the creation, validity, genuineness, enforceability, sufficiency, value or collectibility of any of them, (c) the amount, value or existence of the Collateral, (d) the perfection or priority of any Lien upon the Collateral, or (e) the financial condition of any Loan Party or other obligor or the performance or observance by any Loan Party of its obligations under any of the Loan Documents. Neither Assignor Lender nor Agent has or will have any duty, either initially or on a continuing basis, to make any investigation, evaluation, appraisal of, or any responsibility or liability with respect to the accuracy or completeness of, any information provided to Assignee Lender which has been provided to Assignor Lender or Agent by any Loan Party. Nothing in this Agreement or in the Loan Documents shall impose upon the Assignor Lender or Agent any fiduciary relationship in respect of the Assignee Lender.

#### 5. FAILURE TO ENFORCE

No failure or delay on the part of Agent or Assignor Lender in the exercise of any power, right, or privilege hereunder or under any Loan Document will impair such power, right, or privilege or be construed to be a waiver of any default or acquiescence therein. No single or partial exercise of any such power, right, or privilege will preclude further exercise thereof or of any other right, power, or privilege. All rights and remedies existing under this Agreement are cumulative with, and not exclusive of, any rights or remedies otherwise available.

#### 6. NOTICES

Unless otherwise specifically provided herein, any notice or other communication required or permitted to be given will be in writing and addressed to the respective party as set forth below its signature hereunder, or to such other address as the party may designate in writing to the other.

#### 7. AMENDMENTS AND WAIVERS

No amendment, modification, termination, or waiver of any provision of this Agreement will be effective without the written concurrence of Assignor Lender, Agent and Assignee Lender.

#### 8. SEVERABILITY

Whenever possible, each provision of this Agreement will be interpreted in such manner as to be effective and valid under applicable law. In the event any provision of this Agreement is or is held to be invalid, illegal, or unenforceable under applicable law, such provision will be ineffective only to the extent of such invalidity, illegality, or unenforceability, without invalidating the remainder of such provision or the remaining provisions of the Agreement. In addition, in the event any provision of or obligation under this Agreement is or is held to be invalid, illegal, or unenforceable in any jurisdiction, the validity, legality, and enforceability of the remaining provisions or obligations in any other jurisdictions will not in any way be affected or impaired thereby.

#### 9. SECTION TITLES

Section and Subsection titles in this Agreement are included for convenience of reference only, do not constitute a part of this Agreement for any other purpose, and have no substantive effect.

#### 10. SUCCESSORS AND ASSIGNS

This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns.

#### 11. APPLICABLE LAW

THIS AGREEMENT WILL BE CONSTRUED IN ACCORDANCE WITH AND GOVERNED BY THE LAWS OF THE STATE OF NEW YORK APPLICABLE TO CONTRACTS MADE AND PERFORMED IN THAT STATE.

#### 12. COUNTERPARTS

This Agreement and any amendments, waivers, consents, or supplements may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which, when so executed and delivered, will be deemed an original and all of which shall together constitute one and the same instrument.

[Signature page follows]

ASSIGNEE LENDER:	ASSIGNOR LENDER:	
By: Title:	By: Title:	
Notice Address:	Notice Address:	
ACKNOWLEDGED AND CONSENTED TO: FREEPORT FINANCIAL LLC		
Ву:		
By: Title:		

# Schedule 4.1(a)

# <u>Jurisdiction of Organization and Qualification</u>

Entity	Jurisdiction of Organization	Other Jurisdictions of Qualification to do Business
Addus Holding Corporation	Delaware	None
Addus Management Corporation	Delaware	None
Addus Acquisition Corporation	Delaware	None
Addus Healthcare, Inc.	Illinois	• Alabama
		<ul> <li>California</li> </ul>
		<ul> <li>Delaware</li> </ul>
		<ul> <li>Idaho</li> </ul>
		• Indiana
		• Iowa
		<ul> <li>Missouri</li> </ul>
		<ul> <li>New Mexico</li> </ul>
		<ul> <li>New York</li> </ul>
		<ul> <li>Oregon</li> </ul>
		<ul> <li>Pennsylvania</li> </ul>
		<ul> <li>Washington</li> </ul>
		<ul> <li>Wisconsin</li> </ul>
Fort Smith Home Health Agency, Inc.	Arkansas	None
Lowell Home Health Agency, Inc.	Arkansas	None
Little Rock Home Health Agency, Inc.	Arkansas	None
Professional Reliable Nursing Service, Inc.	California	None
PHC Acquisition Corporation	California	None
Benefits Assurance Co., Inc.	Delaware	None

# Schedule 4.1(b)

# Capitalization

Owner (Certificate Issued to) See Exhibit 4.1(b)	Issuer Addus Holding Corporation	Type of Organization Corporation	# of Interests Owned	Total Shares Outstanding	Total # of Authorized Shares	% of Interest <u>Pledged</u>	Certificate No. (if uncertificated, please indicate so)	Par Value
Addus Holding Corporation	Addus Management Corporation	Corporation	1,000	1,000	1,000	100%	1	0.001
Addus Management Corporation	Addus Acquisition Corporation (immediately prior to Merger)	Corporation	1,000	1,000	1,000	100%	1	0.001
Addus Management Corporation	Addus HealthCare, Inc. (immediately following Merger)	Corporation	800	1,000	1,000	80%	CS-2	0.001
Addus Holding Corporation	Addus Healthcare, Inc. (immediately following Merger)	Corporation	200	1,000	1,000	20%	CS-1	.001

Addus HealthCare, Inc.	Fort Smith Home Health Agency, Inc.	Corporation	100	100	100	100%	1	.10
Addus HealthCare, Inc.	Lowell Home Health Agency, Inc.	Corporation	100	100	100	100%	1	.10
Addus HealthCare, Inc.	Little Rock Home Health Agency, Inc.	Corporation	100	100	100	100%	1	.10
Addus HealthCare, Inc.	Professional Reliable Nursing Service, Inc.	Corporation	200	200	10,000	100%	4	.10
Addus HealthCare, Inc.	PHC Acquisition Corporation	Corporation	100	100	100	100%	2	.10
Addus HealthCare, Inc.	Benefits Assurance Co., Inc.	Corporation	1,000	1,000	1,000	100%	2	None

# Preemptive Rights, Options, Warrants, Conversion Rights, Etc.

- 1. Rights under the Stockholders' Agreement, dated as of the Closing Date, among Addus Holding Corporation, the Investors, and the Management Stockholders (as such terms are defined therein.)
- 2. Addus Holding Corporation 2006 Stock Incentive Plan.

## **Exhibit 4.1(b)**

# **Capitalization of Holdings**

Stockholder	Shares of Series A Preferred Stock	Shares of Common Stock
Eos Capital Partners III, L.P.	28,940	0
Eos Partners SBIC III, L.P.	8,310	0
Freeport Financial LLC	500	0
W. Andrew Wright, III 281 Steeplechase Rd Barrington, IL 60010	0	71,536
Mark S. Heaney 1340 Inverness Lane Schereville, IN 46375	0	5,285
James A. Wright 79 Spring Creek Rd Barrington Hills, IL 60010	0	944
Courtney E. Panzer 4N 539 Hidden Oaks Road St. Charles, IL 60175	0	944
Addus Term Trust 281 Steeplechase Rd Barrington, IL 60010	0	3,114
W. Andrew Wright Grantor Retained Annuity Trust 281 Steeplechase Rd Barrington, IL 60010	0	12,552

#### Financial Statement Matters

- 1. The December 2005 and June 2006 liabilities associated with Addus HealthCare, Inc's pre-2006 workers compensation claims (Pre-2006 Workers Compensation Claims) were estimated based on the insurance companies loss reports and other information available at the time these financial statements were prepared. The actual liabilities associated with these Pre-2006 Workers Compensation Claims were greater than those reported in Addus HealthCare, Inc.'s consolidated financial statements.
- 2. The December 2005 and July 2006 consolidated balance sheets for Addus Healthcare, Inc. reflect prepaid insurance associated with Addus HealthCare, Inc.'s workers compensation, liability, umbrella, malpractice, property and crime insurance policies as a reduction to accrued insurance rather than as a prepaid expense. Additionally, the Cananwill, Inc. promissory note scheduled on Schedule 6.1 has been recorded on a cash basis as payments have been issued on the note.
- 3. Addus HealthCare, Inc. has not recorded any state or local corporate income taxes, penalties or interest for the Internal Revenue Service audit examination changes related to the tax periods ended December 31,1999 and 2000. These income taxes are indemnified by the sellers of Addus HealthCare, Inc.
- 4. Addus HealthCare, Inc. has not recorded any costs incurred in connection with the transaction contemplated by the consummation of the Related Transactions in its June 2006 consolidated financial statements. These costs include, but are not limited to, professional fees incurred by the Addus HealthCare, Inc.'s attorneys, accountants and advisors.

Sources and Uses

Provided under separate cover.

## Intellectual Property

#### Federally Registered Intellectual Property

A. Issued Patents

NONE

B. Patent Applications

NONE

C. Registered Trademarks: Service Mark:

		Registration	Date	Serial	Filing	
Mark	Jurisdiction	Number	Registered	Number	Date	Owner
ADDUS	U.S.A.	2,161,199	06/02/1998	75/018,761	10/23/1995	Addus Healthcare,
						Inc.

D. Trademark Applications for Registration

NONE

E. Registered Copyrights

		Registration	Date	Date	
Title	Jurisdiction	Number	Registered	Created	Owner
Addus System	U.S.A.	TXu-933-693	09/10/1999	1998	Addus Healthcare, Inc.

F. Copyright Applications for Registration

NONE

# Infringement Disclosures

On February 14, 2006, Addus HealthCare, Inc. received notice from representatives of the Business Software Alliance ("BSA") regarding the BSA's intent to investigate whether Addus HealthCare, Inc. has installed on its computers more copies of certain software companies' proprietary software products than Addus HealthCare, Inc. is licensed to use. Pursuant to the BSA's request Addus HealthCare, Inc. is performing an internal audit of all of the software in question on Addus HealthCare, Inc.'s computers at all of its locations and a review of its software licenses. Addus HealthCare, Inc. is also providing proofs of purchase for those licenses. Addus HealthCare, Inc. expects this matter to be resolved without issue and that this would not cause a Material Adverse Effect.

# <u>Investigations and Audits</u>

- 1. The California Employment Development Department has made limited inquiries of Addus HealthCare, Inc. and PHC Acquisition Corporation and Professional Reliable Nursing Service, Inc. This is connected to an audit that was performed on MHCS, Incorporated, an inactive subsidiary on non-employee payees (Form 1099 payees) for the period from 2003 through 2006.
- Addus Healthcare, Inc. is having a document review performed on its 401(k) Retirement Plan by the U.S. Department of Labor, Employee Benefits Security Administration for the period from January 2000 to present. This is <u>U.S. Department of Labor v. Addus HealthCare, Inc.</u>, Case No. 50-02760(53), U.S. Department of Labor (Chicago, IL) (April 12, 2006) (document review for the period of January 1, 2000 and forward).

# Employee Matters

#### (a) Collective Bargaining Agreements

- Service Employees International Union (SEIU),
- United Domestic Workers of America, AFSCME, AFL-CIO,
- American Federation of State, County and Municipal Employees, ALF-CIO

Union	Local	Area Covered	
Service Employees International Union	Local 775	All of Washington State	
Service Employees International Union	Local 503	All of Oregon	
Service Employees International Union	Local 715	San Mateo County	
AFSCME/ United Domestic Workers		Santa Barbara County, Riverside	
		County, Butte County	
SEIU, United Healthcare Workers West		San Joaquin County	
American Federation of State, County and Municipal Employees	Local 3416	Correctional Health in Illinois	
Service Employees International Union (Chicago, IL)	Local 880	Illinois, Indiana & Missouri	
Service Employees International Union (Harrisburg, PA)	Local 1199P	Pennsylvania & Delaware	

Addus HealthCare, Inc. has received union card checks and is the process of negotiating collective bargaining agreements with the Service Employees International Union at its Wheaton, IL, Niles, IL, York, PA, and Delaware IHSS branches. Addus HealthCare, Inc. is awaiting a pending union card check at its Charleston, IL branch. Additionally, pursuant to Addus HealthCare, Inc.'s National Agreement with the Service Workers International Union all of Addus HealthCare, Inc.'s IHSS offices are to be organized.

(<u>b</u>)

None

(c)

(d)

None

(e)

Addus HealthCare, Inc. is party to employment agreements with the following employees:

EMPLOYEE	EMPLOYMENT AGREEMENT DATE
1. W. Andrew Wright	9-18-06
2. Dave Stasiewicz	9-18-06
3. Mike Brown	New hire with a 9/29/06 start date, currently no Employment
	Agreement
4. Mark S. Heaney	9-18-06
5. Edward Budy	3-1-1999
6. Darby Anderson	9-18-06
7. Donna McNally	9-18-06
8. Kim Kruser	9-18-06
9. Lynn Keller	9-18-06
10. Sharon Kreykes	Not yet executed

**Litigation** 

Ownership of Property

Owned Real Property

None

Leased Real Property

See next page

State IL AL AL CA		Landlord W. Andrew Wright
AL AL		W. Andrew Wright
AL	36303	
		Helen Rolen
$C \Lambda$	36606	Executive Center L.L.P.
CA	94010	Equity Office
	92234-	
CA	6869	Granite/Esplanade, LLC
CA	95926	Robert Aisthorpe
CA	95926	Joseph Nock, Loraine Nock
CA	93612	Sentinel Administrative Corp.
CA	94520	Sutter Square Associates III
CA	92543	Yangon, LLC
CA	95355	Eddie L Wright
CA	92054	JEMM Family Limited Partnership
CA	96080	Orville K. and Diane B. Jacobs
CA	96001	Felice 1996 Trust
CA	92507	Marlborough Square Equities, LLC
CA		Abel & Lupe Montero
CA	92408	Executive Office Park
CA	94402	Cornerstone Properties I, LLC
CA		Katsenes Family Trust
CA	93454	C-Four Investment, Inc.
CA	95203	M - R Properties
CA	93001	Richman Trust
DE	19963	1001 MattLind Way, LLC
DE	19808	Pettinaro Enterprises
ID	83814	Meyer-Morse Ironwood Partners
IL		Ronald G Nowman
IL	62626	Farm Credit Services of Central Illinois
IL	61820-	Henson Place Properties
	7859	•
IL	61920	Coles County Community
IL		Imperial Realty
IL		Westchester Partners, L.P.
	2564	,
IL	60620	Imperial Realty
IL		M.B Real Estate Services
IL	62523	Real Estate Investors of Decatur LLC
IL		C Centre, LLC
	1433	,
IL		William E. Crocker dba Crocker Propertie
IL		Joe H. Scott, Jr & Loretta A. Scott as
	CA C	CA 6869 CA 95926 CA 95926 CA 95926 CA 95926 CA 93612 CA 94520 CA 92543 CA 92555 CA 92054 CA 96080 CA 96001 CA 92507 CA 92408 CA 94402 CA 93103 CA 93454 CA 95203 CA 93001 DE 19963 DE 19808 ID 83814 IL 60915 IL 62626 IL 61820- 7859 IL 61920 IL 60646 IL 60620 IL 60604 IL 62523 IL 62523- 1433 IL 62208

Trustees

Options, Rights of First Refusal, Etc.

**Environmental Matters** 

**ERISA** 

# Deposit and Disbursement Accounts

<u>Company</u>	<u>Bank</u>	Account Name	Account #
Addus HealthCare, Inc.	Fifth Third Bank	[***]	[***]
Addus HealthCare, Inc.	Fifth Third Bank	[***]	[***]
Addus HealthCare, Inc.	Fifth Third Bank	[***]	[***]
Addus HealthCare, Inc.	Fifth Third Bank	[***]	[***]
Addus HealthCare, Inc.	Fifth Third Bank	[***]	[***]
Addus HealthCare, Inc.	Fifth Third Bank	[***]	[***]
Addus HealthCare, Inc.	Fifth Third Bank	[***]	[***]
Addus HealthCare, Inc.	Fifth Third Bank	[***]	[***]
Addus HealthCare, Inc.	Fifth Third Bank	[***]	[***]
Little Rock Home Health Agency, Inc.	Fifth Third Bank	[***]	[***]
Lowell Home Health Agency, Inc.	Fifth Third Bank	[***]	[***]
Fort Smith Home Health Agency, Inc.	Fifth Third Bank	[***]	[***]
Addus HealthCare, Inc.	US Bank	[***]	[***]
Addus HealthCare, Inc.	US Bank	[***]	[***]

<sup>[\*\*\*]</sup> Represents material which has been redacted and filed separately with the Securities and Exchange Commission pursuant to a request for confidential treatment pursuant to Rule 406 under the Securities Act of 1933, as amended.

# Agreements and Other Documents

# Licenses and Permits

LOCATION	CLIA NO. & Expiration Date	AGENCY LICENSE NO. AND EXPIRATION DATE	MEDICAID NO.	Medicare No.	NPI No.
Fort Smith, AR	04D1003471 08 29 08	AR4235 12 31 06	138500750 138510757 138509752 137746514 137748732 137747738 139075742 141491765	04-7339	1285690842
Little Rock, AR	04D0981316	AR4090	138507750 138506752 138499757 137740514 137742732 137741738 139100742 141492765	04-7028	1295790103
Pine Bluff, AR	12 21 06	12 31 06	138507750 138506752 138499757 137740514 137742732 137741738 139100742 141492765	N/A	N/A
Rogers, AR	04D0929463 07 11 08	AR4324 12 31 06	N/A	N/A	N/A
Milford, DE	08D1035260 12 27 06	HHAS-013 03 21 07	519055 761414	08-7033	1366408932
Wilmington, DE	08D0904239 08 10 07	HHAS-014 07 03 07	761414	N/A	N/A
Olympia Fields, IL	14D0907136 10 09 07	1007889 01 31 07	421014070001	14-7683	1942263496
Chicago, IL	14D1034695 12 08 06		N/A	N/A	N/A
Chicago N., IL	14D1034695 07 31 07		N/A	N/A	N/A

LOCATION	CLIA NO. & EXPIRATION DATE	AGENCY LICENSE NO. AND EXPIRATION DATE	MEDICAID NO.	MEDICARE NO.	NPI No.
Westchester, IL	14D1034585 12 06 06		N/A	N/A	N/A
Decatur, IL	14D1028669 07 28 08	1010400	42101470004	14-7799	1689630253
Champaign, IL	14D1044905 08 30 07	12 31 06	N/A	N/A	N/A
Marion, IL	14D0990300 08 19 07		42101470002	14-7716	1902862568
Mt. Vernon, IL	14D1034586 12 06 06	1010177 08 31 07	N/A	N/A	N/A
Fairview Hts., IL	14D1034589 12 06 06		N/A	N/A	N/A
Modesto, CA	05D0610454 08 31 07	100000005 08 26 07	HHA 57272F	55-7272	1497718951
Concord, CA	05D0643815 07 10 07	020000226 08 04 07	HHA 70123G	05-7511	1003879552
Red Bluff, CA			HHA08117F	05-8117	1093771651
Redding, CA	05D0993036 10 01 07	230000338 01 02 07	HHA08 117F	N/A	N/A
Chico, CA			N/A	N/A	N/A
Riverside, CA	05D0937694 06 11 07	250000769 08 07 07	HHA08095F	05-8095	1750347365
Portland, OR	38D1001724 07 17 07	131373 12 31 06	100174	38-7146	1497718944
Lowell	N/A	N/A	138508750 138507352 138504757 138457514 138458732 138459738 138928742 141494765	04-7346	1841252699
AL	N/A	N/A	PVT0080Z	N/A	1952367526
Espanola, NM	N/A	N/A	6759 D3597	N/A	1619930161
Philadelphia, PA	N/A	N/A	100728460	N/A	1124081674
York	N/A	N/A	100728460	N/A	N/A
Schuylkill	N/A	N/A	100728460	N/A	N/A
IN	N/A	N/A	100110010a	N/A	1871559443
St. Louis	N/A	N/A	261711501 281711507 000322 (paper claims)	N/A	N/A

## <u>Instruments evidencing Indebtedness or Guaranteed Indebtedness</u>

Schedule 6.2 and Schedule 6.1 are hereby incorporated by reference.

## <u>Instruments evidencing issuance of equity securities, warranties, rights or options</u>

- 1. Stockholders' Agreement, dated as of the Closing Date, among Addus Holding Corporation, the Investors, and the Management Stockholders (as such terms are defined therein.)
- 2. Addus Holding Corporation 2006 Stock Incentive Plan.

# Agreements with referral sources or referral recipients

None

# <u>Insurance</u>

Coverage	Policy Number	Policy Term	Insurer	\$ Limits of Coverage	Deductible/Self-Insured Retention \$	Retroactive Premium Adjustments	Surety Bond Requirement (Y/N)
Workers	·	-					
Compensation/Employers							
Liability							
Oregon and Wisconsin	WC7-641-435409- 036	1/1/06 - 1/1/07	Liberty Mutual	Statutory	350,000 Per Occurrence	Yes	No
Retro Large Deductible -	WA7-64D-435409- 016	1/1/06 - 1/1/07	Liberty Mutual	WC/1,000,000 EL	350,000 Per Occurrence	No	No
All States			,	Statutory	,		
				WC/1,000,000			
				EL			
Automobile Liability	AS2-641-4354409- 026	1/1/06-1/1/07	Liberty Mutual	1,000,000 CSL	200,000 Per Occurrence	No	No
J			,	ACV Physical	2,500 Per Occurrence		
				Damage	,		
<b>General and Professional</b>							
Liability (including							
medical professional							
liability coverage)							
Primary	AHC2705813	6/1/06-07	Chicago Insurance Co.	1,000,000 Per	50,000 Per Occurrence	No	No
-			_	Occurrence	150,000 Policy		
				3,000,000 Policy	Aggregate		
				Aggregate			
Umbrella/Excess Liability	XSS2001372	6/1/06-07	Chicago Insurance Co.	5,000,000 Policy	SIR 10,000 (where	No	No
·			-	Aggregate	umbrella is broader than		
					underlying)		

Crime (including Employee Dishonesty)	104114664	8/27/06-07	Travelers Casualty & Surety Co. of America	1,000,000 Per Occurrence	50,000	No	No
Property	C2074934126	6/1/06-07	CNA	4,539,500 Blanket	5,000 - All Other Perils 25,000 - Flood 25,000 - Earthquake	No	No
Medical Malpractice Liability	SM-843294	7/1/06-07	Essex Insurance Company	1,000,000 Per Occurrence 3,000,000 Policy Aggregate	10,000 Per Occurrence	No	No

#### Schedule 4.22(a)

## Compliance with Health Care Laws

- An investigation by the Missouri Department of Health and Senior Services, Office of Special Investigations was conducted between December 20, 2005 and approximately January 24, 2006 into whether Addus HealthCare, Inc. was in compliance with the Missouri State Medicaid Plan, Section 4.10, which states, in part, "that any individual eligible under the plan may obtain Medicaid services from any institution, agency, pharmacy, person or organization that is qualified to perform the services." On September 5, 2006, the Missouri Department of Social Services sent a letter to Addus HealthCare, Inc. indicating it had found overpayment of \$9,460.80 by the Missouri Medicaid Program. On September 12, 2006, Addus HealthCare, Inc. responded that it would not appeal the finding and attaching a written plan of correction. Addus HealthCare, Inc. will repay the amount by allowing the Division of Medical Services to withhold certain amounts over time.
- 2. The Office of Field Audits of the Illinois Department of Children and Family Services reviewed Addus HealthCare, Inc.'s 2005 Certified Independent Audit Reporting Package and noted that the Reporting Package did not contain a Report on Internal Control and Compliance. Addus HealthCare, Inc. has responded to this finding and is awaiting a response from the Agency.
- 3. A Registered Nurse at the Wilmington, Delaware branch of Addus HealthCare, Inc., tested positive for drug use on September 2, 2005, resulting in this person's immediate termination in accordance with Addus HealthCare, Inc., 's Drug-Free Policy. A Statement of Complaint was filed by Addus HealthCare, Inc., with the Delaware Department of State's Division of Professional Regulation on September 2, 2005 and, upon information and belief, criminal charges have been filed.

Schedule 4.22(f)

Participation in Health Care Programs

None

## Indebtedness

- 1. Demand Promissory Note between Addus HealthCare, Inc. and Friends Home Care, Inc. in the original amount of \$37,500 and dated June 16, 2005. The total amount outstanding is \$34,375.00, as of September 19, 2006.
- 2. Regions Bank Promissory Notes. Aggregate total amount outstanding: \$136,743, as of September 19, 2006.
  - a) Promissory Note between Addus HealthCare, Inc. and Regions Bank in the original amount of \$67,200 and dated November 17, 2005. The maturity date is November 17, 2007.
  - b) Promissory Note between Addus HealthCare, Inc. and Regions Bank (f/n/a Union Planters Bank, National Association) in the original amount of \$105,855 and dated September 10, 2002. The maturity date is September 10, 2007.
  - c) Promissory Note between Addus HealthCare, Inc. and Regions Bank (f/n/a Union Planters Bank, National Association) in the original amount of \$134,799, dated July 9, 2004. The maturity date is July 9, 2009.
- 3. Commercial Insurance Premium Finance and Security Agreement, dated June 8, 2006, by and between Addus HealthCare, Inc. and Cananwill, Inc. wherein Addus HealthCare, Inc. assigned to Cananwill as security for the total amount payable any gross unearned premiums.
- 4. Addus HealthCare, Inc. issued several checks and other drafts in the ordinary course of business prior to the date of the Securities Purchase Agreement which have not been presented for payment. These checks are not individually scheduled here.
- 5. Lease between Addus HealthCare, Inc. and American Express Business Finance Corporation (Key Equipment Finance is payee) for three Ricoh copiers.

- 6. Lease between Addus HealthCare, Inc. and GE Capital for 13 Ricoh copiers and four Panasonic fax machines. Liens are described on Schedule 6.4.
- 7. Addus HealthCare, Inc. has the following obligations under equipment leases:
  - a) Obligations to and liens in favor of Citicorp Vendor Finance, Inc., for Ricoh copiers, as described on Schedule 6.4.
  - b) Obligations to and lien in favor of Wells Fargo Financial Leasing, Inc. for phone systems, as described on Schedule 6.4.
  - c) Obligations to and lien in favor of Marlin Leasing Corp. for copier systems, as described on Schedule 6.4.
  - d) Obligations to and lien in favor of Konica Minolta Business Solutions U.S.A., Inc. for copier systems, as described on Schedule 6.4.
  - e) Obligations to and lien in favor of American Express Business Finance, for leased equipment, as described on Schedule 6.4.
  - f) Obligations to and lien in favor of Banc of America Leasing & Capital, LLC, for copiers, as described on Schedule 6.4.
  - g) Obligations to and lien in favor of DeLage Landen Financial Services, Inc., for copiers, as described on Schedule 6.4.
  - h) Obligations to and lien in favor of Fleet Business Credit, LLC Financial Services, Inc., for leased equipment, as described on Schedule 6.4.
  - i) Obligations to and lien in favor of Savin Credit Corp for leased equipment, as described on Schedule 6.4.

8. Obligations with respect to the following cash-collateralized letters of credit:

Letter of Credit No.	Amount Outstanding	Applicant	Beneficiary	Issue Date	Issuing Bank
			The Travelers Indemnity		
ILS201299	2,000,000 Ad	ddus HealthCare, Inc.	Company	11/4/2005	Fifth Third Bank (Chicago)
			National Union Fire Insurance Co.		
ILS201297	1,897,223 Ad	ddus HealthCare, Inc.	of Pittsburgh, PA, et al	11/2/2005	Fifth Third Bank (Chicago)
			American Casualty Company		
ILS201309	4,282,876 Ad	ddus HealthCare, Inc.	of Reading, Pennsylvania, et al	11/14/2005	Fifth Third Bank (Chicago)
			Liberty Mutual Insurance		
ILS201061	1,275,000 Ad	ddus HealthCare, Inc.	Company	2/14/2005	Fifth Third Bank (Chicago)
			Liberty Mutual Insurance		
ILS201407	2,300,000 Ad	ddus HealthCare, Inc.	Company	2/24/2006	Fifth Third Bank (Chicago)
Total	11,755,109				

<sup>(1)</sup> Automatically extends for a period of one year unless notified at least 90 days prior to any expiration date.

# Liens

1. Addus Healthcare, Inc. has the following liens in the Secretary of State's office in Illinois.

		File# Case# Book#			Comments/
Jurisdiction	Current Secured Party of Record	Page#	File Date	File Type	Collateral
	CITICORP VENDOR FINANCE, INC	6911080	04/25/2003	Original	Leased equipment
	WELLS FARGO FINANCIAL LEASING, INC.	7504683	09/03/2003	Original	Equipment
	MARLIN LEASING CORP.	7741278	10/24/2003	Original	Leased equipment
	FLEET BUSINESS CREDIT, LLC	8053987	01/02/2004	Original	Leased equipment
	KONICA MINOLTA BUSINESS SOLUTIONS U.S.A., INC.	8386064	03/15/2004	Original	Leased equipment
	GENERAL ELECTRIC CAPITAL CORPORATION	8741646	05/27/2004	Original	Equipment
	GENERAL ELECTRIC CAPITAL CORPORATION	9056394	09/08/2004	Original	Equipment
	AMERICAN EXPRESS BUSINESS FINANCE	9080961	09/16/2004	Original	Leased equipment
	FLEET BUSINESS CREDIT, LLC	9214283	10/28/2004	Original	Leased equipment
	CITICORP VENDOR FINANCE, INC.	9676104	03/29/2005	Original	Leased equipment
	GENERAL ELECTRIC CAPITAL CORPORATION	10242355	10/04/2005	Original	Leased equipment
	BANC OF AMERICA LEASING & CAPITAL, LLC	10432758	12/05/2005	Original	Leased equipment
	DE LAGE LANDEN FINANCIAL SERVICES, INC.	10538017	01/06/2006	Original	Leased equipment
	SAVIN CREDIT CORP.	0127660664	10/01/01	Original	Leased

equipment/true lease

- 2. Liens on motor vehicles under the Regions Bank notes described on Schedule 6.1.
- 3. Liens on gross unearned premiums pursuant to the Cananwill security agreement described on Schedule 6.1.

# Contingent Obligations

- 1. Payment obligations in connection with working capital adjustments under the Purchase Agreement.
- 2. Any payments payable pursuant to Article X of the Purchase Agreement or Section 10 of the Escrow Agreement
- 3. Those arising under the lease for the property located at 2401 South Plum Grove Road, Palatine, Illinois, 66067.

# Affiliate Transactions

- 1. Purchase Agreement, dated as of the date hereof, by and among Holdings, Intermediate Holdings, Merger Sub, the Company, W. Andrew Wright, III, as sellers' representative and the other parties identified as sellers therein.
- 2. Lease with W. Andrew Wright for the property located at 2401 South Plum Grove Road, Palatine, Illinois 66067.

## **Business Description**

Addus HealthCare, Inc. (directly, and through the Subsidiaries) provides home healthcare services, including (i) in-home supportive services, (ii) Medicare-certified skilled home care services, (iii) assisted care services, and (iv) adult day care services. Addus HealthCare, Inc. previously provided (i) healthcare staffing services and (ii) correctional healthcare services.

Addus HealthCare, Inc. provides its services through a network of branch offices in Alabama, Arkansas, California, Delaware, Idaho, Illinois, Indiana, Missouri, New Mexico, Oregon, Pennsylvania and Washington and plans to expand. Addus HealthCare, Inc. derives approximately 47% of its revenue from operations in Illinois, 27% from operations in California and 9% from operations in Washington.

Addus HealthCare, Inc. provides the following services:

*In-Home Support Services.* Primarily government funded non-skilled long-term personal home care and companionship services for the elderly, non-elderly disabled and children. Services include bathing, dressing, grooming, assistance with meal preparation, housekeeping and shopping. These services are available in Alabama, California, Delaware, Idaho, Illinois, Indiana, Missouri, New Mexico, Oregon, Pennsylvania and Washington;

Skilled Home Care Services. Skilled care delivered through licensed and Medicare-certified offices enabling patients to avoid unnecessary hospitalizations and hasten hospital discharges. These services are available in Arkansas, California, Delaware, Illinois and Oregon;

Assisted Care Services. Privately funded arranged care for patients recovering from illness or injury enabling patients to recover and regain their independence. These services are available in Arkansas, California, Delaware, Idaho, Illinois, Oregon and Washington; and

Adult Day Care Services. A variety of part-time health, social and supportive services for the physically, mentally or functionally impaired, or for those simply requiring companionship in supportive service centers. Addus HealthCare, Inc. offers adult day care services at four facilities in Illinois.

74.1% of Addus HealthCare, Inc.'s revenue is generated by in-home support services, 22% is derived from skilled home care services, 2.5% is derived from assisted care services and 1.4% is derived from adult day care services.

#### CONSENT AND FIRST AMENDMENT TO CREDIT AGREEMENT

CONSENT AND FIRST AMENDMENT TO CREDIT AGREEMENT (this "Amendment"), dated as of July 29, 2007, among Addus HealthCare, Inc., an Illinois corporation ("Borrower"), the other persons signatory hereto as "Loan Parties", FREEPORT FINANCIAL LLC, a Delaware limited liability company ("Agent") and the Lenders signatory hereto. Terms not defined herein have the meanings given to them in the Credit Agreement (as hereinafter defined).

#### RECITALS

- A. Borrower, the Loan Parties, the Lenders signatory thereto and Agent are party to that certain Credit Agreement dated as of September 19, 2006 (as amended by this Amendment, the "Credit Agreement").
- B. Borrower, the Loan Parties, the Lenders and Agent are entering into this Amendment in connection with (I) the acquisition by Addus HealthCare (Nevada), Inc., a Delaware corporation ("Addus Nevada"), of certain assets and intellectual property of SuCasa Personal Care, LLC ("SuCasa") and Desert PCA of Nevada LLC ("Desert", and, collectively with SuCasa, the "SuCasa Companies"), for an aggregate purchase price of not more than \$3,500,000 (the "SuCasa Acquisition") and (II) the acquisition within ninety (90) days of the First Amendment Effective Date (the "Moore Acquisition") by the Borrower or a wholly-owned Subsidiary of the Borrower of all of the shares of capital stock of Moore Home Health Care, Inc., an Indiana corporation ("Moore") for an aggregate purchase price not to exceed \$375,000.
  - C. Agent and Lenders are willing to enter into this Amendment upon the terms and conditions set forth below.

NOW THEREFORE, in consideration of the matters set forth in the recitals and the covenants and provisions herein set forth, and other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

#### **AGREEMENT**

Section 1. <u>Definitions</u>. Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed thereto in the Credit Agreement.

Section 2. <u>Consent</u>. Agent and the Lenders hereby (A) consent to Addus Nevada (i) entering into that certain Asset Purchase Agreement, dated as of July 29, 2007, by and among Addus Nevada, the SuCasa Companies, the Sellers' Representative named therein (in such capacity, the "<u>Sellers' Representative</u>") and the Sellers set forth on Exhibit A thereto (the "<u>SuCasa Asset Purchase Agreement</u>"), that certain promissory note in the original principal amount of \$250,000, dated as of July 29, 2007 (as the same may be amended or modified as permitted by the Credit Agreement, the "<u>SuCasa Note</u>"), made by Addus Nevada in favor of Sellers' Representative and such documents executed and delivered to the SuCasa Companies

pursuant to the SuCasa Asset Purchase Agreement (the "SuCasa Acquisition Documents") and (ii) consummating the SuCasa Acquisition on the First Amendment Effective Date pursuant to the terms and conditions of the SuCasa Acquisition Documents and this Amendment for an aggregate purchase price not to exceed \$1,750,000 in cash on the First Amendment Effective Date and no more than an additional \$1,500,000 in cash following the First Amendment Effective Date to be paid at the times and in the manner specified in the SuCasa Asset Purchase Agreement provided that such payments shall not be made in the event that any Default or Event of Default under any of Sections 7.1 or 8.1(a) of the Credit Agreement has occurred and is continuing or would result therefrom; (B) consent to the Borrower or a wholly-owned Subsidiary of Borrower (i) entering into documentation with respect to the Moore Acquisition on terms and conditions satisfactory to the Agent (the "Moore Acquisition Documents"); and (ii) consummating the Moore Acquisition on or after the First Amendment Effective Date for an aggregate purchase price not to exceed \$325,000 in cash on the closing date thereof pursuant to the terms of the Moore Acquisition Documents plus certain deferred purchase price owed to the sellers of Moore in an amount not to exceed \$50,000; and (C) agree that for purposes of calculating any Excess Cash Flow prepayment amount that may be due in respect of any Fiscal Year pursuant to Section 2.5(b) of the Credit Agreement, (i) the unfinanced amount of the purchase price paid in cash during such Fiscal Year for the Moore Acquisition and (ii) the unfinanced installment payments made during such Fiscal Year pursuant to the SuCasa Acquisition Documents shall be deducted from the "Subtotal" determined as part of the calculation for "Excess Cash Flow" set forth on Schedule 2 to Annex E of the Credit Agreement prior to applying the "Required Prepayment Percentage" set forth on such Schedule.

Section 3. Amendment to the Credit Agreement. As of the First Amendment Effective Date, the Credit Agreement is hereby amended as follows:

3.1. Section 1.1 of the Credit Agreement is hereby amended by adding the following new definitions thereto:

Addus Nevada means Addus HealthCare (Nevada), Inc., a Delaware corporation.

First Amendment means that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007 among the Loan Parties, Agent and the Lenders.

First Amendment Effective Date has the meaning set forth in the First Amendment.

Moore has the meaning set forth in the Recitals to the First Amendment.

Moore Acquisition has the meaning set forth in the Recitals to the First Amendment.

Moore Acquisition Documents has the meaning set forth in Section 2 of the First Amendment.

<u>Reaffirmation of Collateral Documents</u> means the Consent and Reaffirmation dated as of July 29, 2007 of the Loan Parties signatory thereto, in respect of the Collateral Documents.

SuCasa Acquisition has the meaning set forth in the Recitals to the First Amendment.

SuCasa Acquisition Documents has the meaning set forth in Section 2 of the First Amendment.

SuCasa Asset Purchase Agreement has the meaning set forth in Section 2 of the First Amendment.

SuCasa Companies has the meaning set forth in the Recitals to the First Amendment.

SuCasa Note has the meaning set forth in Section 2 of the First Amendment.

3.2. Section 1.1 of the Credit Agreement is further amended by amending the definition of "EBITDA" by deleting the word "and" that immediately precedes "(xiv)" and replacing it with "," and adding the following language immediately after clause (xiv) therein:

", and (xv) in connection with calculating any monthly financial statements required by <u>Section 7.2(a)</u> for any period which includes the month of September, 2006, without duplication, the one-time charge of \$146,431 expensed pre closing in September 2006 in connection with an audit conducted by the U.S. Department of Labor;"

3.3. The proviso at the end of the definition of "EBITDA" is amended and restated in its entirety to read as follows:

"provided, that, notwithstanding anything to the contrary contained herein, for each of the Fiscal Quarters listed below, EBITDA shall be deemed to be the amount set forth below opposite such corresponding period:

Period	EBITDA
September 30, 2006	\$ 3,563,347
December 31, 2006	\$ 3,843,068
March 31, 2006	\$3,062,914;

Provided, further, that for any period that includes the month of April, May, June or July of 2007, EBITDA shall, without duplication, be increased with respect to the assets acquired in connection with the SuCasa Acquisition by an amount equal to \$71,897 for each such month."

3.4. Section 6.1 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause (i) thereof, deleting the "." at the end of clause (j) thereof and replacing it with ";" and by adding the following new clauses (k), (l) and (m):

"(k) Indebtedness in respect of that certain 8% Junior Subordinated Promissory Note by Addus Nevada payable to Glen Schlosser, as Seller's Representative (on behalf the Sellers (as defined in the SuCasa Asset Purchase Agreement)), in the original principal amount of \$250,000;

This figure includes the one-time charge of \$146,431 expensed pre closing in September 2006 in connection with an audit conducted by the U.S. Department of Labor.

- "(1) Indebtedness for the deferred purchase price in connection with the Moore Acquisition up to \$50,000; and
- (m) Indebtedness consisting of surety bonds issued to licensing authorities in connection in the ordinary course of business in an aggregate amount for all such surety bonds not to exceed \$250,000 outstanding at any time."
- 3.5. Section 6.3 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause (j) thereof, deleting the "." at the end of clause (k) thereof and replacing it with ";" and by adding the following new clauses (l) and (m):
- "(I) Addus Nevada may consummate the SuCasa Acquisition as of the First Amendment Effective Date pursuant to the SuCasa Acquisition Documents; and
- (m) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Moore Acquisition within ninety (90) days of the First Amendment Effective Date pursuant to the Moore Acquisition Documents and all applicable law, provided that the Moore Acquisition Documents shall have been duly authorized, executed and delivered by each of the respective parties thereto and shall be in full force and effect and provided further that the Agent shall have received a complete copy of the fully executed Moore Acquisition Documents, certified by the Borrower as being true, complete and correct."
  - 3.6. Clause (d) of Section 6.6 of the Credit Agreement is hereby amended and restated in its entirety to read as follows:
- "(d) acquire by purchase or otherwise all or any part of the Stock, business or assets of any other Person, provided that (i) Addus Nevada may consummate the SuCasa Acquisition as of the First Amendment Effective Date pursuant to the SuCasa Acquisition Documents; and (ii) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Moore Acquisition on or after the First Amendment Effective Date pursuant to the Moore Acquisition Documents and all applicable law, provided that the Moore Acquisition Documents shall have been duly authorized, executed and delivered by each of the respective parties thereto and shall be in full force and effect and provided further that the Agent shall have received a complete copy of the fully executed Moore Acquisition Documents, certified by the Borrower as being true, complete and correct."
  - 3.7. Section 6.16 of the Credit Agreement is amended and restated in its entirety to read as follows:

"None of the Loan Parties shall change or amend the terms of the Acquisition Note or SuCasa Note if such change or amendment would be adverse in any material respect to the rights or interests of the Loan Parties, the Agent or the Lenders."

3.8. The address for notice to the Borrower in Section 10.3 of the Credit Agreement is amended and restated in its entirety to read as follows:

"If to Borrower: Addus HealthCare, Inc.

2401 South Plum Grove Road

Palatine, Illinois 60067 ATTN: Mark S. Heaney Fax: (847) 303-5376

With a copy to: Nixon Peabody LLP

437 Madison Avenue New York, NY 10022 ATTN: Peter J. Alfano, Esq. Fax: (212) 940-3111"

Section 4. Representations and Warranties. To induce Agent and Lenders to execute this Amendment, each Loan Party hereby represents and warrants to Agent and Lenders as follows:

- (a) the execution, delivery and performance of each of this Amendment and the SuCasa Acquisition Documents have been duly authorized by all requisite action of each Loan Party thereto, and each of this Amendment and the SuCasa Acquisition Documents constitutes the legal, valid and binding obligation of such Person, enforceable against such Person in accordance with its terms, subject to bankruptcy, insolvency and similar laws affecting the enforceability of creditors' rights generally and to general principles of equity; and
- (b) each of the representations and warranties in the Credit Agreement are true and correct in all material respects with the same effect as though made on and as of the date hereof (except, in each case, to the extent stated to relate to an earlier date, in which case such representation or warranty shall have been true and correct on and as of such earlier date) and no Event of Default or Default exists thereunder or would exist after giving effect to this Amendment or the SuCasa Acquisition.
- Section 5. <u>Conditions Precedent</u>. This Amendment shall be effective as of the date first set forth above, subject to the satisfaction of the following condition precedent (the date of such satisfaction being the "<u>First Amendment Effective Date</u>"):
- 5.1. Execution and Delivery. Borrower, each of the other Loan Parties, Agent and the Requisite Lenders shall have executed and delivered this Agreement; the Borrower shall have executed and delivered an Officers Certificate in the form of Exhibit A attached hereto; and each Loan Party shall have executed and delivered the Consent and Reaffirmation in the form of Exhibit B attached hereto.
- 5.2. Financial Status. As of the First Amendment Effective Date after giving effect to the incurrence on the First Amendment Effective Date to the payment of all related costs and expenses incurred in connection with the SuCasa Acquisition and this Amendment, (a) the

Leverage Ratio shall not exceed 3.60 to 1.00, (b) not more than \$6,000,000 in the aggregate of Revolving Loans and Letter of Credit Obligations shall be outstanding, and (c) Borrowing Availability shall be at least \$6,500,000.

- 5.3. <u>Due Diligence</u>. Agent, Lenders and their counsel and advisors shall have completed their legal, accounting and financial due diligence review (including, without limitation, lien searches on the SuCasa Companies) of the SuCasa Companies and the SuCasa Acquisition, the results of which review shall be reasonably satisfactory to Agent and Requisite Lenders.
- 5.4. <u>Material Adverse Effect</u>. There shall have occurred no Material Adverse Effect upon the financial condition, operations, assets, business or properties of the Loan Parties or the SuCasa Companies since December 31, 2006.
- 5.5. SuCasa Acquisition Documents. On or prior to the First Amendment Effective Date, each of the SuCasa Acquisition Documents shall have been duly authorized, executed and delivered by each of the respective parties thereto and shall be in full force and effect and shall not have been amended or modified in any material manner except for such material amendments and modifications, if any, as may be reasonably satisfactory to Agent and the Required Lenders. The SuCasa Acquisition shall have been, or shall be contemporaneously herewith, consummated substantially in accordance with the SuCasa Acquisition Documents and in accordance with all applicable law, except as may have been consented to in writing by Agent. Agent shall have received a complete copy of the fully executed SuCasa Acquisition Documents, certified by the Borrower as being true, complete and correct.
- 5.6. Approvals. All necessary governmental and third party approvals and/or consents in connection with the SuCasa Acquisition and otherwise referred to herein or therein shall have been obtained and remain in effect, and all applicable waiting periods shall have expired without any action being taken by any competent authority which restrains, prevents or imposes materially adverse conditions upon the consummation of all or any part of the SuCasa Acquisition. Additionally, there shall not exist any judgment, order, injunction or other restraint pending or notified prohibiting or imposing material adverse conditions upon all or any part of the SuCasa Acquisition or the transactions contemplated by this Amendment.
  - 5.7. No Defaults. After giving effect to this Amendment, no Event of Default or Default under the Credit Agreement shall have occurred and be continuing.
- 5.8. Representations and Warranties. After giving effect to this Amendment, the representations and warranties of the Loan Parties contained in this Amendment, the Credit Agreement and the other Loan Documents shall be true and correct in all material respects as of the First Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty shall be true and correct in all material respects as of such earlier date.

- 5.9. <u>Payment of Fees</u>. The Borrower shall have paid to Agent for its own account all other fees specified by that certain fee letter between the Borrower and Agent as of the date hereof.
- 5.10. Secretary's Certificate. On the First Amendment Effective Date, Agent shall have received from each Loan Party a certificate signed by the secretary or an assistant secretary of such Loan Party, dated the First Amendment Effective Date, as to the incumbency and signature of the officers of such Loan Party executing any Loan Document and any certificate or other document or instrument to be delivered pursuant hereto by or on behalf of such Loan Party, together with evidence of the incumbency of such Secretary or Assistant Secretary, as the case may be and certifying that there have been no changes to (i) the resolutions of such Loan Party authorizing and approving, among other things, the execution and delivery of the Loan Documents and (ii) the organizational documents of such Loan Party in each case since the Closing Date.
- 5.11. <u>Resolutions</u>. Agent shall have received a copy of the resolutions (in form and substance reasonably satisfactory to Agent) adopted by the Board of Directors of each Loan Party, authorizing or relating to the execution, delivery and performance of this Amendment and the SuCasa Acquisition Documents and the other documents and instruments provided for therein and the consummation of the transactions contemplated hereby and thereby.
- 5.12. Good Standing Certificates. Agent shall have received good standing certificates for each Loan Party from their respective jurisdictions of incorporation or organization.
- 5.13. <u>Fees</u>. Borrower shall have paid all reasonable and documented costs, fees and expenses (including, without limitation, reasonable legal fees and expenses of Winston & Strawn LLP) of Agent.
- 5.14. Other Matters. Agent shall have received such other instruments and documents as Agent or the Required Lenders may reasonably request in connection with the execution of this Amendment, and all such instruments and documents shall be reasonably satisfactory in form and substance to Agent.

#### Section 6. Miscellaneous.

- 6.1. Effect of Amendment. The execution, delivery and effectiveness of this Amendment shall not operate as a waiver of any right, power or remedy of Agent or any Lender under the Credit Agreement or any Loan Document, or constitute a waiver of any provision of the Credit Agreement or any Loan Document, except as specifically set forth herein.
- 6.2. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and by the different parties on separate counterparts, and each such counterpart shall be deemed to be an original, but all such counterparts shall together constitute but one and the same instrument. Delivery of the executed counterpart of this Amendment by telecopy or electronic mail shall be as effective as delivery of a manually executed counterpart to this Amendment.

- 6.3. <u>Severability</u>. The illegality or unenforceability of any provision of this Amendment or any instrument or agreement required hereunder shall not in any way affect or impair the legality or enforceability of the remaining provisions of this Amendment or any instrument or agreement required hereunder.
  - 6.4. Captions. Section captions used in this Amendment are for convenience only, and shall not affect the construction of this Amendment.
- 6.5. Entire Agreement. This Amendment embodies the entire agreement and understanding among the parties hereto and supersedes all prior or contemporaneous agreements and understandings of such Persons, verbal or written, relating to the subject matter hereof.
- 6.6. <u>References</u>. Any reference to the Credit Agreement contained in any notice, request, certificate, or other document executed concurrently with or after the execution and delivery of this Amendment shall be deemed to include this Amendment unless the context shall otherwise require. Reference in any of this Amendment, the Credit Agreement or any other Loan Document to the Credit Agreement shall be a reference to the Credit Agreement as amended hereby and as further amended, modified, restated, supplemented or extended from time to time.

Section 7. <u>Governing Law</u>. THIS AMENDMENT SHALL BE A CONTRACT MADE UNDER AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF ILLINOIS APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE, WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES.

[signature pages follow]

Witness the due execution hereof by the respective duly authorized officers of the undersigned as of the date first above written.

## ADDUS HEALTHCARE, INC.

 By:
 /s/ Mark Heaney

 Name:
 Mark Heaney

 Title:
 Vice President

## ADDUS HOLDING CORPORATION

By: /s/ Simon Bachleda

Name: Simon Bachleda

Title: Vice President

## ADDUS MANAGEMENT CORPORATION

By: /s/ Simon Bachleda
Name: Simon Bachleda
Title: Vice President

## LOWELL HOME HEALTH AGENCY, INC.

 By:
 /s/ Mark Heaney

 Name:
 Mark Heaney

 Title:
 Vice President

## LITTLE ROCK HOME HEALTH AGENCY, INC.

By: /s/ Mark Heaney
Name: Mark Heaney
Title: Vice President

## FORT SMITH HOME HEALTH AGENCY, INC.

By: /s/ Mark Heaney

Name: Mark Heaney

Title: Vice President

## BENEFITS ASSURANCE CO., INC.

 By:
 /s/ W. Andrew Wright

 Name:
 W. Andrew Wright

 Title:
 President

## PHC ACQUISITION CORPORATION

By: /s/ Mark Heaney
Name: Mark Heaney
Title: Vice President

#### PROFESSIONAL RELIABLE NURSING SERVICE, INC.

 By:
 /s/ Mark Heaney

 Name:
 Mark Heaney

 Title:
 Vice President

## FREEPORT FINANCIAL LLC, as Agent

By: /s/ Chad Blakeman

Title: Senior Managing Director

## FREEPORT LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman

Title: Senior Managing Director

# FREEPORT OFFSHORE LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman

Title: Senior Managing Director

## FREEPORT LOAN TRUST 2006-1, as a Lender

/s/ Chad Blakeman

Title: Senior Managing Director

# FIFTH THIRD BANK, CHICAGO (A Michigan Banking Company)

/s/ Gregory H. Bork

Name: Gregeory H. Bork

Title: Vice President

## RESIDENTIAL FUNDING COMPANY LLC

/s/ Dennis M. Hansen

Name: Dennis M. Hansen

Title: SVP

By:

By:

#### **OFFICER'S CERTIFICATE**

Reference is made to that certain Consent and First Amendment to Credit Agreement, dated as of July 29, 2007 (the "First Amendment"; capitalized terms used herein and not otherwise defined shall have the meanings specified thereto in that certain Credit Agreement dated as of September 19, 2006 among the Loan Parties party thereto, the Agent and the Lenders party thereto (the "Credit Agreement") as amended by the First Amendment), among the Borrower, certain other parties, the financial institutions party thereto and Freeport Financial LLC, as Agent. The undersigned hereby certifies that [s]he is the duly authorized and elected of the Borrower and that, as of the date hereof:

- 1. As of the First Amendment Effective Date, after giving effect to the payment of all related costs and expenses incurred in connection with the SuCasa Acquisition and the First Amendment, (a) the Leverage Ratio does not exceed 3.60 to 1.00, (b) not more than \$6,000,000 in the aggregate of Revolving Loans and Letter of Credit Obligations are outstanding and (c) Borrowing Availability is at least \$6,500,000.
- No Material Adverse Effect has occurred upon the financial condition, operations, assets, business or properties of the Loan Parties or SuCasa Companies since December 31, 2006.
- 3. After giving effect to the First Amendment, no Event of Default under the Credit Agreement has occurred or is continuing.
- 4. After giving effect to the First Amendment, the representations and warranties of the Loan Parties contained in the First Amendment, the Credit Agreement and the other Loan Documents are true and correct in all material respects as of the First Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty was true and correct in all material respects as of such earlier date.
- 5. The Loan Parties have consummated the SuCasa Acquisition substantially in accordance with the SuCasa Acquisition Documents and in accordance with all applicable law, except as may have been consented to in writing by Agent.

[Remainder of Page Intentionally Blank]

ADDUS HEALTHCARE, INC.
By: Name:
Name: Title:

IN WITNESS WHEREOF, I have hereunto set my hand this  $\_$  day of July, 2007.

EXHIBIT B

# **CONSENT AND REAFFIRMATION**

Each of the undersigned ("Loan Parties") hereby (i) acknowledges receipt of a copy of the Consent and First Amendment to Credit Agreement dated July 29,
2007 (the "First Amendment"); (ii) consents to the execution and delivery thereof by the other Loan Parties; (iii) agrees to be bound thereby; (iv) affirms that nothing
contained therein shall modify in any respect whatsoever its guaranty of the obligations of the Loan Parties to Agent and Lenders (the "Guarantee"), (v) reaffirms that
the Guarantee and the other Collateral Documents executed by such Person are and shall continue to remain in full force and effect and (vi) supplements Schedules
, and to the Collateral Documents with the new information attached to this Consent and Reaffirmation as Exhibit C-1. Although each of the
Loan Parties has been informed of the matters set forth herein and in the First Amendment and has acknowledged and agreed to same, such Loan Parties understand
that Agent and Lenders have no obligation to inform any of the Loan Parties of such matters in the future or to seek any of the Loan Parties' acknowledgment or
agreement to future amendments or waivers, and nothing herein shall create such a duty.

This Consent and Reaffirmation shall be governed by and construed in accordance with the laws of the State of Illinois, without reference to principles of conflicts of law.

In witness whereo	f each of the u	indersjoned has e	xecuted this Con-	ent and Reaffirmatio	n on and as of the	late of such First Amendment

# ADDUS HEALTHCARE, INC.

Name:	
Title:	
11110.	
ADDUS	S HOLDING CORPORATION
By:	
Name:	
Title:	
ADDUS	S HEALTHCARE (NEVADA), INC.
By:	
Name:	
Title:	
ADDUS	S MANAGEMENT CORPORATION
By:	
Name:	
Title:	
LOWE	LL HOME HEALTH AGENCY, INC.
Bv:	
By: Name:	
Name:	
Name: Title:	
Name: Title:	E ROCK HOME HEALTH AGENCY, INC.
Name: Title: LITTL By:	E ROCK HOME HEALTH AGENCY, INC.
Name: Title: LITTL By: Name:	E ROCK HOME HEALTH AGENCY, INC.
Name: Title: LITTL By:	E ROCK HOME HEALTH AGENCY, INC.

FORT S	SMITH HOME HEALTH AGENCY, INC.
By:	
Name:	
Title:	
BENEF	ITS ASSURANCE CO., INC.
By:	
NI	
Name:	
Title:	
Title:  PHC AG  By:	CQUISITION CORPORATION
Title:  PHC AG  By: Name: Title:	CQUISITION CORPORATION
Title:  PHC AG  By: Name: Title:	CQUISITION CORPORATION
Title:  PHC AG  By: Name: Title:  PROFE	CQUISITION CORPORATION SSIONAL RELIABLE NURSING SERVICE, INC

#### **CONSENT AND SECOND AMENDMENT TO CREDIT AGREEMENT**

CONSENT AND SECOND AMENDMENT TO CREDIT AGREEMENT (this "<u>Amendment</u>"), dated as of October 15, 2007, among Addus HealthCare, Inc., an Illinois corporation ("<u>Borrower</u>"), the other persons signatory hereto as "Loan Parties", FREEPORT FINANCIAL LLC, a Delaware limited liability company ("<u>Agent</u>") and the Lenders signatory hereto. Terms not defined herein have the meanings given to them in the Credit Agreement (as hereinafter defined).

#### RECITALS

A. Borrower, the Loan Parties, the Lenders signatory thereto and Agent are party to that certain Credit Agreement dated as of September 19, 2006 (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007 and as further amended, restated or otherwise modified including by this Amendment, the "<u>Credit Agreement</u>").

B. Borrower, the Loan Parties, the Lenders and Agent are entering into this Amendment in connection with (I) the acquisition (the "Silver State Acquisition") by the Borrower or a wholly-owned Subsidiary of the Borrower of substantially all of the assets of Silver State Personal Care, Inc., a Nevada corporation ("Silver State") for an aggregate purchase price not to exceed \$2,000,000, (II) the increase of the Revolving Loan Commitment by \$5,000,000, (III) the increase of the Term Loan Commitment by \$17,500,000 (the "Term Loan Commitment Increase"), under which an initial advance in the amount of \$5,000,000 will be made on the Second Amendment Effective Date (the "Second Amendment Term Loan Commitment Increase") and multiple subsequent advances in the aggregate amount of \$12,500,000 (such amount, the "Delayed Draw Term Loan Commitment Increase") are proposed to be made on certain Delayed Draw Dates (as defined in the Credit Agreement) and (IV) the continuation of the existing Term Loans (of which \$42,200,000 in aggregate principal amount remains outstanding as of the Second Amendment Effective Date before giving effect to the Second Amendment).

C. Agent, Lenders and the Loan Parties are willing to enter into this Amendment upon the terms and conditions set forth below.

NOW THEREFORE, in consideration of the matters set forth in the recitals and the covenants and provisions herein set forth, and other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

#### **AGREEMENT**

Section 1. <u>Definitions</u>. Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed thereto in the Credit Agreement.

Section 2. <u>Consent.</u> Agent and the Lenders hereby consent to the Borrower or a wholly-owned Subsidiary of Borrower (i) entering into agreements, instruments and other documents for the purpose of consummating the Silver State Acquisition in each case on terms and conditions satisfactory to the Agent (the "<u>Silver State Acquisition Documents</u>", it being agreed that the forms of such agreements, instruments and other documents provided to the Agent on October 12, 2007 are satisfactory to the Agent); and (ii) consummating the Silver State Acquisition on the Second Amendment Effective Date for an aggregate purchase price not to exceed \$1,500,000 paid in cash on the closing date thereof pursuant to the terms of the Silver State Acquisition Documents, plus a deferred purchase price in an amount not to exceed \$500,000 paid to Silver State pursuant to the Silver State Note; provided that such deferred purchase price shall not be paid in the event that any Default or Event of Default under any of <u>Sections 7.1 or 8.1(a)</u> of the Credit Agreement has occurred and is continuing or would result therefrom or the Borrower does not have the minimum Borrowing Availability required by <u>Section 6.17</u> of the Credit Agreement after giving effect to such payment.

Section 3. Amendment to the Credit Agreement. As of the Second Amendment Effective Date, the Credit Agreement is hereby amended as follows:

3.1. Section 1.1 of the Credit Agreement is hereby amended by adding the following new definitions thereto:

Delayed Draw Date has the meaning set forth in Section 2.1(a).

Delayed Draw Term Loan Commitment means (a) as to any Lender, the commitment of such Lender to make its Pro Rata Share of the Delayed Draw Term Loans (as set forth on Annex A to the Second Amendment or in the most recent Assignment Agreement, if any, executed by such Lender) in the maximum aggregate amount not to exceed the Delayed Draw Term Loan Commitment Increase and (b) as to all Lenders, the aggregate commitment of all Lenders to make the Delayed Draw Term Loans in the maximum aggregate amount not to exceed the Delayed Draw Term Loan Commitment Increase, in each case as reduced by Term Loans made on any Delayed Draw Date or otherwise as reduced pursuant hereto.

<u>Delayed Draw Term Loan Commitment Increase</u> has the meaning set forth in the Recitals to the Second Amendment.

Delayed Draw Term Loans has the meaning set forth in Section 2.1(a).

Original Term Loans has the meaning set forth in Section 2.1(a).

Second Amendment means that certain Consent and Second Amendment to Credit Agreement dated as of October 15, 2007 among the Loan Parties, Agent and the Lenders.

Second Amendment Effective Date has the meaning set forth in the Second Amendment.

<u>Second Amendment Reaffirmation of Collateral Documents</u> means the Consent and Reaffirmation (Second Amendment) dated as of October 15, 2007 of the Loan Parties signatory thereto, in respect of the Collateral Documents.

Second Amendment Term Loan Commitment means (a) as to any Lender, the commitment of such Lender to make its Pro Rata Share of the Second Amendment Term Loans (as set forth on Annex A to the Second Amendment or in the most recent Assignment Agreement, if any, executed by such Lender) in the maximum aggregate amount not to exceed the Second Amendment Term Loan Commitment Increase, and (b) as to all Lenders, the aggregate commitment of all Lenders to make the Second Amendment Term Loans.

Second Amendment Term Loan Commitment Increase has the meaning set forth in the Second Amendment.

Second Amendment Term Loans has the meaning set forth in Section 2.1(a).

Silver State has the meaning set forth in the Recitals to the Second Amendment.

Silver State Acquisition has the meaning set forth in the Recitals to the Second Amendment.

Silver State Acquisition Documents has the meaning set forth in Section 2 of the Second Amendment.

Silver State Note means that certain 8% Junior Subordinated Promissory Note by Borrower payable to Silver State in the original principal amount of \$500,000.

3.2. Section 1.1 of the Credit Agreement is further amended by amending and restating the proviso at the end of the definition of "EBITDA" to read as follows:

"provided, that, notwithstanding anything to the contrary contained herein, for each of the Fiscal Quarters listed below, EBITDA shall be deemed to be the amount set forth below opposite such corresponding period:

Period	EBITDA
December 31, 2006	\$4,024,457
March 31, 2007	\$3,224,724
June 30, 2007	\$3,340,717

Provided, further, that for any period that includes the month of July, August or September of 2007, EBITDA shall, without duplication, be increased with respect to the assets acquired in connection with the Silver State Acquisition by an amount equal to \$63,000 for each such month."

3.3. Section 1.1 of the Credit Agreement is further amended by amending and restating the definition of "Fixed Charge Coverage Ratio" to read as follows:

"Fixed Charge Coverage Ratio" means for any measuring period the ratio of (x) (i) EBITDA less (ii) Capital Expenditures, other than the portion thereof funded by third party financing and less (iii) the sum of all federal, state and local income taxes and

franchise taxes (excluding (1) provisions for taxes in respect of gains on the sale of assets, and extraordinary and non-recurring gains and (2), for purposes of calculating the "Fixed Charge Coverage Ratio" for any measuring period that includes the months of March, April, June, September and December of Fiscal Year 2007, \$1,400,000 for March, 2007 and \$400,000 for each of April, 2007, June, 2007, September, 2007 and December, 2007) paid in cash (net of any credit for such taxes), to (y) Fixed Charges."

- 3.4. Section 1.1 of the Credit Agreement is further amended by amending and restating the definition of "McKesson Add-Back" to read as follows:
- "McKesson Add-Back means an amount equal to the expenses booked by the Borrower and its Subsidiaries with respect to McKesson system implementation services from and after the Closing Date in an amount not to exceed \$1,000,000."
- 3.5. Section 1.1 of the Credit Agreement is further amended by deleting the definition of "Revolving Loan Commitment" and replacing it with the following new definition:

"Revolving Loan Commitment means (a) as to any Lender, the commitment of such Lender to make its Pro Rata Share of Revolving Credit Advances or incur its Pro Rata Share of Letter of Credit Obligations as set forth on Annex A or in the most recent Assignment Agreement, if any, executed by such Lender and (b) as to all Lenders, the aggregate commitment of all Lenders to make the Revolving Credit Advances or incur Letter of Credit Obligations, which aggregate commitment shall be Seventeen Million Five Hundred Thousand Dollars (\$17,500,000) on the Second Amendment Effective Date, as such amount may be adjusted, if at all, from time to time in accordance with the Agreement."

3.6. Section 1.1 of the Credit Agreement is further amended by adding the following new sentence to the definition of "Term Loan Commitment" immediately at the end thereof:

"The Term Loan Commitment shall include each Lender's Delayed Draw Term Loan Commitment and the Second Amendment Term Loan Commitment which, together with the Term Loans outstanding immediately prior to giving effect to the Second Amendment collectively equal \$59,700,000."

- 3.7. Section 2.1(a) of the Credit Agreement is hereby amended and restated in its entirety and replaced with the following new Section 2.1(a):
- "Term Loans. (i) On the Closing Date, each Term Lender funded its Pro Rata Share to Borrower of \$45,000,000 (the "Original Term Loans"). On the Second Amendment Effective Date, Original Term Loans remain outstanding in the principal amount of \$42,200,000.
- (ii) On the Second Amendment Effective Date, each Lender shall make available funds equal to its Pro Rata Share of the aggregate Second Amendment Term Loan Commitment in immediately available funds to the Agent (such loans the "Second Amendment Term Loans").

The Second Amendment Term Loan Commitment shall expire concurrently with the making of the related Term Loans on the Second Amendment Effective Date. Borrower covenants and agrees that it shall use the proceeds of the Second Amendment Term Loans advanced on the Second Amendment Effective Date solely to reduce outstanding Revolving Loans, fund its purchase price obligations in respect of the Silver State Acquisition and for fees and expenses incurred in connection with the closing of the transactions contemplated by the Silver State Acquisition Documents and the Second Amendment.

- (iii) On each Delayed Draw Date, subject to the terms and conditions in clause (iv) of this Section 2.1(a) below, each Lender shall make available funds equal to its Pro Rata Share of the Delayed Draw Term Loan Commitment to be drawn on such date in immediately available funds to the Agent (the Loans made on each such date "Delayed Draw Term Loans" and, collectively with the Second Amendment Term Loans and Original Term Loans, collectively, the "Term Loans"). Each borrowing of a Delayed Draw Term Loan shall be in an aggregate minimum amount of \$2,500,000 and integral multiples of \$500,000 in excess of that amount. The Delayed Draw Term Loan Commitment shall expire on the earlier of (i) the date the Delayed Draw Term Loan Commitment is permanently reduced to zero, (ii) the date of the termination of the Commitments pursuant to Section 8.3 and (iii) the date that is nine (9) months after the Second Amendment Effective Date, and any portion of the Delayed Draw Term Loan Commitment unused by the Borrower as of such date shall be automatically terminated. Borrower covenants and agrees that it shall use the proceeds of the Delayed Draw Term Loans (i) to fund Acquisitions consented to by the Requisite Lenders, (ii) to pay fees and expenses incurred in connection with such Acquisition and any amendment to the Credit Agreement in connection therewith and (iii) contemporaneously with the funding of any Acquisition consented to by the Requisite Lenders, to provide additional working capital for the Borrower in connection with such Acquisition.
- (iv) The obligation of each Lender to make Loans in respect of its Delayed Draw Term Loan Commitment is, in addition to the conditions precedent specified in Section 3.2 hereof, subject to the conditions precedent that (i) the Agent shall have received all of the following, each duly executed and dated as of the date of funding of the Delayed Draw Term Loan requested by the Borrower (or such earlier date as shall be satisfactory to the Agent), as applicable, in form and substance satisfactory to the Agent and (ii) as applicable, each of the following statements shall be true and correct as of such date (each such date on which all such conditions precedent have been satisfied or waived in writing by the Agent is called a "Delayed Draw Date" and, collectively, all such dates shall be the "Delayed Draw Dates"):
  - (a) <u>Acquisition Agreements</u>. The Borrower shall have received the consent of the Requisite Lenders to the Acquisition to be funded by a Delayed Draw Term Loan on the Delayed Draw Date and the Lenders shall have received the acquisition agreement and all other material agreements, instruments and documents executed in connection with any such Acquisition to be consummated on such Delayed Draw Date (including without limitation all schedules and exhibits to the relevant acquisition agreement) in each case in form and substance satisfactory to them. Concurrently with the funding of the Term Loans on such Delayed Draw Date, any such Acquisition shall have been consummated in accordance with the terms of such acquisition agreement in all material respects and in compliance with applicable law and regulatory approvals.

- (b) <u>Covenant Compliance</u>. After giving pro forma effect to the Acquisition to be funded on and the incurrence of the Delayed Draw Term Loan on such date, Borrower is in compliance with each of the covenants set forth in <u>Section 7.1</u>.
- (c) <u>Letter of Direction</u>. Agent shall have received a duly executed letter of direction from Borrower addressed to Agent, on behalf of itself and the Lenders directing the disbursement of the proceeds of the Delayed Draw Term Loans on the applicable Delayed Draw Date.
- (d) Opinions of Counsel. Opinions of counsel for each Loan Party, including local counsel reasonably requested by the Agent.
- (e) <u>Insurance</u>. Evidence of the existence of insurance required to be maintained pursuant to the Credit Agreement, together with evidence that the Agent has been named as a lender's loss payee and an additional insured on all related insurance policies.
- (f) <u>Closing Certificate</u>. A certificate executed by an officer of the Borrower on behalf of the Loan Parties certifying the matters set forth as conditions to the Delayed Draw Date.
- (g) Consents, etc. Certified copies of all documents evidencing any necessary corporate or partnership action, consents and governmental approvals (if any) required for the execution, delivery and performance by the Loan Parties of the Acquisition to be funded on the Delayed Draw Date.
- (h) <u>Filings, Registrations and Recordings</u>. The Agent shall have received each document (including Uniform Commercial Code financing statements) required by Section 5.7(c) of the Credit Agreement as may be necessary or desirable in order to create in favor of the Agent, for the benefit of the Lenders, a perfected Lien on the Collateral acquired in connection with such any such Acquisition described therein, prior to any other Liens (subject only to Permitted Encumbrances).
- (i) Other. Such other documents as the Agent or any Lender may reasonably request.
- (v) Borrower shall repay the Term Loans through periodic payments on the dates and in the amounts indicated below ("Scheduled Installments"), and in any event the entire remaining principal balance shall be repaid on the Commitment Termination Date.

#### Term Loan

<u>Date</u>	Scheduled Installment
December 31, 2007	\$ 1,100,000
March 31, 2008	\$ 1,100,000
June 30, 2008	\$ 1,100,000
September 30, 2008	\$ 1,100,000
December 31, 2008	\$ 1,475,000
March 31, 2009	\$ 1,475,000
June 30, 2009	\$ 1,475,000
September 30, 2009	\$ 1,475,000
December 31, 2009	\$ 1,850,000
March 31, 2010	\$ 1,850,000
June 30, 2010	\$ 1,850,000
September 30, 2010	\$ 1,850,000
December 31, 2010	\$ 2,225,000
March 31, 2011	\$ 2,225,000
June 30, 2011	\$ 2,225,000
September 19, 2011	\$ 22,825,000 or the outstanding principal balance of Term Loans outstanding on such date

The above scheduled installment amounts shall be increased in the manner set forth on Annex B to the Second Amendment to the extent any Delayed Draw Term Loans are funded. The final installment payment shall in all events equal the entire remaining principal balance of the Term Loan (including any remaining principal balance of such Delayed Draw Term Loans). Amounts borrowed under this Section 2.1(a) and repaid may not be reborrowed.

At the request of the applicable Lender, the Term Loans shall be evidenced by promissory notes substantially in the form of Exhibit 2.1(a) (as amended, modified, extended, substituted or replaced from time to time, each a "Term Note" and, collectively, the "Term Notes"), and Borrower shall execute and deliver a Term Note to each such Term Lender. Each Term Note shall represent the obligation of Borrower to pay the amount of the applicable Term Lender's Term Loan Commitment, together with interest thereon."

3.8. Section 2.2(a) of the Credit Agreement is hereby amended by deleting the grid therein and replacing it with the following new grid:

Level of Applicable Margin	Leverage Ratio	Applicable Term Loan Index Margin	Applicable Term Loan LIBOR Margin
Level I	$\geq$ 4.00 to 1.00	3.50%	4.50%
Level II	$\geq$ 2.50 to 1.00, and $\leq$ 4.00 to 1.00	3.00%	4.00%
Level III	< 2.50 to 1.00	2.50%	3.50%

- 3.9. Section 2.3(a) of the Credit Agreement is hereby deleted and replaced with the following new Section 2.3(a):
- "Fee Letter. Borrower shall pay to Freeport, individually, the Fees specified in that certain fee letter dated as of October 15, 2007 (the "Freeport Fee Letter"), at the times specified for payment therein."
  - 3.10. A new Section 2.3(f) is hereby added to the Credit Agreement to read as follows:
- "Unused Delayed Draw Line Fee. As additional compensation for each Term Lender with a Delayed Draw Term Loan Commitment, Borrower shall pay to Agent, for the benefit of each such Lender, in arrears, on the first Business Day of each month prior to the Delayed Draw Term Loan Commitment Termination Date and on such date, a fee for each such month or other period for Borrower's non-use of available funds under the Delayed Draw Term Loan Commitment in an amount equal to one half of one percent (0.50%) per annum multiplied by the unused amount of such Lender's Delayed Draw Term Loan Commitment during such month or other period."
  - 3.11. Section 5.12 of the Credit Agreement is hereby amended by adding the following new paragraph (c) thereto:
- "(c) Following the Second Amendment Effective Date, the Borrower shall use its commercially reasonable efforts to promptly complete the transfer of the healthcare provider numbers from SuCasa Personal Care, LLC and Desert PCA of Nevada, LLC (each, a "New Provider Number") to one of the Loan Parties and upon receipt of a New Provider Number shall notify the Agent of receipt of same and if permitted by applicable law, (i) shall transfer all patients who were previously patients of Silver State (the "Silver State Patients") to such New Provider Number within sixty (60) days of receipt of such New Provider Number and (ii) shall retire the healthcare provider number acquired in connection with the Silver State Acquisition within such sixty (60) day period and cease to use it in the operations of the Borrower and its Subsidiaries, unless as otherwise required by applicable law. Notwithstanding the foregoing, in the event that Borrower determines that it is not permitted by applicable law to effect the transfer of the Silver State Patients to a New Provider Number, Borrower and its Subsidiaries shall promptly notify Agent and shall immediately apply for a new healthcare provider number from the appropriate Governmental Authority and shall transfer the Silver State Patients to such new healthcare provider number within sixty (60) days of receipt of such new healthcare provider number."
- 3.12. Section 6.1 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause (I) thereof, deleting the "." at the end of clause (m) thereof and replacing it with "; and" and by adding the following new clause (n):
  - "(n) Indebtedness in respect of the Silver State Note."

- 3.13. Section 6.3 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause (1) thereof, deleting the "." at the end of clause (m) thereof and replacing it with "; and" and by adding the following new clause (n):
- "(n) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Silver State Acquisition as of the Second Amendment Effective Date pursuant to the Silver State Acquisition Documents."
- 3.14. <u>Clause (d)</u> of <u>Section 6.6</u> of the Credit Agreement is hereby amended by deleting the "and" before clause (ii) thereof and adding the following new language immediately at the end thereof:
- "and (iii) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Silver State Acquisition on the Second Amendment Effective Date pursuant to the Silver State Acquisition Documents and all applicable law, provided that the Silver State Acquisition Documents shall be in full force and effect and provided further that the Agent shall have received a complete copy of the fully executed Silver State Acquisition Documents, certified by the Borrower as being true, complete and correct."
  - 3.15. Section 6.16 of the Credit Agreement is amended and restated in its entirety to read as follows:
- "None of the Loan Parties shall change or amend the terms of the Acquisition Note, Silver State Note or SuCasa Note if such change or amendment would be adverse in any material respect to the rights or interests of the Loan Parties, the Agent or the Lenders."
  - 3.16. A new Section 6.17 is hereby added to the Credit Agreement to read as follows:
- "6.17 <u>Payments of Deferred Purchase Price</u>. None of the Loan Parties shall make any payment of deferred purchase price (including with respect to any earnout or other Indebtedness) with respect to any Acquisition entered into on or after the Second Amendment Effective Date if, after giving effect to such payment, the Borrower would have less than \$2,500,000 of Borrowing Availability."
  - 3.17. Section 7.1(c) of the Credit Agreement is amended and restated in its entirety to read as follows:
- "Minimum Fixed Charge Coverage Ratio. Holdings and its Subsidiaries shall have on a consolidated basis at the end of each Fiscal Quarter set forth below, a Fixed Charge Coverage Ratio for the 12-Fiscal Month period then ended, of not less than the following:
  - 1.20:1:00 for the Fiscal Quarter ending September 30, 2007
  - 1.20:1:00 for the Fiscal Quarter ending December 31, 2007
  - 1.15:1:00 for the Fiscal Quarter ending March 31, 2008
  - 1.15:1:00 for the Fiscal Quarter ending June 30, 2008
  - 1.10:1:00 for the Fiscal Quarter ending September 30, 2008

- 1.10:1:00 for the Fiscal Quarter ending December 31, 2008
- 1.10:1:00 for the Fiscal Quarter ending March 31, 2009
- 1.10:1:00 for the Fiscal Quarter ending June 30, 2009
- 1.10:1:00 for each Fiscal Quarter ending thereafter
- 3.18. Section 7.1(d) of the Credit Agreement is amended and restated in its entirety to read as follows:
- "Maximum Leverage Ratio. Holdings and its Subsidiaries on a consolidated basis shall have, at the end of each Fiscal Quarter set forth below, a Leverage Ratio as of the last day of such Fiscal Quarter and for the 12-Fiscal Month period then ended, of not more than the following:
  - 4.25:1.00 for the Fiscal Quarter ending September 30, 2007;
  - 4.25:1.00 for the Fiscal Quarter ending December 31, 2007;
  - 4.25:1.00 for the Fiscal Quarter ending March 31, 2008;
  - 4.25:1.00 for the Fiscal Quarter ending June 30, 2008;
  - 4.00:1.00 for the Fiscal Quarter ending September 30, 2008;
  - 4.00:1.00 for the Fiscal Quarter ending December 31, 2008;
  - 3.75:1.00 for the Fiscal Quarter ending March 31, 2009;
  - 3.75:1.00 for the Fiscal Quarter ending June 30, 2009;
  - 3.50:1.00 for the Fiscal Quarter ending September 30, 2009;
  - 3.50:1.00 for the Fiscal Quarter ending December 31, 2009; 3.25:1.00 for the Fiscal Quarter ending March 31, 2010;
  - 3.25:1.00 for the Fiscal Quarter ending June 30, 2010;
  - 3.00:1.00 for the Fiscal Quarter ending September 30, 2010;
  - 3.00:1.00 for the Fiscal Quarter ending December 31, 2010; and
  - 2.75 for each Fiscal Quarter ending thereafter."
  - 3.19. Annex A of the Credit Agreement is amended and restated in its entirety to read in the manner set forth as Annex C to this Amendment.
- Section 4. Representations and Warranties. To induce Agent and Lenders to execute this Amendment, each Loan Party hereby represents and warrants to Agent and Lenders as follows:
  - (a) the execution, delivery and performance of each of this Amendment and the Silver State Acquisition Documents have been duly authorized by all requisite action of each Loan Party thereto, and each of this Amendment and the Silver State Acquisition Documents constitutes the legal, valid and binding obligation of such Person, enforceable against such Person in accordance with its terms, subject to bankruptcy, insolvency and similar laws affecting the enforceability of creditors' rights generally and to general principles of equity; and
  - (b) each of the representations and warranties in the Credit Agreement are true and correct in all material respects with the same effect as though made on and as of the date hereof (except, in each case, to the extent stated to relate to an earlier date, in

which case such representation or warranty shall have been true and correct on and as of such earlier date) and no Event of Default or Default exists thereunder or would exist after giving effect to this Amendment or the Silver State Acquisition.

Section 5. <u>Conditions Precedent</u>. This Amendment shall be effective as of the date first set forth above, subject to the satisfaction of the following conditions precedent (the date of such satisfaction being the "<u>Second Amendment Effective Date</u>"):

- 5.1. Execution and Delivery. Borrower, each of the other Loan Parties, Agent, each Lender with a Second Amendment Term Loan Commitment, each Lender with a Delayed Draw Term Loan Commitment, each Lender whose Revolving Loan Commitment is being increased as of the Second Amendment Effective Date and the Requisite Lenders shall have executed and delivered this Agreement; the Borrower shall have executed and delivered an Officers Certificate in the form of Exhibit A attached hereto; and each Loan Party shall have executed and delivered the Consent and Reaffirmation in the form of Exhibit B attached hereto.
- 5.2. <u>Financial Status</u>. As of the Second Amendment Effective Date after giving effect to the incurrence on the Second Amendment Effective Date to the payment of all related costs and expenses incurred in connection with the Silver State Acquisition and this Amendment, (a) the Leverage Ratio shall not exceed 3.90 to 1.00, (b) not more than \$6,500,000 in the aggregate of Revolving Loans and Letter of Credit Obligations shall be outstanding, and (c) Borrowing Availability shall be at least \$6,000,000.
- 5.3. <u>Due Diligence</u>. Agent, Lenders and their counsel and advisors shall have completed their legal, accounting and financial due diligence review (including, without limitation, lien searches on Silver State) of Silver State and the Silver State Acquisition, the results of which review shall be reasonably satisfactory to Agent and Requisite Lenders.
- 5.4. <u>Material Adverse Effect</u>. There shall have occurred no Material Adverse Effect upon the financial condition, operations, assets, business or properties of the Loan Parties or Silver State since December 31, 2006.
- 5.5. Silver State Acquisition Documents. On or prior to the Second Amendment Effective Date, each of the Silver State Acquisition Documents shall have been duly authorized by each of the Loan Parties party thereto and executed and delivered by each of the respective parties thereto and shall be in full force and effect and shall not have been amended or modified in any material manner except for such material amendments and modifications, if any, as may be reasonably satisfactory to Agent and the Requisite Lenders. The Silver State Acquisition shall have been, or shall be contemporaneously herewith, consummated substantially in accordance with the Silver State Acquisition Documents and in accordance with all applicable law, except as may have been consented to in writing by Agent. Agent shall have received a complete copy of the fully executed Silver State Acquisition Documents, certified by the Borrower as being true, complete and correct.
- 5.6. Approvals. All necessary governmental and third party approvals and/or consents in connection with the Silver State Acquisition and otherwise referred to herein or therein shall

have been obtained and remain in effect, and all applicable waiting periods shall have expired without any action being taken by any competent authority which restrains, prevents or imposes materially adverse conditions upon the consummation of all or any part of the Silver State Acquisition, in each case as of the Second Amendment Effective Date. Additionally, there shall not exist any judgment, order, injunction or other restraint pending or notified prohibiting or imposing material adverse conditions upon all or any part of the Silver State Acquisition or the transactions contemplated by this Amendment as of the Second Amendment Effective Date.

- 5.7. No Defaults. After giving effect to this Amendment, no Event of Default or Default under the Credit Agreement shall have occurred and be continuing.
- 5.8. Representations and Warranties. After giving effect to this Amendment, the representations and warranties of the Loan Parties contained in this Amendment, the Credit Agreement and the other Loan Documents shall be true and correct in all material respects as of the Second Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty shall be true and correct in all material respects as of such earlier date.
- 5.9. Payment of Fees. The Borrower shall have paid (i) to Agent for its own account all fees payable on the Second Amendment Effective Date and specified by that certain fee letter between the Borrower and Agent as of the date hereof and (ii) shall have paid to Agent on behalf of each Lender who executes this Amendment on or prior to October 15, 2007 an amendment fee in an amount equal to 25 bps times the sum of such Lender's Revolving Commitment as of the Second Amendment Effective Date and the principal amount of such Lender's Term Loans outstanding on the Second Amendment Effective Date, in each case prior to giving effect to the Second Amendment.
- 5.10. Secretary's Certificate. Agent shall have received from each Loan Party a certificate signed by the secretary or an assistant secretary of such Loan Party, dated as of the Second Amendment Effective Date, as to the incumbency and signature of the officers of such Loan Party executing any Loan Document and any certificate or other document or instrument to be delivered pursuant hereto by or on behalf of such Loan Party, together with evidence of the incumbency of such Secretary or Assistant Secretary, as the case may be and certifying that there have been no changes to (i) the resolutions of such Loan Party authorizing and approving, among other things, the execution and delivery of the Loan Documents and (ii) the organizational documents of such Loan Party in each case since the Closing Date.
- 5.11. <u>Resolutions</u>. Agent shall have received a copy of the resolutions (in form and substance reasonably satisfactory to Agent) adopted by the Board of Directors of each Loan Party, authorizing or relating to the execution, delivery and performance of this Amendment and the Silver State Acquisition Documents and the other documents and instruments provided for therein and the consummation of the transactions contemplated hereby and thereby.
- 5.12. Good Standing Certificates. Agent shall have received good standing certificates for each Loan Party from their respective jurisdictions of incorporation or organization.

- 5.13. <u>Fees</u>. Borrower shall have paid all reasonable and documented costs, fees and expenses (including, without limitation, reasonable legal fees and expenses of Winston & Strawn LLP) of Agent.
  - 5.14. Opinions of Counsel. Agent shall have received opinions of counsel for each Loan Party, including local counsel reasonably requested by the Agent.
- 5.15. Other Matters. Agent shall have received such other instruments and documents as Agent or the Requisite Lenders may reasonably request in connection with the execution of this Amendment, and all such instruments and documents shall be reasonably satisfactory in form and substance to Agent.

#### Section 6. Miscellaneous.

- 6.1. Effect of Amendment. The execution, delivery and effectiveness of this Amendment shall not operate as a waiver of any right, power or remedy of Agent or any Lender under the Credit Agreement or any Loan Document, or constitute a waiver of any provision of the Credit Agreement or any Loan Document, except as specifically set forth herein.
- 6.2. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and by the different parties on separate counterparts, and each such counterpart shall be deemed to be an original, but all such counterparts shall together constitute but one and the same instrument. Delivery of the executed counterpart of this Amendment by telecopy or electronic mail shall be as effective as delivery of a manually executed counterpart to this Amendment.
- 6.3. <u>Severability</u>. The illegality or unenforceability of any provision of this Amendment or any instrument or agreement required hereunder shall not in any way affect or impair the legality or enforceability of the remaining provisions of this Amendment or any instrument or agreement required hereunder.
  - 6.4. Captions. Section captions used in this Amendment are for convenience only, and shall not affect the construction of this Amendment.
- 6.5. Entire Agreement. This Amendment embodies the entire agreement and understanding among the parties hereto and supersedes all prior or contemporaneous agreements and understandings of such Persons, verbal or written, relating to the subject matter hereof.
- 6.6. <u>References</u>. Any reference to the Credit Agreement contained in any notice, request, certificate, or other document executed concurrently with or after the execution and delivery of this Amendment shall be deemed to include this Amendment unless the context shall otherwise require. Reference in any of this Amendment, the Credit Agreement or any other Loan Document to the Credit Agreement shall be a reference to the Credit Agreement as amended hereby and as further amended, modified, restated, supplemented or extended from time to time.

Section 7. <u>Governing Law</u>. THIS AMENDMENT SHALL BE A CONTRACT MADE UNDER AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF ILLINOIS APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE, WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES.

[signature pages follow]

Witness the due execution hereof by the respective duly authorized officers of the undersigned as of the date first above written.

#### ADDUS HEALTHCARE, INC.

 By:
 /s/ W. Andrew Wright

 Name:
 W. Andrew Wright

 Title:
 President & CEO

#### ADDUS HOLDING CORPORATION

 By:
 /s/ W. Andrew Wright

 Name:
 W. Andrew Wright

 Title:
 President & CEO

#### ADDUS MANAGEMENT CORPORATION

 By:
 /s/ W. Andrew Wright

 Name:
 W. Andrew Wright

 Title:
 President & CEO

# LOWELL HOME HEALTH AGENCY, INC.

By: /s/ W. Andrew Wright

Name: W. Andrew Wright

Title: President & CEO

#### LITTLE ROCK HOME HEALTH AGENCY, INC.

By: /s/ W. Andrew Wright Name: W. Andrew Wright

Title: President & CEO

#### FORT SMITH HOME HEALTH AGENCY, INC.

By: /s/ W. Andrew Wright

Name: W. Andrew Wright Title: President & CEO

#### BENEFITS ASSURANCE CO., INC.

/s/ W. Andrew Wright

Name: W. Andrew Wright

By:

By:

President & CEO Title:

#### PHC ACQUISITION CORPORATION

/s/ W. Andrew Wright

Name: W. Andrew Wright

Title: President & CEO

#### PROFESSIONAL RELIABLE NURSING SERVICE, INC.

/s/ W. Andrew Wright

By: Name: W. Andrew Wright

Title: President & CEO

#### FREEPORT FINANCIAL LLC, as Agent

By: /s/ Chad Blakeman
Title: Sr. Managing Director

# FREEPORT LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman
Title: Sr. Managing Director

# FREEPORT OFFSHORE LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman
Title: Sr. Managing Director

# FREEPORT LOAN TRUST 2006-1, as a Lender

By: /s/ Chad Blakeman
Title: Sr. Managing Director

# FIFTH THIRD BANK, CHICAGO (A Michigan Banking Company)

By: /s/ Michael E. May
Name: Michael E. May
Title: Vice President

#### CF BLACKBURN LLC

By: /s/ Dennis M. Hansen

Name: Dennis M. Hansen

Title: Vice President

#### OFFICER'S CERTIFICATE

Reference is made to that certain Consent and Second Amendment to Credit Agreement, dated as of October 15, 2007 (the "Second Amendment"; capitalized terms used herein and not otherwise defined shall have the meanings specified thereto in that certain Credit Agreement dated as of September 19, 2006 among the Loan Parties party thereto, the Agent and the Lenders party thereto (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007 and as further amended, restated or otherwise modified from time to time including by the Second Amendment, the "Credit Agreement"), among the Borrower, certain other parties, the financial institutions party thereto and Freeport Financial LLC, as Agent. The undersigned hereby certifies that [s]he is the duly authorized and elected \_\_\_\_\_\_\_ of the Borrower and that, as of the date hereof:

- As of the Second Amendment Effective Date, after giving effect to the payment of all related costs and expenses incurred in connection with the Silver State Acquisition and the Second Amendment, (a) the Leverage Ratio does not exceed 3.90 to 1.00, (b) not more than \$6,500,000 in the aggregate of Revolving Loans and Letter of Credit Obligations are outstanding and (c) Borrowing Availability is at least \$6,000,000.
- 2. No Material Adverse Effect has occurred upon the financial condition, operations, assets, business or properties of the Loan Parties or Silver State since December 31, 2006.
- 3. After giving effect to the Second Amendment, no Event of Default or Default under the Credit Agreement has occurred or is continuing.
- 4. After giving effect to the Second Amendment, the representations and warranties of the Loan Parties contained in the Second Amendment, the Credit Agreement and the other Loan Documents are true and correct in all material respects as of the Second Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty was true and correct in all material respects as of such earlier date.
- 5. The Loan Parties have consummated the Silver State Acquisition concurrently with the Second Amendment substantially in accordance with the Silver State Acquisition Documents and in accordance with all applicable law, except as may have been consented to in writing by Agent.

6.	Attached hereto are true, complete and correct copies of the fully executed Silver State Acquisition Documents.			
	[Remainder of Page Intentionally Blank]			

IN WITNESS WHEREOF, I have hereunto set my hand this	day of October, 2007.	
		ADDUS HEALTHCARE, INC.
		By: Name:
		Name: Title:

#### **EXHIBIT B**

# **CONSENT AND REAFFIRMATION**

Each of the undersigned ("Loan Parties") hereby (i) acknowledges receipt of a copy of the Consent and Second Amendment to Credit Agreement dated as of October 15, 2007 (the "Second Amendment"); (ii) consents to the execution and delivery thereof by the other Loan Parties; (iii) agrees to be bound thereby; (iv) affirms that nothing contained therein shall modify in any respect whatsoever its guaranty of the obligations of the Loan Parties to Agent and Lenders (the "Guarantee") and (v) reaffirms that the Guarantee and the other Collateral Documents executed by such Person are and shall continue to remain in full force and effect. Although each of the Loan Parties has been informed of the matters set forth herein and in the Second Amendment and has acknowledged and agreed to same, such Loan Parties understand that Agent and Lenders have no obligation to inform any of the Loan Parties of such matters in the future or to seek any of the Loan Parties' acknowledgment or agreement to future amendments or waivers, and nothing herein shall create such a duty.

This Consent and Reaffirmation shall be governed by and construed in accordance with the laws of the State of Illinois, without reference to principles of conflicts of law.

In witness whereof	f each of the und	lersioned has exe	cuted this Conse	ent and Reaffirm:	ation on and as	of the date of s	such Second Amendmer	nt

# ADDUS HEALTHCARE, INC.

Name:	
Title:	
11110.	
ADDUS	S HOLDING CORPORATION
By:	
Name:	
Title:	
ADDUS	S HEALTHCARE (NEVADA), INC.
By:	
Name:	
Title:	
ADDUS	S MANAGEMENT CORPORATION
By:	
Name:	
Title:	
LOWE	LL HOME HEALTH AGENCY, INC.
Bv:	
By: Name:	
Name:	
Name: Title:	
Name: Title:	E ROCK HOME HEALTH AGENCY, INC.
Name: Title: LITTL By:	E ROCK HOME HEALTH AGENCY, INC.
Name: Title: LITTL By: Name:	E ROCK HOME HEALTH AGENCY, INC.
Name: Title: LITTL By:	E ROCK HOME HEALTH AGENCY, INC.

By:	
Name:	-
Title:	
BENE	FITS ASSURANCE CO., INC.
By:	
Name:	
Title:	
PHC A	CQUISITION CORPORATION
	equipment com outmon
By:	
Name: Title:	
Name: Title:	ESSIONAL RELIABLE NURSING SERVICE, INC
Name: Title: PROFI By:	ESSIONAL RELIABLE NURSING SERVICE, INC
Name: Title: PROF	ESSIONAL RELIABLE NURSING SERVICE, INC
Name: Title: PROFI By: Name: Title:	ESSIONAL RELIABLE NURSING SERVICE, INC
Name: Title: PROFI By: Name: Title:	ESSIONAL RELIABLE NURSING SERVICE, INC

# PRO RATA SHARES AND COMMITMENT AMOUNTS

	SECOND AMENDMENT TERM LOAN COMMITMENT		DELAYED DRAW TERM LOAN COMMITMENT	
	\$	%	\$	%
Lender				
Freeport Loan Fund LLC	1,714,285.71	34.2857143	4,285,714.29	34.2857143
Residential Funding Corporation	-0-	-0-	-0-	-0-
Fifth Third Bank (Chicago)	3,285,714.29	65.7142857	8,214,285.71	65.7142857
Totals	\$ 5,000,000	100%	\$ 12,500,000	100%

The scheduled installments set forth in Section 2.1(a) of the Credit Agreement shall be increased by the following amounts for each \$500,000 of Delayed Draw Term Loans funded after the Second Amendment Effective Date:

#### Term Loan

Date	Scheduled Installment
December 31, 2007	\$5,000
March 31, 2008	\$5,000
June 30, 2008	\$5,000
September 30, 2008	\$5,000
December 31, 2008	\$7,500
March 31, 2009	\$7,500
June 30, 2009	\$7,500
September 30, 2009	\$7,500
December 31, 2009	\$10,000
March 31, 2010	\$10,000
June 30, 2010	\$10,000
September 30, 2010	\$10,000
December 31, 2010	\$12,500
March 31, 2011	\$12,500
June 30, 2011	\$12,500
September 19, 2011	\$372,500 or
	remaining principal
	balance

# ANNEX A to CREDIT AGREEMENT

# PRO RATA SHARES AND COMMITMENT AMOUNTS

	REVOLVER		ORIGINAL TE	RM LOAN
	\$	%	\$	%
Lender				
Freeport Loan Fund LLC	5,978,260.87	34.1614907	21,521,739.14	47.8260870
Residential Funding Corporation	3,260,869.56	18.6335404	11,739,130.43	26.0869565
Fifth Third Bank (Chicago)	8,260,869.57	47.2049689	11,739,130.43	26.0869565
Totals	\$ 17,500,000.00	100.00%	\$ 45,000,000.00	100.00%

#### CONSENT AND THIRD AMENDMENT TO CREDIT AGREEMENT

CONSENT AND THIRD AMENDMENT TO CREDIT AGREEMENT (this "Amendment"), dated as of November 13, 2007, among Addus HealthCare, Inc., an Illinois corporation ("Borrower"), the other persons signatory hereto as "Loan Parties", FREEPORT FINANCIAL LLC, a Delaware limited liability company ("Agent") and the Lenders signatory hereto. Terms not defined herein have the meanings given to them in the Credit Agreement (as hereinafter defined).

#### RECITALS

- A. Borrower, the Loan Parties, the Lenders signatory thereto and Agent are party to that certain Credit Agreement dated as of September 19, 2006 (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007, that certain Consent and Second Amendment to Credit Agreement dated as of October 15, 2007 and as further amended, restated or otherwise modified including by this Amendment, the "<u>Credit Agreement</u>").
- B. Borrower, the Loan Parties, the Lenders and Agent are entering into this Amendment in connection with (I) the acquisition (the "Vegas Valley Acquisition") by Addus HealthCare (Nevada), Inc., a Delaware corporation and a wholly-owned Subsidiary of the Borrower ("Addus Nevada"), of substantially all of the assets of Vegas Valley Personal Care, LLC, a Nevada limited liability company ("Vegas Valley") for an aggregate purchase price not to exceed \$1,550,000 and (II) the acquisition (the "Greater Vegas Acquisition" and, together with the Vegas Valley Acquisition, the "Vegas Acquisitions")) by Addus Nevada of substantially all of the assets of Greater Vegas Personal Care, LLC, a Nevada limited liability company ("Greater Vegas") for an aggregate purchase price not to exceed \$1,550,000.
  - C. Agent, Lenders and the Loan Parties are willing to enter into this Amendment upon the terms and conditions set forth below.

NOW THEREFORE, in consideration of the matters set forth in the recitals and the covenants and provisions herein set forth, and other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

#### **AGREEMENT**

- Section 1. <u>Definitions</u>. Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed thereto in the Credit Agreement.
- Section 2. <u>Consent</u>. Agent and the Lenders hereby consent to Addus Nevada (i) entering into (v) that certain Asset Purchase Agreement, dated as of even date herewith (the "<u>Greater Vegas Purchase Agreement</u>"), by and between Addus Nevada, Greater Vegas, Fabiana Alfaro and Ema Gomez Mitchell, (w) that certain Asset Purchase Agreement, dated as of even date herewith (the "<u>Vegas Valley Purchase Agreement</u>"), by and between Addus Nevada, Vegas

Valley and Ema Gomez Mitchell, (x) that certain Consulting Agreement, dated as of even date herewith (the "Vegas Consulting Agreement"), by and between Addus Nevada and Ema Gomez Mitchell, (y) the Vegas Earn Out Agreement and (z) the other agreements, instruments and documents for the purpose of consummating the Vegas Acquisitions in each case on terms and conditions satisfactory to the Agent (the "Vegas Acquisition Documents", it being agreed that the forms of the Greater Vegas Purchase Agreement, the Vegas Valley Purchase Agreement, the Vegas Consulting Agreement, the Vegas Earn Out Agreement and such other agreements, instruments and documents provided to the Agent on November 12, 2007 are satisfactory to the Agent); (ii) consummating the Vegas Valley Acquisition on the Third Amendment Effective Date for an aggregate purchase price not to exceed \$950,000 paid in cash on the closing date thereof pursuant to the terms of the Vegas Acquisition Documents, plus a deferred purchase price in an amount not to exceed \$950,000 paid in cash on the closing date thereof pursuant to the terms of the Vegas Acquisition Documents, plus a deferred purchase price in an amount not to exceed \$950,000 paid in cash on the closing date thereof pursuant to the terms of the Vegas Acquisition Documents, plus a deferred purchase price in an amount not to exceed \$950,000 paid to Greater Vegas pursuant to the Vegas Earn Out Agreement; provided that in each case such deferred purchase price in an amount not to exceed \$600,000 paid to Greater Vegas pursuant to the Vegas Earn Out Agreement; provided that in each case such deferred purchase price in an amount not be paid in the event that any Default or Event of Default under any of Sections 7.1 or 8.1(a) of the Credit Agreement has occurred and is continuing or would result thereform or the Borrower does not have the minimum Borrowing Availability required by Section 6.17 of the Credit Agreement after giving effect to such payment.

Section 3. Amendment to the Credit Agreement. As of the Third Amendment Effective Date, the Credit Agreement is hereby amended as follows:

3.1. Section 1.1 of the Credit Agreement is hereby amended by adding the following new definitions thereto:

Greater Vegas has the meaning set forth in the Recitals to the Third Amendment.

Greater Vegas Acquisition has the meaning set forth in the Recitals to the Third Amendment.

Medicaid Investigation means the matters which were the subject of the investigation pursuant to (i) the Search Warrant, dated as of April 10, 2007, issued by the State of Nevada to any Peace Officer in the County of Clark related to Vegas Valley and (ii) the Search Warrant, dated as of April 10, 2007, issued by the State of Nevada to any Peace Officer in the County of Clark related to the Greater Vegas.

Third Amendment means that certain Consent and Third Amendment to Credit Agreement dated as of November 13, 2007 among the Loan Parties, Agent and the Lenders.

Third Amendment Effective Date has the meaning set forth in the Third Amendment.

Third Amendment Reaffirmation of Collateral Documents means the Consent and Reaffirmation (Third Amendment) dated as of November 13, 2007 of the Loan Parties signatory thereto, in respect of the Collateral Documents.

Vegas Acquisition Documents has the meaning set forth in Section 2 of the Third Amendment.

Vegas Acquisitions has the meaning set forth in the Recitals to the Third Amendment.

<u>Vegas Earn Out Agreement</u> means that certain Earn Out Agreement dated as of November 13, 2007 by and between Addus HealthCare (Nevada), Inc., Fabiana Alfaro, Ema Gomez Mitchell, Vegas Valley and Greater Vegas.

<u>Vegas Valley</u> has the meaning set forth in the Recitals to the Third Amendment.

Vegas Valley Acquisition has the meaning set forth in the Recitals to the Third Amendment.

3.2. Section 1.1 of the Credit Agreement is further amended by amending and restating the proviso at the end of the definition of "EBITDA" to read as follows:

"provided, that, notwithstanding anything to the contrary contained herein, for each of the Fiscal Quarters listed below, EBITDA shall be deemed to be the amount set forth below opposite such corresponding period:

Period	EBITDA
December 31, 2006	\$4,331,076
March 31, 2007	\$3,574,099
June 30, 2007	\$3,690,092

Provided, further, that (i) for any period that includes the month of July of 2007, EBITDA shall, without duplication, be increased with respect to the assets acquired in connection with the SuCasa Acquisition, the Silver State Acquisition and the Vegas Acquisitions by an amount equal to \$246,500 for such month, (ii) for any period that includes the month of August or September of 2007, EBITDA shall, without duplication, be increased with respect to the assets acquired in connection with the Silver State Acquisition and the Vegas Acquisitions by an amount equal to \$180,000 for each such month and (iii) for any period that includes the month of October of 2007 to be increased with respect to the assets acquired in the Vegas Acquisitions by an amount equal to \$117,000 for such month."

- 3.3. Section 2.1(a)(y) of the Credit Agreement is hereby amended and restated in its entirety and replaced with the following new Section 2.1(a)(y):
- (v) Borrower shall repay the Term Loans through periodic payments on the dates and in the amounts indicated below ("Scheduled Installments"), and in any event the entire remaining principal balance shall be repaid on the Commitment Termination Date.

#### Term Loan

Date	Sc	heduled Installment
December 31, 2007	\$	1,140,000
March 31, 2008	\$	1,140,000
June 30, 2008	\$	1,140,000
September 30, 2008	\$	1,140,000
December 31, 2008	\$	1,535,000
March 31, 2009	\$	1,535,000
June 30, 2009	\$	1,535,000
September 30, 2009	\$	1,535,000
December 31, 2009	\$	1,930,000
March 31, 2010	\$	1,930,000
June 30, 2010	\$	1,930,000
September 30, 2010	\$	1,930,000
December 31, 2010	\$	2,325,000
March 31, 2011	\$	2,325,000
June 30, 2011	\$	2,325,000
September 19, 2011	\$	25,805,000
	•	or the outstanding
		principal balance
		of Term Loans
		outstanding on
		such date

The above scheduled installment amounts reflect the incurrence by the Borrower of the Delayed Draw Term Loans funded on the Third Amendment Effective Date and shall thereafter be increased in the manner set forth on Annex B to the Second Amendment to the extent any further Delayed Draw Term Loans are funded. The final installment payment shall in all events equal the entire remaining principal balance of the Term Loan (including any remaining principal balance of such Delayed Draw Term Loans). Amounts borrowed under this Section 2.1(a) and repaid may not be reborrowed.

At the request of the applicable Lender, the Term Loans shall be evidenced by promissory notes substantially in the form of Exhibit 2.1(a) (as amended, modified, extended, substituted or replaced from time to time, each a "Term Note" and, collectively, the "Term Notes"), and Borrower shall execute and deliver a Term Note to each such Term Lender. Each Term Note shall represent the obligation of Borrower to pay the amount of the applicable Term Lender's Term Loan Commitment, together with interest thereon."

3.4. Section 5.12 of the Credit Agreement is hereby amended by adding the following new paragraph (d) thereto:

"(d) In connection with the businesses acquired in the Vegas Acquisitions, the Borrower and its Subsidiaries shall use the new healthcare provider number obtained by the Borrower

after the Second Amendment Effective Date and prior to the Third Amendment Effective Date. For purposes of clarification, neither Borrower nor any Subsidiary of the Borrower shall acquire or otherwise utilize any healthcare provider number used by Vegas Valley or Greater Vegas."

- 3.5. Section 6.1 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause (m) thereof, deleting the "." at the end of clause (n) thereof and replacing it with "; and" and by adding the following new clause (o):
  - "(o) Indebtedness in respect of the Vegas Earn Out Agreements."
- 3.6. Section 6.3 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause (m) thereof, deleting the "." at the end of clause (n) thereof and replacing it with "; and" and by adding the following new clause (o):
- "(o) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Vegas Valley Acquisition and the Greater Vegas Acquisition as of the Third Amendment Effective Date pursuant to the Vegas Acquisition Documents."
- 3.7. <u>Clause (d)</u> of <u>Section 6.6</u> of the Credit Agreement is hereby amended by deleting the "and" before clause (iii) thereof and adding the following new language immediately at the end thereof:
- "and (iv) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Vegas Valley Acquisition and the Greater Vegas Acquisition on the Third Amendment Effective Date pursuant to the Vegas Acquisition Documents and all applicable law, provided that the Vegas Acquisition Documents shall be in full force and effect and provided further that the Agent shall have received a complete copy of the fully executed Vegas Valley Acquisition Documents, certified by the Borrower as being true, complete and correct."
  - 3.8. Section 6.16 of the Credit Agreement is amended and restated in its entirety to read as follows:
- "None of the Loan Parties shall change or amend the terms of the Acquisition Note, Silver State Note, Vegas Earn Out Agreement, or SuCasa Note if such change or amendment would be adverse in any material respect to the rights or interests of the Loan Parties, the Agent or the Lenders."
  - 3.9. Section 6.17 of the Credit Agreement is hereby amended and restated in its entirety to read as follows:
- "6.17 <u>Payments of Deferred Purchase Price</u>. None of the Loan Parties shall make any payment of deferred purchase price (including with respect to any earnout or other Indebtedness) with respect to any Acquisition entered into on or after the Second Amendment Effective Date if, after giving effect to such payment, the Borrower would have less than \$3,000,000 of Borrowing Availability."

3.10. Section 7.1(b) of the Credit Agreement is amended and restated in its entirety to read as follows:

"Minimum EBITDA. Holdings and its Subsidiaries on a consolidated basis shall have, at the end of each Fiscal Quarter set forth below, Minimum EBITDA for the 12-Fiscal Month period then ended calculated of not less than the following:

Period	Mir	imum EBITDA
September 30, 2007	\$	12,500,000
December 31, 2007	\$	12,750,000
March 31, 2008	\$	13,250,000
June 30, 2008	\$	13,250,000
September 30, 2008	\$	13,500,000
December 31, 2008	\$	13,500,000
March 31, 2009	\$	14,000,000
June 30, 2009	\$	14,000,000
September 30, 2009	\$	14,500,000
December 31, 2009	\$	14,750,000
March 31, 2010	\$	14,750,000
June 30, 2010	\$	14,750,000
September 30, 2010	\$	15,000,000
December 31, 2010	\$	15,000,000
March 31, 2011 and each Fiscal Quarter ending thereafter	\$	15,250,000

Section 4. Representations and Warranties. To induce Agent and Lenders to execute this Amendment, each Loan Party hereby represents and warrants to Agent and Lenders as follows:

- (a) the execution, delivery and performance of each of this Amendment and the Vegas Acquisition Documents have been duly authorized by all requisite action of each Loan Party thereto, and each of this Amendment and the Vegas Acquisition Documents constitutes the legal, valid and binding obligation of such Person, enforceable against such Person in accordance with its terms, subject to bankruptcy, insolvency and similar laws affecting the enforceability of creditors' rights generally and to general principles of equity; and
- (b) each of the representations and warranties in the Credit Agreement are true and correct in all material respects with the same effect as though made on and as of the date hereof (except, in each case, to the extent stated to relate to an earlier date, in which case such representation or warranty shall have been true and correct on and as of such earlier date) and no Event of Default or Default exists thereunder or would exist after giving effect to this Amendment or the Vegas Acquisitions.

- Section 5. <u>Conditions Precedent</u>. This Amendment shall be effective as of the date first set forth above, subject to the satisfaction of the following conditions precedent (the date of such satisfaction being the "<u>Third Amendment Effective Date</u>"):
- 5.1. Execution and Delivery. Borrower, each of the other Loan Parties, Agent and the Requisite Lenders shall have executed and delivered this Agreement; the Borrower shall have executed and delivered an Officers Certificate in the form of Exhibit A attached hereto; each Loan Party shall have executed and delivered the Consent and Reaffirmation in the form of Exhibit B attached hereto.
- 5.2. <u>Covenant Compliance</u>. After giving pro forma effect to the Vegas Acquisitions to and the incurrence of the Delayed Draw Term Loans in connection therewith, Borrower is in compliance with each of the covenants set forth in <u>Section 7.1</u>.
- 5.3. <u>Letter of Direction</u>. Agent shall have received a duly executed letter of direction from Borrower addressed to Agent, on behalf of itself and the Lenders directing the disbursement of the proceeds of the Delayed Draw Term Loans to fund the Vegas Acquisitions.
- 5.4. <u>Insurance</u>. Agent shall have received evidence of the existence of insurance required to be maintained pursuant to the Credit Agreement, together with evidence that the Agent has been named as a lender's loss payee and an additional insured on all related insurance policies to the extent there is any change in such insurance in effect prior to the Third Amendment Effective Date as a result of or in connection with the Vegas Acquisitions.
- 5.5. <u>Closing Certificate</u>. Agent shall have received a certificate executed by an officer of the Borrower on behalf of the Loan Parties certifying the matters set forth herein as conditions.
- 5.6. <u>Filings, Registrations and Recordings</u>. The Agent shall have received each document (including Uniform Commercial Code financing statements) required by Section 5.7(c) of the Credit Agreement as may be necessary or desirable in order to create in favor of the Agent, for the benefit of the Lenders, a perfected Lien on the Collateral acquired in connection with the Vegas Acquisitions prior to any other Liens (subject only to Permitted Encumbrances).
- 5.7. <u>Due Diligence</u>. Agent, Lenders and their counsel and advisors shall have completed their legal, accounting and financial due diligence review (including, without limitation, lien searches on Vegas Valley and Greater Vegas) of Vegas Valley and Greater Vegas and the Vegas Acquisitions, the results of which review shall be reasonably satisfactory to Agent and Requisite Lenders.
- 5.8. <u>Material Adverse Effect</u>. There shall have occurred no Material Adverse Effect upon the financial condition, operations, assets, business or properties of (i) the Loan Parties, Greater Vegas or Vegas Valley, taken as a whole, since December 31, 2006; or (ii) Greater Vegas and Vegas Valley (in each case, other than the Medicaid Investigation) since December 31, 2006.

- 5.9. <u>Vegas Acquisition Documents</u>. On or prior to the Third Amendment Effective Date, each of the Vegas Acquisition Documents shall have been duly authorized by each of the Loan Parties party thereto and executed and delivered by each of the respective parties thereto and shall be in full force and effect and shall not have been amended or modified in any manner adverse to the Lenders except for such amendments and modifications, if any, as may be reasonably satisfactory to Agent and the Requisite Lenders. The Vegas Acquisitions shall have been, or shall be contemporaneously herewith, consummated substantially in accordance with the Vegas Acquisition Documents (as amended in accordance with the immediately preceding sentence) and in accordance with all applicable law, except as may have been consented to in writing by Agent. Agent shall have received a complete copy of the fully executed Vegas Acquisition Documents, certified by the Borrower as being true, complete and correct.
- 5.10. <u>Approvals</u>. All necessary governmental and third party approvals and/or consents in connection with the Vegas Acquisitions and otherwise referred to herein or therein shall have been obtained and remain in effect, and all applicable waiting periods shall have expired without any action being taken by any competent authority which restrains, prevents or imposes materially adverse conditions upon the consummation of all or any part of the Vegas Acquisitions, in each case as of the Third Amendment Effective Date. Additionally, there shall not exist any judgment, order, injunction or other restraint pending or notified prohibiting or imposing material adverse conditions upon all or any part of the Vegas Acquisitions or the transactions contemplated by this Amendment as of the Third Amendment Effective Date.
  - 5.11. No Defaults. After giving effect to this Amendment, no Event of Default or Default under the Credit Agreement shall have occurred and be continuing.
- 5.12. Representations and Warranties. After giving effect to this Amendment, the representations and warranties of the Loan Parties contained in this Amendment, the Credit Agreement and the other Loan Documents shall be true and correct in all material respects as of the Third Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty shall be true and correct in all material respects as of such earlier date.
- 5.13. Secretary's Certificate. Agent shall have received from each Loan Party a certificate signed by the secretary or an assistant secretary of such Loan Party, dated as of the Third Amendment Effective Date, as to the incumbency and signature of the officers of such Loan Party executing any Loan Document and any certificate or other document or instrument to be delivered pursuant hereto by or on behalf of such Loan Party, together with evidence of the incumbency of such Secretary or Assistant Secretary, as the case may be and certifying that there have been no changes to (i) the resolutions of such Loan Party authorizing and approving, among other things, the execution and delivery of the Loan Documents and (ii) the organizational documents of such Loan Party in each case since the Closing Date.

- 5.14. <u>Resolutions</u>. Agent shall have received a copy of the resolutions (in form and substance reasonably satisfactory to Agent) adopted by the Board of Directors of each Loan Party, authorizing or relating to the execution, delivery and performance of this Amendment and the Vegas Acquisition Documents and the other documents and instruments provided for therein and the consummation of the transactions contemplated hereby and thereby.
- 5.15. Good Standing Certificates. Agent shall have received good standing certificates for each Loan Party from their respective jurisdictions of incorporation or organization.
- 5.16. <u>Fees</u>. Borrower shall have paid all reasonable and documented costs, fees and expenses (including, without limitation, reasonable legal fees and expenses of Winston & Strawn LLP) of Agent.
  - 5.17. Opinions of Counsel. Agent shall have received opinions of counsel for each Loan Party, including local counsel reasonably requested by the Agent.
- 5.18. Other Matters. Agent shall have received such other instruments and documents as Agent or the Requisite Lenders may reasonably request in connection with the execution of this Amendment, and all such instruments and documents shall be reasonably satisfactory in form and substance to Agent.

#### Section 6. Miscellaneous.

- 6.1. Effect of Amendment. The execution, delivery and effectiveness of this Amendment shall not operate as a waiver of any right, power or remedy of Agent or any Lender under the Credit Agreement or any Loan Document, or constitute a waiver of any provision of the Credit Agreement or any Loan Document, except as specifically set forth herein.
- 6.2. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and by the different parties on separate counterparts, and each such counterpart shall be deemed to be an original, but all such counterparts shall together constitute but one and the same instrument. Delivery of the executed counterpart of this Amendment by telecopy or electronic mail shall be as effective as delivery of a manually executed counterpart to this Amendment.
- 6.3. <u>Severability</u>. The illegality or unenforceability of any provision of this Amendment or any instrument or agreement required hereunder shall not in any way affect or impair the legality or enforceability of the remaining provisions of this Amendment or any instrument or agreement required hereunder.
  - 6.4. Captions. Section captions used in this Amendment are for convenience only, and shall not affect the construction of this Amendment.
- 6.5. Entire Agreement. This Amendment embodies the entire agreement and understanding among the parties hereto and supersedes all prior or contemporaneous agreements and understandings of such Persons, verbal or written, relating to the subject matter hereof.

6.6. <u>References</u>. Any reference to the Credit Agreement contained in any notice, request, certificate, or other document executed concurrently with or after the execution and delivery of this Amendment shall be deemed to include this Amendment unless the context shall otherwise require. Reference in any of this Amendment, the Credit Agreement or any other Loan Document to the Credit Agreement shall be a reference to the Credit Agreement as amended hereby and as further amended, modified, restated, supplemented or extended from time to time.

Section 7. <u>Governing Law</u>. THIS AMENDMENT SHALL BE A CONTRACT MADE UNDER AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF ILLINOIS APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE, WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES.

[signature pages follow]

Witness the due execution hereof by the respective duly authorized officers of the undersigned as of the date first above written.

#### ADDUS HEALTHCARE, INC.

By: /s/ David W. Stasiewicz

Name: David W. Stasiewicz

Title: Chief Financial Officer

#### ADDUS HOLDING CORPORATION

By: /s/ Simon Bachleda

Name: Simon Bachleda

Title: Vice President

#### ADDUS MANAGEMENT CORPORATION

By: /s/ Simon Bachleda
Name: Simon Bachleda
Title: Vice President

# LOWELL HOME HEALTH AGENCY, INC.

By: /s/ David W. Stasiewicz
Name: David W. Stasiewicz
Title: Chief Financial Officer

LITTLE ROCK	HOME H	EALTH A	GENCY, INC.
-------------	--------	---------	-------------

By: /s/ David W. Stasiewicz
Name: David W. Stasiewicz

Title: Chief Financial Officer

#### FORT SMITH HOME HEALTH AGENCY, INC.

By: /s/ David W. Stasiewicz
Name: David W. Stasiewicz

Title:

By:

By:

#### BENEFITS ASSURANCE CO., INC.

Chief Financial Officer

By: /s/ David W. Stasiewicz

Name: David W. Stasiewicz

Title: Chief Financial Officer

#### PHC ACQUISITION CORPORATION

/s/ David W. Stasiewicz

Name: David W. Stasiewicz
Title: Chief Financial Officer

#### PROFESSIONAL RELIABLE NURSING SERVICE, INC.

/s/ David W. Stasiewicz

Name: David W. Stasiewicz
Title: Chief Financial Officer

#### FREEPORT FINANCIAL LLC, as Agent

By: /s/ Ian Fowler

Title: Duly Authorized Signatory

# FREEPORT LOAN FUND LLC, as a Lender

By: /s/ Ian Fowler

Title: Duly Authorized Signatory

# FREEPORT OFFSHORE LOAN FUND LLC, as a Lender

By: /s/ Ian Fowler

Title: Duly Authorized Signatory

#### FREEPORT LOAN TRUST 2006-1, as a Lender

/s/ Ian Fowler

By:

Title:

Duly Authorized Signatory

# FIFTH THIRD BANK, CHICAGO (A Michigan Banking Company)

By: /s/ Michael E. May

Name: Michael E. May
Title: Vice President

#### CF BLACKBURN LLC

/s/ Dennis M. Hansen

Name: Dennis M. Hansen

Title: VP

By:

#### EXHIBIT A

#### OFFICER'S CERTIFICATE

Reference is made to that certain Consent and Third Amendment to Credit Agreement, dated as of November 13, 2007 (the "Third Amendment"; capitalized terms used herein and not otherwise defined shall have the meanings specified thereto in that certain Credit Agreement dated as of September 19, 2006 among the Loan Parties party thereto, the Agent and the Lenders party thereto (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007, that certain Consent and Second Amendment to Credit Agreement dated as of October 15, 2007 and as further amended, restated or otherwise modified from time to time including by the Third Amendment, the "Credit Agreement"), among the Borrower, certain other parties, the financial institutions party thereto and Freeport Financial LLC, as Agent. The undersigned hereby certifies that [s]he is the duly authorized and elected \_\_\_\_\_\_\_ of the Borrower and that, as of the date hereof:

- 1. As of the Third Amendment Effective Date, after giving pro forma effect to the Vegas Acquisitions to and the incurrence of the Delayed Draw Term Loans in connection therewith, Borrower is in compliance with each of the covenants set forth in Section 7.1.
- 2. No Material Adverse Effect shall have occurred upon the financial condition, operations, assets, business or properties of (i) the Loan Parties, Greater Vegas or Vegas Valley, taken as a whole, since December 31, 2006; or (ii) Greater Vegas and Vegas Valley (in each case, other than the Medicaid Investigation) since December 31, 2006.
- 3. After giving effect to the Third Amendment, no Event of Default or Default under the Credit Agreement has occurred or is continuing.
- 4. After giving effect to the Third Amendment, the representations and warranties of the Loan Parties contained in the Third Amendment, the Credit Agreement and the other Loan Documents are true and correct in all material respects as of the Third Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty was true and correct in all material respects as of such earlier date.
- 5. The Loan Parties have consummated the Vegas Acquisitions concurrently with the Third Amendment substantially in accordance with the Vegas Acquisition Documents and in accordance with all applicable law, except as may have been consented to in writing by Agent.

6. Attached hereto are true, complete and correct copies of the fully executed Vegas Acquisition Documents.

[Remainder of Page Intentionally Blank]

ADDUS HEALTHCARE, INC.
Ву:
Name:
Title:

IN WITNESS WHEREOF, I have hereunto set my hand this \_\_\_\_\_ day of November, 2007.

**EXHIBIT B** 

#### **CONSENT AND REAFFIRMATION**

Each of the undersigned ("Loan Parties") hereby (i) acknowledges receipt of a copy of the Consent and Third Amendment to Credit Agreement dated as of November 13, 2007 (the "Third Amendment"); (ii) consents to the execution and delivery thereof by the other Loan Parties; (iii) agrees to be bound thereby; (iv) affirms that nothing contained therein shall modify in any respect whatsoever its guaranty of the obligations of the Loan Parties to Agent and Lenders (the "Guarantee") and (v) reaffirms that the Guarantee and the other Collateral Documents executed by such Person are and shall continue to remain in full force and effect. Although each of the Loan Parties has been informed of the matters set forth herein and in the Third Amendment and has acknowledged and agreed to same, such Loan Parties understand that Agent and Lenders have no obligation to inform any of the Loan Parties of such matters in the future or to seek any of the Loan Parties' acknowledgment or agreement to future amendments or waivers, and nothing herein shall create such a duty.

This Consent and Reaffirmation shall be governed by and construed in accordance with the laws of the State of Illinois, without reference to principles of conflicts of law.

In witness whereof	each of the n	indersioned has e	executed this Co	insent and Reaffi	rmation on and as	s of the date of si	uch Third Amendment

# ADDUS HEALTHCARE, INC.

By:	
Name:	
Title:	
11010.	
ADDUS	S HOLDING CORPORATION
By:	
Title:	
Title.	<del></del>
ADDUS	S HEALTHCARE (NEVADA), INC.
By:	
Name:	
Title:	
ADDU	S MANAGEMENT CORPORATION
By:	
Title:	
LOWE	LL HOME HEALTH AGENCY, INC.
Bv:	
5	
Title:	
Title.	
LITTL	E ROCK HOME HEALTH AGENCY, INC.
By:	
Name:	
Title:	
TILIC.	

	SMITH HOME HEALTH AGENCY, INC.
By:	
Title:	
BENE	FITS ASSURANCE CO., INC.
By:	
Name:	
Title:	
PHC A	ACQUISITION CORPORATION
By:	
Name:	
Name: Title:	
Title:	
Title:	
Title:	ESSIONAL RELIABLE NURSING SERVICE, INC
Title:  PROF  By:	ESSIONAL RELIABLE NURSING SERVICE, INC
PROFI By: Name: Title:	ESSIONAL RELIABLE NURSING SERVICE, INC
PROFI By: Name: Title:	ESSIONAL RELIABLE NURSING SERVICE, INC
PROFI By: Name: Title:	ESSIONAL RELIABLE NURSING SERVICE, INC  S HEALTHCARE (NEW JERSEY), INC.

	THIRD AMENDM DRAW FUNDI	
	\$	%
Lender		
Freeport Loan Fund LLC	1,371,428.57	34.2857143
Residential Funding Corporation	-0-	-0-
Fifth Third Bank (Chicago)	2,628,571.43	65.7142857
Totals	\$ 4,000,000	100%

#### CONSENT AND FOURTH AMENDMENT TO CREDIT AGREEMENT

CONSENT AND FOURTH AMENDMENT TO CREDIT AGREEMENT (this "Amendment"), dated as of April 1, 2008, among Addus HealthCare, Inc., an Illinois corporation ("Borrower"), the other persons signatory hereto as "Loan Parties", FREEPORT FINANCIAL LLC, a Delaware limited liability company ("Agent") and the Lenders signatory hereto. Terms not defined herein have the meanings given to them in the Credit Agreement (as hereinafter defined).

#### **RECITALS**

- A. Borrower, the Loan Parties, the Lenders signatory thereto and Agent are party to that certain Credit Agreement dated as of September 19, 2006 (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007, that certain Consent and Second Amendment to Credit Agreement dated as of October 15, 2007, that certain Consent and Third Amendment to Credit Agreement dated as of November 13, 2007 and as further amended, restated or otherwise modified including by this Amendment, the "Credit Agreement").
- B. Borrower, the Loan Parties, the Lenders and Agent are entering into this Amendment in connection with the acquisition (the "Full Life Acquisition") by Addus HealthCare (Idaho), Inc., a Delaware corporation and a wholly-owned Subsidiary of the Borrower ("Addus Idaho"), of substantially all of the assets of each of A Full Life Agency, Inc., Life's Alternatives, Inc., A Full Life of Montana, Inc., A Full Life Home Care, Inc., A Full Life Home Health, Lewiston, Inc., and A Full Life Home Health, Boise, Inc. (collectively "Full Life") for an aggregate purchase price not to exceed \$4,325,000.
  - C. Agent, Lenders and the Loan Parties are willing to enter into this Amendment upon the terms and conditions set forth below.

NOW THEREFORE, in consideration of the matters set forth in the recitals and the covenants and provisions herein set forth, and other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

#### **AGREEMENT**

Section 1. <u>Definitions</u>. Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed thereto in the Credit Agreement.

Section 2. <u>Consent</u>. Agent and the Lenders hereby consent to Addus Idaho (i) entering into (w) that certain Asset Purchase Agreement, dated as of even date herewith (the "<u>Full Life Purchase Agreement</u>"), by and between Addus Idaho, the Borrower, Donald Gross, Sandra Gross, A Full Life Agency, Inc., Life's Alternatives, Inc., Alternatives For Life, Inc., A Full Life of Montana, Inc., A Full Life Home Care, Inc., A Full Life Home Health, Lewiston, Inc., and A Full Life Home Health, Boise, Inc., (x) the Idaho Note, (y) the Idaho Earn Out Agreement and

(z) the other agreements, instruments and documents for the purpose of consummating the Full Life Acquisition in each case on terms and conditions satisfactory to the Agent (the "Full Life Acquisition Documents", it being agreed that the forms of the Full Life Purchase Agreement, the Idaho Note, the Idaho Earn Out Agreement and such other agreements, instruments and documents provided to the Agent on April 1, 2008 are satisfactory to the Agent); (ii) consummating the Full Life Acquisition on the Fourth Amendment Effective Date for an aggregate purchase price not to exceed \$2,000,000 paid in cash on the closing date thereof pursuant to the terms of the Full Life Acquisition Documents, plus a deferred purchase price in an amount not to exceed \$950,000 paid pursuant to the Idaho Earn Out Agreement, plus the issuance of the Idaho Note and up to \$125,000 in deferred payments (the "Idaho Deferred Payments") within fifteen days of the delivery of the 2008 audited financial statements of Borrower paid pursuant to Section 3.1(d) of the Full Life Purchase Agreement; provided that neither such deferred purchase price paid pursuant to Section 3.1(d) of the Full Life Purchase Agreement, nor any amount paid pursuant to the Idaho Earn Out Agreement nor any payments on the Idaho Note shall be paid in the event that any Default or Event of Default under any of Sections 7.1 or 8.1(a) of the Credit Agreement has occurred and is continuing or would result therefrom or the Borrower does not have the minimum Borrowing Availability required by Section 6.17 of the Credit Agreement after giving effect to such payment.

Section 3. Amendment to the Credit Agreement. As of the Fourth Amendment Effective Date, the Credit Agreement is hereby amended as follows:

3.1. Section 1.1 of the Credit Agreement is hereby amended by adding the following new definitions thereto:

Fourth Amendment means that certain Consent and Fourth Amendment to Credit Agreement dated as of April 1, 2008 among the Loan Parties, Agent and the Lenders.

Fourth Amendment Effective Date has the meaning set forth in the Fourth Amendment.

Fourth Amendment Reaffirmation of Collateral Documents means the Consent and Reaffirmation (Fourth Amendment) dated as of April 1, 2008 of the Loan Parties signatory thereto, in respect of the Collateral Documents.

Full Life has the meaning set forth in the Recitals to the Fourth Amendment.

Full Life Acquisition has the meaning set forth in the Recitals to the Fourth Amendment.

Full Life Acquisition Documents has the meaning set forth in Section 2 of the Fourth Amendment.

<u>Idaho Deferred Payments</u> has the meaning set forth in Section 2 of the Fourth Amendment.

Idaho Earn Out Agreement means that certain Earn Out Agreement dated as of April 1, 2008 by and among Addus Idaho, A Full Life Agency, Inc., Life's Alternatives, Inc., Alternatives For Life, Inc., A Full Life of Montana, Inc., A Full Life Home Care, Inc., A Full Life Home Health, Lewiston, Inc., A Full Life Home Health, Boise, Inc., Donald Gross and Sandra Gross.

<u>Idaho Note</u> means that certain 8% Junior Subordinated Promissory Note by Addus Idaho payable to Donald and Sandra Gross in the original principal amount of \$1,250,000.

3.2. Section 1.1 of the Credit Agreement is further amended by amending and restating the proviso at the end of the definition of "EBITDA" to read as follows:

"provided, that, notwithstanding anything to the contrary contained herein, for each of the Fiscal Quarters listed below, EBITDA shall be deemed to be the amount set forth below opposite such corresponding period:

Period March 31, 2007	EBITDA \$3,795,286
June 30, 2007	\$3,911,279
September 30, 2007	\$3,933,970
December 31, 2007	\$4,099,889

Provided, further, that for any period that includes the months of January, February or March of 2008, EBITDA shall, without duplication, be increased with respect to the assets acquired in connection with the Full Life Acquisition by an amount equal to \$74,000 for each such month."

- 3.3. Section 2.1(a)(y) of the Credit Agreement is hereby amended and restated in its entirety and replaced with the following new Section 2.1(a)(y):
- (v) Borrower shall repay the Term Loans through periodic payments on the dates and in the amounts indicated below ("Scheduled Installments"), and in any event the entire remaining principal balance shall be repaid on the Commitment Termination Date.

#### Term Loan

<u>Date</u>	Scheduled Installment
June 30, 2008	\$1,165,000
September 30, 2008	\$1,165,000
December 31, 2008	\$1,572,500
March 31, 2009	\$1,572,500
June 30, 2009	\$1,572,500
September 30, 2009	\$1,572,500
December 31, 2009	\$1,980,000

March 31, 2010	\$ 1,980,000
June 30, 2010	\$ 1,980,000
September 30, 2010	\$ 1,980,000
December 31, 2010	\$ 2,387,500
March 31, 2011	\$ 2,387,500
June 30, 2011	\$ 2,387,500
September 19, 2011	\$ 27,717,500 or the outstanding principal balance of Term Loans outstanding on such date

The above scheduled installment amounts reflect the incurrence by the Borrower of the Delayed Draw Term Loans funded on the Fourth Amendment Effective Date and shall thereafter be increased in the manner set forth on Annex B to the Second Amendment to the extent any further Delayed Draw Term Loans are funded. The final installment payment shall in all events equal the entire remaining principal balance of the Term Loan (including any remaining principal balance of such Delayed Draw Term Loans). Amounts borrowed under this Section 2.1(a) and repaid may not be reborrowed.

At the request of the applicable Lender, the Term Loans shall be evidenced by promissory notes substantially in the form of Exhibit 2.1(a) (as amended, modified, extended, substituted or replaced from time to time, each a "Term Note" and, collectively, the "Term Notes"), and Borrower shall execute and deliver a Term Note to each such Term Lender. Each Term Note shall represent the obligation of Borrower to pay the amount of the applicable Term Lender's Term Loan Commitment, together with interest thereon."

- 3.4. Section 6.1 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause (n) thereof, deleting the "." at the end of clause (o) thereof and replacing it with "; and" and by adding the following new clause (p):
  - "(p) Indebtedness in respect of the Idaho Earn Out Agreements, the Idaho Deferred Payments and Idaho Note."
- 3.5. Section 6.3 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause (n) thereof, deleting the "." at the end of clause (o) thereof and replacing it with "; and" and by adding the following new clause (p):
- "(p) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Full Life Acquisition as of the Fourth Amendment Effective Date pursuant to the Full Life Acquisition Documents."
- 3.6. <u>Clause (d)</u> of <u>Section 6.6</u> of the Credit Agreement is hereby amended by deleting the "and" before clause (iv) thereof and adding the following new language immediately at the end thereof:

"and (v) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Full Life Acquisition on the Fourth Amendment Effective Date pursuant to the Full Life Acquisition Documents and all applicable law, provided that the Full Life Acquisition Documents shall be in full force and effect and provided further that the Agent shall have received a complete copy of the fully executed Full Life Acquisition Documents, certified by the Borrower as being true, complete and correct."

- 3.7. Section 6.16 of the Credit Agreement is amended and restated in its entirety to read as follows:
- "None of the Loan Parties shall change or amend the terms of the Acquisition Note, Silver State Note, Vegas Earn Out Agreement, Idaho Earn Out Agreement, Idaho Note or SuCasa Note if such change or amendment would be adverse in any material respect to the rights or interests of the Loan Parties, the Agent or the Lenders."
  - 3.8. Section 6.17 of the Credit Agreement is hereby amended by replacing the reference to "\$3,000,000" therein with a reference to "\$3,500,000".
  - 3.9. Section 7.1(b) of the Credit Agreement is amended and restated in its entirety to read as follows:

"Minimum EBITDA. Holdings and its Subsidiaries on a consolidated basis shall have, at the end of each Fiscal Quarter set forth below, Minimum EBITDA for the 12-Fiscal Month period then ended calculated of not less than the following:

Period	Minimum EBITDA
March 31, 2008	\$13,500,000
June 30, 2008	\$13,750,000
September 30, 2008	\$14,000,000
December 31, 2008	\$14,250,000
March 31, 2009	\$14,750,000
June 30, 2009	\$15,000,000
September 30, 2009	\$15,500,000
December 31, 2009	\$15,500,000
March 31, 2010	\$15,500,000
June 30, 2010	\$15,500,000
September 30, 2010	\$16,000,000
December 31, 2010	\$16,000,000
March 31, 2011 and each Fiscal Quarter ending thereafter	\$16,000,000

- 3.10. Schedule 2 to Annex E of the Credit Agreement is amended to include the following new paragraph in the column underneath the word Less: and prior to the word Subtotal:
- "an amount equal to any unfinanced cash payments during such Fiscal Year with respect to the Idaho Earn Out and Idaho Note; provided that the term unfinanced when used in this paragraph shall mean any such payment that the Borrower has not directly or indirectly drawn Revolving Loans (or used other Indebtedness) to finance such payment with the determination of whether any such payment is unfinanced being acknowledged and consented to by Agent such consent not to be unreasonably withheld."
- Section 4. Representations and Warranties. To induce Agent and Lenders to execute this Amendment, each Loan Party hereby represents and warrants to Agent and Lenders as follows:
  - (a) the execution, delivery and performance of each of this Amendment and the Full Life Acquisition Documents have been duly authorized by all requisite action of each Loan Party thereto, and each of this Amendment and the Full Life Acquisition Documents constitutes the legal, valid and binding obligation of such Person, enforceable against such Person in accordance with its terms, subject to bankruptcy, insolvency and similar laws affecting the enforceability of creditors' rights generally and to general principles of equity; and
  - (b) each of the representations and warranties in the Credit Agreement are true and correct in all material respects with the same effect as though made on and as of the date hereof (except, in each case, to the extent stated to relate to an earlier date, in which case such representation or warranty shall have been true and correct on and as of such earlier date) and no Event of Default or Default exists thereunder or would exist after giving effect to this Amendment or the Full Life Acquisition.
- Section 5. <u>Conditions Precedent</u>. This Amendment shall be effective as of the date first set forth above, subject to the satisfaction of the following conditions precedent (the date of such satisfaction being the "<u>Fourth Amendment Effective Date</u>"):
- 5.1. Execution and Delivery. Borrower, each of the other Loan Parties, Agent and the Requisite Lenders shall have executed and delivered this Agreement; the Borrower shall have executed and delivered an Officers Certificate in the form of Exhibit A attached hereto; each Loan Party shall have executed and delivered the Consent and Reaffirmation in the form of Exhibit B attached hereto.
- 5.2. <u>Covenant Compliance</u>. After giving pro forma effect to the Full Life Acquisition to and the incurrence of the Delayed Draw Term Loans in connection therewith, Borrower is in compliance with each of the covenants set forth in <u>Section 7.1</u>.
- 5.3. Letter of Direction. Agent shall have received a duly executed letter of direction from Borrower addressed to Agent, on behalf of itself and the Lenders directing the disbursement of the proceeds of the Delayed Draw Term Loans to fund the Full Life Acquisition.

- 5.4. <u>Insurance</u>. Agent shall have received evidence of the existence of insurance required to be maintained pursuant to the Credit Agreement, together with evidence that the Agent has been named as a lender's loss payee and an additional insured on all related insurance policies to the extent there is any change in such insurance in effect prior to the Fourth Amendment Effective Date as a result of or in connection with the Full Life Acquisition.
- 5.5. <u>Closing Certificate</u>. Agent shall have received a certificate executed by an officer of the Borrower on behalf of the Loan Parties certifying the matters set forth herein as conditions.
- 5.6. Filings, Registrations and Recordings. The Agent shall have received each document (including Uniform Commercial Code financing statements) required by Section 5.7(c) of the Credit Agreement as may be necessary or desirable in order to create in favor of the Agent, for the benefit of the Lenders, a perfected Lien on the Collateral acquired in connection with the Full Life Acquisition prior to any other Liens (subject only to Permitted Encumbrances).
- 5.7. <u>Due Diligence</u>. Agent, Lenders and their counsel and advisors shall have completed their legal, accounting and financial due diligence review (including, without limitation, lien searches on Full Life) of Full Life and the Full Life Acquisition, the results of which review shall be reasonably satisfactory to Agent and Requisite Lenders.
- 5.8. <u>Material Adverse Effect</u>. There shall have occurred no Material Adverse Effect upon the financial condition, operations, assets, business or properties of (i) the Loan Parties or Full Life, taken as a whole, since December 31, 2006; or (ii) Full Life since December 31, 2006.
- 5.9. Full Life Acquisition Documents. On or prior to the Fourth Amendment Effective Date, each of the Full Life Acquisition Documents shall have been duly authorized by each of the Loan Parties party thereto and executed and delivered by each of the respective parties thereto and shall be in full force and effect and shall not have been amended or modified in any manner adverse to the Lenders except for such amendments and modifications, if any, as may be reasonably satisfactory to Agent and the Requisite Lenders. The Full Life Acquisition shall have been, or shall be contemporaneously herewith, consummated substantially in accordance with the Full Life Acquisition Documents (as amended in accordance with the immediately preceding sentence) and in accordance with all applicable law, except as may have been consented to in writing by Agent. Agent shall have received a complete copy of the fully executed Full Life Acquisition Documents, certified by the Borrower as being true, complete and correct.
- 5.10. Approvals. All necessary governmental and third party approvals and/or consents in connection with the Full Life Acquisition and otherwise referred to herein or therein shall have been obtained and remain in effect, and all applicable waiting periods shall have expired without any action being taken by any competent authority which restrains, prevents or imposes materially adverse conditions upon the consummation of all or any part of the Full Life Acquisition, in each case as of the Fourth Amendment Effective Date. Additionally, there shall not exist any judgment, order, injunction or other restraint pending or notified prohibiting or imposing material adverse conditions upon all or any part of the Full Life Acquisition or the transactions contemplated by this Amendment as of the Fourth Amendment Effective Date.

- 5.11. No Defaults. After giving effect to this Amendment, no Event of Default or Default under the Credit Agreement shall have occurred and be continuing.
- 5.12. Representations and Warranties. After giving effect to this Amendment, the representations and warranties of the Loan Parties contained in this Amendment, the Credit Agreement and the other Loan Documents shall be true and correct in all material respects as of the Fourth Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty shall be true and correct in all material respects as of such earlier date.
- 5.13. Secretary's Certificate. Agent shall have received from each Loan Party a certificate signed by the secretary or an assistant secretary of such Loan Party, dated as of the Fourth Amendment Effective Date, as to the incumbency and signature of the officers of such Loan Party executing any Loan Document and any certificate or other document or instrument to be delivered pursuant hereto by or on behalf of such Loan Party, together with evidence of the incumbency of such Secretary or Assistant Secretary, as the case may be and certifying that there have been no changes to (i) the resolutions of such Loan Party authorizing and approving, among other things, the execution and delivery of the Loan Documents and (ii) the organizational documents of such Loan Party in each case since the Closing Date.
- 5.14. <u>Resolutions</u>. Agent shall have received a copy of the resolutions (in form and substance reasonably satisfactory to Agent) adopted by the Board of Directors of each Loan Party, authorizing or relating to the execution, delivery and performance of this Amendment and the Full Life Acquisition Documents and the other documents and instruments provided for therein and the consummation of the transactions contemplated hereby and thereby.
- 5.15. Good Standing Certificates. Agent shall have received good standing certificates for each Loan Party from their respective jurisdictions of incorporation or organization.
- 5.16. <u>Fees</u>. Borrower shall have paid all reasonable and documented costs, fees and expenses (including, without limitation, reasonable legal fees and expenses of Winston & Strawn LLP) of Agent.
  - 5.17. Opinions of Counsel. Agent shall have received opinions of counsel for each Loan Party, including local counsel reasonably requested by the Agent.
- 5.18. Other Matters. Agent shall have received such other instruments and documents as Agent or the Requisite Lenders may reasonably request in connection with the execution of this Amendment, and all such instruments and documents shall be reasonably satisfactory in form and substance to Agent.

#### Section 6. Miscellaneous.

- 6.1. Effect of Amendment. The execution, delivery and effectiveness of this Amendment shall not operate as a waiver of any right, power or remedy of Agent or any Lender under the Credit Agreement or any Loan Document, or constitute a waiver of any provision of the Credit Agreement or any Loan Document, except as specifically set forth herein.
- 6.2. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and by the different parties on separate counterparts, and each such counterpart shall be deemed to be an original, but all such counterparts shall together constitute but one and the same instrument. Delivery of the executed counterpart of this Amendment by telecopy or electronic mail shall be as effective as delivery of a manually executed counterpart to this Amendment.
- 6.3. <u>Severability</u>. The illegality or unenforceability of any provision of this Amendment or any instrument or agreement required hereunder shall not in any way affect or impair the legality or enforceability of the remaining provisions of this Amendment or any instrument or agreement required hereunder.
  - 6.4. Captions. Section captions used in this Amendment are for convenience only, and shall not affect the construction of this Amendment.
- 6.5. Entire Agreement. This Amendment embodies the entire agreement and understanding among the parties hereto and supersedes all prior or contemporaneous agreements and understandings of such Persons, verbal or written, relating to the subject matter hereof.
- 6.6. <u>References</u>. Any reference to the Credit Agreement contained in any notice, request, certificate, or other document executed concurrently with or after the execution and delivery of this Amendment shall be deemed to include this Amendment unless the context shall otherwise require. Reference in any of this Amendment, the Credit Agreement or any other Loan Document to the Credit Agreement shall be a reference to the Credit Agreement as amended hereby and as further amended, modified, restated, supplemented or extended from time to time.

Section 7. <u>Governing Law</u>. THIS AMENDMENT SHALL BE A CONTRACT MADE UNDER AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF ILLINOIS APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE, WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES.

[signature pages follow]

Witness the due execution hereof by the respective duly authorized officers of the under	ersigned	as of the date first above written.
	ADDUS	S HEALTHCARE, INC.
]	By:	/s/ Mark Heaney
	Name:	Mark Heaney
•	Title:	VP & CEO
	ADDUS	S HOLDING CORPORATION
1	By:	/s/ Simon A. Bachleda
	Name:	Simon A. Bachleda
•	Title:	
	ADDUS	S MANAGEMENT CORPORATION
	By:	/s/ Simon A. Bachleda
	Name:	Simon A. Bachleda
•	Title:	

By:

Name:

Title:

LOWELL HOME HEALTH AGENCY, INC.

/s/ Mark Heaney

Mark Heaney
VP & CEO

LITT	LE ROCK HOME HEALTH AGENCY, INC.
By:	/s/ Mark Heaney
Name	: Mark Heaney
Title:	VP & COO
FORT	T SMITH HOME HEALTH AGENCY, INC.
By:	/s/ Mark Heaney
Name	: Mark Heaney
Title:	VP & COO
BENI	EFITS ASSURANCE CO., INC.
BENI By:	
BENI By: Name	EFITS ASSURANCE CO., INC. /s/ David W. Stasiewicz
BENI By: Name Title:	EFITS ASSURANCE CO., INC. /s/ David W. Stasiewicz : David W. Stasiewicz Secretary  ACQUISITION CORPORATION
BENE By: Name Title: PHC -	/s/ David W. Stasiewicz : David W. Stasiewicz Secretary

PROFESSIONAL RELIABLE NURSING SERVICE, INC.

By: /s/ Mark Heaney
Name: Mark Heaney
Title: VP & COO

## FREEPORT FINANCIAL LLC, as Agent

By: /s/ Chad Blakeman
Title: Sr. Managing Director

## FREEPORT LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman
Title: Sr. Managing Director

## FREEPORT OFFSHORE LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman
Title: Sr. Managing Director

## FREEPORT LOAN TRUST 2006-1, as a Lender

By: /s/ Chad Blakeman
Title: Sr. Managing Director

# FIFTH THIRD BANK, CHICAGO (A Michigan Banking Company)

By: /s/ Gregory H. Bork
Name: Gregory H. Bork

Title: Vice President

#### CF BLACKBURN LLC

By: /s/ Dennis M. Hansen

Name: Dennis M. Hansen

Title: V.P.

#### **OFFICER'S CERTIFICATE**

Reference is made to that certain Consent and Fourth Amendment to Credit Agreement, dated as of April 1, 2008 (the "Fourth Amendment"; capitalized terms used herein and not otherwise defined shall have the meanings specified thereto in that certain Credit Agreement dated as of September 19, 2006 among the Loan Parties party thereto, the Agent and the Lenders party thereto (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007, that certain Consent and Second Amendment to Credit Agreement dated as of October 15, 2007, that certain Consent and Third Amendment to Credit Agreement dated as of November 13, 2007 and as further amended, restated or otherwise modified from time to time including by the Fourth Amendment, the "Credit Agreement"), among the Borrower, certain other parties, the financial institutions party thereto and Freeport Financial LLC, as Agent. The undersigned hereby certifies that [s]he is the duly authorized and elected \_\_\_\_\_\_\_ of the Borrower and that, as of the date hereof:

- 1. As of the Fourth Amendment Effective Date, after giving pro forma effect to the Full Life Acquisition to and the incurrence of the Delayed Draw Term Loans in connection therewith, Borrower is in compliance with each of the covenants set forth in Section 7.1.
- 2. No Material Adverse Effect shall have occurred upon the financial condition, operations, assets, business or properties of (i) the Loan Parties or Full Life, taken as a whole, since December 31, 2006; or (ii) Full Life since December 31, 2006.
- 3. After giving effect to the Fourth Amendment, no Event of Default or Default under the Credit Agreement has occurred or is continuing.
- 4. After giving effect to the Fourth Amendment, the representations and warranties of the Loan Parties contained in the Fourth Amendment, the Credit Agreement and the other Loan Documents are true and correct in all material respects as of the Fourth Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty was true and correct in all material respects as of such earlier date.
- 5. The Loan Parties have consummated the Full Life Acquisition concurrently with the Fourth Amendment substantially in accordance with the Full Life Acquisition Documents and in accordance with all applicable law, except as may have been consented to in writing by Agent.

6.	Attached hereto are true, complete and correct copies of the fully executed Full Life Acquisition Documents.
	[Remainder of Page Intentionally Blank]

ADDUS HEALTHCARE, INC.
Ву:
Name:
Title:

IN WITNESS WHEREOF, I have hereunto set my hand this \_\_\_\_\_ day of April, 2008.

**EXHIBIT B** 

## **CONSENT AND REAFFIRMATION**

Each of the undersigned ("Loan Parties") hereby (i) acknowledges receipt of a copy of the Consent and Fourth Amendment to Credit Agreement dated as of April 1, 2008 (the "Fourth Amendment"); (ii) consents to the execution and delivery thereof by the other Loan Parties; (iii) agrees to be bound thereby; (iv) affirms that nothing contained therein shall modify in any respect whatsoever its guaranty of the obligations of the Loan Parties to Agent and Lenders (the "Guarantee") and (v) reaffirms that the Guarantee and the other Collateral Documents executed by such Person are and shall continue to remain in full force and effect. Although each of the Loan Parties has been informed of the matters set forth herein and in the Fourth Amendment and has acknowledged and agreed to same, such Loan Parties understand that Agent and Lenders have no obligation to inform any of the Loan Parties of such matters in the future or to seek any of the Loan Parties' acknowledgment or agreement to future amendments or waivers, and nothing herein shall create such a duty.

This Consent and Reaffirmation shall be governed by and construed in accordance with the laws of the State of Illinois, without reference to principles of conflicts of law.

In witness whereof	each of the	indercioned has	executed this Con	cent and Reaffirms	ation on and ac of	the date of such Fourt	h Amendment

## ADDUS HEALTHCARE, INC.

HOLDING CORPORATION
HEALTHCARE (NEVADA), INC.
HEALTHCARE (IDAHO), INC.
MANACEMENT CORRORATION
MANAGEMENT CORPORATION
LL HOME HEALTH AGENCY, INC.

By:	
Name:	
Title:	
FORT	SMITH HOME HEALTH AGENCY, INC.
By:	
Name:	
Title:	
BENEI	FITS ASSURANCE CO., INC.
By:	
Name:	
	CQUISITION CORPORATION
PHC A By: Name:	CQUISITION CORPORATION
PHC A By: Name: Title:	
PHC A By: Name: Title:	
PHC A By: Name: Title: PROFI	
PHC A By: Name: Title: PROFI By:	ESSIONAL RELIABLE NURSING SERVICE, INC.
PHC A By: Name: Title: PROFI By: Name:	ESSIONAL RELIABLE NURSING SERVICE, INC.
PHC A By: Name: Title: PROFI By: Name: Title:	ESSIONAL RELIABLE NURSING SERVICE, INC.
By: Name: Title: PROFI By: Name: Title:	ESSIONAL RELIABLE NURSING SERVICE, INC.  S HEALTHCARE (NEW JERSEY), INC.
PHC A By: Name: Title: PROFI By: Name: Title:	ESSIONAL RELIABLE NURSING SERVICE, INC.

		FOURTH AMENDMENT DELAYED DRAW FUNDING AMOUNT	
	_	\$	%
Lender			
Freeport Loan Fund LLC	\$	857,142.86	34.2857143%
Residential Funding Corporation		-0-	-0-
Fifth Third Bank (Chicago)	\$	1,642,857.14	65.7142857%
Totals	\$	2,500,000	100%

#### CONSENT AND FIFTH AMENDMENT TO CREDIT AGREEMENT

CONSENT AND FIFTH AMENDMENT TO CREDIT AGREEMENT (this "Amendment"), dated as of June 9, 2008, among ADDUS HEALTHCARE, INC., an Illinois corporation ("Borrower"), the other persons signatory hereto as "Loan Parties", FREEPORT FINANCIAL LLC, a Delaware limited liability company ("Agent") and the Lenders signatory hereto. Terms not defined herein have the meanings given to them in the Credit Agreement (as hereinafter defined).

#### RECITALS

A. Borrower, the Loan Parties, the Lenders signatory thereto and Agent are party to that certain Credit Agreement dated as of September 19, 2006 (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007, that certain Consent and Second Amendment to Credit Agreement dated as of October 15, 2007, that certain Consent and Fourth Amendment to Credit Agreement dated as of November 13, 2007, that certain Consent and Fourth Amendment to Credit Agreement dated as of April 1, 2008 and as further amended, restated or otherwise modified including by this Amendment, the "Credit Agreement").

B. Borrower, the Loan Parties, the Lenders and Agent are entering into this Amendment in connection with (i) the acquisition (the "<u>Down East Acquisition</u>") by Addus HealthCare (North Carolina), Inc., a Delaware corporation and a wholly-owned Subsidiary of the Borrower ("<u>Addus North Carolina</u>"), of substantially all of the assets of Down East Health Care, LLC, a North Carolina limited liability company ("<u>Down East</u>"), for an aggregate purchase price not to exceed \$1,625,000 and (ii) the acquisition (the "<u>New Life Acquisition</u>") by Addus HealthCare (Nevada), Inc., a Delaware corporation and a wholly-owned Subsidiary of the Borrower ("<u>Addus Nevada</u>"), of substantially all of the assets of New Life Personal Care LLC, a Nevada limited liability company ("<u>New Life</u>"), for an aggregate purchase price not to exceed \$1,500,000, which acquisition shall occur no later than 30 days following the Fifth Amendment Effective Date (as hereinafter defined).

C. Agent, Lenders and the Loan Parties are willing to enter into this Amendment upon the terms and conditions set forth below.

NOW THEREFORE, in consideration of the matters set forth in the recitals and the covenants and provisions herein set forth, and other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

#### **AGREEMENT**

Section 1. Definitions. Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed thereto in the Credit Agreement.

Section 2. Consent. Agent and the Lenders hereby consent to the following:

2.1. Addus North Carolina (i) entering into (x) an Asset Purchase Agreement, dated as of even date herewith (the "Down East Purchase Agreement"), by and between Addus North Carolina, the sellers party thereto and Down East, for the purpose of acquiring substantially all of the assets of Down East and (y) the other agreements, instruments and documents for the purpose of consummating the Down East Acquisition in each case on terms and conditions reasonably satisfactory to the Agent (collectively, the "Down East Acquisition Documents"); (ii) consummating the Down East Acquisition on the Fifth Amendment Effective Date for an aggregate purchase price not to exceed \$1,000,000 paid in cash on the closing date thereof pursuant to the terms of the Down East Acquisition Documents, plus contingent consideration in an amount not to exceed \$500,000 paid pursuant to the Down East Purchase Agreement, plus up to \$125,000 in deferred payments (together with contingent consideration, the "Down East Deferred Payments"); provided that Down East Deferred Payments shall not be paid in the event that any Default or Event of Default under any of Sections 7.1 or 8.1(a) of the Credit Agreement has occurred and is continuing or would result therefrom or the Borrower does not have the minimum Borrowing Availability required by Section 6.17 of the Credit Agreement after giving effect to such payment.

- 2.2. Addus Nevada (i) entering into (x) that certain Asset Purchase Agreement (the "New Life Purchase Agreement"), by and between Addus Nevada, the sellers party thereto and New Life and (y) the other agreements, instruments and documents for the purpose of consummating the New Life Acquisition in each case on terms and conditions reasonably satisfactory to the Agent (collectively, the "New Life Acquisition Documents"), it being agreed that the forms of the New Life Purchase Agreement and the other New Life Acquisition Documents provided to the Agent on June 9, 2008 are satisfactory to the Agent; (ii) consummating the New Life Acquisition, subject to the satisfaction of the conditions set forth on Annex C attached hereto, on the New Life Acquisition Effective Date for an aggregate purchase price not to exceed \$300,000 paid in cash on the closing date thereof pursuant to the terms of the New Life Acquisition Documents, plus installment consideration ("New Life Installment Consideration") in an amount not to exceed \$1,200,000 paid pursuant to the New Life Purchase Agreement; provided that New Life Installment Consideration shall not be paid in the event that any Default or Event of Default under any of Sections 7.1 or 8.1(a) of the Credit Agreement has occurred and is continuing or would result therefrom or the Borrower does not have the minimum Borrowing Availability required by Section 6.17 of the Credit Agreement after giving effect to such payment.
- 2.3. The merger of Intermediate Holdings with and into the Borrower, with the Borrower being the surviving entity, which merger shall occur no later than 30 days following the Fifth Amendment Effective Date.

Section 3. Amendment to the Credit Agreement. As of the Fifth Amendment Effective Date, the Credit Agreement is hereby amended as follows:

3.1. Section 1.1 of the Credit Agreement is hereby amended by adding the following new definitions thereto:

Down East has the meaning set forth in the Recitals to the Fifth Amendment.

Down East Acquisition has the meaning set forth in the Recitals to the Fifth Amendment.

<u>Down East Acquisition Documents</u> has the meaning set forth in Section 2 of the Fifth Amendment.

Down East Deferred Payments has the meaning set forth in Section 2 of the Fifth Amendment.

Fifth Amendment means that certain Consent and Fifth Amendment to Credit Agreement dated as of June 9, 2008 among the Loan Parties, Agent and the Lenders.

Fifth Amendment Effective Date has the meaning set forth in the Fifth Amendment.

<u>Fifth Amendment Reaffirmation of Collateral Documents</u> means the Consent and Reaffirmation (Fifth Amendment), dated as of June 9, 2008 of the Loan Parties signatory thereto, in respect of the Collateral Documents.

New Life has the meaning set forth in the Recitals to the Fifth Amendment.

New Life Acquisition has the meaning set forth in the Recitals to the Fifth Amendment.

New Life Acquisition Documents has the meaning set forth in Section 2 of the Fifth Amendment.

New Life Acquisition Effective Date has the meaning set forth in the Fifth Amendment.

New Life Installment Consideration has the meaning set forth in Section 2 of the Fifth Amendment.

3.2. On the Fifth Amendment Effective Date, <u>Section 1.1</u> of the Credit Agreement shall be <u>amended</u> by amending and restating the proviso at the end of the definition of "EBITDA" to read as follows:

"provided, that, notwithstanding anything to the contrary contained herein, for each of the Fiscal Quarters listed below, EBITDA shall be deemed to be the amount set forth below opposite such corresponding period:

Period_	EBITDA
June 30, 2007	\$4,096,280
September 30, 2007	\$4,106,470
December 31, 2007	\$4,272,389
March 31, 2008	\$3,810,479

Provided, further, that for any period that includes the months of April or May of 2008, EBITDA shall, without duplication, be increased with respect to the assets acquired in connection with the Down East Acquisition by an amount equal to \$57,500 for each such month."

3.3. On the New Life Acquisition Effective Date, Section 1.1 of the Credit Agreement shall be further amended by amending and restating the proviso at the end of the definition of "EBITDA" to read as follows:

"provided, that, notwithstanding anything to the contrary contained herein, for each of the Fiscal Quarters listed below, EBITDA shall be deemed to be the amount set forth below opposite such corresponding period:

Period June 30, 2007	EBITDA \$4,242,030
September 30, 2007	\$4,252,220
December 31, 2007	\$4,418,139
March 31, 2008	\$3,956,229

Provided, further, that for any period that includes the months of April or May of 2008, EBITDA shall, without duplication, be increased with respect to the assets acquired in connection with the Down East Acquisition and New Life Acquisition by an amount equal to \$106,000 for each such month."

- 3.4. On the Fifth Amendment Effective Date, Section 2.1(a)(v) of the Credit Agreement shall be amended and restated in its entirety and replaced with the following new Section 2.1(a)(v):
- (v) Borrower shall repay the Term Loans through periodic payments on the dates and in the amounts indicated below ("Scheduled Installments"), and in any event the entire remaining principal balance shall be repaid on the Commitment Termination Date.

#### Term Loan

Date	Scheduled Installment
June 30, 2008	\$1,180,000
September 30, 2008	\$1,180,000
December 31, 2008	\$1,595,000
March 31, 2009	\$1,595,000
June 30, 2009	\$1,595,000
September 30, 2009	\$1,595,000
December 31, 2009	\$2,010,000
March 31, 2010	\$2,010,000

June 30, 2010	\$	2,010,000
September 30, 2010	\$	2,010,000
December 31, 2010	\$	2,425,000
March 31, 2011	\$	2,425,000
June 30, 2011	\$	2,425,000
September 19, 2011	pr	28,865,000 the outstanding incipal balance of Term Loans outstanding on such date

The above scheduled installment amounts reflect the incurrence by the Borrower of the Delayed Draw Term Loans funded on the Fifth Amendment Effective Date as set forth on Annex A attached hereto and shall thereafter be increased in the manner set forth on Annex B to the Second Amendment to the extent any further Delayed Draw Term Loans are funded. The final installment payment shall in all events equal the entire remaining principal balance of the Term Loan (including any remaining principal balance of such Delayed Draw Term Loans). Amounts borrowed under this Section 2.1(a) and repaid may not be reborrowed.

At the request of the applicable Lender, the Term Loans shall be evidenced by promissory notes substantially in the form of Exhibit 2.1(a) (as amended, modified, extended, substituted or replaced from time to time, each a "Term Note" and, collectively, the "Term Notes"), and Borrower shall execute and deliver a Term Note to each such Term Lender. Each Term Note shall represent the obligation of Borrower to pay the amount of the applicable Term Lender's Term Loan Commitment, together with interest thereon."

3.5. Section 2.1(a)(iii) of the Credit Agreement is hereby amended and restated in its entirety and replaced with the following new Section 2.1(a)(iii):

"(iii) On each Delayed Draw Date, subject to the terms and conditions in <a href="clause(iv)">clause(iv)</a> of this <a href="Section 2.1(a)">Section 2.1(a)</a> below, each Lender shall make available funds equal to its Pro Rata Share of the Delayed Draw Term Loan Commitment to be drawn on such date in immediately available funds to the Agent (the Loans made on each such date "<a href="Delayed Draw Term Loans">Delayed Draw Term Loans</a> and, collectively with the Second Amendment Term Loans and Original Term Loans, the "<a href="Term Loans">Term Loans</a>"). Except as otherwise agreed by Agent, each borrowing of a Delayed Draw Term Loan shall be in an aggregate minimum amount of \$2,500,000 and integral multiples of \$500,000 in excess of that amount. The Delayed Draw Term Loan Commitment shall expire on the earlier of (i) the date the Delayed Draw Term Loan Commitment is permanently reduced to zero, (ii) the date of the termination of the Commitments pursuant to <a href="Section 8.3">Section 8.3</a> and (iii) the date that is fifteen (15) months after the Second Amendment Effective Date, and any portion of the Delayed Draw Term Loan Commitment unused by the Borrower as of such date shall be automatically terminated. Borrower covenants and agrees that it shall use the proceeds of the Delayed Draw Term Loans (i) to fund Acquisitions consented to by the Requisite Lenders, (ii) to pay fees and expenses incurred in connection with such Acquisition and any amendment to the Credit Agreement in connection therewith and (iii) contemporaneously with the funding of any Acquisition consented to by the Requisite Lenders, to provide additional working capital for the Borrower in connection with such Acquisition."

- 3.6. On the New Life Acquisition Effective Date, Section 2.1(a)(y) of the Credit Agreement shall be amended and restated in its entirety and replaced with the following new Section 2.1(a)(y):
- (v) Borrower shall repay the Term Loans through periodic payments on the dates and in the amounts indicated below ("Scheduled Installments"), and in any event the entire remaining principal balance shall be repaid on the Commitment Termination Date.

#### Term Loan

Date Occord	heduled Installment
June 30, 2008	\$ 1,192,000
September 30, 2008	\$ 1,192,000
December 31, 2008	\$ 1,613,000
March 31, 2009	\$ 1,613,000
June 30, 2009	\$ 1,613,000
September 30, 2009	\$ 1,613,000
December 31, 2009	\$ 2,034,000
March 31, 2010	\$ 2,034,000
June 30, 2010	\$ 2,034,000
September 30, 2010	\$ 2,034,000
December 31, 2010	\$ 2,455,000
March 31, 2011	\$ 2,455,000
June 30, 2011	\$ 2,455,000
September 19, 2011	29,783,000 or the outstanding principal balance of Term Loans outstanding on such date

The above scheduled installment amounts reflect the incurrence by the Borrower of the Delayed Draw Term Loans funded on the New Life Acquisition Effective Date as set forth on Annex B attached hereto and shall thereafter be increased in the manner set forth on Annex B to the Second Amendment to the extent any further Delayed Draw Term Loans are funded. The final installment payment shall in all events equal the entire remaining principal balance of the Term Loan (including any remaining principal balance of such Delayed Draw Term Loans). Amounts borrowed under this Section 2.1(a) and repaid may not be reborrowed.

At the request of the applicable Lender, the Term Loans shall be evidenced by promissory notes substantially in the form of <u>Exhibit 2.1(a)</u> (as amended, modified, extended, substituted or replaced from time to time, each a "<u>Term Note</u>" and, collectively, the "<u>Term Notes</u>"), and

Borrower shall execute and deliver a Term Note to each such Term Lender. Each Term Note shall represent the obligation of Borrower to pay the amount of the applicable Term Lender's Term Loan Commitment, together with interest thereon."

- 3.7. Section 6.1 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause (o) thereof, deleting the "." at the end of clause (p) thereof and replacing it with "; and" and by adding the following new clause (q):
  - "(q) Indebtedness in respect of the Down East Deferred Payments and the New Life Installment Consideration."
- 3.8. Section 6.3 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause (o) thereof, deleting the "." at the end of clause (p) thereof and replacing it with "; and" and by adding the following new clauses (q) and (r):
- "(q) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Down East Acquisition as of the Fifth Amendment Effective Date pursuant to the Down East Acquisition Documents; and
- (r) Borrower or a wholly-owned Subsidiary of Borrower may consummate the New Life Acquisition as of the New Life Acquisition Effective Date pursuant to the New Life Acquisition Documents."
- 3.9. <u>Clause (d)</u> of <u>Section 6.6</u> of the Credit Agreement is hereby amended by deleting the "and" before clause (v) thereof and adding the following new language immediately at the end thereof:
- "(vi) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Down East Acquisition on the Fifth Amendment Effective Date pursuant to the Down East Acquisition Documents and all applicable law, provided that the Down East Acquisition Documents shall be in full force and effect and provided further that the Agent shall have received a complete copy of the fully executed Down East Acquisition Documents, certified by the Borrower as being true, complete and correct; and
- (vii) Borrower or a wholly-owned Subsidiary of Borrower may consummate the New Life Acquisition on the New Life Acquisition Effective Date pursuant to the New Life Acquisition Documents and all applicable law, provided that the New Life Acquisition Documents shall be in full force and effect and provided further that the Agent shall have received a complete copy of the fully executed New Life Acquisition Documents, certified by the Borrower as being true, complete and correct.
  - 3.10. Section 6.16 of the Credit Agreement is amended and restated in its entirety to read as follows:

"None of the Loan Parties shall change or amend the terms of the Acquisition Note, Silver State Note, Vegas Earn Out Agreement, Idaho Earn Out Agreement, Idaho Note, SuCasa Note, Down East Purchase Agreement or New Life Purchase Agreement if such change or amendment would be adverse in any material respect to the rights or interests of the Loan Parties, the Agent or the Lenders."

- 3.11. Section 6.17 of the Credit Agreement is hereby amended by replacing the reference to "\$3,500,000" therein with a reference to "\$4,000,000".
- 3.12. On the Fifth Amendment Effective Date, Section 7.1(b) of the Credit Agreement shall be amended and restated in its entirety to read as follows:

"Minimum EBITDA. Holdings and its Subsidiaries on a consolidated basis shall have, at the end of each Fiscal Quarter set forth below, Minimum EBITDA for the 12-Fiscal Month period then ended calculated of not less than the following:

Period	Minimum EBITDA
June 30, 2008	\$14,500,000
September 30, 2008	\$14,750,000
December 31, 2008	\$15,000,000
March 31, 2009	\$15,500,000
June 30, 2009	\$15,750,000
September 30, 2009	\$16,000,000
December 31, 2009	\$16,000,000
March 31, 2010	\$16,250,000
June 30, 2010	\$16,250,000
September 30, 2010	\$16,500,000
December 31, 2010	\$16,500,000
March 31, 2011 and each Fiscal Quarter ending thereafter	\$16,750,000

3.13. On the New Life Acquisition Effective Date, Section 7.1(b) of the Credit Agreement shall be amended and restated in its entirety to read as follows:

"Minimum EBITDA. Holdings and its Subsidiaries on a consolidated basis shall have, at the end of each Fiscal Quarter set forth below, Minimum EBITDA for the 12-Fiscal Month period then ended calculated of not less than the following:

Period	Minimum EBITDA
June 30, 2008	\$14,750,000
September 30, 2008	\$15,250,000
December 31, 2008	\$15,250,000
March 31, 2009	\$16,000,000

June 30, 2009	\$16,250,000
September 30, 2009	\$16,500,000
December 31, 2009	\$16,500,000
March 31, 2010	\$16,500,000
June 30, 2010	\$16,750,000
September 30, 2010	\$17,000,000
December 31, 2010	\$17,000,000
March 31, 2011 and each Fiscal Quarter ending thereafter	\$17,250,000

3.14. On the Fifth Amendment Effective Date, <u>Schedule 2</u> to <u>Annex E</u> of the Credit Agreement shall be amended to include the following new paragraph in the column underneath the word <u>Less</u>: and immediately prior to the word <u>Subtotal</u>:

"at such time after the cash consideration paid by the Borrower in connection with the Down East Acquisition exceeds \$1,500,000, an amount equal to any unfinanced cash payments with respect to the Down East Purchase Agreement, such amount not to exceed \$125,000 in the aggregate; provided that the term unfinanced when used in this paragraph shall mean any such payment that the Borrower has not directly or indirectly drawn Revolving Loans (or used other Indebtedness) to finance such payment with the determination of whether any such payment is unfinanced being acknowledged and consented to by Agent such consent not to be unreasonably withheld."

3.15. On the New Life Acquisition Effective Date, <u>Schedule 2</u> to <u>Annex E</u> of the Credit Agreement shall be amended to include the following new paragraph in the column underneath the word <u>Less</u>: and immediately prior to the word <u>Subtotal</u>:

"at such time after the cash consideration paid by the Borrower in connection with the New Life Acquisition exceeds \$1,200,000, an amount equal to any unfinanced cash payments with respect to the New Life Purchase Agreement, such amount not to exceed \$300,000 in the aggregate; provided that the term unfinanced when used in this paragraph shall mean any such payment that the Borrower has not directly or indirectly drawn Revolving Loans (or used other Indebtedness) to finance such payment with the determination of whether any such payment is unfinanced being acknowledged and consented to by Agent such consent not to be unreasonably withheld."

Section 4. Representations and Warranties. To induce Agent and Lenders to execute this Amendment, each Loan Party hereby represents and warrants to Agent and Lenders as follows:

(a) the execution, delivery and performance of each of this Amendment and the Down East Acquisition Documents have been duly authorized by all requisite action of each Loan Party thereto, and each of this Amendment and the Down East Acquisition Documents constitutes the legal, valid and binding obligation of such Person, enforceable against such Person in accordance with its terms, subject to bankruptcy, insolvency and similar laws affecting the enforceability of creditors' rights generally and to general principles of equity; and

(b) each of the representations and warranties in the Credit Agreement are true and correct in all material respects with the same effect as though made on and as of the date hereof (except, in each case, to the extent stated to relate to an earlier date, in which case such representation or warranty shall have been true and correct on and as of such earlier date) and no Event of Default or Default exists thereunder or would exist after giving effect to this Amendment or the Down East Acquisition.

Section 5. <u>Conditions Precedent</u>. This Amendment shall be effective as of the date first set forth above, subject to the satisfaction of the following conditions precedent (the date of such satisfaction being the "<u>Fifth Amendment Effective Date</u>") (which conditions, for purposes of clarification, shall not include those set forth on <u>Annex C</u> attached hereto):

- 5.1. Execution and Delivery. Borrower, each of the other Loan Parties, Agent and the Requisite Lenders shall have executed and delivered this Agreement; the Borrower shall have executed and delivered an Officer's Certificate in the form of Exhibit A attached hereto; each Loan Party shall have executed and delivered the Consent and Reaffirmation in the form of Exhibit B attached hereto.
- 5.2. <u>Covenant Compliance</u>. After giving pro forma effect to the Down East Acquisition to and the incurrence of the Delayed Draw Term Loans in connection therewith, Borrower is in compliance with each of the covenants set forth in <u>Section 7.1</u>.
- 5.3. Letter of Direction. Agent shall have received a duly executed letter of direction from Borrower addressed to Agent, on behalf of itself and the Lenders directing the disbursement of the proceeds of the Delayed Draw Term Loans to fund the Down East Acquisition.
- 5.4. <u>Insurance</u>. Agent shall have received evidence of the existence of insurance required to be maintained pursuant to the Credit Agreement, together with evidence that the Agent has been named as a lender's loss payee and an additional insured on all related insurance policies to the extent there is any change in such insurance in effect prior to the Fifth Amendment Effective Date as a result of or in connection with the Down East Acquisition.
- 5.5. Filings, Registrations and Recordings. The Agent shall have received each document (including Uniform Commercial Code financing statements) required by Section 5.7(c) of the Credit Agreement as may be necessary or desirable in order to create in favor of the Agent, for the benefit of the Lenders, a perfected Lien on the Collateral acquired in connection with the Down East Acquisition prior to any other Liens (subject only to Permitted Encumbrances).
- 5.6. <u>Due Diligence</u>. Agent, Lenders and their counsel and advisors shall have completed their legal, accounting and financial due diligence review (including, without limitation, lien searches on Down East) of Down East and the Down East Acquisition, the results of which review shall be reasonably satisfactory to Agent and Requisite Lenders.

- 5.7. <u>Material Adverse Effect</u>. There shall have occurred no Material Adverse Effect upon the financial condition, operations, assets, business or properties of (i) the Loan Parties or Down East, taken as a whole, since December 31, 2007; or (ii) Down East since December 31, 2007.
- 5.8. <u>Down East Acquisition Documents</u>. On or prior to the Fifth Amendment Effective Date, each of the Down East Acquisition Documents shall have been duly authorized by each of the Loan Parties party thereto and executed and delivered by each of the respective parties thereto and shall be in full force and effect and shall not have been amended or modified in any manner adverse to the Lenders except for such amendments and modifications, if any, as may be reasonably satisfactory to Agent and the Requisite Lenders. The Down East Acquisition shall have been, or shall be contemporaneously herewith, consummated substantially in accordance with the Down East Acquisition Documents (as amended in accordance with the immediately preceding sentence) and in accordance with all applicable law, except as may have been consented to in writing by Agent. Agent shall have received a complete copy of the fully executed Down East Acquisition Documents, certified by the Borrower as being true, complete and correct.
- 5.9. Approvals. All necessary governmental and third party approvals and/or consents in connection with the Down East Acquisition and otherwise referred to herein or therein shall have been obtained and remain in effect, and all applicable waiting periods shall have expired without any action being taken by any competent authority which restrains, prevents or imposes materially adverse conditions upon the consummation of all or any part of the Down East Acquisition, in each case as of the Fifth Amendment Effective Date. Additionally, there shall not exist any judgment, order, injunction or other restraint pending or notified prohibiting or imposing material adverse conditions upon all or any part of the Down East Acquisition or the transactions contemplated by this Amendment as of the Fifth Amendment Effective Date.
  - 5.10. No Defaults. After giving effect to this Amendment, no Event of Default or Default under the Credit Agreement shall have occurred and be continuing.
- 5.11. Representations and Warranties. After giving effect to this Amendment, the representations and warranties of the Loan Parties contained in this Amendment, the Credit Agreement and the other Loan Documents shall be true and correct in all material respects as of the Fifth Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty shall be true and correct in all material respects as of such earlier date.
- 5.12. Secretary's Certificate. Agent shall have received from each Loan Party a certificate signed by the secretary or an assistant secretary of such Loan Party, dated as of the Fifth Amendment Effective Date, as to the incumbency and signature of the officers of such Loan Party executing any Loan Document and any certificate or other document or instrument to be delivered pursuant hereto by or on behalf of such Loan Party, together with evidence of the incumbency of such Secretary or Assistant Secretary, as the case may be and certifying that there have been no changes to (i) the resolutions of such Loan Party authorizing and approving, among

other things, the execution, delivery and performance of this Amendment and the other Loan Documents and (ii) the organizational documents of such Loan Party in each case since the date on which they were most recently delivered to Agent.

- 5.13. Good Standing Certificates. Agent shall have received good standing certificates for each Loan Party from their respective jurisdictions of incorporation or organization.
- 5.14. Addus North Carolina Secretary's Certificate. Agent shall have received from Addus North Carolina a certificate signed by the secretary or an assistant secretary of such corporation, dated as of the Fifth Amendment Effective Date, as to (i) the incumbency and signature of the officers of such corporation executing any Loan Document and any certificate or other document or instrument to be delivered pursuant hereto by or on behalf of such corporation, (ii) the resolutions of such corporation authorizing and approving, among other things, the execution and delivery of the Loan Documents, the Down East Acquisition Documents and the other documents and instruments provided for therein and the consummation of the transactions contemplated hereby and thereby, (ii) the organizational documents of such corporation, and (iii) the good standing certificate for such corporation from its jurisdiction of incorporation.
- 5.15. <u>Fees</u>. Borrower shall have paid all reasonable and documented costs, fees and expenses (including, without limitation, reasonable legal fees and expenses of Winston & Strawn LLP) of Agent.
  - 5.16. Opinions of Counsel. Agent shall have received opinions of counsel for each Loan Party, including local counsel reasonably requested by the Agent.
- 5.17. Other Matters. Agent shall have received such other instruments and documents as Agent or the Requisite Lenders may reasonably request in connection with the execution of this Amendment, and all such instruments and documents shall be reasonably satisfactory in form and substance to Agent.

Notwithstanding the foregoing, the conditions precedent with respect to the New Life Acquisition shall be those set forth on Annex C attached hereto.

#### Section 6. Miscellaneous.

- 6.1. Effect of Amendment. The execution, delivery and effectiveness of this Amendment shall not operate as a waiver of any right, power or remedy of Agent or any Lender under the Credit Agreement or any Loan Document, or constitute a waiver of any provision of the Credit Agreement or any Loan Document, except as specifically set forth herein.
- 6.2. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and by the different parties on separate counterparts, and each such counterpart shall be deemed to be an original, but all such counterparts shall together constitute but one and the same instrument. Delivery of the executed counterpart of this Amendment by telecopy or electronic mail shall be as effective as delivery of a manually executed counterpart to this Amendment.

- 6.3. <u>Severability</u>. The illegality or unenforceability of any provision of this Amendment or any instrument or agreement required hereunder shall not in any way affect or impair the legality or enforceability of the remaining provisions of this Amendment or any instrument or agreement required hereunder.
  - 6.4. Captions. Section captions used in this Amendment are for convenience only, and shall not affect the construction of this Amendment.
- 6.5. Entire Agreement. This Amendment embodies the entire agreement and understanding among the parties hereto and supersedes all prior or contemporaneous agreements and understandings of such Persons, verbal or written, relating to the subject matter hereof.
- 6.6. <u>References</u>. Any reference to the Credit Agreement contained in any notice, request, certificate, or other document executed concurrently with or after the execution and delivery of this Amendment shall be deemed to include this Amendment unless the context shall otherwise require. Reference in any of this Amendment, the Credit Agreement or any other Loan Document to the Credit Agreement shall be a reference to the Credit Agreement as amended hereby and as further amended, modified, restated, supplemented or extended from time to time.

Section 7. <u>Governing Law</u>. THIS AMENDMENT SHALL BE A CONTRACT MADE UNDER AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF ILLINOIS APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE, WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES.

[signature pages follow]

Witness the due execution hereof by the respective duly authorized officers of the undersigned as of the date first above written.

## ADDUS HOLDING CORPORATION

## ADDUS MANAGEMENT CORPORATION

By: /s/ Simon Bachleda

Name: Simon Bachleda

Title: Vice President and Secretary

## ADDUS HEALTHCARE, INC.

By: /s/ Mark S. Heaney

Name: Mark S. Heaney

Title: President and Chief Executive Officer

FORT SMITH HOME HEALTH AGENCY, INC.

LITTLE ROCK HOME HEALTH AGENCY, INC.

LOWELL HOME HEALTH AGENCY, INC.

## PHC ACQUISITION CORPORATION

## PROFESSIONAL RELIABLE NURSING SERVICE, INC.

By: /s/ Mark S. Heaney
Name: Mark S. Heaney
Title: Vice President

## BENEFITS ASSURANCE CO., INC.

/s/ David W. Stasiewicz

Name: David W. Stasiewicz

Title: Chief Financial Officer and Secretary

## FREEPORT FINANCIAL LLC, as Agent

By: /s/ Chad Blakeman
Title: Sr. Managing Director

## FREEPORT LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman
Title: Sr. Managing Director

## FREEPORT OFFSHORE LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman
Title: Sr. Managing Director

## FREEPORT LOAN TRUST 2006-1, as a Lender

By: /s/ Chad Blakeman

Title: Sr. Managing Director

# FIFTH THIRD BANK, CHICAGO (A Michigan Banking Company)

By: /s/ Michael E. May
Name: Michael E. May

Title: Vice President

CF BLACKBURN LLC							
	$\sim$ T	TOI	•	DI	TIDAT	TT	•

By: /s/ Dennis M. Hansen
Name: Dennis M. Hansen

Title: V.P.

### OFFICER'S CERTIFICATE

Reference is made to that certain Consent and Fifth Amendment to Credit Agreement, dated as of June 9, 2008 (the "Fifth Amendment"; capitalized terms used herein and not otherwise defined shall have the meanings specified thereto in that certain Credit Agreement dated as of September 19, 2006 among the Loan Parties party thereto, the Agent and the Lenders party thereto (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007, that certain Consent and Second Amendment to Credit Agreement dated as of October 15, 2007, that certain Consent and Third Amendment to Credit Agreement dated as of November 13, 2007, that certain Consent and Fourth Amendment to Credit Agreement dated as of April 1, 2008 and as further amended, restated or otherwise modified from time to time including by the Fifth Amendment, the "Credit Agreement"), among the Borrower, certain other parties, the financial institutions party thereto and Freeport Financial LLC, as Agent. The undersigned hereby certifies that he is the duly authorized and elected President & Chief Executive Officer of the Borrower and that, as of the date hereof:

- 1. As of the Fifth Amendment Effective Date, after giving pro forma effect to the Down East Acquisition to and the incurrence of the Delayed Draw Term Loans in connection therewith, Borrower is in compliance with each of the covenants set forth in Section 7.1.
- 2. No Material Adverse Effect has occurred upon the financial condition, operations, assets, business or properties of (i) the Loan Parties or Down East, taken as a whole, since December 31, 2007; or (ii) Down East since December 31, 2007.
- 3. After giving effect to the Fifth Amendment, no Event of Default or Default under the Credit Agreement has occurred or is continuing.
- 4. After giving effect to the Fifth Amendment, the representations and warranties of the Loan Parties contained in the Fifth Amendment, the Credit Agreement and the other Loan Documents are true and correct in all material respects as of the Fifth Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty was true and correct in all material respects as of such earlier date.
- 5. The Loan Parties have consummated the Down East Acquisition concurrently with the Fifth Amendment substantially in accordance with the Down East Acquisition Documents and in accordance with all applicable law, except as may have been consented to in writing by Agent.

6.	Attached hereto as Exhibit A are true, complete and correct copies of the fully executed Down East Acquisition Documents.					
	[Remainder of Page Intentionally Blank]					

IN WITNESS WHEREOF, I have hereunto set my hand this day of June, 2008.	
	ADDUS HEALTHCARE, INC.
	By:  Name: Mark S. Heaney Title: President & Chief Executive Officer

**EXHIBIT B** 

# **CONSENT AND REAFFIRMATION (FIFTH AMENDMENT)**

Each of the undersigned ("Loan Parties") hereby (i) acknowledges receipt of a copy of the Consent and Fifth Amendment to Credit Agreement dated as of June 9, 2008 (the "Fifth Amendment"); (ii) consents to the execution and delivery thereof by the other Loan Parties; (iii) agrees to be bound thereby; (iv) affirms that nothing contained therein shall modify in any respect whatsoever its guaranty of the obligations of the Loan Parties to Agent and Lenders (the "Guarantee") and (v) reaffirms that the Guarantee and the other Collateral Documents executed by such Person are and shall continue to remain in full force and effect. Although each of the Loan Parties has been informed of the matters set forth herein and in the Fifth Amendment and has acknowledged and agreed to same, such Loan Parties understand that Agent and Lenders have no obligation to inform any of the Loan Parties of such matters in the future or to seek any of the Loan Parties' acknowledgment or agreement to future amendments or waivers, and nothing herein shall create such a duty.

This Consent and Reaffirmation shall be governed by and construed in accordance with the laws of the State of Illinois, without reference to principles of conflicts of law.

In witness whereof, each of the undersigned has executed this Consent and Reaffirmation on and as of the date of such Fifth Amendment.

### ADDUS HOLDING CORPORATION

# ADDUS MANAGEMENT CORPORATION

By:

Name: Simon Bachleda

Title: Vice President and Secretary

# ADDUS HEALTHCARE, INC.

By:

Name: Mark S. Heaney

Title: President and Chief Executive Officer

ADDUS HEALTHCARE (IDAHO), INC.

ADDUS HEALTHCARE (NEVADA), INC.

ADDUS HEALTHCARE (NEW JERSEY), INC.

FORT SMITH HOME HEALTH AGENCY, INC.

LITTLE ROCK HOME HEALTH AGENCY, INC.

LOWELL HOME HEALTH AGENCY, INC.

PHC ACQUISITION CORPORATION

PROFESSIONAL RELIABLE NURSING SERVICE, INC.

By:

Name: Mark S. Heaney Title: Vice President

BENEFITS ASSURANCE CO., INC.

By:

Name: David W. Stasiewicz

Title: Chief Financial Officer and Secretary

### OFFICER'S CERTIFICATE

Reference is made to that certain Consent and Fifth Amendment to Credit Agreement, dated as of June 9, 2008 (the "Fifth Amendment"; capitalized terms used herein and not otherwise defined shall have the meanings specified thereto in that certain Credit Agreement dated as of September 19, 2006 among the Loan Parties party thereto, the Agent and the Lenders party thereto (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007, that certain Consent and Second Amendment to Credit Agreement dated as of October 15, 2007, that certain Consent and Third Amendment to Credit Agreement dated as of November 13, 2007, that certain Consent and Fourth Amendment to Credit Agreement dated as of April 1, 2008 and as further amended, restated or otherwise modified from time to time including by the Fifth Amendment, the "Credit Agreement"), among the Borrower, certain other parties, the financial institutions party thereto and Freeport Financial LLC, as Agent. The undersigned hereby certifies that he is the duly authorized and elected President & Chief Executive Officer of the Borrower and that, as of the date hereof:

- 1. As of the New Life Acquisition Effective Date, after giving pro forma effect to the New Life Acquisition to and the incurrence of the Delayed Draw Term Loans in connection therewith, Borrower is in compliance with each of the covenants set forth in Section 7.1.
- 2. No Material Adverse Effect shall have occurred upon the financial condition, operations, assets, business or properties of (i) the Loan Parties or New Life, taken as a whole, since December 31, 2007; or (ii) New Life since December 31, 2007.
- 3. After giving effect to the Fifth Amendment and the New Life Acquisition, no Event of Default or Default under the Credit Agreement has occurred or is continuing.
- 4. After giving effect to the Fifth Amendment and the New Life Acquisition, the representations and warranties of the Loan Parties contained in the Fifth Amendment, the Credit Agreement and the other Loan Documents are true and correct in all material respects as of the New Life Acquisition Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty was true and correct in all material respects as of such earlier date.
- 5. The Loan Parties have consummated the New Life Acquisition substantially in accordance with the New Life Acquisition Documents and in accordance with all applicable law, except as may have been consented to in writing by Agent.

(	5.	Attached hereto as Exhibit A are true, complete and correct copies of the fully executed New Life Acquisition Documents.					
		[Remainder of Page Intentionally Blank]					

WITNESS WHEREOF, I have hereunto set my hand this day of	, 2008.
	ADDUS HEALTHCARE, INC.
	Name: Mark S. Heaney Title: President & Chief Executive Officer

# ANNEX A FIFTH AMENDMENT EFFECTIVE DATE DELAYED DRAW FUNDING AMOUNTS

FIFTH AMENDMENT EFFECTIVE DATE DELAYED DRAW FUNDING

	AMOUNT		
	\$	%	
Lender			
Freeport Loan Fund LLC	\$ 514,	285.71 34.2857143%	
CF Blackburn LLC	0	0	
Fifth Third Bank (Chicago)	\$ 985,	714.29 65.7142857%	
Totals	\$ 1,500,	000.00 100%	

# ANNEX B NEW LIFE ACQUISITION EFFECTIVE DATE DELAYED DRAW FUNDING AMOUNTS

	NEW LIFE AC EFFECTIVE DA DRAW FUNDIN	TE DELAYED
	\$	%
Lender		
Freeport Loan Fund LLC	\$ 411,428.57	34.2857143%
CF Blackburn LLC	0	0
Fifth Third Bank (Chicago)	\$ 788,571.43	65.7142857%
Totals	\$1,200,000.00	100%

Section 1. The date upon which all conditions set forth on this <u>Annex C</u> are satisfied shall be the "<u>New Life Acquisition Effective Date</u>" (which date, for purpose of clarification, shall be a date that is not more than 30 days following the Fifth Amendment Effective Date):

- 1.1. Execution and Delivery. The Borrower shall have executed and delivered an Officer's Certificate in the form of Exhibit C attached to the Fifth Amendment.
- 1.2. <u>Covenant Compliance</u>. After giving pro forma effect to the New Life Acquisition to and the incurrence of the Delayed Draw Term Loans in connection therewith, Borrower is in compliance with each of the covenants set forth in <u>Section 7.1</u>.
- 1.3. Enforceability. The execution, delivery and performance of the New Life Acquisition Documents have been duly authorized by all requisite action of Addus Nevada, and the New Life Acquisition Documents constitutes the legal, valid and binding obligation of Addus Nevada, enforceable against Addus Nevada in accordance with its terms, subject to bankruptcy, insolvency and similar laws affecting the enforceability of creditors' rights generally and to general principles of equity.
- 1.4. Representations and Warranties. Each of the representations and warranties in the Credit Agreement are true and correct in all material respects with the same effect as though made on and as of the date hereof (except, in each case, to the extent stated to relate to an earlier date, in which case such representation or warranty shall have been true and correct on and as of such earlier date) and no Event of Default or Default exists thereunder or would exist after giving effect to the New Life Acquisition.
- 1.5. Letter of Direction. Agent shall have received a duly executed letter of direction from Borrower addressed to Agent, on behalf of itself and the Lenders directing the disbursement of the proceeds of the Delayed Draw Term Loans to fund the New Life Acquisition.
- 1.6. Filings, Registrations and Recordings. The Agent shall have received each document (including Uniform Commercial Code financing statements) required by Section 5.7(c) of the Credit Agreement as may be necessary or desirable in order to create in favor of the Agent, for the benefit of the Lenders, a perfected Lien on the Collateral acquired in connection with the New Life Acquisition prior to any other Liens (subject only to Permitted Encumbrances).
- 1.7. <u>Due Diligence</u>. Agent, Lenders and their counsel and advisors shall have completed their legal, accounting and financial due diligence review (including, without limitation, lien searches on New Life) of New Life and the New Life Acquisition, the results of which review shall be reasonably satisfactory to Agent.

- 1.8. <u>Material Adverse Effect</u>. There shall have occurred no Material Adverse Effect upon the financial condition, operations, assets, business or properties of (i) the Loan Parties or New Life, taken as a whole, since December 31, 2007; or (ii) New Life since December 31, 2007.
- 1.9. New Life Acquisition Documents. On or prior to the New Life Acquisition Effective Date, each of the New Life Acquisition Documents shall have been duly authorized by Addus Nevada and executed and delivered by each of the respective parties thereto and shall be in full force and effect and shall not have been amended or modified in any manner adverse to the Lenders except for such amendments and modifications, if any, as may be reasonably satisfactory to Agent. The New Life Acquisition shall have been, or shall be contemporaneously herewith, consummated substantially in accordance with the New Life Acquisition Documents (as amended in accordance with the immediately preceding sentence) and in accordance with all applicable law, except as may have been consented to in writing by Agent. Agent shall have consented to the terms and conditions of the New Life Acquisition and received a complete copy of the fully executed New Life Acquisition Documents, certified by the Borrower as being true, complete and correct.
- 1.10. <u>Approvals</u>. All necessary governmental and third party approvals and/or consents in connection with the New Life Acquisition and otherwise referred to herein or therein shall have been obtained and remain in effect, and all applicable waiting periods shall have expired without any action being taken by any competent authority which restrains, prevents or imposes materially adverse conditions upon the consummation of all or any part of the New Life Acquisition, in each case as of the New Life Acquisition Effective Date. Additionally, there shall not exist any judgment, order, injunction or other restraint pending or notified prohibiting or imposing material adverse conditions upon all or any part of the New Life Acquisition or the transactions contemplated thereby as of the New Life Acquisition Effective Date.
- 1.11. No Defaults. After giving effect to the Fifth Amendment and the New Life Acquisition, no Event of Default or Default under the Credit Agreement shall have occurred and be continuing.
- 1.12. Secretary's Certificate. Agent shall have received from Addus Nevada a certificate signed by the secretary or an assistant secretary of such corporation, dated as of the closing of the New Life Acquisition, as to (i) the incumbency and signature of the officers of such corporation executing any New Life Acquisition Document by or on behalf of such corporation, (ii) the resolutions of such corporation authorizing and approving, among other things, the execution and delivery of the New Life Acquisition Documents, (ii) the organizational documents of such corporation, and (iii) the good standing certificate for such corporation from its jurisdiction of incorporation.
- 1.13. <u>Fees</u>. Borrower shall have paid all reasonable and documented costs, fees and expenses (including, without limitation, reasonable legal fees and expenses of Winston & Strawn LLP) of Agent.

1.14. Other Matte Amendment and the	ers. Agent shall have re he New Life Acquisition	ceived such other instrum on Documents, and all suc	nents and documents a	as Agent may reasona cuments shall be reason	oly request in connect	ion with the execution of and substance to A

# CONSENT AND SIXTH AMENDMENT TO CREDIT AGREEMENT

CONSENT AND SIXTH AMENDMENT TO CREDIT AGREEMENT (this "Amendment"), dated as of September 25, 2008, among ADDUS HEALTHCARE, INC., an Illinois corporation ("Borrower"), the other persons signatory hereto as "Loan Parties", FREEPORT FINANCIAL LLC, a Delaware limited liability company ("Agent") and the Lenders signatory hereto. Terms not defined herein have the meanings given to them in the Credit Agreement (as hereinafter defined).

#### **RECITALS**

- A. Borrower, the Loan Parties, the Lenders signatory thereto and Agent are party to that certain Credit Agreement dated as of September 19, 2006 (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007, that certain Consent and Second Amendment to Credit Agreement dated as of October 15, 2007, that certain Consent and Fourth Amendment to Credit Agreement dated as of November 13, 2007, that certain Consent and Fourth Amendment to Credit Agreement dated as of June 9, 2008 and as further amended, restated or otherwise modified including by this Amendment, the "Credit Agreement").
- B. Borrower, the Loan Parties, the Lenders and Agent are entering into this Amendment in connection with the acquisition (the "Awakened Alternatives Acquisition") by Addus HealthCare (Indiana), Inc., a Delaware corporation and a wholly-owned Subsidiary of the Borrower ("Addus Indiana"), of substantially all of the assets of Awakened Alternatives Enterprises, LLC, an Illinois limited liability company ("Awakened Alternatives"), for an aggregate purchase price not to exceed \$450.000.
  - C. Agent, Lenders and the Loan Parties are willing to enter into this Amendment upon the terms and conditions set forth below.

NOW THEREFORE, in consideration of the matters set forth in the recitals and the covenants and provisions herein set forth, and other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

#### **AGREEMENT**

- Section 1. Definitions. Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed thereto in the Credit Agreement.
- Section 2. Consent. Agent and the Lenders hereby consent to the following:
- 2.1. Addus Indiana (i) entering into (x) an Asset Purchase Agreement, dated as of even date herewith (the "Awakened Alternatives Purchase Agreement"), by and between Addus Indiana, Awakened Alternatives and certain Members (as defined in the Awakened Alternatives

Purchase Agreement) of Awakened Alternatives party thereto, for the purpose of acquiring substantially all of the assets of Awakened Alternatives and (y) the other agreements, instruments and documents for the purpose of consummating the Awakened Alternatives Acquisition, in each case on terms and conditions reasonably satisfactory to the Agent (collectively, the "Awakened Alternatives Acquisition Documents"); and (ii) consummating the Awakened Alternatives Acquisition on the Sixth Amendment Effective Date for an aggregate purchase price not to exceed \$300,000 paid in cash on the closing date thereof pursuant to the terms of the Awakened Alternatives Acquisition Documents, plus a deferred purchase price in an amount not to exceed \$50,000 paid pursuant to the Awakened Alternatives Purchase Agreement, plus the issuance of the Awakened Alternatives Note; provided that neither such deferred purchase price paid pursuant to the Awakened Alternatives Purchase Agreement nor any payments on the Awakened Alternatives Note shall be paid in the event that any Default or Event of Default under any of Sections 7.1 or 8.1(a) of the Credit Agreement has occurred and is continuing or would result therefrom or the Borrower does not have the minimum Borrowing Availability required by Section 6.17 of the Credit Agreement after giving effect to such payment.

Section 3. Amendment to the Credit Agreement. As of the Sixth Amendment Effective Date, the Credit Agreement is hereby amended as follows:

3.1. Section 1.1 of the Credit Agreement is hereby amended by adding the following new definitions thereto:

Awakened Alternatives has the meaning set forth in the Recitals to the Sixth Amendment.

Awakened Alternatives Acquisition has the meaning set forth in the Recitals to the Sixth Amendment.

Awakened Alternatives Acquisition Documents has the meaning set forth in Section 2 of the Sixth Amendment.

Awakened Alternatives Note means that certain 6% Junior Subordinated Promissory Note by Borrower payable to Awakened Alternatives in the original principal amount of \$100,000.

Sixth Amendment means that certain Consent and Sixth Amendment to Credit Agreement dated as of September 25, 2008 among the Loan Parties, Agent and the Lenders.

Sixth Amendment Effective Date has the meaning set forth in the Sixth Amendment.

Sixth Amendment Reaffirmation of Collateral Documents means the Consent and Reaffirmation (Sixth Amendment), dated as of September 25, 2008 of the Loan Parties signatory thereto, in respect of the Collateral Documents.

- 3.2. On the Sixth Amendment Effective Date, Section 2.1(a)(v) of the Credit Agreement shall be amended and restated in its entirety and replaced with the following new Section 2.1(a)(v):
- (v) Borrower shall repay the Term Loans through periodic payments on the dates and in the amounts indicated below ("Scheduled Installments"), and in any event the entire remaining principal balance shall be repaid on the Commitment Termination Date.

#### Term Loan

Date		Scheduled Installment
September 30, 2008	\$	1,197,000
December 31, 2008	\$	1,620,500
March 31, 2009	\$	1,620,500
June 30, 2009	\$	1,620,500
September 30, 2009	\$	1,620,500
December 31, 2009	\$	2,044,000
March 31, 2010	\$	2,044,000
June 30, 2010	\$	2,044,000
September 30, 2010	\$	2,044,000
December 31, 2010	\$	2,467,500
March 31, 2011	\$	2,467,500
June 30, 2011	\$	2,467,500
September 19, 2011	\$	30,170,500
		the outstanding
	p	rincipal balance
		of Term Loans
		outstanding on such date
		such date

The above scheduled installment amounts reflect the incurrence by the Borrower of the Delayed Draw Term Loans funded on the Sixth Amendment Effective Date as set forth on Annex A attached hereto and shall thereafter be increased in the manner set forth on Annex B to the Second Amendment to the extent any further Delayed Draw Term Loans are funded. The final installment payment shall in all events equal the entire remaining principal balance of the Term Loan (including any remaining principal balance of such Delayed Draw Term Loans). Amounts borrowed under this Section 2.1(a) and repaid may not be reborrowed.

At the request of the applicable Lender, the Term Loans shall be evidenced by promissory notes substantially in the form of Exhibit 2.1(a) (as amended, modified, extended, substituted or replaced from time to time, each a "Term Note" and, collectively, the "Term Notes"), and Borrower shall execute and deliver a Term Note to each such Term Lender. Each Term Note shall represent the obligation of Borrower to pay the amount of the applicable Term Lender's Term Loan Commitment, together with interest thereon."

3.3. Section 6.1 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause  $(\underline{q})$  thereof, deleting the "." at the end of clause  $(\underline{q})$  thereof and replacing it with "; and" and by adding the following new clause  $(\underline{r})$ :

"(r) Indebtedness in respect of the Awakened Alternatives Purchase Agreement and Awakened Alternatives Note."

- 3.4. Section 6.3 of the Credit Agreement is hereby amended by deleting the "and" at the end of clause (q) thereof, deleting the "." at the end of clause (r) thereof and replacing it with "; and" and by adding the following new clauses (s):
- "(s) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Awakened Alternatives Acquisition as of the Sixth Amendment Effective Date pursuant to the Sixth Amendment Acquisition Documents."
- 3.5. <u>Clause (d)</u> of <u>Section 6.6</u> of the Credit Agreement is hereby amended by deleting the "and" before clause (vii) thereof and adding the following new language immediately at the end thereof:
- "(viii) Borrower or a wholly-owned Subsidiary of Borrower may consummate the Awakened Alternatives Acquisition on the Sixth Amendment Effective Date pursuant to the Awakened Alternatives Acquisition Documents and all applicable law, provided that the Awakened Alternatives Acquisition Documents shall be in full force and effect and provided further that the Agent shall have received a complete copy of the fully executed Awakened Alternatives Acquisition Documents, certified by the Borrower as being true, complete and correct."
  - 3.6. Section 6.16 of the Credit Agreement is amended and restated in its entirety to read as follows:
- "None of the Loan Parties shall change or amend the terms of the Acquisition Note, Silver State Note, Vegas Earn Out Agreement, Idaho Earn Out Agreement, Idaho Note, SuCasa Note, Down East Purchase Agreement, New Life Purchase Agreement, Awakened Alternatives Purchase Agreement or Awakened Alternatives Note if such change or amendment would be adverse in any material respect to the rights or interests of the Loan Parties, the Agent or the Lenders."
- Section 4. Representations and Warranties. To induce Agent and Lenders to execute this Amendment, each Loan Party hereby represents and warrants to Agent and Lenders as follows:
- (a) the execution, delivery and performance of each of this Amendment and the Awakened Alternatives Acquisition Documents have been duly authorized by all requisite action of each Loan Party thereto, and each of this Amendment and the Awakened Alternatives Acquisition Documents constitutes the legal, valid and binding obligation of such Person, enforceable against such Person in accordance with its terms, subject to bankruptcy, insolvency and similar laws affecting the enforceability of creditors' rights generally and to general principles of equity; and
- (b) each of the representations and warranties in the Credit Agreement are true and correct in all material respects with the same effect as though made on and as of the date hereof (except, in each case, to the extent stated to relate to an earlier date, in which case such

representation or warranty shall have been true and correct on and as of such earlier date) and no Event of Default or Default exists thereunder or would exist after giving effect to this Amendment or the Awakened Alternatives Acquisition.

Section 5. <u>Conditions Precedent</u>. This Amendment shall be effective as of the date first set forth above, subject to the satisfaction of the following conditions precedent (the date of such satisfaction being the "<u>Sixth Amendment Effective Date</u>"):

- 5.1. Execution and Delivery. Borrower, each of the other Loan Parties, Agent and the Requisite Lenders shall have executed and delivered this Agreement; the Borrower shall have executed and delivered an Officer's Certificate in the form of Exhibit A attached hereto; each Loan Party shall have executed and delivered the Consent and Reaffirmation in the form of Exhibit B attached hereto.
- 5.2. <u>Covenant Compliance</u>. After giving pro forma effect to the Awakened Alternatives Acquisition to and the incurrence of the Delayed Draw Term Loans in connection therewith, Borrower is in compliance with each of the covenants set forth in <u>Section 7.1</u>.
- 5.3. Letter of Direction. Agent shall have received a duly executed letter of direction from Borrower addressed to Agent, on behalf of itself and the Lenders directing the disbursement of the proceeds of the Delayed Draw Term Loans to fund the Awakened Alternatives Acquisition.
- 5.4. <u>Insurance</u>. Agent shall have received evidence of the existence of insurance required to be maintained pursuant to the Credit Agreement, together with evidence that the Agent has been named as a lender's loss payee and an additional insured on all related insurance policies to the extent there is any change in such insurance in effect prior to the Sixth Amendment Effective Date as a result of or in connection with the Awakened Alternatives Acquisition.
- 5.5. Filings, Registrations and Recordings. The Agent shall have received each document (including Uniform Commercial Code financing statements) required by Section 5.7(c) of the Credit Agreement as may be necessary or desirable in order to create in favor of the Agent, for the benefit of the Lenders, a perfected Lien on the Collateral acquired in connection with the Awakened Alternatives Acquisition prior to any other Liens (subject only to Permitted Encumbrances).
- 5.6. <u>Due Diligence</u>. Agent, Lenders and their counsel and advisors shall have completed their legal, accounting and financial due diligence review (including, without limitation, lien searches on Awakened Alternatives) of Awakened Alternatives and the Awakened Alternatives Acquisition, the results of which review shall be reasonably satisfactory to Agent and the Requisite Lenders.
- 5.7. <u>Material Adverse Effect</u>. There shall have occurred no Material Adverse Effect upon the financial condition, operations, assets, business or properties of (i) the Loan Parties or Awakened Alternatives, taken as a whole, since December 31, 2007; or (ii) Awakened Alternatives since December 31, 2007.

- 5.8. Awakened Alternatives Acquisition Documents. On or prior to the Sixth Amendment Effective Date, each of the Awakened Alternatives Acquisition Documents shall have been duly authorized by each of the Loan Parties party thereto and executed and delivered by each of the respective parties thereto and shall be in full force and effect and shall not have been amended or modified in any manner adverse to the Lenders except for such amendments and modifications, if any, as may be reasonably satisfactory to Agent and the Requisite Lenders. The Awakened Alternatives Acquisition shall have been, or shall be contemporaneously herewith, consummated substantially in accordance with the Awakened Alternatives Acquisition Documents (as amended in accordance with the immediately preceding sentence) and in accordance with all applicable law, except as may have been consented to in writing by Agent. Agent shall have received a complete copy of the fully executed Awakened Alternatives Acquisition Documents, certified by the Borrower as being true, complete and correct.
- 5.9. Approvals. All necessary governmental and third party approvals and/or consents in connection with the Awakened Alternatives Acquisition and otherwise referred to herein or therein shall have been obtained and remain in effect, and all applicable waiting periods shall have expired without any action being taken by any competent authority which restrains, prevents or imposes materially adverse conditions upon the consummation of all or any part of the Awakened Alternatives Acquisition, in each case as of the Sixth Amendment Effective Date. Additionally, there shall not exist any judgment, order, injunction or other restraint pending or notified prohibiting or imposing material adverse conditions upon all or any part of the Awakened Alternatives Acquisition or the transactions contemplated by this Amendment as of the Sixth Amendment Effective Date.
  - 5.10. No Defaults. After giving effect to this Amendment, no Event of Default or Default under the Credit Agreement shall have occurred and be continuing.
- 5.11. Representations and Warranties. After giving effect to this Amendment, the representations and warranties of the Loan Parties contained in this Amendment, the Credit Agreement and the other Loan Documents shall be true and correct in all material respects as of the Sixth Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty shall be true and correct in all material respects as of such earlier date.
- 5.12. Secretary's Certificate. Agent shall have received from each Loan Party a certificate signed by the secretary or an assistant secretary of such Loan Party, dated as of the Sixth Amendment Effective Date, as to the incumbency and signature of the officers of such Loan Party executing any Loan Document and any certificate or other document or instrument to be delivered pursuant hereto by or on behalf of such Loan Party, together with evidence of the incumbency of such Secretary or Assistant Secretary, as the case may be and certifying that there have been no changes to (i) the resolutions of such Loan Party authorizing and approving, among other things, the execution, delivery and performance of this Amendment and the other Loan Documents and (ii) the organizational documents of such Loan Party in each case since the date on which they were most recently delivered to Agent.

- 5.13. <u>Good Standing Certificates</u>. Agent shall have received good standing certificates for each Loan Party from their respective jurisdictions of incorporation or organization.
- 5.14. Addus Indiana Secretary's Certificate. Agent shall have received from Addus Indiana a certificate signed by the secretary or an assistant secretary of such corporation, dated as of the Sixth Amendment Effective Date, as to (i) the incumbency and signature of the officers of such corporation executing any Loan Document and any certificate or other document or instrument to be delivered pursuant hereto by or on behalf of such corporation, (ii) the resolutions of such corporation authorizing and approving, among other things, the execution and delivery of the Loan Documents, the Awakened Alternatives Acquisition Documents and the other documents and instruments provided for therein and the consummation of the transactions contemplated hereby and thereby, (iii) the organizational documents of such corporation, and (iv) the good standing certificate for such corporation from its jurisdiction of incorporation.
- 5.15. <u>Fees</u>. Borrower shall have paid all reasonable and documented costs, fees and expenses (including, without limitation, reasonable legal fees and expenses of Winston & Strawn LLP) of Agent.
  - 5.16. Opinions of Counsel. Agent shall have received opinions of counsel for each Loan Party, including local counsel reasonably requested by the Agent.
- 5.17. Other Matters. Agent shall have received such other instruments and documents as Agent or the Requisite Lenders may reasonably request in connection with the execution of this Amendment, and all such instruments and documents shall be reasonably satisfactory in form and substance to Agent.

#### Section 6. Miscellaneous.

- 6.1. Effect of Amendment. The execution, delivery and effectiveness of this Amendment shall not operate as a waiver of any right, power or remedy of Agent or any Lender under the Credit Agreement or any Loan Document, or constitute a waiver of any provision of the Credit Agreement or any Loan Document, except as specifically set forth herein.
- 6.2. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and by the different parties on separate counterparts, and each such counterpart shall be deemed to be an original, but all such counterparts shall together constitute but one and the same instrument. Delivery of the executed counterpart of this Amendment by telecopy or electronic mail shall be as effective as delivery of a manually executed counterpart to this Amendment.
- 6.3. <u>Severability</u>. The illegality or unenforceability of any provision of this Amendment or any instrument or agreement required hereunder shall not in any way affect or impair the legality or enforceability of the remaining provisions of this Amendment or any instrument or agreement required hereunder.
  - 6.4. Captions. Section captions used in this Amendment are for convenience only, and shall not affect the construction of this Amendment.

- 6.5. Entire Agreement. This Amendment embodies the entire agreement and understanding among the parties hereto and supersedes all prior or contemporaneous agreements and understandings of such Persons, verbal or written, relating to the subject matter hereof.
- 6.6. <u>References</u>. Any reference to the Credit Agreement contained in any notice, request, certificate, or other document executed concurrently with or after the execution and delivery of this Amendment shall be deemed to include this Amendment unless the context shall otherwise require. Reference in any of this Amendment, the Credit Agreement or any other Loan Document to the Credit Agreement shall be a reference to the Credit Agreement as amended hereby and as further amended, modified, restated, supplemented or extended from time to time.
- Section 7. <u>Governing Law</u>. THIS AMENDMENT SHALL BE A CONTRACT MADE UNDER AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF ILLINOIS APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE, WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES.

[signature pages follow]

Witness the due execution hereof by the respective duly authorized officers of the undersigned as of the date first above written.

# ADDUS HOLDING CORPORATION

By: /s/ Simon Bachleda
Name: Simon Bachleda

Title: Vice President and Secretary

ADDUS HEALTHCARE, INC.

FORT SMITH HOME HEALTH AGENCY, INC.

LITTLE ROCK HOME HEALTH AGENCY, INC.

LOWELL HOME HEALTH AGENCY, INC.

PHC ACQUISITION CORPORATION

PROFESSIONAL RELIABLE NURSING SERVICE, INC.

BENEFITS ASSURANCE CO., INC.

By: /s/ Frank Leonard

Name: Frank Leonard

Title: Chief Financial Officer and Secretary

# FREEPORT FINANCIAL LLC, as Agent

By: /s/ Chad Blakeman

Title: Senior Managing Director

# FREEPORT LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman

Title: Senior Managing Director

# FREEPORT OFFSHORE LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman

Title: Senior Managing Director

# FREEPORT LOAN TRUST 2006-1, as a Lender

/s/ Chad Blakeman

By:

Title: Senior Managing Director

# FIFTH THIRD BANK, CHICAGO (A Michigan Banking Company)

By: /s/ Michael E. May
Name: Michael E. May

Title: Vice President

CF				

By: /s/ Angela D. Brown Name: Angela D. Brown
Title: Managing Director

### **OFFICER'S CERTIFICATE**

Reference is made to that certain Consent and Sixth Amendment to Credit Agreement, dated as of September 25, 2008 (the "Sixth Amendment"; capitalized terms used herein and not otherwise defined shall have the meanings specified thereto in that certain Credit Agreement dated as of September 19, 2006 among the Loan Parties party thereto, the Agent and the Lenders party thereto (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007, that certain Consent and Second Amendment to Credit Agreement dated as of October 15, 2007, that certain Consent and Third Amendment to Credit Agreement dated as of November 13, 2007, that certain Consent, Fourth Amendment to Credit Agreement dated as of April 1, 2008 and Fifth Amendment to Credit Agreement dated as of June 9, 2008 and as further amended, restated or otherwise modified from time to time including by the Sixth Amendment, the "Credit Agreement"), among the Borrower, certain other parties, the financial institutions party thereto and Freeport Financial LLC, as Agent. The undersigned hereby certifies that he is the duly authorized and elected President & Chief Executive Officer of the Borrower and that, as of the date hereof:

- 1. As of the Sixth Amendment Effective Date, after giving pro forma effect to the Awakened Alternatives Acquisition to and the incurrence of the Delayed Draw Term Loans in connection therewith, Borrower is in compliance with each of the covenants set forth in Section 7.1.
- 2. No Material Adverse Effect has occurred upon the financial condition, operations, assets, business or properties of (i) the Loan Parties or Awakened Alternatives, taken as a whole, since December 31, 2007; or (ii) Awakened Alternatives since December 31, 2007.
- 3. After giving effect to the Sixth Amendment, no Event of Default or Default under the Credit Agreement has occurred or is continuing.
- 4. After giving effect to the Sixth Amendment, the representations and warranties of the Loan Parties contained in the Sixth Amendment, the Credit Agreement and the other Loan Documents are true and correct in all material respects as of the Sixth Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty was true and correct in all material respects as of such earlier date.
- 5. The Loan Parties have consummated the Awakened Alternatives Acquisition concurrently with the Sixth Amendment substantially in accordance with the Awakened Alternatives Acquisition Documents and in accordance with all applicable law, except as may have been consented to in writing by Agent.

6.	Attached hereto as Exhibit A are true, complete and correct copies of the fully executed Awakened Alternatives Acquisition Documents.
	[Remainder of Page Intentionally Blank]

IN WITNESS WHEREOF, I have hereunto set my hand this	day of September, 2008.	
		ADDUS HEALTHCARE, INC.
		By: Name: Mark S. Heaney
		Name: Mark S. Heaney Title: President & Chief Executive Officer

**EXHIBIT B** 

# **CONSENT AND REAFFIRMATION (SIXTH AMENDMENT)**

Each of the undersigned ("Loan Parties") hereby (i) acknowledges receipt of a copy of the Consent and Sixth Amendment to Credit Agreement dated as of September 25, 2008 (the "Sixth Amendment"); (ii) consents to the execution and delivery thereof by the other Loan Parties; (iii) agrees to be bound thereby; (iv) affirms that nothing contained therein shall modify in any respect whatsoever its guaranty of the obligations of the Loan Parties to Agent and Lenders (the "Guarantee") and (v) reaffirms that the Guarantee and the other Collateral Documents executed by such Person are and shall continue to remain in full force and effect. Although each of the Loan Parties has been informed of the matters set forth herein and in the Sixth Amendment and has acknowledged and agreed to same, such Loan Parties understand that Agent and Lenders have no obligation to inform any of the Loan Parties of such matters in the future or to seek any of the Loan Parties' acknowledgment or agreement to future amendments or waivers, and nothing herein shall create such a duty.

This Consent and Reaffirmation shall be governed by and construed in accordance with the laws of the State of Illinois, without reference to principles of conflicts of law.

In witness whereof, each of the undersigned has executed this Consent and Reaffirmation on and as of the date of such Sixth Amendment.

### ADDUS HOLDING CORPORATION

By:

Name: Simon Bachleda

Title: Vice President and Secretary

ADDUS HEALTHCARE, INC.

ADDUS HEALTHCARE (NORTH CAROLINA), INC.

ADDUS HEALTHCARE (IDAHO), INC.

ADDUS HEALTHCARE (NEVADA), INC.

ADDUS HEALTHCARE (NEW JERSEY), INC.

FORT SMITH HOME HEALTH AGENCY, INC.

LITTLE ROCK HOME HEALTH AGENCY, INC.

LOWELL HOME HEALTH AGENCY, INC.

PHC ACQUISITION CORPORATION

PROFESSIONAL RELIABLE NURSING SERVICE, INC.

BENEFITS ASSURANCE CO., INC.

By:

Name: Frank Leonard
Title: Secretary

# ANNEX A SIXTH AMENDMENT EFFECTIVE DATE DELAYED DRAW FUNDING AMOUNTS

SIXTH AMENDMENT EFFECTIVE DATE DELAYED DRAW FUNDING

		AMOUNT	
		\$	%
Lender			
Freeport Loan Fund LLC	\$ 1	171,428.57	34.2857143%
CF Blackburn LLC		0	0
Fifth Third Bank (Chicago)	\$ 3	328,571.43	65.7142857%
Totals	\$ 5	500,000.00	100%

### CONSENT AND SEVENTH AMENDMENT TO CREDIT AGREEMENT

CONSENT AND SEVENTH AMENDMENT TO CREDIT AGREEMENT (this "<u>Amendment</u>"), dated as of October 21, 2008, among ADDUS HEALTHCARE, INC., an Illinois corporation ("<u>Borrower</u>"), the other persons signatory hereto as "Loan Parties", FREEPORT FINANCIAL LLC, a Delaware limited liability company ("<u>Agent</u>") and the Lenders signatory hereto. Terms not defined herein have the meanings given to them in the Credit Agreement (as hereinafter defined).

#### RECITALS

A. Borrower, the Loan Parties, the Lenders signatory thereto and Agent are party to that certain Credit Agreement dated as of September 19, 2006 (as amended by that certain Consent and First Amendment to Credit Agreement dated as of July 29, 2007, that certain Consent and Second Amendment to Credit Agreement dated as of October 15, 2007, that certain Consent and Fourth Amendment to Credit Agreement dated as of April 1, 2008, that certain Consent and Fifth Amendment to Credit Agreement dated as of June 9, 2008, that certain Consent and Sixth Amendment to Credit Agreement dated as of September 25, 2008 and as further amended, restated or otherwise modified including by this Amendment, the "Credit Agreement").

- B. Borrower, the Loan Parties, the Lenders and Agent are entering into this Amendment to amend certain provisions of the Credit Agreement.
- C. Agent, Lenders and the Loan Parties are willing to enter into this Amendment upon the terms and conditions set forth below.

NOW THEREFORE, in consideration of the matters set forth in the recitals and the covenants and provisions herein set forth, and other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

### **AGREEMENT**

Section 1. <u>Definitions</u>. Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed thereto in the Credit Agreement.

Section 2. <u>Consent</u>. Borrower, Agent and the Lenders hereby (A) consent to the assignment of certain Loans pursuant to the Assignment Agreement (the "<u>Seventh Amendment Assignment</u>") attached to this Amendment as <u>Exhibit A</u> notwithstanding the minimum assignment provisions of Section 9.1(a) of the Credit Agreement; and (B) consent to the drawing of all remaining Delayed Draw Term Loan Commitments immediately prior to the Seventh Amendment Effective Date notwithstanding the conditions to borrowing set forth in <u>Section 2.1(a)(iv)</u> of the Credit Agreement have not been met; provided that, notwithstanding the terms set forth in <u>Section 2.1(a)(iii)</u> of the Credit Agreement with respect to the use of proceeds of the

Delayed Draw Term Loans, the proceeds of the Delayed Draw Term Loans so borrowed are immediately applied to pay the amendment fee pursuant to Section 5.4 hereof and repay the outstanding principal amount of the Revolving Loans.

Section 3. Amendment to the Credit Agreement. As of the Seventh Amendment Effective Date, the Credit Agreement is hereby amended as follows:

3.1. Section 1.1 of the Credit Agreement is hereby amended by adding the following new definitions thereto:

Seventh Amendment means that certain Consent and Seventh Amendment to Credit Agreement dated as of October 21, 2008 among the Loan Parties, Agent and the Lenders.

Seventh Amendment Assignment has the meaning set forth in the Seventh Amendment.

Seventh Amendment Effective Date has the meaning set forth in the Seventh Amendment.

Seventh Amendment Reaffirmation of Collateral Documents means the Consent and Reaffirmation (Seventh Amendment), dated as of October 21, 2008 of the Loan Parties signatory thereto, in respect of the Collateral Documents.

- 3.2. On the Seventh Amendment Effective Date, <u>Section 2.1(a)(v)</u> of the Credit Agreement shall be amended and restated in its entirety and replaced with the following new <u>Section 2.1(a)(v)</u>:
- (v) Borrower shall repay the Term Loans through periodic payments on the dates and in the amounts indicated below ("Scheduled Installments"), and in any event the entire remaining principal balance shall be repaid on the Commitment Termination Date.

# Term Loan

December 31, 2008	Scheduled Installment \$1,662,500	
March 31, 2009	\$1,662,500	
June 30, 2009	\$1,662,500	
September 30, 2009	\$1,662,500	
December 31, 2009	\$2,100,000	
March 31, 2010	\$2,100,000	
June 30, 2010	\$2,100,000	
September 30, 2010	\$2,100,000	
December 31, 2010	\$2,537,500	
March 31, 2011	\$2,537,500	
June 30, 2011	\$2,537,500	
September 19, 2011	\$32,368,500 or the outstanding principal balance of	

\$32,368,500 or the outstanding principal balance of Term Loans outstanding on such date

The above scheduled installment amounts reflect the incurrence by the Borrower of the Delayed Draw Term Loans funded as of the Seventh Amendment Effective Date. The final installment payment shall in all events equal the entire remaining principal balance of the Term Loan (including any remaining principal balance of such Delayed Draw Term Loans). Amounts borrowed under this Section 2.1(a) and repaid may not be reborrowed.

At the request of the applicable Lender, the Term Loans shall be evidenced by promissory notes substantially in the form of Exhibit 2.1(a) (as amended, modified, extended, substituted or replaced from time to time, each a "Term Note" and, collectively, the "Term Notes"), and Borrower shall execute and deliver a Term Note to each such Term Lender. Each Term Note shall represent the obligation of Borrower to pay the amount of the applicable Term Lender's Term Loan Commitment, together with interest thereon."

- 3.3. On the Seventh Amendment Effective Date, Section 7.1(d) of the Credit Agreement shall be amended by replacing the Leverage Ratios currently referenced in such Section (i) for the Fiscal Quarters ending December 31, 2008 and March 31, 2009 with references in each case to 4.25:1.00; (ii) for the Fiscal Quarter ending June 30, 2009 with a reference to 4:00:1.00; and (iii) for the Fiscal Quarter ending September 30, 2009 with a reference to 3.75:1.00.
- Section 4. Representations and Warranties. To induce Agent and Lenders to execute this Amendment, each Loan Party hereby represents and warrants to Agent and Lenders as follows:
- (a) the execution, delivery and performance of this Amendment has been duly authorized by all requisite action of each Loan Party thereto, and this Amendment constitutes the legal, valid and binding obligation of such Person, enforceable against such Person in accordance with its terms, subject to bankruptcy, insolvency and similar laws affecting the enforceability of creditors' rights generally and to general principles of equity; and
- (b) each of the representations and warranties in the Credit Agreement are true and correct in all material respects with the same effect as though made on and as of the date hereof (except, in each case, to the extent stated to relate to an earlier date, in which case such representation or warranty shall have been true and correct on and as of such earlier date) and no Event of Default or Default exists thereunder or would exist after giving effect to this Amendment.
- Section 5. <u>Conditions Precedent</u>. This Amendment shall be effective as of the date first set forth above, subject to the satisfaction of the following conditions precedent (the date of such satisfaction being the "<u>Seventh Amendment Effective Date</u>"):
- 5.1. Execution and Delivery. Borrower, each of the other Loan Parties, Agent, the Requisite Lenders, each holder of Delayed Draw Term Loans and the Supermajority Revolving Lenders shall have executed and delivered this Agreement and each Loan Party shall have executed and delivered the Consent and Reaffirmation in the form of Exhibit B attached hereto.

- 5.2. Assignment Agreements. The Seventh Amendment Assignment shall be effective and the settlement of amounts due pursuant thereto shall have occurred.
- 5.3. <u>Delayed Draw Term Loans</u>. The Lenders holding Delayed Draw Term Loan Commitments shall have funded all such remaining Commitments and the proceeds of such borrowing shall have been immediately applied to pay the amendment fee pursuant to Section 5.4 hereof and repay the outstanding principal amount of the Revolving Loans.
- 5.4. Amendment Fee. Borrower shall have paid to Agent for the benefit of each Lender an amendment fee in an amount equal to \$271,991.25 which amount shall be payable to each such Lender based on such Lender's Pro Rata Share of such amount in each case immediately prior to giving effect to the Seventh Amendment Assignment but after giving effect to the drawing of the Delayed Draw Term Loans referenced in Section 5.3 of this Amendment.
  - 5.5. No Defaults. After giving effect to this Amendment, no Event of Default or Default under the Credit Agreement shall have occurred and be continuing.
- 5.6. Representations and Warranties. After giving effect to this Amendment, the representations and warranties of the Loan Parties contained in this Amendment, the Credit Agreement and the other Loan Documents shall be true and correct in all material respects as of the Seventh Amendment Effective Date, with the same effect as though made on such date, except to the extent that any such representation or warranty relates to an earlier date, in which case such representation or warranty shall be true and correct in all material respects as of such earlier date.
- 5.7. <u>Fees</u>. Borrower shall have paid all reasonable and documented costs, fees and expenses (including, without limitation, reasonable legal fees and expenses of Winston & Strawn LLP) of Agent.
- 5.8. Other Matters. Agent shall have received such other instruments and documents as Agent or the Requisite Lenders may reasonably request in connection with the execution of this Amendment, and all such instruments and documents shall be reasonably satisfactory in form and substance to Agent.

#### Section 6. <u>Miscellaneous</u>

6.1. Effect of Amendment. The execution, delivery and effectiveness of this Amendment shall not operate as a waiver of any right, power or remedy of Agent or any Lender under the Credit Agreement or any Loan Document, or constitute a waiver of any provision of the Credit Agreement or any Loan Document, except as specifically set forth herein.

- 6.2. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and by the different parties on separate counterparts, and each such counterpart shall be deemed to be an original, but all such counterparts shall together constitute but one and the same instrument. Delivery of the executed counterpart of this Amendment by telecopy or electronic mail shall be as effective as delivery of a manually executed counterpart to this Amendment.
- 6.3. <u>Severability</u>. The illegality or unenforceability of any provision of this Amendment or any instrument or agreement required hereunder shall not in any way affect or impair the legality or enforceability of the remaining provisions of this Amendment or any instrument or agreement required hereunder.
  - 6.4. Captions. Section captions used in this Amendment are for convenience only, and shall not affect the construction of this Amendment.
- 6.5. Entire Agreement. This Amendment embodies the entire agreement and understanding among the parties hereto and supersedes all prior or contemporaneous agreements and understandings of such Persons, verbal or written, relating to the subject matter hereof.
- 6.6. <u>References</u>. Any reference to the Credit Agreement contained in any notice, request, certificate, or other document executed concurrently with or after the execution and delivery of this Amendment shall be deemed to include this Amendment unless the context shall otherwise require. Reference in any of this Amendment, the Credit Agreement or any other Loan Document to the Credit Agreement shall be a reference to the Credit Agreement as amended hereby and as further amended, modified, restated, supplemented or extended from time to time.

Section 7. <u>Governing Law</u>. THIS AMENDMENT SHALL BE A CONTRACT MADE UNDER AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF ILLINOIS APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE, WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES.

[signature pages follow]

Witness the due execution hereof by the respective duly authorized officers of the undersigned as of the date first above written.

# ADDUS HOLDING CORPORATION

By: /s/ Simon Bachleda
Name: Simon Bachleda

Title: Vice President and Secretary

ADDUS HEALTHCARE, INC.

FORT SMITH HOME HEALTH AGENCY, INC.

LITTLE ROCK HOME HEALTH AGENCY, INC.

LOWELL HOME HEALTH AGENCY, INC.

PHC ACQUISITION CORPORATION

PROFESSIONAL RELIABLE NURSING SERVICE, INC.

BENEFITS ASSURANCE CO., INC.

By: /s/ Frank Leonard

Name: Frank Leonard

Title: Chief Financial Officer and Secretary

# FREEPORT FINANCIAL LLC, as Agent

By: /s/ Chad Blakeman
Title: Sr. Managing Director

# FREEPORT LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman
Title: Sr. Managing Director

# FREEPORT OFFSHORE LOAN FUND LLC, as a Lender

By: /s/ Chad Blakeman
Title: Sr. Managing Director

# FREEPORT LOAN TRUST 2006-1, as a Lender

By: /s/ Chad Blakeman

Title: Sr. Managing Director

# FIFTH THIRD BANK, CHICAGO (A Michigan Banking Company)

By: /s/ Michael E. May

Name: Michael E. May

Title: Vice President

CF BLACKBURN LLC
By: Name: Title:

## ASSIGNMENT AGREEMENT

This Assignment Agreement (this "<u>Agreement</u>") is made as of October 21, 2008 by and between FREEPORT LOAN FUND LLC ("<u>Assignor Lender</u>") and FIFTH THIRD BANK, CHICAGO (A MICHIGAN BANKING COMPANY) ("<u>Assignee Lender</u>"). All capitalized terms used in this Agreement and not otherwise defined herein will have the respective meanings set forth in the Credit Agreement as hereinafter defined.

#### **RECITALS**:

WHEREAS, Addus HealthCare, Inc., an Illinois corporation, ("Borrower"), Freeport Financial LLC, as agent ("Agent"), Assignor Lender and other Persons signatory thereto as Lenders have entered into that certain Credit Agreement dated as of September 19, 2006 (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") pursuant to which Assignor Lender has agreed to make certain Loans to, and incur certain Letter of Credit Obligations for, Borrower;

WHEREAS, Assignor Lender desires to assign to Assignee Lender a portion of its interest in the Loans (as described below), the Letter of Credit Obligations and the Collateral and to delegate to Assignee Lender a portion of its Commitments and other duties with respect to such Loans, Letter of Credit Obligations and Collateral;

WHEREAS, Assignee Lender desires to continue as a Lender under the Credit Agreement and to accept such assignment and delegation from Assignor Lender; and

NOW, THEREFORE, in consideration of the premises and the agreements, provisions, and covenants herein contained, Assignor Lender and Assignee Lender agree as follows:

## 1. ASSIGNMENT, DELEGATION, AND ACCEPTANCE

1.1 Assignment. Assignor Lender hereby transfers and assigns to Assignee Lender, without recourse and without representations or warranties of any kind (except as set forth in Section 3.2), such percentage of Assignor Lender's right, title, and interest in the Delayed Draw Term Loan Commitment, Loan Documents and the Collateral, as of the Effective Date (as hereinafter defined), as follows:

Assignee Lender's LoansCommitment AmountPro Rata ShareDelayed Draw Term Loan Commitment\$ 960,000.0034.28571428%

- 1.2 Delegation. Assignor Lender hereby irrevocably assigns and delegates to Assignee Lender a portion of its Commitments and its other duties and obligations as a Lender under the Loan Documents equivalent to the Pro Rata Share set forth above.
- 1.3 Acceptance by Assignee Lender. By its execution of this Agreement, Assignee Lender irrevocably purchases, assumes and accepts such assignment and delegation and agrees to be a Lender with respect to the delegated interest under the Loan Documents and to be bound by the terms and conditions thereof. By its execution of this Agreement, Assignor Lender agrees, to the extent provided herein, to relinquish its rights and be released from its obligations and duties under the Credit Agreement.
- 1.4 Effective Date. Such assignment and delegation by Assignor Lender and acceptance by Assignee Lender will be effective and Assignee Lender will become a Lender under the Loan Documents as of the date of this Agreement ("<u>Effective Date</u>"). Interest and Fees accrued prior to the Effective Date are for the account of Assignor Lender, and Interest and Fees accrued from and after the Effective Date are for the account of Assignee Lender. The Assignor Lender hereby agrees to waive payment of the assignment fee required by the Credit Agreement for this assignment. This is a one-time waiver and should not be construed as creating a course of conduct.
  - 2. INITIAL PAYMENT AND DELIVERY OF NOTES
  - 2.1 [Reserved]
  - 2.2 [Reserved]
- 2.3 Execution and Delivery of Notes. Assignor Lender will deliver to Agent the Notes previously delivered to Assignor Lender for redelivery to Borrower and, if requested by Assignee Lender, Agent will obtain from Borrower for delivery to Assignor Lender and Assignee Lender, new executed Notes evidencing Assignee Lender's and Assignor Lender's respective Pro Rata Shares in the Loans after giving effect to the assignment described in Section 1. Each new Note will be issued in the aggregate maximum principal amount of the applicable Commitment of the Lender to whom such Note is issued.
  - 3. REPRESENTATIONS, WARRANTIES AND COVENANTS
- 3.1 Assignee Lender's Representations, Warranties and Covenants. Assignee Lender hereby represents, warrants, and covenants the following to Assignor Lender and Agent:
  - (a) This Agreement is a legal, valid, and binding agreement of Assignee Lender, enforceable according to its terms;
  - (b) The execution and performance by Assignee Lender of its duties and obligations under this Agreement and the Loan Documents will not require any registration with, notice to, or consent or approval by any Governmental Authority;

- (c) Assignee Lender is familiar with transactions of the kind and scope reflected in the Loan Documents and in this Agreement;
- (d) Assignee Lender has made its own independent investigation and appraisal of the financial condition and affairs of each Loan Party, has conducted its own evaluation of the Loans and Letter of Credit Obligations, the Loan Documents and each Loan Party's creditworthiness, has made its decision to become a Lender to Borrower under the Credit Agreement independently and without reliance upon Assignor Lender or Agent, and will continue to do so;
- (e) Assignee Lender is entering into this Agreement in the ordinary course of its business, and is acquiring its interest in the Loans and Letter of Credit Obligations for its own account and not with a view to or for sale in connection with any subsequent distribution; provided, however, that at all times the distribution of Assignee Lender's property shall, subject to the terms of the Credit Agreement, be and remain within its control;
- (f) No future assignment or participation granted by Assignee Lender pursuant to Section 9.1 of the Credit Agreement will require Assignor Lender, Agent, or Borrower to file any registration statement with the Securities and Exchange Commission or to apply to qualify under the blue sky laws of any state;
  - (g) Assignee Lender has no loans (other than the Loans) to, written or oral agreements with, or equity or other ownership interest in any Loan Party;
- (h) Assignee Lender will not enter into any written or oral agreement with, or acquire any equity or other ownership interest in, any Loan Party without the prior written consent of Agent; and
- (i) As of the Effective Date, Assignee Lender (i) is entitled to receive payments of principal and interest in respect of the Obligations without deduction for or on account of any taxes imposed by the United States of America or any political subdivision thereof, (ii) is not subject to capital adequacy or similar requirements under Section 2.8(a) of the Credit Agreement, (iii) does not require the payment of any increased costs under Section 2.8(b) of the Credit Agreement, and (iv) is not unable to fund LIBOR Loans under Section 2.8(b) of the Credit Agreement, and Assignee Lender will indemnify Agent from and against all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, or expenses that result from Assignee Lender's failure to fulfill its obligations under the terms of Section 2.9(c) of the Credit Agreement or from any other inaccuracy in the foregoing.

- 3.2 Assignor Lender's Representations, Warranties and Covenants. Assignor Lender hereby represents, warrants and covenants the following to Assignee Lender:
  - (a) Assignor Lender is the legal and beneficial owner of the Delayed Draw Term Loan Commitment;
  - (b) This Agreement is a legal, valid and binding agreement of Assignor Lender, enforceable according to its terms;
  - (c) The execution and performance by Assignor Lender of its duties and obligations under this Agreement and the Loan Documents will not require any registration with, notice to or consent or approval by any Governmental Authority;
  - (d) Assignor Lender has full power and authority, and has taken all action necessary to execute and deliver this Agreement and to fulfill the obligations hereunder and to consummate the transactions contemplated hereby;
  - (e) Assignor Lender is the legal and beneficial owner of the interests being assigned hereby, free and clear of any adverse claim, lien, encumbrance, security interest, restriction on transfer, purchase option, call or similar right of a third party; and
    - (f) This Assignment by Assignor Lender to Assignee Lender complies, in all material respects, with the terms of the Loan Documents.

## 4. LIMITATIONS OF LIABILITY

Neither Assignor Lender (except as provided in Section 3.2) nor Agent makes any representations or warranties of any kind, nor assumes any responsibility or liability whatsoever, with regard to (a) the Loan Documents or any other document or instrument furnished pursuant thereto or the Loans, Letter of Credit Obligations or other Obligations, (b) the creation, validity, genuineness, enforceability, sufficiency, value or collectibility of any of them, (c) the amount, value or existence of the Collateral, (d) the perfection or priority of any Lien upon the Collateral, or (e) the financial condition of any Loan Party or other obligor or the performance or observance by any Loan Party of its obligations under any of the Loan Documents. Neither Assignor Lender nor Agent has or will have any duty, either initially or on a continuing basis, to make any investigation, evaluation, appraisal of, or any responsibility or liability with respect to the accuracy or completeness of, any information provided to Assignee Lender which has been provided to Assignor Lender or Agent by any Loan Party. Nothing in this Agreement or in the Loan Documents shall impose upon the Assignor Lender or Agent any fiduciary relationship in respect of the Assignee Lender.

## 5. FAILURE TO ENFORCE

No failure or delay on the part of Agent or Assignor Lender in the exercise of any power, right, or privilege hereunder or under any Loan Document will impair such power, right, or privilege or be construed to be a waiver of any default or acquiescence therein. No single or partial exercise of any such power, right, or privilege will preclude further exercise thereof or of any other right, power, or privilege. All rights and remedies existing under this Agreement are cumulative with, and not exclusive of, any rights or remedies otherwise available.

#### 6. NOTICES

Unless otherwise specifically provided herein, any notice or other communication required or permitted to be given will be in writing and addressed to the respective party as set forth below its signature hereunder, or to such other address as the party may designate in writing to the other.

## 7. AMENDMENTS AND WAIVERS

No amendment, modification, termination, or waiver of any provision of this Agreement will be effective without the written concurrence of Assignor Lender, Agent and Assignee Lender.

#### 8. SEVERABILITY

Whenever possible, each provision of this Agreement will be interpreted in such manner as to be effective and valid under applicable law. In the event any provision of this Agreement is or is held to be invalid, illegal, or unenforceable under applicable law, such provision will be ineffective only to the extent of such invalidity, illegality, or unenforceability, without invalidating the remainder of such provision or the remaining provisions of the Agreement. In addition, in the event any provision of or obligation under this Agreement is or is held to be invalid, illegal, or unenforceable in any jurisdiction, the validity, legality, and enforceability of the remaining provisions or obligations in any other jurisdictions will not in any way be affected or impaired thereby.

#### 9. SECTION TITLES

Section and Subsection titles in this Agreement are included for convenience of reference only, do not constitute a part of this Agreement for any other purpose, and have no substantive effect.

## 10. SUCCESSORS AND ASSIGNS

This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns.

## 11. APPLICABLE LAW

THIS AGREEMENT WILL BE CONSTRUED IN ACCORDANCE WITH AND GOVERNED BY THE LAWS OF THE STATE OF NEW YORK APPLICABLE TO CONTRACTS MADE AND PERFORMED IN THAT STATE.

# 12. COUNTERPARTS

This Agreement and any amendments, waivers, consents, or supplements may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which, when so executed and delivered, will be deemed an original and all of which shall together constitute one and the same instrument.

[Signature page follows]

ASSIGNEE LENDER:	ASSIGNOR LENDER:
FIFTH THIRD BANK, CHICAGO (A MICHIGAN BANKING COMPANY)	FREEPORT LOAN FUND LLC
By:Title:	By:Title:

IN WITNESS WHEREOF, this Agreement has been duly executed as of the date first written above.

222 South Riverside Plaza 33rd Floor, MD GRVR3I Chicago, IL 60606 Attn: Gregory H. Bork

Fax: (312) 704-4374

Notice Address:

Notice Address:

500 West Madison Street, Suite 7200 Chicago, Illinois 60661 Attn: Addus HealthCare Inc. Account Officer Fax: (312) 281-4646

## ASSIGNMENT AGREEMENT

This Assignment Agreement (this "Agreement") is made as of October 21, 2008 by and between FREEPORT OFFSHORE LOAN FUND LLC ("Assignor Lender") and FIFTH THIRD BANK, CHICAGO (A MICHIGAN BANKING COMPANY) ("Assignee Lender"). All capitalized terms used in this Agreement and not otherwise defined herein will have the respective meanings set forth in the Credit Agreement as hereinafter defined.

#### RECITALS

WHEREAS, Addus HealthCare, Inc., an Illinois corporation, ("Borrower"), Freeport Financial LLC, as agent ("Agent"), Assignor Lender and other Persons signatory thereto as Lenders have entered into that certain Credit Agreement dated as of September 19, 2006 (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") pursuant to which Assignor Lender has agreed to make certain Loans to, and incur certain Letter of Credit Obligations for, Borrower;

WHEREAS, Assignor Lender desires to assign to Assignee Lender a portion of its interest in the Loans (as described below), the Letter of Credit Obligations and the Collateral and to delegate to Assignee Lender a portion of its Commitments and other duties with respect to such Loans, Letter of Credit Obligations and Collateral:

WHEREAS, Assignee Lender desires to continue as a Lender under the Credit Agreement and to accept such assignment and delegation from Assignor Lender; and

NOW, THEREFORE, in consideration of the premises and the agreements, provisions, and covenants herein contained, Assignor Lender and Assignee Lender agree as follows:

## 1. ASSIGNMENT, DELEGATION, AND ACCEPTANCE

1.1 Assignment. Assignor Lender hereby transfers and assigns to Assignee Lender, without recourse and without representations or warranties of any kind (except as set forth in Section 3.2), such percentage of Assignor Lender's right, title, and interest in the Revolving Loan Commitment, Loan Documents and the Collateral, as of the Effective Date (as hereinafter defined), as follows:

Assignee Lender's LoansCommitment AmountPro Rata ShareRevolving Loan Commitment\$ 4,040,000.0023.08571428%

On the Effective Date the funded portion of the Revolving Loan Commitment so assigned is equal to \$1,731,429.00 (the "Funded Revolving Loan Amount").

1.2 Delegation. Assignor Lender hereby irrevocably assigns and delegates to Assignee Lender a portion of its Commitments and its other duties and obligations as a Lender under the Loan Documents equivalent to the Pro Rata Share set forth above.

- 1.3 Acceptance by Assignee Lender. By its execution of this Agreement, Assignee Lender irrevocably purchases, assumes and accepts such assignment and delegation and agrees to be a Lender with respect to the delegated interest under the Loan Documents and to be bound by the terms and conditions thereof. By its execution of this Agreement, Assignor Lender agrees, to the extent provided herein, to relinquish its rights and be released from its obligations and duties under the Credit Agreement.
- 1.4 Effective Date. Such assignment and delegation by Assignor Lender and acceptance by Assignee Lender will be effective and Assignee Lender will become a Lender under the Loan Documents as of the date of this Agreement ("<u>Effective Date</u>") and upon payment of the Assigned Amount (as such term is defined below). Interest and Fees accrued prior to the Effective Date are for the account of Assignor Lender, and Interest and Fees accrued from and after the Effective Date are for the account of Assignee Lender. The Assignor Lender hereby agrees to waive payment of the assignment fee required by the Credit Agreement for this assignment. This is a one-time waiver and should not be construed as creating a course of conduct.

## 2. INITIAL PAYMENT AND DELIVERY OF NOTES

2.1 Payment of the Assigned Amount. Assignee Lender will pay to Assignor Lender, in immediately available funds, not later than 4:00 p.m. (Chicago time) on the Effective Date, an amount equal to the Funded Revolving Loan Amount (the "Assigned Amount").

#### 2.2 [Reserved]

2.3 Execution and Delivery of Notes. Following payment of the Assigned Amount, Assignor Lender will deliver to Agent the Notes previously delivered to Assignor Lender for redelivery to Borrower and, if requested by Assignee Lender, Agent will obtain from Borrower for delivery to Assignor Lender and Assignee Lender, new executed Notes evidencing Assignee Lender's and Assignor Lender's respective Pro Rata Shares in the Loans after giving effect to the assignment described in Section 1. Each new Note will be issued in the aggregate maximum principal amount of the applicable Commitment of the Lender to whom such Note is issued.

# 3. REPRESENTATIONS, WARRANTIES AND COVENANTS

- 3.1 Assignee Lender's Representations, Warranties and Covenants. Assignee Lender hereby represents, warrants, and covenants the following to Assignor Lender and Agent:
  - (a) This Agreement is a legal, valid, and binding agreement of Assignee Lender, enforceable according to its terms;
  - (b) The execution and performance by Assignee Lender of its duties and obligations under this Agreement and the Loan Documents will not require any registration with, notice to, or consent or approval by any Governmental Authority;

- (c) Assignee Lender is familiar with transactions of the kind and scope reflected in the Loan Documents and in this Agreement;
- (d) Assignee Lender has made its own independent investigation and appraisal of the financial condition and affairs of each Loan Party, has conducted its own evaluation of the Loans and Letter of Credit Obligations, the Loan Documents and each Loan Party's creditworthiness, has made its decision to become a Lender to Borrower under the Credit Agreement independently and without reliance upon Assignor Lender or Agent, and will continue to do so;
- (e) Assignee Lender is entering into this Agreement in the ordinary course of its business, and is acquiring its interest in the Loans and Letter of Credit Obligations for its own account and not with a view to or for sale in connection with any subsequent distribution; provided, however, that at all times the distribution of Assignee Lender's property shall, subject to the terms of the Credit Agreement, be and remain within its control;
- (f) No future assignment or participation granted by Assignee Lender pursuant to Section 9.1 of the Credit Agreement will require Assignor Lender, Agent, or Borrower to file any registration statement with the Securities and Exchange Commission or to apply to qualify under the blue sky laws of any state;
  - (g) Assignee Lender has no loans (other than the Loans) to, written or oral agreements with, or equity or other ownership interest in any Loan Party;
- (h) Assignee Lender will not enter into any written or oral agreement with, or acquire any equity or other ownership interest in, any Loan Party without the prior written consent of Agent; and
- (i) As of the Effective Date, Assignee Lender (i) is entitled to receive payments of principal and interest in respect of the Obligations without deduction for or on account of any taxes imposed by the United States of America or any political subdivision thereof, (ii) is not subject to capital adequacy or similar requirements under Section 2.8(a) of the Credit Agreement, (iii) does not require the payment of any increased costs under Section 2.8(b) of the Credit Agreement, and (iv) is not unable to fund LIBOR Loans under Section 2.8(b) of the Credit Agreement, and Assignee Lender will indemnify Agent from and against all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, or expenses that result from Assignee Lender's failure to fulfill its obligations under the terms of Section 2.9(c) of the Credit Agreement or from any other inaccuracy in the foregoing.

- 3.2 Assignor Lender's Representations, Warranties and Covenants. Assignor Lender hereby represents, warrants and covenants the following to Assignee Lender:
  - (a) Assignor Lender is the legal and beneficial owner of the Assigned Amount;
  - (b) This Agreement is a legal, valid and binding agreement of Assignor Lender, enforceable according to its terms;
  - (c) The execution and performance by Assignor Lender of its duties and obligations under this Agreement and the Loan Documents will not require any registration with, notice to or consent or approval by any Governmental Authority;
  - (d) Assignor Lender has full power and authority, and has taken all action necessary to execute and deliver this Agreement and to fulfill the obligations hereunder and to consummate the transactions contemplated hereby;
  - (e) Assignor Lender is the legal and beneficial owner of the interests being assigned hereby, free and clear of any adverse claim, lien, encumbrance, security interest, restriction on transfer, purchase option, call or similar right of a third party; and
    - (f) This Assignment by Assignor Lender to Assignee Lender complies, in all material respects, with the terms of the Loan Documents.

#### 4. LIMITATIONS OF LIABILITY

Neither Assignor Lender (except as provided in Section 3.2) nor Agent makes any representations or warranties of any kind, nor assumes any responsibility or liability whatsoever, with regard to (a) the Loan Documents or any other document or instrument furnished pursuant thereto or the Loans, Letter of Credit Obligations or other Obligations, (b) the creation, validity, genuineness, enforceability, sufficiency, value or collectibility of any of them, (c) the amount, value or existence of the Collateral, (d) the perfection or priority of any Lien upon the Collateral, or (e) the financial condition of any Loan Party or other obligor or the performance or observance by any Loan Party of its obligations under any of the Loan Documents. Neither Assignor Lender nor Agent has or will have any duty, either initially or on a continuing basis, to make any investigation, evaluation, appraisal of, or any responsibility or liability with respect to the accuracy or completeness of, any information provided to Assignee Lender which has been provided to Assignor Lender or Agent by any Loan Party. Nothing in this Agreement or in the Loan Documents shall impose upon the Assignor Lender or Agent any fiduciary relationship in respect of the Assignee Lender.

#### 5. FAILURE TO ENFORCE

No failure or delay on the part of Agent or Assignor Lender in the exercise of any power, right, or privilege hereunder or under any Loan Document will impair such power, right, or privilege or be construed to be a waiver of any default or acquiescence therein. No single or partial exercise of any such power, right, or privilege will preclude further exercise thereof or of any other right, power, or privilege. All rights and remedies existing under this Agreement are cumulative with, and not exclusive of, any rights or remedies otherwise available.

## 6. NOTICES

Unless otherwise specifically provided herein, any notice or other communication required or permitted to be given will be in writing and addressed to the respective party as set forth below its signature hereunder, or to such other address as the party may designate in writing to the other.

#### 7. AMENDMENTS AND WAIVERS

No amendment, modification, termination, or waiver of any provision of this Agreement will be effective without the written concurrence of Assignor Lender, Agent and Assignee Lender.

## 8. SEVERABILITY

Whenever possible, each provision of this Agreement will be interpreted in such manner as to be effective and valid under applicable law. In the event any provision of this Agreement is or is held to be invalid, illegal, or unenforceable under applicable law, such provision will be ineffective only to the extent of such invalidity, illegality, or unenforceability, without invalidating the remainder of such provision or the remaining provisions of the Agreement. In addition, in the event any provision of or obligation under this Agreement is or is held to be invalid, illegal, or unenforceable in any jurisdiction, the validity, legality, and enforceability of the remaining provisions or obligations in any other jurisdictions will not in any way be affected or impaired thereby.

#### 9. SECTION TITLES

Section and Subsection titles in this Agreement are included for convenience of reference only, do not constitute a part of this Agreement for any other purpose, and have no substantive effect.

## 10. SUCCESSORS AND ASSIGNS

This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns.

## 11. APPLICABLE LAW

THIS AGREEMENT WILL BE CONSTRUED IN ACCORDANCE WITH AND GOVERNED BY THE LAWS OF THE STATE OF NEW YORK APPLICABLE TO CONTRACTS MADE AND PERFORMED IN THAT STATE.

# 12. COUNTERPARTS

This Agreement and any amendments, waivers, consents, or supplements may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which, when so executed and delivered, will be deemed an original and all of which shall together constitute one and the same instrument.

[Signature page follows]

ASSIGNEE LENDER:	ASSIGNOR LENDER:
FIFTH THIRD BANK, CHICAGO (A MICHIGAN BANKING COMPANY)	FREEPORT LOAN FUND LLC
By:	By:

IN WITNESS WHEREOF, this Agreement has been duly executed as of the date first written above.

Notice Address:

222 South Riverside Plaza 33rd Floor, MD GRVR3I Chicago, IL 60606 Attn: Gregory H. Bork Fax: (312) 704-4374 Notice Address:

500 West Madison Street, Suite 7200 Chicago, Illinois 60661 Attn: Addus HealthCare Inc. Account Officer Fax: (312) 281-4646

**EXHIBIT B** 

# **CONSENT AND REAFFIRMATION (SEVENTH AMENDMENT)**

Each of the undersigned ("Loan Parties") hereby (i) acknowledges receipt of a copy of the Consent and Seventh Amendment to Credit Agreement dated as of October 21, 2008 (the "Seventh Amendment"); (ii) consents to the execution and delivery thereof by the other Loan Parties; (iii) agrees to be bound thereby; (iv) affirms that nothing contained therein shall modify in any respect whatsoever its guaranty of the obligations of the Loan Parties to Agent and Lenders (the "Guarantee") and (v) reaffirms that the Guarantee and the other Collateral Documents executed by such Person are and shall continue to remain in full force and effect. Although each of the Loan Parties has been informed of the matters set forth herein and in the Seventh Amendment and has acknowledged and agreed to same, such Loan Parties understand that Agent and Lenders have no obligation to inform any of the Loan Parties of such matters in the future or to seek any of the Loan Parties' acknowledgment or agreement to future amendments or waivers, and nothing herein shall create such a duty.

This Consent and Reaffirmation shall be governed by and construed in accordance with the laws of the State of Illinois, without reference to principles of conflicts of law.

In witness whereof, each of the undersigned has executed this Consent and Reaffirmation on and as of the date of such Seventh Amendment.

# ADDUS HOLDING CORPORATION

By:

Name: Simon Bachleda

Title: Vice President and Secretary

ADDUS HEALTHCARE, INC.

ADDUS HEALTHCARE (NORTH CAROLINA), INC.

ADDUS HEALTHCARE (IDAHO), INC.

ADDUS HEALTHCARE (NEVADA), INC.

ADDUS HEALTHCARE (NEW JERSEY), INC.

FORT SMITH HOME HEALTH AGENCY, INC.

LITTLE ROCK HOME HEALTH AGENCY, INC.

LOWELL HOME HEALTH AGENCY, INC.

PHC ACQUISITION CORPORATION

PROFESSIONAL RELIABLE NURSING SERVICE, INC.

BENEFITS ASSURANCE CO., INC.

By:

Name: Frank Leonard
Title: Secretary